Electrophysiological indices of the Violence Inhibition Mechanism and their associations with physical aggression, callous-unemotional traits, and dietary omega-3

DEAN FIDO

A thesis submitted in partial fulfilment of the requirements of Nottingham Trent University for the degree of Doctor of Philosophy

August 2015

This work is the intellectual property of the author. You may copy up to 5% of this work for private study, or personal, non-commercial research. Any re-use of the information contained within this document should be fully referenced, quoting the author, title, university, degree level and pagination. Queries or requests for any other use, or if a more substantial copy is required, should be directed in the owner(s) of the Intellectual Property Rights.

Abstract

Each year, aggressive behaviour contributes to a substantial number of criminal offenses - resulting in severe personal, social, and financial ramifications. As such, there is importance for understanding the underlying mechanisms of aggression and how it might best be managed. The violence inhibition mechanism, thought to be dysfunctional in individuals characterised by aggressive behaviour and associated callous-unemotional traits, is used to explain how the perception of facial distress might inhibit an ongoing aggressive act. Despite a sizeable literature characterising the violence inhibition mechanism, to date, empirical research has overlooked this investigation on an electrophysiological level.

This thesis developed a novel facial-affect stopping task in order to tease apart the distinct stages of face processing and distress-induced motor extinction using electroencephalography. Results suggested that whilst callous-unemotional traits, specifically uncaring traits, were associated with electrophysiological indices of structural/featural face processing (N170, P200), aggressive traits, specifically physical aggression, were associated with electrophysiological indices of distress-induced motor extinction (stop-N200, stop-P300). Furthermore, in light of a growing literature suggesting a benefit of omega-3 dietary intake for both aggressive and callous-unemotional traits, correlational analysis suggested an association between omega-3 and physical aggression/distress-induced motor extinction, but not callous-unemotional traits/face processing.

These results have theoretical implications for understanding and investigating the violence inhibition mechanism on an electrophysiological level, as well as practical utility for better understanding how omega-3 might benefit aggressive traits and motor extinction. Specifically, the importance of distinguishing between both [1] aggressive and callous-unemotional traits and [2] face processing and motor extinction ability, when investigating the violence inhibition mechanism.

Ethical Approval

The Nottingham Trent University College Research Ethics Committee granted ethical approval for this thesis:

No. 2012/01 – 9th January 2012

- Modeling central inhibitory mechanisms in relation to aggression and impulsivity

No. 2012/09 – 23rd February 2012

- Developing a correlation perspective of Omega-3 intake, Aggression, Depression and markers of prenatal testosterone

No. 2012/16 – 8th February 2012

 Effect of Omega-3 supplementation on neuro-correlates: A mechanism study

Acknowledgments

I would like to express my sincere appreciation to Dr. Alexander L. Sumich, my Director of Studies, for his support throughout my academic career. His support, knowledge, and expertise were invaluable. His criticisms were well received and often drove my wider reading. I would like to thank him for his supervision over the past five years and look forward to working with him again in the future.

I would also like to thank Dr. Claire A. J. Bloxsom and Prof. Michael Gregson for their valuable advice and for taking the time to discuss the wider issues and application from which this thesis has benefited greatly. Furthermore, I would like to thank the staff and students of Nottingham Trent University for all of their assistance. In particular, I would like to thank Phil Banyard who has opened my eyes to teaching and taught me how to best engage students in academia. Thanks also to Prof. Irene Tracey and colleagues at the FMRIB Centre, Oxford, for their support and understanding during the final stages of editing.

I would also like to thank Elaine and Michael Fido for their unconditional love and support, as well as Brenda Clarke and Vera Fido, to whom this thesis is dedicated. My special thanks also go to my family and close friends for their encouragement and faith throughout the entire process, especially when it was needed the most.

Finally, I thank my participants, without which this thesis would not have been possible.

Dean Fido was funded by a Vice Chancellor's Bursary from Nottingham Trent University. Development of the MacBrain Face Stimulus Set was overseen by Nim Tottenham and supported by the John D. and Catherine T. MacArthur Foundation Research Network on Early Experience and Brain Development. Please contact Nim Tottenham at tott0006@tc.umn.edu for more information concerning the stimulus set.

Table of Contents

Abstract	iii
Ethical Approval	iv
Acknowledgments	v
List of Tables	xi
List of Figures	xii
List of Abbreviations	xiii
Foreword	XV
Chapter One: Conceptualisation and cognitive explanations of aggression	1
1.1 Introduction	1
1.2 Motivation and Manifestation of Aggression	1
1.2.1 Motivation	2
1.2.1.1 Reactive Aggression	2
1.2.1.2 Instrumental Aggression	3
1.2.2 Manifestation	5
1.2.2.1 Overt/Covert Aggression	6
1.2.2.2 Direct/Indirect Aggression	6
1.3 Stages of a Reactive Aggressive Encounter	7
1.3.1 Face Processing	7
1.3.2 Generation of Negative Affect	9
1.3.3 Executive Control	11
1.3.4 Inhibition by Cues of Distress	12
1.3.5 Behaviour Modulation	19
1.4 Chapter Summary	21
Chapter Two: Neurobiology of reactive aggression	22
2.2 Face Processing	22
2.3 Facial Threat	24
2.4 HPA-axis	26
2.5 Facial Distress	29
2.6 Executive Functioning	31
2.7 Chapter Summary	36
Chapter Three: Electrophysiological indices of aggression	37
3.1 Electroencephalography and Event-Related Potentials	37
3.2 Face Processing	39
3.2.1 N170	39
3.2.2 P200	41
3.2.3 N200	43
3.2.4 P300	43
3.3 Facial Threat and Distress Processing	44
3.3.1 N170	45
3.3.2 P200	47
3.3.3 N200	48
3.3.4 P300	49
3.3.5 Threat/Distress Processing Summary	49

vi

3.4 Executive Function 3.4.1 N200 3.4.2 P300 3.5 Negative Affect 3.6 CU and Aggressive Traits 3.6.1 N170 3.6.2 P200 3.6.3 N200 3.6.4 P300 3.7 Chapter Summary	50 50 51 52 54 55 56 56 56 58 61
Chapter Four: Dietary impact on aggression	63
4.1 Introduction	63
4.2 Dietary Modulators of Aggression	63
4.3 Polyunsaturated Fatty Acids	64
4.3.1 Composition, Intake, and Cell-level Importance	64
4.3.2 Polyunsaturated Fatty Acids and Aggression	68
4.5.2.1 Confetational Studies 4.3.2.2 Group Comparison Studies	60 69
4.3.2.3 Supplementation Studies	71
4.3.2.4 Fatty Acids/Aggression Summary	74
4.4 Face Processing	74
4.5 Negative Affect	76
4.5.1 Correlational Studies	76
4.5.2 Supplementation Studies	77
4.5.3 Underlying Mechanisms	/8
4.0 EXECUTIVE FUNCTIONING	00
4.7 Chanter Summary	Q1
4.7 Chapter Summary	81
4.7 Chapter Summary Chapter Five: Thesis aims and the delineation and justification of methods	81 82
4.7 Chapter Summary Chapter Five: Thesis aims and the delineation and justification of methods 5.1 Introduction	81 82 82
4.7 Chapter Summary Chapter Five: Thesis aims and the delineation and justification of methods 5.1 Introduction 5.2 Thesis Aims 5.2 Developmentation and Quantitative Measures	81 82 82 82
4.7 Chapter Summary Chapter Five: Thesis aims and the delineation and justification of methods 5.1 Introduction 5.2 Thesis Aims 5.3 Psychometric and Quantitative Measures 5.3 1 Aggression Questionnaire y 2	81 82 82 83 83
 4.7 Chapter Summary Chapter Five: Thesis aims and the delineation and justification of methods 5.1 Introduction 5.2 Thesis Aims 5.3 Psychometric and Quantitative Measures 5.3.1 Aggression Questionnaire v.2 5.3 2 Inventory of Callous-Unemotional Traits 	81 82 82 83 83 83
 4.7 Chapter Summary Chapter Five: Thesis aims and the delineation and justification of methods 5.1 Introduction 5.2 Thesis Aims 5.3 Psychometric and Quantitative Measures 5.3.1 Aggression Questionnaire v.2 5.3.2 Inventory of Callous-Unemotional Traits 5.3.3 Food Frequency Questionnaire 	81 82 82 83 83 83 85 87
 4.7 Chapter Summary Chapter Five: Thesis aims and the delineation and justification of methods 5.1 Introduction 5.2 Thesis Aims 5.3 Psychometric and Quantitative Measures 5.3.1 Aggression Questionnaire v.2 5.3.2 Inventory of Callous-Unemotional Traits 5.3.3 Food Frequency Questionnaire 5.3.4 Alcohol Use Disorders Identification Test 	81 82 82 83 83 83 85 87 88
 4.7 Chapter Summary Chapter Five: Thesis aims and the delineation and justification of methods 5.1 Introduction 5.2 Thesis Aims 5.3 Psychometric and Quantitative Measures 5.3.1 Aggression Questionnaire v.2 5.3.2 Inventory of Callous-Unemotional Traits 5.3.3 Food Frequency Questionnaire 5.3.4 Alcohol Use Disorders Identification Test 5.3.5 Cannabis Use Disorders Identification Test 	81 82 82 83 83 83 85 85 87 88 88
 4.7 Chapter Summary Chapter Five: Thesis aims and the delineation and justification of methods 5.1 Introduction 5.2 Thesis Aims 5.3 Psychometric and Quantitative Measures 5.3.1 Aggression Questionnaire v.2 5.3.2 Inventory of Callous-Unemotional Traits 5.3.3 Food Frequency Questionnaire 5.3.4 Alcohol Use Disorders Identification Test 5.3.5 Cannabis Use Disorders Identification Test Chapter Six: Correlations between physical aggression & callous-unemotional traits 	81 82 82 83 83 83 85 87 88 88 88 90
 4.7 Chapter Summary Chapter Five: Thesis aims and the delineation and justification of methods 5.1 Introduction 5.2 Thesis Aims 5.3 Psychometric and Quantitative Measures 5.3.1 Aggression Questionnaire v.2 5.3.2 Inventory of Callous-Unemotional Traits 5.3.3 Food Frequency Questionnaire 5.3.4 Alcohol Use Disorders Identification Test 5.3.5 Cannabis Use Disorders Identification Test Chapter Six: Correlations between physical aggression & callous-unemotional traits 6.1 Introduction 	81 82 82 83 83 83 85 87 88 88 88 90
 4.7 Chapter Summary Chapter Five: Thesis aims and the delineation and justification of methods 5.1 Introduction 5.2 Thesis Aims 5.3 Psychometric and Quantitative Measures 5.3.1 Aggression Questionnaire v.2 5.3.2 Inventory of Callous-Unemotional Traits 5.3.3 Food Frequency Questionnaire 5.3.4 Alcohol Use Disorders Identification Test 5.3.5 Cannabis Use Disorders Identification Test Chapter Six: Correlations between physical aggression & callous-unemotional traits 6.1 Introduction 6.2 Methods 	81 82 82 83 83 83 85 85 87 88 88 88 90 92
 4.7 Chapter Summary Chapter Five: Thesis aims and the delineation and justification of methods 5.1 Introduction 5.2 Thesis Aims 5.3 Psychometric and Quantitative Measures 5.3.1 Aggression Questionnaire v.2 5.3.2 Inventory of Callous-Unemotional Traits 5.3.3 Food Frequency Questionnaire 5.3.4 Alcohol Use Disorders Identification Test 5.3.5 Cannabis Use Disorders Identification Test Chapter Six: Correlations between physical aggression & callous-unemotional traits 6.1 Introduction 6.2 Methods 6.2.1 Participants 	81 82 82 83 83 83 85 87 88 88 88 90 90 92 92
 4.7 Chapter Summary Chapter Five: Thesis aims and the delineation and justification of methods 5.1 Introduction 5.2 Thesis Aims 5.3 Psychometric and Quantitative Measures 5.3.1 Aggression Questionnaire v.2 5.3.2 Inventory of Callous-Unemotional Traits 5.3.3 Food Frequency Questionnaire 5.3.4 Alcohol Use Disorders Identification Test 5.3 cannabis Use Disorders Identification Test Chapter Six: Correlations between physical aggression & callous-unemotional traits 6.1 Introduction 6.2 Methods 6.2.1 Participants 6.2.2 Measures 	81 82 82 83 83 83 85 87 88 88 88 90 90 92 92 92 92
 4.7 Chapter Summary Chapter Five: Thesis aims and the delineation and justification of methods 5.1 Introduction 5.2 Thesis Aims 5.3 Psychometric and Quantitative Measures 5.3.1 Aggression Questionnaire v.2 5.3.2 Inventory of Callous-Unemotional Traits 5.3.3 Food Frequency Questionnaire 5.3.4 Alcohol Use Disorders Identification Test 5.3.5 Cannabis Use Disorders Identification Test Chapter Six: Correlations between physical aggression & callous-unemotional traits 6.1 Introduction 6.2 Methods 6.2.1 Participants 6.2.2 Measures 6.2.3 Procedure 	81 82 82 83 83 83 85 87 88 88 88 90 90 92 92 92 92 92 92
 4.7 Chapter Summary Chapter Five: Thesis aims and the delineation and justification of methods 5.1 Introduction 5.2 Thesis Aims 5.3 Psychometric and Quantitative Measures 5.3.1 Aggression Questionnaire v.2 5.3.2 Inventory of Callous-Unemotional Traits 5.3.3 Food Frequency Questionnaire 5.3.4 Alcohol Use Disorders Identification Test 5.3.5 Cannabis Use Disorders Identification Test Chapter Six: Correlations between physical aggression & callous-unemotional traits 6.1 Introduction 6.2 Methods 6.2.1 Participants 6.2.2 Measures 6.2.3 Procedure 6.2.4 Statistical Analysis 	81 82 82 83 83 85 87 88 88 88 90 90 92 92 92 92 92 92 92 93
 4.7 Chapter Summary Chapter Five: Thesis aims and the delineation and justification of methods 5.1 Introduction 5.2 Thesis Aims 5.3 Psychometric and Quantitative Measures 5.3.1 Aggression Questionnaire v.2 5.3.2 Inventory of Callous-Unemotional Traits 5.3.3 Food Frequency Questionnaire 5.3.4 Alcohol Use Disorders Identification Test 5.3 Cannabis Use Disorders Identification Test 5.3 Correlations between physical aggression & callous-unemotional traits 6.1 Introduction 6.2 Methods 6.2.1 Participants 6.2.2 Measures 6.3 Procedure 6.2.4 Statistical Analysis 6.3 Results 	81 82 82 83 83 85 87 88 88 88 90 92 92 92 92 92 92 92 92 93 93
 4.7 Chapter Summary Chapter Five: Thesis aims and the delineation and justification of methods 5.1 Introduction 5.2 Thesis Aims 5.3 Psychometric and Quantitative Measures 5.3.1 Aggression Questionnaire v.2 5.3.2 Inventory of Callous-Unemotional Traits 5.3.3 Food Frequency Questionnaire 5.3.4 Alcohol Use Disorders Identification Test 5.3.5 Cannabis Use Disorders Identification Test 5.3 Correlations between physical aggression & callous-unemotional traits 6.1 Introduction 6.2 Methods 6.2.1 Participants 6.2.2 Measures 6.3.3 Procedure 6.3.4 Statistical Analysis 6.3 Results 6.3.1 Psychometric Measures 	81 82 82 83 83 83 85 87 88 88 88 90 92 92 92 92 92 92 93 93 93 93
 4.7 Chapter Summary Chapter Five: Thesis aims and the delineation and justification of methods 5.1 Introduction 5.2 Thesis Aims 5.3 Psychometric and Quantitative Measures 5.3.1 Aggression Questionnaire v.2 5.3.2 Inventory of Callous-Unemotional Traits 5.3.3 Food Frequency Questionnaire 5.3.4 Alcohol Use Disorders Identification Test 5.3.5 Cannabis Use Disorders Identification Test Chapter Six: Correlations between physical aggression & callous-unemotional traits 6.1 Introduction 6.2 Methods 6.2.1 Participants 6.2.2 Measures 6.3.1 Psychometric Measures 6.3.1 Correlations 	81 82 82 83 83 85 87 88 88 88 90 90 92 92 92 92 92 92 92 92 93 93 93 93 95
 4.7 Chapter Summary Chapter Five: Thesis aims and the delineation and justification of methods 5.1 Introduction 5.2 Thesis Aims 5.3 Psychometric and Quantitative Measures 5.3.1 Aggression Questionnaire v.2 5.3.2 Inventory of Callous-Unemotional Traits 5.3.3 Food Frequency Questionnaire 5.3.4 Alcohol Use Disorders Identification Test 5.3.5 Cannabis Use Disorders Identification Test 5.3 Correlations between physical aggression & callous-unemotional traits 6.1 Introduction 6.2 Methods 6.2.1 Participants 6.2.2 Measures 6.3.1 Psychometric Measures 6.3.1 Psychometric Measures 6.3.2 Correlations 	81 82 82 83 83 85 87 88 88 90 90 92 92 92 92 92 92 92 93 93 93 93 95 96 96
 4.7 Chapter Summary Chapter Five: Thesis aims and the delineation and justification of methods 5.1 Introduction 5.2 Thesis Aims 5.3 Psychometric and Quantitative Measures 5.3.1 Aggression Questionnaire v.2 5.3.2 Inventory of Callous-Unemotional Traits 5.3.3 Food Frequency Questionnaire 5.3.4 Alcohol Use Disorders Identification Test 5.3.5 Cannabis Use Disorders Identification Test Chapter Six: Correlations between physical aggression & callous-unemotional traits 6.1 Introduction 6.2 Methods 6.2.1 Participants 6.2.2 Measures 6.3.1 Psychometric Measures 6.3.1 Psychometric Measures 6.3.2 Correlations 6.3.2.1 Total Sample 6.3.2.3 Females 	81 82 82 83 83 85 87 88 88 90 90 92 92 92 92 92 92 92 93 93 93 93 95 96 96 96
 4.7 Chapter Summary Chapter Five: Thesis aims and the delineation and justification of methods 5.1 Introduction 5.2 Thesis Aims 5.3 Psychometric and Quantitative Measures 5.3.1 Aggression Questionnaire v.2 5.3.2 Inventory of Callous-Unemotional Traits 5.3.3 Food Frequency Questionnaire 5.3.4 Alcohol Use Disorders Identification Test 5.3.5 Cannabis Use Disorders Identification Test 5.3.5 Cannabis Use Disorders Identification Test Chapter Six: Correlations between physical aggression & callous-unemotional traits 6.1 Introduction 6.2 Methods 6.2.1 Participants 6.2.2 Measures 6.3.2 Procedure 6.3.1 Psychometric Measures 6.3.2 Correlations 6.3.2.1 Total Sample 6.3.2.3 Females 6.3.2.4 Sex Differences 	81 82 82 83 83 85 87 88 88 90 90 92 92 92 92 92 92 92 92 93 93 93 93 93 95 96 96 96 96 97
 4.7 Chapter Summary Chapter Five: Thesis aims and the delineation and justification of methods 5.1 Introduction 5.2 Thesis Aims 5.3 Psychometric and Quantitative Measures 5.3.1 Aggression Questionnaire v.2 5.3.2 Inventory of Callous-Unemotional Traits 5.3.3 Food Frequency Questionnaire 5.3.4 Alcohol Use Disorders Identification Test 5.3 Cannabis Use Disorders Identification Test Chapter Six: Correlations between physical aggression & callous-unemotional traits 6.1 Introduction 6.2 Methods 6.2.1 Participants 6.2.2 Measures 6.2.3 Procedure 6.2.4 Statistical Analysis 6.3 Results 6.3.2 Correlations 6.3.2 Correlations 6.3.2 Males 6.3.2.3 Females 6.3.2.4 Sex Differences 6.4 Discussion 	81 82 82 83 83 85 87 88 88 90 92 92 92 92 92 92 92 93 93 93 93 93 95 96 96 96 96 97 98

Chapter Seven: Baseline assessment of the Facial Affect Stop-Go Task	102
7.1 General Introduction	102
7.2 Development of the Facial Affect Stop-Go Task	102
7.2.1 Task Development	102
7.2.1.1 Stimuli	103
7.2.1.2 Trials	103
7.2.1.3 Stimulus Presentation and Filtering	104
7.2.2 FAST v.1	105
7.2.2.1 Participants	105
7.2.2.2 Procedure	105
7.2.2.3 Physiological Data Acquisition	106
7.2.2.4 Signal Processing	107
7.2.2.5 Results and Discussion	107
7.2.3 FAST v.2	108
7.2.3.1 Participants	108
7.2.3.2 Procedure	108
7.2.3.3 Physiological Data Acquisition	109
7.2.3.4 Signal Processing	109
7.2.3.5 Results and Discussion	110
7.2.4 FAST v.3 and Experimental Methods	112
7.2.4.1 Participants	112
7.2.4.2 Procedure	112
7.2.4.5 Physiological Data Acquisition	114
7.2.4.4 Signal Flocessing	114
7.3 Testing of the Facial Affect Ston-Co Task	114
7.3.1 Introduction	115
7.3.1 Introduction	11J 117
7.3.2 Methous 7.3.2.1 Participants	117
7 3 2 2 Procedure	117
7 3 2 3 Physiological Data Acquisition	117
7.3.2.4 Signal Processing	117
7.3.2.5 Statistics	118
7.3.3 Results	118
7.3.3.1 Behavioural	118
7.3.3.2 N170	119
7.3.3.3 P200	120
7.3.3.4 N200	122
7.3.3.4.1 Main Effects	122
7.3.3.4.2 Interactions	124
7.3.3.5 P300	130
7.3.3.5.1 Main Effects	130
7.3.3.5.2 Interactions	131
7.4 Discussion	141
7.4.1 N170	141
7.4.2 P200	143
7.4.3 N200	144
7.4.4 P300	145
7.4.5 Limitations	146
7.4.6 Summary	148
Chapter Eight: Correlations between physical aggression/callousness-unemo	otional traits
and face-evoked ERPs	149
8.1 Introduction	149
8.2 Methodology	152

viii

8.2.1 Measures	152
8.2.2 Statistical Analysis	152
8.3 Results	153
8.3.1 Behavioural	154
8.3.2 Correlations	154
8.3.2.1 N170	154
8.3.2.2 P200	154
8.4 Discussion	155
8.4.1 N170	155
8.4.2 P200	156
8.4.3 Implications, Alternative Explanations, and Future Research	157
8.4.4 Summary	159
Chapter Nine: Correlations between physical aggression/callous-unemotional traits a	ind
motor extinction ERPs	160
9.1 Introduction	160
9.2 Methods	162
9.2.1 Measures	162
9.2.2 Statistical Analysis	162
9.3 Results	163
9.3.1 Psychometrics	163
9.3.2 Behavioural data	163
9.3.3 Correlations	164
9.3.3.1 N200	164
9.3.3.2 P300	164
9.4 Discussion	165
9.4.1 N200	166
9.4.2 P300	167
9.4.3 Chapter Summary	168
Chapter Ten: Relationships between omega-3 intake, callous-unemotional traits, and	
physical aggression	170
10.1 Introduction	170
10.2 Methods	172
10.2.1 Participants	172
10.2.2 Measures	172
10.2.3 Procedure	173
10.2.4 Statistical Analysis	173
10.3 Results	173
10.4 Discussion	175
Chanter Fleven: Correlations between omega-3 intake and the electrophysiological in	dices
of the Violence Inhibition Mechanism	178
11.1 Introduction	178
11.2 Methods	180
11.2.1 Measures	180
11.2.2 Statistical Analysis	180
11.3 Results	181
11.3.1 Behavioural Data	181
11.3.2 N170 and P200	181
11.3.3 Stop-N200 and Stop-P300	181
11.4 Discussion	183

Chapter Twelve: Conclusion	186	
12.1 Introduction	186	
12.2 Empirical Findings and Theoretical Implication	187	
12.2.1 Research Question One	187	
12.2.2 Research Question Two	188	
12.2.3 Research Question Three	190	
12.2.4 Research Questions Four and Five	192	
12.3 Scientific Implications	193	
12.4 Recommendation for Future Research and Limitations	194	
12.4.1 Sample	194	
12.4.2 Reported Diet	195	
12.4.3 Statistical Considerations	196	
12.5 Policy Implication	197	
12.6 Thesis Conclusion	198	
References	200	

List of Tables

- Table 1
 Examples of scenarios associated with motivations for and manifestations of aggression
- Table 2
 Means and standard deviations of psychometric measures
- Table 3
 Bivariate correlation coefficients between psychometric measures and sex
- Table 4-Descriptions of N170 amplitudes across emotion, hemisphere,
site location, and task
- Table 5-Descriptions of P200 amplitudes across emotion, hemisphere,
site location, and task
- Table 6-Descriptions of N200 amplitudes across emotion, hemisphere,
site location, and task
- Table 7-Descriptions of P300 amplitudes across emotion, hemisphere,
site location, and task
- Table 8
 Statistically significant ANOVA summary table for main effects and interactions across ERPs
- Table 9
 Correlations between psychometric measures and behavioural data
- Table 10-Partial correlations between fatty acid intake (EPA, DHA) and
psychometric scores (physical aggression, callousness,
uncaring, unemotional)

List of Figures

- Figure 1 Means and standard deviations of psychometric responses for males and females
- Figure 2 Scatter plots of partial correlations between
- Figure 3 FAST version 1
- Figure 4 FAST version 2
- Figure 5 FAST version 3
- Figure 6 ERP waveforms to fearful, sad, angry, and neutral facial expressions
- Figure 7 ERP waveforms to fearful and sad stop facial expressions during background and stop task conditions
- Figure 8 Scatter plots of correlations between ICU-measured uncaring traits and ERPs evoked over left temporo-parietal sites
- Figure 9 Correlations between physical aggression and stop-N200 and stop-P300 amplitudes
- Figure 10 Correlations between physical aggression and *In*-transformed EPA and DHA intake
- Figure 11 Partial correlations between central-midline stop-N200 and stop-P300 responses to sad and fearful facial expressions, and *In*-EPA and *In*-DHA intake

List of Abbreviations

ADJAR	-	Adjacent response technique
AHR	-	Adrenocorticotropic hormone release
AQ	-	Aggression Questionnaire
AUDIT	-	Alcohol Use Disorder Identification Test
ALA	-	Alpha-linoleic acid
ANOVA	-	Analyses of variance
ACC	-	Anterior cingulate cortex
ASPD	-	Antisocial personality disorder
APSD	-	Antisocial Process Screening Device
AA	-	Arachidonic acid
ADHD	-	Attention deficit hyperactivity disorder
BPD	-	Borderline personality disorder
BPAQ	-	Buss and Perry Aggression Questionnaire
CD	-	Conduct disorder
CU	-	Callous-unemotional
CUDIT	-	Cannabis Use Disorder Identification Test
CRFs	-	Corticotropin-releasing factors
DHA	-	Docosahexaenoic acid
dACC	-	Dorsal anterior cingulate cortex
dIPFC	-	Dorsolateral prefrontal cortex
dmPFC	-	Dorsomedial prefrontal cortex
EPA	-	Eicosapentaenoic acid
EEG	-	Electroencephalography
EFAs	-	Essential fatty acids
ERP	-	Event-related potentials
FAST	-	Facial Affect Stop Task
FFQ	-	Food Frequency Questionnaire
(f)MRI	-	(Functional) magnetic resonance imaging
GAM	-	General aggression model
HRM	-	Horse race model
HPA-axis	-	Hypothalamus, pituitary gland, and adrenal axis
IFG	-	Inferior frontal gyrus
IES	-	Integrated emotion system
ISI	-	Inter-stimulus intervals
IED	-	Intermittent explosive disorder
ICU	-	Inventory of Callous–Unemotional Traits
ІНур	-	Lateral hypothalamus
LA	-	Linoleic acid
LCPUFA	-	Long-chain polyunsaturated fatty acids
MEG	-	Magnetoencephalography
MAPPA	-	Multi-Agency Public Protection Arrangement
OFA	-	Occipital face area
OFC	-	Orbitofrontal cortices

PAG	-	Periaqueductal gray
PUFAs	-	Polyunsaturated fatty acids
Pre-SMA	-	Pre-supplementary motor area
PFC	-	Prefrontal cortex
PPI–R	-	Psychopathic Personality Inventory - Revised
PCL-R	-	Psychopathy Checklist - Revised
RNA	-	Ribonucleic acid
SNR	-	Signal to noise ratio
SCP	-	Social-cognitive processing model
SST	-	Stop-signal task
STG	-	Superior temporal gyrus
STS	-	Superior temporal sulcus
vACC	-	Ventral anterior cingulate cortex
vmHyp	-	Ventromedial hypothalamus
vmPFC	-	Ventromedial prefrontal cortex
VPP	-	Vertex positive potential
VIM	-	Violence inhibition mechanism

Foreword

Aggression is a natural construct adaptive to the requirements of social context (Nelson & Trainor, 2007; Rosel & Siever, 2015). Under certain circumstances, aggression provides a means of successfully negotiating threat, protecting resources, and acquiring mates. However, persistent and extreme aggression can be detrimental to oneself and problematic for wider society (Anderson & Bushman, 2001; Buydens-Branchey & Branchey, 2008; Nelson & Trainor, 2007).

Each year, an estimated 1.3 million people die as a result of violence (World Health Organisation, 2014), a term used to denote extreme, physical aggression (Anderson & Bushman, 2002; Reidy, Shelley-Tremblay, & Lilienfeld, 2011). In England, though lower than its peak in 1995, aggression was associated with 634,462 criminal offences between 2013 and 2014 (Office for National Statistics: Crime Statistics, 2014). Clearly, there is need to better understand the underlying mechanisms contributing to aggression, and how they are best managed.

One such mechanism, the violence inhibition mechanism (VIM; Blair, 1995), suggests that humans are typically able to extinguish antisocial and aggressive behaviour following the perception of facial distress in others. Thought to be a function of empathy, the VIM appears dysfunctional in psychopathic individuals, characterised by aggressive and callous-unemotional (CU) traits (Blair, 2001; Hare, 1991). Furthermore, due to the growing interest in using diet, specifically omega-3, as a moderator of aggressive behaviour (Buydens-Branchey & Branchey, 2008), there is reason to expect omega-3 intake might moderate the VIM response. For the first time, this thesis explores the VIM in relation to physical aggression, CU traits, and omega-3 intake.

Chapter one delineates motivations underpinning, and manifestations of, aggressive behaviour, before outlining key evolutionary and cognitive theories of aggression. In this chapter, aggression is framed around the core stages of the VIM (i.e. face processing and distress-induced motor extinction), with supporting knowledge drawn from forensic, clinical, and community samples characterised by aggressive and CU traits, in which the VIM is considered dysfunctional. Chapter two focuses discussion on reactive aggression - expanding on the cognitive models outlined in chapter one - using evidence obtained through neuroimaging (e.g. and lesion studies). Here, neurobiological underpinnings of reactive aggression, distress perception, and motor extinction are delineated.

Chapter three explains how the neurobiological mechanisms described in chapter two might be indexed via temporally accurate electrophysiological measurements, and suggests how they might manifest in samples characterised by aggressive and CU traits. Chapter four draws these multidisciplinary themes together and suggests a role of diet, particularly omega-3 fatty acids, in the moderation of physical aggression, CU traits, and distress-induced motor extinction. Furthermore, primary and secondary research aims are delineated in chapter five, before outlining/justifying the key research methods used throughout the thesis.

Six experimental chapters follow. Chapter six identifies baseline associations between physical aggression and CU traits in a large community cohort. Chapter seven documents the development of the Facial Affect Stop Task (FAST) used to investigate how distinct face processing and motor extinction stages of the VIM might be indexed on an electrophysiological level. Chapters eight and nine correlate physical aggression and CU traits with FAST-evoked ERPs and chapters ten and eleven examine these ERP/psychometric associations in the context of omega-3 dietary intake. Specifically, chapter ten explores baseline correlations between omega-3 intake and self-reported measures of physical aggression and CU traits, and chapter eleven explores associations between omega-3 intake and the electrophysiological indices of the FAST.

Finally, chapter twelve draws together the experimental findings by discussing their theoretical and practical utility, and how said results might provide a baseline for future investigation. Subsequently, task-specific and global research limitations are addressed before summarising the novel contributions of the thesis. "Let's sit down and have a discussion about this queer animal," the second blind man said, breaking the silence.

"A very good idea. Very good." the other two agreed for they also had this in mind. Without waiting for anyone to be properly seated, the second one blurted out, "This queer animal is like our straw fans swinging back and forth to give us a breeze. However, it's not so big or well made. The main portion is rather wispy."

"No, no!" the first blind man shouted in disagreement. "This queer animal resembles two big trees without any branches."

"You're both wrong." the third man replied. "This queer animal is similar to a snake; it's long and round, and very strong."

How they argued! Each one insisted that he alone was correct. Of course, there was no conclusion for not one had thoroughly examined the whole elephant. How can anyone describe the whole until he has learned the total of the parts."

(Kuo & Kuo, 1976, pp. 85)

Chapter One: Conceptualisation and cognitive explanations of aggression

1.1 Introduction

As with the parabolic elephant described on page xvii, the many facets of aggression have complicated adoption of a universal definition (Barratt, Stanford, Kent, & Felthous, 1997) and have lead to extensive debate of its conceptualisation. Leading contemporary definitions classify aggression as the threat of, or actual behaviour towards another, with the intention to cause harm, and a belief that the other is motivated to avoid said harm (Anderson & Huesmann, 2003; Bushman & Anderson, 2001; Paschall & Fishbein, 2002). From an evolutionary perspective, successful traits that increase gene distribution and solve environmental problems are preserved and reproduced (Buss & Shackelford, 1997; Darwin, 1895). A lack of outbreeding for humans to exhibit the capacity to aggress implies prior and on going utility for aggression (Archer, 2009; Raine, 2013).

Understanding the multifarious nature of aggression and its neurobiological underpinnings has benefit for identifying neurobiological markers of aggression and potential treatment measures. This chapter begins by delineating the motivations underpinning, and manifestations of aggression, before exploring how aggression might be explained through cognitive theory.

1.2 Motivation and Manifestation of Aggression

Aggression is a complex phenomenon often discussed through specific dichotomies. As shown in Table 1 (page 4), whilst some dichotomies emphasise the motivations behind an aggressive act (*reactive/instrumental*; Baron & Richardson, 1994), others indicate how aggression might manifestation (*covert/overt, direct/indirect*; Bendig, 1962; Berkowitz, 1993).

1.2.1 Motivation

Emerging from an earlier classification of proactive and reactive behaviour (Feshbach, 1964), motivation to aggress is commonly considered to fall within the reactive/instrumental dichotomy (Baron & Richardson, 1994).

1.2.1.1 Reactive Aggression

Reactive aggression is an unregulated behavioural outburst without intention, goal, or profit (Raine, 2013; Reidy et al., 2011) that likely evolved as a strategy for defending resources, social status, and oneself (Wallner & Machatschke, 2009). In most instances, reactive aggression is accompanied by physiological changes such as heart pounding, a racing pulse, shallow breathing, and muscle tension (Wahlund & Kristiansson, 2009). In animal models, reactive aggression is referred to as affective defence or defensive rage (Blair, 2002; Gregg & Siegel, 2001; Meloy, 2006; Wahlund & Kristiansson, 2009), and features behaviours such as kicking, pawing, and baring teeth (Hess et al., 1928; Hess & Brugger, 1943; Hunsperger, 1956; Kruk et al., 1998; Nakao, 1958).

Reactive aggression is driven by negative affective states, such as anger or anxiety (Feshbach, 1964; Raine, 2013; Spielberger, Reheiser, & Sydeman, 1995), and serves to abolish aversive feelings such as threat (Averill, 1977; Baron & Richardson, 1994; Tyson, 1998, *see row 1 of Table 1*). Threat can be conveyed through many visual and/or auditory modalities, but in humans, is commonly indicated through angry facial expressions (Öhman, Flykt, & Esteves, 2001; Pourtois et al., 2004). Whilst moderate provocation (i.e. a challenge over resources) might evoke anger-induced aggression, a more severe, lifethreatening challenge might result in an unrestrained, fear-induced, aggressive response (Blanchard & Blanchard, 1984). At its extreme, homicide might reflect an evolutionary mechanism for solving dangerous, gene-damaging conflicts (Daly & Wilson, 1988).

When used in certain contexts, such as the workplace or when outnumbered, aggression may be detrimental to social status and/or gene survival (Buss &

Shackelford, 1997). In such contexts, aggression might be negatively perceived, and so the loss of social stake from inhibiting aggression is likely outweighed by the benefits of inhibiting an aggressive act (Cashdan & Downes, 2012). Whilst contradicting instinct theories, which frame aggression as a natural instinct manifesting invariantly of context (Buss & Shackelford, 1997), the ability to adapt behaviour to the requirements of social context is vital for gene survival.

In some literature, the term reactive aggression has been used interchangeably with hostility (Feshbach, 1964; Spielberger et al., 1995; Tsytsarev & Grodnitzky, 1995). More accurately, hostility denotes a negative evaluation, accompanied by a desire to inflict harm (Buss, 1961). The conceptual framework of reactive and hostile aggression has been related to impulsivity in both clinical and empirical literature (Atkins et al., 1993; Barratt et al., 1997; Lee & Coccaro, 2001; Moeller et al., 2001). Impulsivity is a multifaceted concept, characterised by acting without thinking (Smith, 1952), an inability to delay gratification (Best, 2002; Logan, Schachar, & Tannock, 1997; Logue, 1988), and poor inhibition of behaviour or thought (Frick & Hare, 2001; Logan et al., 1997). Impulsive aggression frames the use of aggression as a rapid response born from an agitated state (Barratt, 1991; Connor, 2002).

1.2.1.2 Instrumental Aggression

Opposed to impulsive, affect-driven, reactive aggression (Meloy, 2000; Stanford et al., 2003), instrumental aggression represents cold-blooded, controlled behaviour, used to achieve a purpose (Baron & Richardson, 1994; Reidy, et al., 2011; Stanford et al., 2003). Instrumental aggression may manifest following a cooling-off period (Wahlund & Kristiansson, 2009), and as evidenced in Table 1, row 2, might be used to gain advantage over rivals by accruing social status (*see cells A2 and B2*), resources (*see cell C2*), or sexual mates (*see cell D2*).

Table 1.

Examples of scenarios associated with motivations for and manifestations of aggression

	Co	vert	Overt		
	Direct (A)	Indirect (B)	Direct (C)	Indirect (D)	
Reactive	A wife angers her husband by	A man is told by his wife's friends	At a busy bar, a man	A man learns from a group of friends	
(1)	accusing him of having an affair.	that he is suspected of having an	knocks into another	that another friend is having an affair	
	In response, the man <u>yells at his</u>	affair. Angered by these	man, subsequently	with his wife. Without the friend	
	wife – suggesting that he works	accusation, and in the hope that	spilling his drinks. With	present to confront, anger drives	
	hard to support their family and	they will tell his wife, the man	no word of an apology	<i>him</i> to spread unfavourable and	
	that he is hurt by her	<u>confides in these friends</u> that he	for the act, the second	character-damaging rumours	
	accusations. The wife is	is hurt by the accusations and	man <i>feels anger</i> and	about the friend as retribution. <u>The</u>	
	manipulated to feel guilt.	only wants the best for his family.	swears <u>at the male</u>	group exclude the absent friend in	
		The wife is made to feel guilt.	before punching <u>him</u>	<u>the future.</u>	
			to the ground.		
Instrumental	An individual asks advice from a	Two co-workers are interviewing	<u>A man is stopped in</u>	A male has long been in love with	
(2)	co-worker when rehearsing for a	for the same position. During the	<u>the street late at night</u>	his female friend, who is currently in	
	job interview. Unbeknown to	first candidate's interview, they	by an attacker with a	a relationship with another man.	
	them, their co-worker is applying	mention the other candidate is	knife. The attacker	With the aim of achieving a	
	for the same job and tries to	aggressive and a poor team	demands the victim's	relationship with this friend, the male	
	<i>gain an advantage</i> by giving	player. The employer is	phone and wallet by	tells her that her partner is	
	poor advice about what to say	manipulated into taking a less	threatening harm,	cheating. This results in the female	
	during the interview. This	favourable view on the second	before escaping with	leaving her partner and finding	
	causes the candidate to fail.	candidate. Ultimately, the first	their possessions.	comfort with her friend, eventually	
		candidate is hired.		leading to a relationship.	

Note. Information relating to **Covert/Overt** manifestations is emboldened, <u>Direct/Indirect</u> manifestations are underlined, and *Reactive/Instrumental* motivations are italicised.

4

Resources, including food and shelter, are pivotal to survival (Raine, 2013) and facilitate the acquisition of sexual mates (Archer, 2009). From an evolutionary perspective, genes might benefit from others failing to gather resources (Dawkins, 2006) and so aggression might present a successful strategy for gene preservation (Raine, 2013). However, although winning conflicts benefits gene survival, frequent acts of aggression increase the risk of failure – possibly resulting in harm, retaliation, or the shunning of sexual mates.

Perceived by some as risky and detrimental (Farthing, 2005), the use of aggression might project bravery and dominance during sexual mate acquisition (Ferguson & Dyck, 2012; Kelly & Dunbar, 2001; Krämer, Jansma, Tempelmann, & Münte, 2007). Dominance has been described as a vehicle for reproductive success (Elliot, 1995) and might facilitate the retention of sexual mates by warding off current and future challengers (Buss & Shackelford, 1997; Raine, 2013). Intra-sexual aggression is observed less frequently in females (Campbell, 1995), and may partially account for the evolutionary development of increased body size and strength in males (Archer, 2009).

Instrumental aggression is discussed in conjunction with proactive, premeditated, and predatory aggression. Whilst both proactive and premeditated aggression encapsulate goal-driven behaviours used to detriment another, only proactive aggression occurs solely without provocation (Connor, 2002; Feshbach, 1964). In animals, predatory aggression might be defined through hunting and survival behaviours such as stalking and neck biting (Kruk, 2014; Kruk et al., 1998; Wahlund & Kristiansson, 2009; Wasman & Flynn, 1962), with instrumental aggression in humans thought to include stalking behaviour, and at the extreme, rape (Apostolou, 2013).

1.2.2 Manifestation

Affect- or goal-driven aggression manifests both verbally and physically, as a function of context, sex, and individual differences (Buss & Perry, 1992). This section delineates overt/covert and direct/indirect manifestations of aggression.

1.2.2.1 Overt/Covert Aggression

The overt/covert dichotomy of aggression was coined in Bendig (1962), and later discussed in Loeber & Schmaling (1985). Covert aggressors are predominantly female (Campbell, 1999), and seek to manipulate social hierarchy through charm, guilt, and rumormongering (Björkqvist, Österman, & Kaukiainen, 1992; Crick & Grotpeter, 1995, *see columns A and B of Table 1 for examples*). Typically, covert behaviours are used in an attempt to aggress unnoticed, whilst decreasing the likelihood of counterattack (Björkqvist et al., 1992; Ramirez & Andreu, 2006).

In contrast, overt aggressors use outward, face-to-face aggression against individuals or objects with the intention of causing harm or damage (Baron & Richardson, 1994; Bendig, 1962; Berkowitz, 1993). Overt aggression comprises verbal or physical responses (Alderman, Knight, & Morgan, 1997; Bendig, 1962; Kassinove & Sukhodolsky, 1995; *see cell C1 of Table 1*), and is closely associated with direct aggression (Garandeau & Cillessen, 2006). Though overt aggression might manifest indirectly, such as through spreading characterdamaging rumours or secret telling, it is likely to reveal the aggressor's involvement and so might be considered detrimental.

1.2.2.2 Direct/Indirect Aggression

Covert aggression has been associated with indirect aggression (Coyne, Archer, & Eslea, 2004; Garandeau & Cillessen, 2006) whereby damage is caused through the destruction of resources and the elicitation of negative reactions/disapproval of others (Archer, 2001; Berkowitz, 1993; Buss, 1961; Richardson & Green, 2003). For example, negative discourse about an individual to their peers might result in shunning and subsequent harm to their social relationships. Indirect aggression develops during later stages of development and carries only minimal personal risk (Björkqvist et al., 1992; Björkqvist, Österman, & Lagerspetz, 1994). In contrast, direct aggression refers to actual aggression directed towards a target. As shown in Table 1, direct aggression may be threatened through a reactive manner in response to perceived provocation (*see cell C1*), or through an instrumental manner in order to obtain resources (*see cell C2*).

1.3 Stages of a Reactive Aggressive Encounter

The following section focuses discussion on reactive aggression, and delineates key cognitive theories that might underpin the stages of a reactive aggressive response. First, processes associated with the perception of threat will be outlined. As noted in section 1.2.1.1 (page 2), threat in humans is commonly conveyed through angry facial expressions and so this overview will be restricted to the perception of facial affect, as opposed to other situational (e.g. environment, presence of a weapon) or interpersonal (e.g. voice, body posture) manifestations of threat. Subsequently, cognitive processes associated with threat- and anger-induced negative affect will be discussed along with decision making processes that might moderate the manifestation, or inhibition of aggressive behaviour. Finally, theory thought to underpin the activation of the VIM will be delineated. In particular, how the perception of social cues, specifically facial distress, might engage an empathic response resulting in the halting of aggressive behaviour.

1.3.1 Face Processing

Humans explore the world and attend to environmental stimuli in both a conscious and unconscious manner (Ögmen & Breitmeyer, 2006). Faces are *special* to human perceivers as they capture attention, convey intention, and facilitate the formation of relationships (Apicella, Sicca, Federico, Campatelli, & Muratori, 2013; Meaux, Roux, & Batty, 2014; Stenberg, Wiking, & Dahl, 1998; Vuilleumier, 2000). Efficient face processing is required to rapidly extract key social information such as sex, ethnic origin, gaze direction, and emotional state (Fei-Fei, Iyer, Koch, & Perona, 2007).

Facial expressions impart emotionally salient information (Buck, 1984; Darwin, 1872; Izard, 2007) and are critical for adaptive functioning and the avoidance of

adverse experiences (Anderson, Christoff, Panitz, de Rosa, & Gabrieli, 2003; Hartley & Phelps, 2009; Hinojosa, Mercado, & Carretié, 2015). Humans have evolved a core set of primary emotions, namely fear, anger, sadness, disgust, happiness, and surprise (Ekman, 1993) and are able to decode and understanding these emotions by adulthood (Calder et al., 1996). Following the initial structural processing of faces, facial identities and expressions are thought to process along two parallel, yet independent pathways (Bruce & Young, 1986). Originally conceptualised by the social cognitive informationprocessing model (Crick & Dodge, 1994), emotion processing is thought to develop through socialisation, whereby information relating to the face is reinforced by social knowledge (Cusi, Nazarov, Holshausen, Macqueen, & McKinnon, 2012; Ekman, 1999; Green et al., 2008).

In particular, angry facial expressions denoting threat demand the rapid reallocation of attention and are rated more arousing than neutral faces (Blanchette, 2006; Fox, Griggs, & Mouchlianitis, 2007; Nummenmaa, Hyönä, & Calvo, 2010; Öhman et al., 2001; Pourtois et al., 2004; Sawada, Sato, Uono, Kochiyama, & Toichi, 2014; Vuilleumier & Schwartz, 2001). From an evolutionary perspective, a propensity to orient attention to threat benefits harm avoidance and protection of social stake (Holmes, Mogg, de Fockert, Nielsen, & Bradley, 2014; LeDoux, 1996; Öhman, Flykt, & Lundqvist, 2000). This preferential response, the threat-superiority effect (Gilbert, 2005; Gilbert, Price, & Allan, 1995), can be thought of as a self-protection system; alerting organisms to potential threat or danger. Here, over-, as opposed to underreaction, is favourable for survival (Anderson & Carnagey, 2004; Gilbert, 2005). In community samples, angry, relative to happy and neutral facial expressions are processed faster, with fewer errors, when presented both individually or among a crowd of faces (Fox et al., 2000; Hansen & Hansen, 1988; Horstmann & Bauland, 2006; Pinkham, Griffin, Baron, Sasson, & Gur, 2010; Sawada et al., 2014; Williams, McGlone, Abbott, & Mattingley, 2005).

Perception of facial threat is moderated by aggressive traits. Although global face processing deficits have been reported in a moderately large cohort of incarcerated, relative to non-incarcerated male adolescents during a forced-choice paradigm (McCown, Johnson, & Austin, 1986), threat-specific face processing errors have been reported elsewhere (Best et al., 2002; Sato, Uono, Matsuura, & Toichi, 2009). In Sato et al. (2009), incarcerated, relative to non-incarcerated males were more likely to misattribute facial disgust as anger, even when presented with several response options. Similarly, patients with, relative to those without, intermittent explosive disorder (IED), a disorder characterised by recurrent acts of reactive aggression disproportionate to actual provocation (Coccaro, 2003), evidenced anger processing deficits and a greater likelihood of miscategorising neutral facial expressions as being negatively valenced (Best et al., 2002).

Taken together, angry facial expressions are considered a key vehicle in conveying threat, and demand the rapid reallocation of neuronal resources as a function of their social and biological importance. This importance is exemplified by response biases towards threat in typically developing humans. However, in individuals characterised by aggressive traits, this threat perception bias can present atypically – possibly impacting wider face processing.

1.3.2 Generation of Negative Affect

Aggression is often preceded by anger, a negative affective state that serves to index the propensity to aggress (Buss & Warren, 2000). Anger can be shortlived, or persistent, and is thought to play a role in mood disorders, anxiety, and reactive aggression (Wilkowski & Robinson, 2010). Furthermore, anger can be elicited to environmental triggers, memories, and rumination (Berkowitz, 1993; Eckhardt & Deffenbacher, 1995) and is known to disrupt communication, cognition, inhibition, and physiological change (Berkowitz, 1993; Berkowitz & Harmon-Jones, 2004; Demenescu, Kortekaas, den Boer, & Aleman, 2010; Kassinove & Sukhodolsky, 1995; Melcher, Born, & Gruber, 2011; Pillutla & Murnighan, 1996; Tsytsarev & Grodnitzky, 1995). Over an extended period of time, anger can be detrimental to health and social relationships (Baron et al., 2007; Bettencourt, Talley, Benjamin, & Valentine, 2006; Shorey et al., 2011).

Cumulative effects might also affect anger. Physiological arousal associated with anger dissipates slowly, allowing anger to last over long periods of time. As suggested in the excitation transfer theory (Zillmann, 1983), stimuli presented shortly after the induction of an anger state might be misattributed as hostile and further disrupt cognition. In Cohen, Eckhardt, & Schagat (1998), following the induction of anger in the form of a participant-directed insult, individuals took longer to respond to target words when surrounded by non-target anger-related distractors, suggesting an attentional bias towards angry stimuli. This effect was augmented in individuals with high trait anger. In contrast, using a different, yet qualitatively similar anger induction paradigm, Eckhardt & Cohen (1997) did not observe a bias towards anger-related words. However, although this investigation controlled for word-specific factors (e.g. length), it did not quantify state anger prior to anger induction, and so cannot conclusively state that anger was induced.

Consolidating several cognitive theories of aggression, the general aggression model (GAM) outlines how social stimuli in the form of personal (e.g. personality traits, attitudes, biological predispositions) and situational (e.g. environmental triggers, social cues) inputs might moderate affect, arousal, and cognition (Anderson & Bushman, 2002). On a cognitive level, aggressive thoughts, feelings, emotions, and behaviours are thought to interconnect – forming a composite knowledge structure associated with aggression (Berkowitz, 1993; Guerra & Huesmann, 2004). This structure, elaborated on in section 1.3.3 (page 11), is thought to respond to social cues, resulting in the retrieval of anger-/aggression-related information and in turn sensations of anger (Anderson & Carnagey, 2004). A greater yield of aggressive memories, thoughts, and feelings might be gained as a function of aggressive personality traits, increased attention, and stimulus salience (Berkowitz, 1993; Guerra & Huesmann, 2004)

1.3.3 Executive Control

Through the use of executive functioning, humans are typically able to divert their attention away from task-irrelevant stimuli and regulate their affective feelings in order to inhibit an aggressive response (Chen, Muggleton, & Chang, 2014; Davidson, Putnam, & Larson, 2000; Holmes et al., 2014; Lewis et al., 2008; Raaijmakers et al., 2008; Schachar & Logan, 1990; Vilà-Balló, Hdez-Lafuente, Rostan, Cunillera, & Rodriguez-Fornells, 2014). Prior to the execution of a behavioural response, behavioural scripts, defined as an expected series of behaviours for a given situation, are consciously evaluated against one another (Anderson & Carnagey, 2004). As outlined in the social-cognitive processing model (SCP; Huesmann, 1998), behavioural scripts are acquired through direct and indirect social learning, reinforcement, and punishment. Disruption to this learning process, possibly as a function of social exclusion and/or the absence of positive role models (Maughan & Rutter, 2001), is likely to increase the risk of antisocial and aggressive behaviour (DeWall, Anderson, & Bushman, 2011).

Pro-aggressive scripts (i.e. scripts that promote the use of aggression) are formed, and subsequently reinforced, when aggression is observed to benefit the acquisition of resources, sexual mates, and social status, or as a successful means of defence. An increased availability of pro-aggressive scripts has been inversely associated with cognitive resources (Finkel, DeWall, Slotter, Oaten, & Foshee, 2009), and likely facilitates incidents of aggression (Lochman & Dodge, 1994). Importantly, access to pro-aggressive scripts does not necessarily determine whether aggressive behaviour will ensue. A well-developed cognitive system generates both pro- and non-aggressive responses, and applies them appropriately depending on the given context (Guerra & Huesmann, 2004). As discussed in section 1.3.2 (page 9), the ability to suppress aggressive behaviour is vital for long-term organism survival (Guerra & Huesmann, 2004).

When a response is not instantly required, these aforementioned scripts are considered, evaluated, and re-appraised in accordance with normative beliefs and self-schemas (Crick & Dodge, 1994). Whilst normative beliefs encapsulate

beliefs about what is socially acceptable in the given context (Anderson & Carnagey, 2004; DeWall et al., 2011), self-schemas describe how an individual wants to be perceived (Guerra & Huesmann, 2004). Although limited to self-reports in the absence of behavioural measures, increased physical and verbal inter-partner aggression has been associated with positive attitudes towards inter-partner violence in students (Fincham, Cui, Braithwaite, & Pasley, 2008).

Behavioural responses are further filtered through the assessment of both the consequences of, and actual ability to, enact a certain script (Anderson & Huesmann, 2003). Although the benefit of using aggression as an effective strategy for warding off the advances of others has been discussed in section 1.2.1.2 (page 3), aggression may be personally detrimental. For example, aggressing against a dangerous opponent may lead to repercussions (DiGiuseppe, 1995). Conceptualised within the frustration-anger-displacement theory (Dollard, Doob, Miller, Mowrer, & Sears, 1939), frustration is displaced elsewhere (i.e. an object or another) when it is considered detrimental to aggress against the original source of frustration. However, this theory fails to acknowledge why, and indeed how, subsequent targets are selected.

1.3.4 Inhibition by Cues of Distress

Typically, aggressive behaviour results in a social response, such as facial distress (e.g. fear, sadness). This section explores how the facial distress of another might evoke an inhibitory response. Previously, animal models have considered the role of distress and submission in relation to aggression inhibition. An innate aggression control mechanism was proposed in Eibl-Eibesfeldt (1970) and Lorenz (1966, 1981), suggesting same-species aggression might terminate following victim submission. A similar mechanism, the VIM (Blair, 1995; Blair et al., 1995, 1999), may exist in humans.

The VIM assumes socially acceptable behaviours, which allow individuals to function within society, are formed through instrumental learning and aversive conditioning (Blair, 2003). Over time, positive reactions help forge stimulus-

reward associations with a given behaviour, resulting in increased use of said behaviour (Baxter & Murray, 2002). Conversely, negative reactions may make individuals '*feel bad*' and so are likely to result in behaviours being used less frequently (Blair, 1995). When negative reactions succeed antisocial behaviours, individuals begin to learn moral (i.e. victim-based) and conventional (i.e. social order-based) transgressions (Blair, 2006). Negative reactions to moral transgression create stimulus-punishment associations (Baxter & Murray, 2002), reducing the likelihood of said behaviours being used in the future for fear of negative consequences (Blair et al., 2006; Raine, 2013). However, if bad behaviour goes unpunished, or is rewarded (e.g. through material gain), behaviour attenuation is not effectively learned and might even override previously forged, negative associations (Blair, 1995; Mineka, Davidson, Cook, & Keir, 1984).

In particular, the VIM is thought to respond to the perception of distress, including tears (Blair, 1993) and fearful and sad facial or vocal expressions (Blair et al., 1997; Decety & Chaminade, 2003). In contrast to angry faces, which convey direct threat, facial distress conveys ambiguous threat and the need for avoidance (Blair, 1995; Ewbank et al., 2009; Mineka & Cook, 1993). Distress likely inhibits aggression by activating cognitive withdrawal schemas (Blair, 1993; Nichols, 2001), which result in individuals learning to avoid behaviour that evokes pain and distress in others (Blair, 1993, 1995, 2004; Blair et al., 1999). Distress cues stabilise social interactions among healthy functioning individuals (Marsh & Ambady, 2007), with children as young as four showing reduced stealing behaviour after associating said behaviours with sad facial responses (Camras, 1977).

Salient negative social cues form a malleable database of VIM triggers capable of guiding behaviour across contexts (Blair, 1995, 2003; Blair et al., 1997). Poor responses to distress-related social cues may result in a failure to associate distress with negative affect (Blair, 2001, 2003, 2005) and subsequent failure to trigger the VIM (Blair, 2005). As evidenced in both children and adults, deficits

in distress recognition have been associated with inappropriate response choices, conduct problems, and violent behaviour (Marsh & Blair, 2008; Perry et al., 2001).

Distress-induced behavioural inhibition is likely underpinned by empathy (Decety & Jackson, 2004; Eisenberg, 2000; Marsh & Ambady, 2007). Empathy is a multidimensional phenomenon defined as an ability to understand the feelings and emotional states of others in reference to the self (Decety & Jackson, 2004; Decety & Moriguchi, 2007; Hoffman, 2000; Preston & de Waal, 2002). Empathy is thought to develop through socialisation and moral transgressions (Morton & Frith, 1993).

Empathy comprises both affective and cognitive components (Blair, 2013; Carr & Lutjemeier, 2005; Davis, 1983). Sometimes referred to as empathic concern (Davis, 1983), affective empathy is the vicarious emotional experience of the internal emotional state of another (Bryant, 1982) coupled with the ability to verbally describe said state (Blair, 2013). Affective empathy is said to emerge from the recognition and representation of emotion (Blair, 1995; Feshbach & Feshbach, 1982; Hoffman, 1987). The ability to identify and respond to an emotional state is a vital aspect of everyday interaction (Siedel et al., 2013). In contrast, cognitive empathy, sometimes referred to as perspective taking (Batson, Fultz, & Schoenrade, 1987; Davis, 1983), is the ability to recognise and comprehend the feelings of others as being separate from one's own (Decety, 2011; Decety & Jackson, 2004; Decety, Michalska, Akitsuki, & Lahey, 2009).

Empathic dysfunction is considered detrimental to the VIM because of its associations with deficient processing of fearful faces (Carr & Lutjemeier, 2005). Self-reported cognitive and affective empathic responses to the suffering of others are inversely related to physical aggression. Increased empathy has been associated with pro-social behaviour and moral reasoning (Batson, 1991; Decety & Meyer, 2008), and decreased indirect aggression, verbal aggression,

14

assault, and irritability (Eisenberg, 2000; Richardson, Hammock, Smith, Gardner, & Signo, 1994). However, this association may only last under moderate, but not intense provocation (Blair, 2005; Davidson et al., 2000; Frick et al., 2003; Miller & Eisenberg, 1988; Zahn-Waxler, Cole, Welsh, & Fox, 1995). Under intense provocation, cognitive inhibitors become ineffective as a result of acute impairment of cognitive functioning (Zillmann, 1988).

The VIM is dysfunctional in psychopathy (Blair, 1995, 2001; Blair et al., 2004), a developmental disorder first described in Cleckley (1941, 1976). Psychopathy is thought to occur in less than 1% of the general population (Hare, 2003), and worsens with age (Kubak & Salekin, 2009). Alongside machiavellianism and narcissism, psychopathy forms one of the components of the Dark Triad (Paulhus & Williams, 2002), and is characterised by a constellation of affective, interpersonal, and behavioural malfunction (Cleckley, 1976; Frick, O'Brien, Wootton, & McBurnett, 1994; Hare, 2003; Hare & Neumann, 2006; Harpur, Hare, & Hakstian, 1989). Specifically, the affective component of psychopathy comprises a lack of guilt, low empathy, callousness, and unemotional and uncaring traits (Hare, 1991, 2003), with the interpersonal component encapsulating manipulative behaviours, such as lying, cheating, and superficial charm (Harpending & Sobus, 1987; Raine, 1993). The behavioural component manifests through impulsive actions and sensation seeking with disregard for the completion of long-term goals (Barry et al., 2000; Dawel, O'Kearney, McKone, & Palermo, 2012; Hart & Dempster, 1997).

The affective dimension of psychopathy is encapsulated by CU traits, and is prominent in most conceptualisations of psychopathy (Cleckley, 1976; Hare, 1993). Although CU traits are frequently used as a proxy measure of psychopathy (Dawel et al., 2012), with their presence in childhood predictive of psychopathic development and poor treatment outcomes (Frick, Ray, Thornton, & Kahn, 2014; Frick & White, 2008), CU traits have only recently been accepted as a diagnostic criterion of antisocial behaviour disorders, such as conduct disorder (CD) (American Psychological Association, 2013; Waller, Gardner, & Hyde, 2013).

CU traits are associated with behavioural aspects of psychopathy. Neatly summarised by a meta-analysis of 24 studies sampling children and adolescents, CU traits predict future delinquency and aggression severity (Frick & Dickens, 2006). Specifically, presence of CU traits during childhood and adolescence positively correlates with future incidents of physical aggression (Coid et al., 2009; Frick et al., 2003; Krischer & Sevecke, 2008; Loney, Frick, Clements, Ellis, & Kerlin, 2003; Woodworth & Porter, 2002), violent sexual offending (Caputo, Frick, & Brodsky, 1999), and long-lasting patterns of antisocial behaviour (Essau et al., 2006; Frick & White, 2008).

Although psychopathic populations frequently aggress in an instrumental manner (Cornell et al., 1996; Meyer-Lindenberg et al., 2006), they also engage in reactive aggressive behaviour (Frick et al., 2003). Such aggression and antisocial behaviour manifests more diversely, and of a greater severity, than presented in non-CU counterparts (Christian, Frick, Hill, Tyler, & Frazer, 1997; Frick et al., 2003; Kruh, Frick, & Clements, 2005; Lynam, 1997). However, some findings (e.g. Kruh et al., 2005) are based on self-report assessments, which are inherently open to inflation, and as restricted to male samples in most cases, should be discussed with caution.

Psychopaths, and children evidencing psychopathic-like traits, are thought to have difficulty understanding the emotional significance, but not the meaning, of emotion (Hervé, Hayes, & Hare, 2003), and evidence both a decreased sensitivity to punishment (Fisher & Blair, 1998) and a lack of victim-based moral reasoning (Blair, 2003; Montagne et al., 2005). Such deficits are likely facilitated by an inability to [1] process stimuli, [2] form cue associations, and [3] condition an emotional response (Flor, Birbaumer, Hermann, Ziegler, & Patrick, 2002). Specifically, psychopaths express insensitivity (Blair et al., 2004) and low physiological responses to basic emotions (Verona, Patrick, Curtin, Bradley, & Lang, 2004).

Although a general face processing deficit spanning sadness, fear, disgust, and anger has previously been reported in violent, relative to non-violent offenders and controls (Robinson et al., 2012; Seidel et al., 2013), two meta-analyses support distress-specific face processing deficits in individuals characterised by antisocial behaviour (e.g. psychopaths, criminals, delinquents, and individuals displaying high trait aggression) (Marsh & Blair, 2008; Wilson, Juodis, & Porter, 2011). Distress recognition deficits extend to children with psychopathic-like traits who also exhibit deficits in processing body distress (Blair, Colledge, Murray, & Mitchell, 2001; Dadds et al., 2006; Stevens, Charman, & Blair, 2001). Recognition of body distress is also poor in adult psychopaths (Blair et al., 1997). Whilst a key analysis in the distress processing literature, Wilson et al. (2011) also offer some evidence for an association between antisocial behaviour and deficits in processing non-distress emotions, though this could be an artefact of the use of lenient alpha levels (.10). During aggressive encounters, deficient processing of distress may lead to a prolonged, uninhibited aggressive response (Marsh & Ambady, 2007).

Processing of fearful and sad faces is thought to share similar neurobiological underpinnings (Adolphs & Tranel, 2004, *see chapter two, section 2.5 [page 29]*). Although deficits in sadness recognition have been reported elsewhere (Dolan & Fullam, 2006; Eisenbarth, Alpers, Segrè, Calogero, & Angrilli, 2008; Fairchild, van Goozen, Calder, Stollery, & Goodyer, 2009, 2010; Fullam & Dolan, 2006; Hastings et al., 2008; Woodworth & Waschbusch, 2008), CU traits have been predominantly associated with fear recognition deficits both visually (Blair, 2000; Blair et al., 2004; Dadds et al., 2008; Fairchild et al., 2009; Iria & Barbosa, 2009; Leist & Dadds, 2009; Muñoz, 2009) and vocally (Stevens et al., 2001). Whilst poor fear recognition has been attributed to ambiguity of facial fear as a whole (Elfenbein & Ambady, 2002; Rapcsak et al., 2000; Russell, 1994; Stockdale, Morrison, Kmiecik, Garbarino, & Silton, 2015), intact

17

recognition of facial fear in healthy controls suggests a specific association with CU traits (Blair et al., 2004).

Deficits in distress recognition do not always coincide with the presence of CU traits. Psychopaths and individuals with psychopathic traits have shown an intact ability to process verbal and facial fear, alongside anger and happiness (Eisenbarth et al., 2008; Fairchild et al., 2010; Glass & Newman, 2006; Hansen, Farivar, Thompson, & Hess, 2008; Hastings et al., 2008; Hiatt, Lorenz, & Newman, 2002; Kosson, Suchy, Mayer, & Libby, 2002). In some cases, psychopathy has been associated with *increased* fear recognition (Del Gaizo & Falkenbach, 2008), with similar findings reported at trend level in children with psychopathic traits (Woodworth & Waschbusch, 2008). However, these results might reflect variation in methodology, including stimuli used, frequency thereof, and measures recorded (*see Glass & Newman, 2006*).

An attempt to understand the disparity of results relating to distress processing was undertaken in Dawel et al. (2012). Impairments in processing positive and negative emotions were found across both facial and vocal modalities. Adults, children, and adolescents exhibited emotion recognition deficits for fear, happiness, sadness, and surprise, but not anger or disgust. Whilst previous meta-analyses considered psychopathy as a united construct (Marsh & Blair, 2008; Wilson et al., 2011), Dawel et al. separated out processing impairments associated with the affective factor of psychopathy only; implicating poor distress, but not threat, recognition in association with CU traits.

There have been mixed results when investigating associations between CU traits and empathy. Although inversely related to CU traits in some child samples (Anastassiou-Hadjicharalambous & Warden, 2008; Dadds et al., 2009; Pasalich, Dadds, & Hawes, 2014), cognitive empathy has been shown to remain intact in adults with psychopathy, and children and adolescents with psychopathic-like traits (Hare, 2003; Jones, Laurens, Herba, Barker, & Viding, 2010; Richell et al., 2005; Schwenck et al., 2012). Conversely, an inverse

association between CU traits and affective empathy has been evidenced in adults (Seidel et al., 2013), as well as children and adolescents (Anastassiou-Hadjicharalambous & Warden, 2008; Dadds et al., 2009; Jones et al., 2010; Pasalich, 2014; Schwenck et al., 2012). Although the results reported in Pasalich et al. were based on parent reports, the stability of this finding across samples suggests this is a reliable measure. Taken together, whilst individuals characterised by CU traits display atypical responses to the emotion of others, their overall cognitive understanding of emotion, in most cases, remains intact (Hare, 2003).

The inability to experience and generate an empathic response to distress likely impedes avoidance learning and subsequent halting behaviours (Bird & Viding, 2014; Blair, 2004). An absence of inhibitory factors might facilitate prolonged aggression (Blair, Morton, Leonard, & Blair, 2006; Hare, 1991, 2003; Richardson et al., 1994). Poor empathy may account for the greater number, and severity of crimes committed by psychopathic, relative to non-psychopathic individuals (Frick et al., 2003; Hare, 2003; Marshall et al., 1995)

1.3.5 Behaviour Modulation

Activation of the VIM is thought to result in the suppression of a pre-initiated motor response (Hughes, Fulham, Johnston, & Michie, 2012). Whilst individuals may not always halt their motor behaviour, successful stopping relies on cognitive control, defined as the stopping or overriding of a mental process with or without intention (MacLeod, 2007; Miyake et al., 2000). Cognitive control represents a cluster of processes underpinning the adaption and planning of actions within an ever-changing social context (Enriquez-Geppert et al., 2010; Oldenburg et al., 2012). On a cognitive level, inhibitory control is modelled using the horse race model (HRM; Logan, 1994), whereby independent go and nogo responses are thought to compete prior to execution (Logan & Cowan, 1984).

Greater motor inhibition is required to suppress a motor response when imminent motor inhibition is required. During target detection tasks where
individuals are asked to respond to target stimuli, but inhibit a response to similar non-target stimuli, motor responses are inhibited prior to elicitation. Alternatively, stop-signal (SST) and stop-go paradigms (Lappin & Eriksen, 1966; Logan & Cowan, 1984), which require motor suppression of a pre-potent, or pre-initiated motor response (Aron, 2007; Bekker, Kenemans, Hoeksma, Talsma, & Verbaten, 2005), require greater inhibition due to the increased cognitive load required to successfully suppress a response (Johnstone et al., 2007; Rubia, Smith, Brammer, & Taylor, 2003). Whilst target detection paradigms require automatic action, stop paradigms require greater cognitive control (Verbruggen & Logan, 2008). Successful stopping is referenced by the HRM (Logan, 1994) and reflects the outcome of competition between independent go and stop operations (Boucher, Palmeri, Logan, & Schall, 2007).

The ability to suppress a planned or on going motor response as a function of environmental stimuli is a prerequisite for socially adaptive behaviour (Huster, Pilis, Lavallee, Calhoun, & Hermann, 2014; Oldenburg et al., 2012). Stopping times are shortened by impulsive traits (Marsh et al., 2002) and emotional imagery (Pessoa, Padmala, Kenzer, & Bauer, 2012). In relation to the VIM, there is currently a dearth of investigation into behavioural extinction sampling from aggressive individuals or using distress stimuli as a stopping agent. In a single study where individuals were asked to classify, but not respond to, angry and neutral facial expressions, individuals characterised by high self-reported aggressive traits exhibited longer stopping times than their low aggressive trait counterparts (Pawliczek, Derntl, Kellermann, Gur, & Schneider, 2013). Although facial affect was incorporated into this paradigm, it was not used as a stopping stimulus. Instead, a stop response was indicated via a change of colour to the image frame. As aggressive and CU traits are associated with poorer distress processing, it is likely further detriments would be observed if facial affect, particularly distress, signalled the need to stop.

1.4 Chapter Summary

This chapter delineated motivations underpinning, and manifestations of aggression in humans, before tailoring discussion to how various stages of a reactive aggressive response might be explained through cognitive theory. In particular, facial anger was identified as core means of conveying threat in humans because of its utility to capture attention and facilitate a negative state of affect. Regulation of this affective state was described by executive functioning, whereby individuals are typically able to select an appropriate behavioural response for a given situation, based on prior learning and contextual information. Moreover, increased attention to threat, and executive functioning deficits are evidenced in samples characterised by aggressive traits. The latter part of the chapter attended to the VIM, and how facial distress, through generation of an empathic response, might allow individuals to inhibit antisocial and aggressive responses in order to better function within society. Chapter two delineates the known neurobiological underpins of facial affect processing and distress-induced motor extinction, and how these networks might show dysfunction in samples characterised by aggressive and CU traits.

Chapter Two: Neurobiology of reactive aggression

2.1 Introduction

In chapter one, facial expressions were described as vehicles for both reactive aggressive responses to threat (e.g. angry faces) and inhibitory responses to distress (e.g. fearful and sad faces) (Ögmen & Breitmeyer, 2006). Chapter two builds on the cognitive perspective of aggression described in chapter one, through delineation of key neurobiological networks thought to be involved in the generation/inhibition of an aggressive response. Cortical networks of face processing are outlined, before discussing how facial threat and distress, in particular, might process along fast subcortical routes as a function of their biological significance (Amaral, Price, Pitknen, & Carmichael, 1992). Moreover, this chapter describes how anger is generated through the activation of core stress systems, and how distress might engage frontal-limbic networks involved in executive functioning, behavioural inhibition, and motor suppression.

2.2 Face Processing

In humans, faces evoke an immediate, and automatic process consisting of visuospatial and attentional networks - comprising both cortical (i.e. dorsolateral prefrontal, anterior and posterior cingulate, inferior/middle occipital gyri, fusiform gyrus, insula, primary and secondary visual cortices) and subcortical (i.e. thalamus, basal ganglia, pulvinar, putamen, amygdala, hippocampus) brain regions (Palermo & Rhodes, 2007; Vuilleumier & Driver, 2007, *but see Fusar-Poli et al., 2009 for a review*).

Initially, retinal information is transferred to the lateral geniculate nucleus, a core relay centre in the thalamus (Fox, 2008; Morris, Öhman, & Dolan, 1999), before being projected to the occipital (i.e. primary and secondary visual cortices, inferior/middle occipital and lingual gyri) and temporal (i.e. fusiform, superior temporal) lobes (Adolphs, 2002; Gauthier et al., 2000; Haxby, Hoffman, & Gobbini, 2000; Vuilleumier & Driver, 2007). At occipital sites, the inferior

occipital gyrus houses the occipital face area (OFA), a region shown to represent facial aspects such as the eyes and mouth (Nichols, Betts, & Wilson, 2010; Pitcher et al., 2007). The OFA shares connections with the middle occipital gyrus (Renier et al., 2010) and posterior parietal cortex (Potts & Tucker, 2001) where spatial information is processed. Renier et al. reported increased middle occipital gyrus activation - as measured by functional magnetic resonance imaging (fMRI) - in participants tasked with comparing the location of geometric shapes between trials. However, as this region also responded to tactile and auditory stimuli, it is not limited to visual processing.

In the temporal lobe, though responsive to non-facial stimuli such as houses (Haxby et al., 2001), activation of the fusiform gyrus is maximal to faces (Apicella et al., 2013; Vuilleumier et al., 2001). The fusiform gyrus is functionally and anatomically connected to the superior temporal sulcus (STS), where facial detail is extracted (Haxby et al., 2000; Haxby, Hoffman, & Gobbini, 2002; Puce & Perrett, 2003) in order to process social relevance and intent (Adolphs, 2002; Allison, Puce, & McCarthy, 2000; Nummenmaa, Hyönä, & Calvo, 2010). Attenuated activation of the fusiform-STS pathway is associated with facial affect recognition deficits (Sato, Toichi, Uono, & Kochiyama, 2012; Schultz et al., 2003). However, as fusiform gyrus volume is unchanged (Pierce et al., 2001) or even enlarged (Waiter et al., 2004) in autistic samples characterised by poor affect recognition, disruption to this pathway might be driven by a poorly developed STS (Boddaert et al., 2004).

The fusiform gyrus and STS share direct (Amaral et al., 1992; Fairhall & Ishai, 2007; Puce & Perrett, 2003; Pujol et al., 2009), and indirect (Petrides & Pandya, 2007), connections with the amygdala. The amygdala is present in both mammalian and non-mammalian creatures (e.g. reptiles, birds, fish) (Gross, 2007; Haxby et al., 2000; Janak & Tye, 2015), and in humans comprises three interconnected nuclei thought to play a core role in emotion processing (Fox, Oler, Tromp, Fudge, & Kalin, 2015; Sah, Faber, Lopez, & Power, 2003). During face processing, the amygdala encodes emotionally valenced facial features

(Adolphs, 2010; Fitzgerald et al., 2006; Sergerie, Chochol, & Armony, 2008; Winston, Gottfried, Kilner, & Dolan, 2005; Zald, 2003; Zink et al., 2008) and feeds this information back into the fusiform-STS pathway (Allison et al., 2000; Pessoa & Adolphs, 2010; Sabatinelli, Bradley, Fitzsimmons, & Lang, 2005; Vuilleumier, Richardson, Armony, Driver, & Dolan, 2004). This feedback mechanism serves to both increase the cellular response to facial affect and aid the resolution of visual ambiguity (Adolphs, 2001). Amygdala activation likely explains increased fusiform-STS connectivity when viewing high intensity/emotional, relative to static/neutral, facial stimuli (Carr, Iacoboni, Dubeau, Mazziotta, & Lenzi, 2003; Fusar-Poli et al., 2009; Winston et al., 2003a).

2.3 Facial Threat

Angry facial expressions denote threat and signal provocation (Blair, 2003; Blair et al., 1999; *but see chapter one, section 1.3.1 [page 7]*). On a neurobiological level, threat-relevant information is transferred from the thalamus to the amygdala, along a fast, yet coarse subcortical pathway (Amaral et al., 1992; LeDoux, 1996; Öhman & Mineka, 2001; Pontius, 2002; Vuilleumier, Armony, Driver, & Dolan, 2003). This pathway, comprising the superior colliculus and pulvinar, has been shown to activate in response to angry facial stimuli masked in order to inhibit cortical processing (Morris et al., 1999). Furthermore, case studies of patients with bilateral (Adolphs, Tranel, Damasio, & Damasio, 1994, 1995; Calder et al., 1996), but not unilateral (Adolphs et al., 1995) amygdala lesions have reported impaired recognition of facial and vocal expressions of anger. However, whilst this network has evolutionarily benefit for harm avoidance (Helmut, 2003; Vuilleumier et al., 2003), these patients also evidenced impaired fear processing, and so it is not threat-specific.

Being able to appropriately respond to threat is essential for everyday social interaction (*see chapter one, section 1.3.3 [page 9]*). Atypical amygdala structure and activation has been evidenced in populations characterised by aggressive traits. In Matthies et al. (2012), bilateral amygdala volume was

smaller in individuals with high, relative to low, lifetime histories of aggression. This effect survived after controlling for anxiety and depression; disorders related to reduced amygdala volume (Hamilton, Siemer, & Gotlib, 2008). Similarly, relative to non-clinical controls, smaller amygdala volumes were observed in male (Soloff, Nutche, Goradia, & Diwadkar, 2008) and female (Rüsch et al., 2003; Soloff et al., 2008) patients with borderline personality disorder (BPD), a disorder characterised by intense, anger-driven behaviour (American Psychological Association, 2013). Furthermore, bilateral amygdala volume was inversely associated with current and childhood aggressive traits in individuals characterised by a range of aggression-related psychopathologies (Pardini, Raine, Erickson, & Loeber, 2014). Moreover, after correcting for baseline aggression, low amygdala volume in this sample was predictive of aggressive incidents over a three-year follow up period.

Elsewhere, relative to controls, males with IED (Coccaro, McCloskey, Fitzgerald, & Phan, 2007; Coccaro, Sripada, Yanowitch, & Phan, 2011) and BPD (Donegan et al., 2003; Thomas et al., 2013) have evidenced increased amygdala activation to angry facial expressions. However, these investigations are not without limitations. Amygdala activation in Coccaro et al. (2007) was calculated by contrasting haemodynamic responses to angry facial expressions with those evoked during rest periods - not to neutral facial expressions. Whilst amygdala responses to angry faces were positively correlated with lifetime histories of aggression, group contrasts might have been augmented as a function of not subtracting the residual amygdala activation to faces during the analysis. Furthermore, in the BPD samples, amygdala hyperactivation was also reported to positive, negative, and neutral expressions (Donegan et al., 2003) as well as fearful and neutral faces (Thomas et al., 2013), suggesting amygdala hyperactivity is not anger-specific in BPD.

Mixed findings have been observed in samples characterised by CU traits. Though not threat-specific, psychopathic offenders, relative to nonpsychopathic offenders and non-offender controls, have evidenced attenuated left amygdala activity during an emotional word memory task (Kiehl et al., 2001). Conversely, Müller et al. (2003) reported *increased* right hemispheric amygdala activity to negative images in a small cohort of psychopathic offenders, relative to non-offender controls. Although Müller et al. argue increased activation might reflect task modality, left hemispheric amygdala hypoactivity to negative images has been observed in adolescent males with CD, relative to age-matched controls, using a comparable paradigm (Sterzer, Stadler, Krebs, Kleinschmidt, & Poustka, 2005). It should be noted that none of these investigations used facial threat as stimuli.

Together, these results suggest that aggressive and CU traits might differentially associate with amygdala reactivity to threat. Whilst amygdala hyperactivity might contribute to an over generation of negative affect, amygdala hypoactivity suggests impaired processing of threat-relevant social cues (Davidson et al., 2000).

2.4 HPA-axis

Facial threat contributes to negative affective states, such as anger (*see chapter one, section 1.3.2 [page 9]*). On a neurobiological level, anger is generated as a function of threat-relevant information being projected from the amygdala to the hypothalamus, pituitary gland, and adrenal axis (HPA-axis). Within this network, threat information projects via the stria terminalis, which activates as a function of threat intensity (Somerville, Whalen, & Kelley, 2010).

Though not fully understood in humans (Toth et al., 2010), stimulation of the feline hypothalamus has long been understood to elicit an aggressive response (Hess, 1928). As reviewed in Kruk (2014), current feline models of aggression suggest neuronal convergence from the amygdala to the ventromedial (vmHyp) and lateral (IHyp) sections of the hypothalamus initiates defensive rage and predatory aggression, respectively (Delville, de Vries, & Ferris, 2000; Greenberg et al., 1984; Gregg & Siegel, 2001; Summers & Greenberg, 1994, 1995). These pathways are mutually inhibitive (Gregg & Siegel, 2001). During

defensive rage, the vmHyp projects to the dorsal midbrain periaqueductal gray (PAG) and brainstem (Gregg & Siegel, 2001; Siegel & Victoroff, 2009), resulting in the manifestation of physical and verbal aggression (Sewards & Sewards, 2003). This aggression is promoted over time as a function of PAG-hypothalamic feedback (Gregg & Siegel, 2001).

The HPA-axis is considered the central stress system in humans (Hawes, Brennan, & Dadds, 2009). Here, threat-induced hypothalamic activation evokes the release of corticotropin-releasing factors (CRFs), which enter the anterior pituitary blood supply and facilitate adrenocorticotropic hormone release (AHR). AHR prepares the release of adrenaline, used to boost energy and physical ability, and so facilitates the preparation of a behavioural response (Fox, 2008). The HPA-axis is regulated by the hippocampus, which decreases CRF output (Gregg & Siegel, 2001; Raine, 2013) and aids affect processing by signalling the declarative fact of threat (Blanchard, Hynd, Minke, Minemoto, & Blanchard, 1991; Raine, 2013; Williams et al., 2001).

Amygdala hyperactivity to facial threat might detriment the amygdala-HPA network by over stimulating the hypothalamus (Davidson et al., 2000; Vyas, Mitra, Shankaranarayana, & Chattarji, 2002). Frequent threat-induced activation of the hypothalamus likely habituates the hypothalamic response, and increases the threat-arousal threshold required to engage hippocampal modulation (Kruk et al., 1998; Strauss et al., 2005; Viau, 2002). Consequently, deficits in hippocampal inhibition likely contribute to increased instances of threat-induced aggressive behaviour (Delville, David, Taravosh-Lahn, & Wommack, 2003; Lopez-Duran, Kovacs, & George, 2009; van Goozen & Fairchild, 2008).

HPA-axis efficacy is typically indexed by cortisol, with increased cortisol related to increased adrenalin output (Hawes et al., 2009). In rodent samples, though not evidenced in Oakes & Coover (1997), low basal cortisol (i.e. cortisol measured over several days) has been associated with increased aggression (Kruk, Halasz, Meelis, & Haller, 2004), suggesting an inverse relationship between HPA-axis functionality and aggressive behaviour. The null result reported in Oakes & Coover (1997) might reflect low baseline aggression scores restricting variance in the data.

In humans, though based on a small effect size, a meta-analysis of 82 investigations between 1978 and 2006 proposed an inverse relationship between basal cortisol and antisocial behaviour (Alink et al., 2008). This small effect size might reflect an amalgamation of positive and negative relationships with the literature. For example, whilst resting state cortisol has been negatively associated with antisocial behaviour in some investigations (Dolan, Anderson, & Deakin, 2001; McBurnett et al., 2000; Oosterlaan et al., 2005; Pajer, Gardner, Rubin, Perel, & Neal, 2001; van Goozen et al., 1998; Virkkunen, 1985), positive associations with reactive aggression have been reported elsewhere (van Bokhoven et al., 2005).

More recently, basal, but not acute (i.e. current/short term) cortisol, has been inversely associated with paradigm-induced aggression, indexed by the duration and intensity of aversive blasts of white noise (Böhnke, Bertsch, Kruk, & Naumann, 2010). Additionally, whilst not reaching statistical significance, lower cortisol levels have been observed in adolescent offenders detained for property, drug, and violent crimes, relative to non-offender controls (Feilhauer, Cima, Korebrits, & Nicolson, 2013). These findings suggest individuals with low basal HPA-axis activation might be more sensitive to provocation and more vulnerable to aggressive reactions.

Inconsistent results have also been observed in relation to cortisol levels and CU traits. In an adolescent cohort, basal cortisol levels were lower in males with high, relative to low parent-rated CU traits, after controlling for conduct problem severity (Loney, Butler, Lima, Counts, & Eckel, 2006). Similarly, resting state cortisol levels were lower in subgroups of offenders reporting high, relative to low psychopathic traits (Cima, Smeets, & Jelicic, 2008). However, measures of

basal cortisol did not correlate with CU traits in adolescent (Azar et al., 2004; Poustka et al., 2010) or adult (Feilhauer et al., 2013) samples. Feilhauer et al. failed to observe associations between basal cortisol and either callousness, uncaring, unemotional, or narcissistic traits. Variation in findings might reflect population heterogeneity and diverse definitions of antisocial behaviour (Hawes et al. 2009).

2.5 Facial Distress

Facial distress is thought to activate a similar subcortical pathway to that described in section 2.3 (page 24) (Amaral et al., 1992; LeDoux, 2000; Morris et al., 1996; Pessoa et al., 2002; Williams et al., 2001). Bilateral amygdala augmentation has been observed to fearful (Fusar-Poli et al., 2009; Vuilleumier et al., 2001) and sad (Fusar-Poli et al., 2009), relative to neutral facial expressions. Furthermore, in relation to threat, greater amygdala activation has been observed to fearful, relative to angry, facial expressions (Breiter et al., 1996; Phillips et al., 1997, 1998, 2004, *but see Murphy, Nimmo-Smith, & Lawrence, 2003 for a review*), and eyes (Morris, deBonis, & Dolan, 2002; Whalen et al., 2004), suggesting a greater amygdala bias towards distress.

Distress-evoked amygdala activation is central to the integrated emotion system (IES) associated with the VIM – reliant on intact amygdala functioning to extinguish antisocial behaviour (Blair, 2005). Patients with amygdala lesions evidence impaired recognition of fearful and sad (Adolphs et al., 1995, 1999; Blair, 1995, 2006; Calder, Lawrence, & Young, 2001; Lykken, 1957; Vuilleumier et al., 2004), but not happy (Adolphs et al., 1999) facial expressions. Furthermore, as the ability to label afraid body postures remains intact following amygdala lesioning (Adolphs & Tranel, 2003; Atkinson, Heberlein, & Adolphs, 2007), amygdala damage may impair recognition of *facial* distress, specifically.

Amygdala hypoactivation to fearful faces has been observed in adolescents exhibiting CU traits (Blair, 2003; Dadds, El Masry, Wimalaweera, & Guastella, 2008; Jones, Laurens, Herba, Barker, & Viding, 2009; Sterzer et al., 2005; Sylvers, Brennan, & Lilienfeld, 2011), patients with antisocial personality disorder (ASPD) (Birbaumer et al., 2005), and schizophrenia (Dolan & Fullam, 2009) scoring high on psychopathy measures. Amygdala hypoactivation is also observed in children with conduct problems and attention deficit hyperactivity disorder (ADHD) scoring high on CU traits, relative to controls (Marsh et al., 2008). One recent investigation failed to observe differences in fear-evoked amygdala activation between adolescents with and without conduct problems (Lozier, Cardinale, Vanmeter, & Marsh, 2014). However, *post hoc* analysis, controlling for externalisation, reported an inverse correlation between CU traits and right hemispheric amygdala activity to fearful faces in the overall cohort suggesting importance for teasing apart aggressive and CU traits.

As described in chapter one, section 1.3.4 (page 12), the successful perception of distress typically evokes an empathic response, critical to the activation of the VIM. On a neurobiological level, empathy is related to activation of the ventromedial prefrontal cortex (vmPFC) and inferior frontal gyrus (IFG). The vmPFC is a composite of the medial frontal and orbitofrontal cortices (OFC) (Bechara, Damasio & Damasio, 2000; Damasio, 1994) and is considered the limbic portion of the prefrontal cortex (PFC) (Pietrini, Guazzelli, Vasso, Jaffe, & Grafman, 2000). During face processing, the OFC, in particular, moderates amygdala activity by providing meaning to stimuli when resolving ambiguity (Ploghaus et al., 2000; Raine, 2013).

The vmPFC is a key brain region involved in the emergence of emotional selfreflection (Blair, 2013; Goubert et al., 2005; Shamay-Tsoory, Tomer, Goldsher, Berger, & Aharon-Peretz, 2004; *but see van der Meer, Johnson, Schmitzer-Torbert, & Redish, 2010 for review*). Lesioning of the vmPFC is related to deficits in cognitive empathy (Shamay-Tsoory, Aharon-Peret, & Perry, 2009; Shamay-Tsoory, Tomer, Berger, & Aharon-Peretz, 2003), with affective empathy/emotion recognition left intact (Shamay-Tsoory et al., 2009). Conversely, lesioning of the IFG, a region that receives projections from the fusiform gyrus and amygdala (Fairhall & Ishai, 2007), is associated with impaired affective empathy and emotion recognition (Shamay-Tsoory et al., 2009). Self-reported affective empathy has been shown to predict IFG activation in response to affective facial stimuli (Jabbi, Swart, & Keysers, 2007). The IFG is typically activated during the passive viewing or imitation of emotional faces (Catmur, Walsh, & Heyes, 2009; Dapretto et al., 2006; Gallese, 2007; Heiser, Iacoboni, Maeda, Marcus, & Mazziotta, 2003; Schulte-Ruther, Markowitsch, Fink, & Piefke, 2007; Seitz et al., 2008), as well as tasks involving empathising with the suffering of others (Nummenmaa et al., 2008).

2.6 Executive Functioning

Although this chapter has framed the amygdala as a key brain region involved in affect-driven reactive aggression, humans are typically able to regulate aggressive behaviour through higher-order cognitive systems underpinned by prefrontal networks (Moffitt, 1993; Nauta, 1971). PFC activation is observed during response preparation, sustained motivation, self-control, and affect regulation (Giancola, Mezzich, & Tarter, 1998; Perdeci et al., 2010; Sotres-Bayon, Cain, & LeDoux, 2006; Tranel, Bechara, & Denburg, 2002) and so is thought to contribute to the containment of affect-driven reactive aggression (*see Davidson et al., 2000 for review*).

Previously, composite PFC hypoactivation has been observed in samples of murderers (Raine et al., 1994, 1998; Raine, Venables, & Mednick, 1997), ASPD (Anckarsater, 2006; Bassarath, 2001; Brower & Price, 2001; Hoptman, 2003; Pridmore, Chambers, & McArthur, 2005) and psychopaths exhibiting reactive, but not instrumental, aggression (Kiehl, 2006). Furthermore, use of diffusion tensor MRI, a method of imaging white matter connectivity, has shown dysfunctional fronto-limbic connectivity in antisocial individuals (Sundram et al., 2012). However, it is important to disentangle the underlying frontal networks to better understand their impact on aggression and CU traits. This section delineates the role of the anterior cingulate cortex (ACC), OFC, and IFG in executive functioning and inhibitory processes. Typically functioning humans show considerable anatomical and functional connectivity between the OFC and amygdala (Amaral & Price, 1984; Freese & Amaral, 2009; Marsh et al., 2011; Pietrini et al., 2000). This network likely underpins script theory (Huesmann, 1988) and the appraisal mechanism described in the GAM (Anderson & Bushman, 2002, see chapter one, section 1.3.2 [page 9]). Specifically, the OFC is thought to moderate affect-driven amygdala responses by providing contextual meaning to social cues (Bechara et al., 2000; Damasio, 2003; Davidson, Jackson, & Kalin, 2000; Davidson, Putman, et al., 2000; Etkin et al., 2011; Horn, Dolan, Elliott, Deakin, & Woodruff, 2003; Rolls, 1999; Saddoris, Gallagher, & Schoenbaum, 2005; Schiller & Delgado, 2010). OFC lesioning has been associated with increased reactive aggression, socially deviant behaviours, and poor affect recognition (Anderson, Bechara, Damasio, Tranel, & Damasio, 1999; Grafman et al., 1996; Winstanley, Theobald, Cardinal, & Robbins, 2004). Contralateral amygdala-OFC connectivity during resting state inversely associated with trait anger (Fulwiler, King, & Zhang, 2012) and aggression (Hoptman et al., 2010).

Although OFC activation has been observed when experiencing anger in Dougherty et al. (2004), other evidence suggests healthy functioning individuals require OFC *deactivation* in order to imagine aggressive behaviour and to aggress in a video game setting (Pietrini et al., 2000). Baseline OFC hypoactivation might indicate vulnerability to poor affect regulation and antisocial tendencies (Antonucci et al., 2006; Beyer, Münte, & Krämer, 2014; Blair, 2001; Blair et al., 2002; Raine, Fung, & Lam, 2011) and has been observed in samples with comorbid schizophrenia/ASPD (Joyal, Dubreucq, Gendron, & Millaud, 2007), BPD (Goyer et al., 1994; Soloff et al., 2003), IED (Coccaro et al., 2007; Izquierdo & Murray, 2004; Izquierdo, Suda, & Murray, 2004; New et al., 2009), comorbid depression/aggression (Dougherty et al., 2004), as well as both adolescent (Marsh et al., 2011) and adult (Dolan & Fullam, 2009) samples with psychopathic traits. Furthermore, OFC activity evoked to angry facial expressions has been found to inversely correlate with aggression - measured by the delivery of an aversive noise - in a community sample (Beyer et al., 2014).

The OFC has dense projections to the ACC (Bush, Luu, & Posner, 2000), a region thought to be involved in attention, conflict detection, motor inhibition, and affect regulation (Botvinick et al., 2001; Botvinick, Cohen, & Carter, 2004; Braver et al., 2001; Brown et al., 2006; Bush et al., 2000; Cardinal, Parkinson, Hall, & Everitt, 2002; Devinsky, Morrell, & Vogt, 1995; Garavan, Ross, Murphy, Roche, & Stein, 2002). Whilst the dorsal section of the ACC (dACC) is thought to be involved in decision-making processes such as conflict monitoring (Holroyd & Coles, 2002), choice selection, and performance monitoring (Frith, Friston, Liddle, & Frackowiak, 1991; van Veen & Carter, 2002), the ventral section of the ACC (vACC) is thought to be involved in representing and making judgements on stimuli, as well as regulating emotional behaviour (Bush et al., 2000; Devinsky et al., 1995; Wager, Davidson, Hughes, Lindquist, & Ochsner, 2008).

The ACC regulates affect via connections to the amygdala (Etkin, Egner, Peraza, Kandel, & Hirsch, 2006; Ghashghaei et al., 2007). As such, ACC activation has been observed in response to facial threat (Blair et al., 1999) and distress (Blair et al., 1999; Morris et al., 1998), and during states of anger (Dougherty et al., 1999). Dougherty et al. evoked anger in a community sample using anger-inducing scripts, and found increased cerebral blood flow in the right dACC during affect regulation. However, although anger was confirmed using a psychometric assessment, typical physiological changes such as those described in chapter one, section 1.2.1.1 (page 3) were not observed.

ACC dysfunction has been reported in patients with schizophrenia (Reske et al., 2007), BPD (Tebartz van Elst et al., 2003), and CD (Stadler et al., 2007; Sterzer et al., 2005), characterised by aggressive traits. In both Sterzer et al. (2005) and Stadler et al. (2007), dACC deactivation was observed in aggressive children and adolescents, in response to negative affective images. Together, these

results suggest ACC dysfunction might predict vulnerability for poorer affect regulation, and in turn, antisocial and aggressive behaviour.

In response to a letter-based go/nogo paradigm, where participants are tasked with responding to one stimulus but not another, motor inhibition has been associated with neuronal transmission between the ACC, IFG, basal ganglia, and pre-motor regions (Steele et al., 2014). In addition to its role in the generation of empathy, the IFG has been shown to drive attention (Corbetta & Shulman, 2002), stop-signal detection (Garavan, Hester, Murphy, Fassbender, & Kelly, 2006; Mars et al., 2009; Rubia, Smith, Brammer, & Taylor, 2003; Schall et al., 2002), and motor impulse suppression (Antonucci et al., 2006; Aron et al., 2003). Though evident in the left hemisphere during simple inhibitory tasks (Konishi et al., 1999; Menon, Aldeman, White, Glover, & Reiss, 2001; Swick, Ashley, & Turken, 2008), IFG activity is maximal in the right hemisphere during tasks requiring extinction of a pre-initiated motor response (Aron, Robins, & Poldrack, 2004).

The IFG extracts stop-relevant information and uses it to both update working memory (Mars et al., 2008) and signal the need for motor extinction (Mostofsky & Simmonds, 2008; Verbruggen et al., 2010). As a function of pre-initiated behaviour being more difficult to disrupt than planned behaviour, IFG activity is thought to be greater to stop than nogo stimuli (Hughes, Fulham, Johnston, & Michie, 2012). During behaviour extinction, the IFG directly stimulates the superior temporal gyrus (STG) in order to quickly interrupt on going motor signals (Aron, Behrens, Smith, Frank, & Poldrack, 2007; Nambu et al., 2002). Subsequently, the IFG suppresses the basal ganglia, a region involved in motor behaviour (Aron, 2007; Aron & Poldrack, 2006; Rubia et al., 2001). As measured by a subdural electrode array, successful, relative to unsuccessful stopping is associated with greater activation of the right IFG (Swann et al., 2009).

The role of the IFG in motor extinction has been compounded across lesioning studies as well as task-based and structural MRI. Swick et al. (2008) reported a greater number of inhibitory errors in individuals with, relative to without IFG lesions. Furthermore, in samples with schizophrenia, characterised by inhibitory deficits, IFG activation has been inversely related to stopping times in response to auditory tones (Hughes et al., 2012). IFG grey matter has also been inversely associated with schizophrenic symptom severity (Suga et al., 2010).

Additional to IFG suppression of the basal ganglia, a comparison of haemodynamic responses to nogo and stop stimuli suggests a comparable network whereby the IFG moderates regions of the dorsomedial prefrontal cortex (dmPFC) such as the pre-supplementary motor area (Pre-SMA; Dum & Strick, 2005; Rubia et al., 2001). The Pre-SMA is critical for motor extinction (Aron et al., 2003, 2007; Chamberlain et al., 2008; Chambers et al., 2006; Floden & Stuss, 2006; Li, Huang, Constable, & Sinha, 2006). In particular, the dorsal Pre-SMA is associated with suppression of arm movements in monkeys (Mirabella, Pani, & Ferraina, 2011) and motor responses in humans (Aron et al., 2007; Floden & Stuss, 2006; Garavan et al., 2006; Nachev, Wydell, O'Neill, Husain, & Kennard, 2007; Rubia et al., 2003; Schall, Stuphorn, & Brown, 2002).

The dmPFC is thought to mediate executive control functions such as motor attention, conflict monitoring, and response selection (Rubia et al., 2001). A motor suppression task requiring the classification of angry and neutral faces was used to demonstrate increased Pre-SMA activation in samples of male students characterised by low (lower 15th percentile), relative to high (upper 15th percentile) aggressive scores (Rubia et al., 2003). However, stop signals in this task were not affect-driven, and instead relied upon a change in stimulus frame colour. Additionally, individuals exhibiting high, relative to low trait aggression evidenced Pre-SMA activation attenuation and longer reaction times when asked to classify angry and neutral facial expressions - inhibiting said response when followed by a change of frame colour (Pawliczek et al., 2013).

2.7 Chapter Summary

At its core, chapter two has framed reactive aggression as an interaction between a hyperactive amygdala-HPA axis response to threat and a hypoactive amygdala-PFC response to facial distress (McEllistrem, 2004; Morgan & Lilienfeld, 2000; Raine, 2013; Slever, 2002). However, this response is far from linear and not always guaranteed to result in the manifestation of aggression (Golden, Jackson, Peterson-Rohne, & Gontkovsky, 1996). Informed by the VIM, chapter two used supporting evidence from community, forensic, and clinical samples to describe how aggressive and CU traits might differentially modulate the amygdala responses to threat/distress, and in turn, how this might result in the disengagement of prefrontal brain networks associated with motor extinction. Still, key to the VIM, it remains to be seen how facial distress might moderate motor extinction when used as a stopping mechanism. Chapter three describes how these neurobiological networks might be indexed on an electrophysiological level, and how these markers might differ in samples characterised by aggressive/CU traits.

Chapter Three: Electrophysiological indices of aggression

3.1 Electroencephalography and Event-Related Potentials

Chapter three builds on the VIM-related neurobiological networks discussed in chapter two. Specifically, it describes how face processing, both in general and in response to facial threat/distress, as well as distress-induced motor extinction, might be indexed on an electrophysiological level using event-related potentials (ERP).

First described in humans by Berger (1929), amplified extracellular activity surrounding postsynaptic neurons can be measured at the scalp using electroencephalography (EEG). EEG serves as an inexpensive, non-invasive, and easily implementable method of indexing neuronal processes (Delle-Vigne, Wang, Kornreich, Verbanck, & Campanella, 2014). Compared to neuroimaging techniques reliant upon haemodynamic measures of brain activity, such as MRI, EEG commands high temporal - but low spatial - resolution and reflects the response from brain networks, rather than precise brain structures (Luck, 2012). High temporal resolution allows investigation of dynamic changes and neural network connectivity (Luck, 2005).

For effective measurement of EEG, neural generators need to be orientated in the same direction and share similar excitatory/inhibitory inputs. A collection of positively and negatively orientated generators would result in voltage being cancelled out to near zero. Therefore, summated voltage likely occurs in pyramidal cells; the main input-output cells of the cortex, perpendicularly aligned to the cortical surface (Luck, 2005, 2012).

Stimulus-evoked electrical voltage within a given generator travels through the brain, via the least resistant pathway. Upon reaching the skull, electrical activity spreads laterally as a function of high electrical resistance (Luck, 2005). Like a

voltmeter, EEG is measured between an active electrode and reference site, which can be adjusted for optimal voltage contrast.

EEG that is time-locked to the presentation of a stimulus or behavioural response - the ERP - has been mapped onto distinct information processing stages (Coles et al., 1990; Connor, 2002; Halgren, Marinkovic, & Chauvel, 1998). Components of the ERP encompass stimulus-related processing, alongside background electrical activity from baseline brain function. Several components make up an ERP waveform (Luck, 2006, 2013) and are visualised as negative or positive deflections around a baseline, which can be assessed for latency (measured in milliseconds) and amplitude (measured in millivolts).

Components temporally overlap, such that any given component may visually influence subsequent waves (Luck, 2006, 2013). This influence is predominant across earlier components and likely results in amplitude attenuation or augmentation, as well as latency elongation. By contrasting two experimental conditions, ERPs can be used to determine how specific cognitive processes might be influenced by a given experimental variable and so are useful assays of cognitive dysfunction (Donchin, 1979; Luck, 2013; Rugg & Coles, 1995).

Due to low signal to noise ratio (SNR), it is difficult to discriminate task-specific components from resting state EEG using individual trials (Luck, 2006). Noise is defined as variation in the data, which is not explained by a statistical model, and so should be reduced where possible. As noise is considered constant between trials, averaging data across several trials reduces trial-by-trial background noise to near-zero voltage (Halgren et al., 1998; Luck, 2005). Although trial-by-trial variation is lost, averaging acts to reduce noise and boost statistical power by removing unwanted fluctuations in the data. The residual ERP, mapped over multiple trials, sums task-related signal coherently, and results in a single task-specific response (Luck, 2006).

Typically, the first 50 ms of a given ERP waveform is thought to index basic brainstem activity, with electrophysiological activity between 50 and 100 ms

representing the processing of sensory perceptual information evoked in the primary sensory and frontal cortices. This is followed by the incorporation of primitive limbic processing at around 100 to 300 ms, before higher cognitive processing after 300 ms (Connor, 2002). The following sections discuss ERP components associated with the processes underpinning reactive aggressive behaviour (e.g. face processing, executive function, and distress-induced motor extinction), in addition to how they might be moderated by negative affect and aggressive/CU traits. Where known, ERP generators are related to the neurobiological underpinnings of reactive aggression discussed in chapter two.

3.2 Face Processing

Due to their high temporal resolution, ERPs provide a powerful tool to measure the cognitive efficiency of face processing, and allow for the capture of emotional and attentional processing at several stages (Chai, Castañón, Ongür, & Whitfield-Gabrieli, 2012; de Pascalis et al., 2005; Luck & Kappenman, 2013; Vuilleumier & Pourtois, 2007). Conscious processing of visual stimuli predominantly occurs between 150 and 800 ms, and is indexed by N170, P200, N200, and P300 ERP components. This section maps these components onto brain networks associated with face processing.

3.2.1 N170

Early ERP investigation reported large positive responses to visual stimuli at central-dorsal points of the scalp (i.e. the vertex) between 140 and 180 ms (Bötzel & Grüsser, 1989; Jeffreys, 1989). Coined the vertex positive potential (VPP), this positivity was accompanied by a negative deflection, the N170 (Bentin et al., 1996), at bilateral, occipito-temporal sites with origins in the temporal cortex (Jeffreys, 1996). Jeffreys suggested earlier ignorance of occipito-temporal ERPs was an artefact of EEG output being referenced to the mastoids, which attenuated occipito-temporal voltage. More recently, this claim has been supported by a meta-analysis of studies evoking N170 responses to facial expressions (Hinojosa et al., 2015). Here, Hinojosa et al. reported greater

effect sizes when referencing to the common average, relative to mastoid sites. Measurement of the N170 is preferential to the VPP as [1] electrodes are placed in closer proximity to generating structures, and [2] bilateral electrode placement allows for observation of lateralisation effects (Luck & Kappenman, 2013).

The neural origin of the N170 is thought to lie within the OFA, STS, and lateral and posterior (Deffke et al., 2007; Rossion, Joyce, Cottrell, & Tarr, 2003; Shibata et al., 2002) sections of the fusiform gyrus (Adolphs, 2002; Itier & Taylor, 2004; Nummenmaa, Hyönä, et al., 2010; Rossion et al., 2003; Sadeh et al., 2010; Watanabe, Kakigi, & Puce, 2003). The fusiform gyrus, in particular, is considered critical during face processing (Vuilleumier & Pourtois, 2007). Furthermore, intracranial recordings within the right OFA and fusiform gyrus have shown activation during the N170 time-window (Jonas et al., 2012; Parvizi et al., 2012).

Further support for the involvement of the fusiform gyrus and STS in N170 generation is provided by magnetoencephalography (MEG) and fMRI. Generation of the M170, a magnetic counterpart of the N170, has been localised within the fusiform gyrus (Deffke et al., 2007; Halgren, Raij, Marinkovic, Jousmäki, & Hari, 2000; Linkenkaer-Hansen et al., 1998). Additionally, positive correlations between N170 amplitude and activation of both the fusiform gyrus (Horovitz et al., 2004; Nguyen & Cunnington, 2014; Sadeh et al., 2010) and STS (Nguyen & Cunnington, 2014; Sadeh et al., 2010) and STS (Nguyen & Cunnington, 2014; Sadeh et al., 2010; Sato, Kochiyama, Uono, & Yoshikawa, 2008) have been measured using concurrent EEG and fMRI in response to neutral faces.

The N170 is evoked to visual stimuli including words, objects, and faces. In particular, faces, compared to objects such as houses, increase postsynaptic activity (Luck & Kappenman, 2013) and evoke larger N170 deflections (Allison et al., 1999; Crist, Wu, Karp, & Woldorff, 2008; Holmes et al., 2005). N170 augmentation is observed to a variety of facial stimuli including line drawings,

cartoons, schematic representations, gray-scale, and isolated eyes (Bentin et al., 1996; Carmel & Bentin, 2002; Eger, Sterzer, Russ, Giraud, & Kleinschmidt, 2003; Krombholz, Schaefer, & Boucsein, 2007; Rossion, 2014; Rossion & Jacques, 2011) and interacts positively with increased stimulus intensity (Sprengelmeyer & Jentzsch, 2006). The face-evoked N170 is characterised by right hemispheric dominance (Bentin et al., 1996; Hinojosa et al., 2015; McCarthy, Puce, Gore, & Allison, 1997; Rossion et al., 2003; Stockdale, Morrison, Kmiecik, Garbarino, & Silton, 2015), regardless of the presence of eyes (Magnuski & Gola, 2013), valence (Polich & Herbst, 2000), or whether the face is static or dynamic (Caharel, et al., 2013).

The N170 can be modulated through experimental manipulations. Presentation of stimuli altered through contrast reversal (Itier & Taylor, 2002), inversion (Bentin et al., 1996; Eimer, 2000b; Jacques, d'Arripe, & Rossion, 2007; Rossion et al., 2000), degradation of quality (Holmes et al., 2005), and the scrambling of facial features (Zion-Golumbic & Bentin, 2007), both prolong and augment the N170 response. Conversely, although not always evident (Huddy et al., 2003; Mnatsakanian & Tarkka, 2004), repeatedly presenting the same facial stimulus acts to attenuate the N170 response (Campanella et al., 2000; Itier & Taylor, 2002). N170 attenuation is also observed in tasks where stimuli are presented over varying sizes and modalities (Jacques et al., 2007). Taken together, these findings suggest that a greater number of neural resources are recruited when faces are more difficult to identify (Rossion et al., 2000).

3.2.2 P200

Directly following the N170, a positive deflection occurs in the ERP waveform around 200 ms post stimulus onset - the P200. The P200 is thought to index the encoding of task salient features (Potts & Tucker, 2001) and is evoked to both visual and auditory stimuli (Potts, Dien, Hartry-Speiser, McDougal, & Tucker, 1998; Potts, Liotti, Tucker, & Posner, 1996). Although distributed over anterior electrode sites during target detection tasks (Carretié et al., 2004; Potts, 2004; Potts & Tucker, 2001), the P200 shows an occipito-temporal distribution, with maximal activation over the right hemisphere during face processing (Farah, 1990; Watanabe, Kakigi, Koyama, & Kirino, 1999).

Neural generators of the face-evoked P200 are largely unknown (González-Roldán et al., 2011). However, one investigation has measured brain activity of epilepsy patients using depth electrodes implanted into temporal, occipital, and parietal brain regions (Halgren et al., 1994). Approximately 180 ms post stimulus onset, Halgren et al. reported increased bilateral activation in the fusiform gyrus in response to faces. Although the colour and size of facial stimuli were not controlled for, fusiform gyrus activation was observed in all six patients. Two patients showed fusiform gyrus activation exclusively for faces, and three patients evidenced increased fusiform gyrus activation to faces, relative to words, letters, or symbols. Furthermore, whilst not specific to facial stimuli, activation was recorded in the posterior cingulate, as well as supramarginal, posterior superior and middle temporal, and parahippocampal gyri, between 140 and 280 ms post stimulus onset.

Previously, P200 augmentation has been observed in response to affective pictures (Carretié et al., 2001, 2004; Delplanque, Lavoie, Hot, Silvert, & Sequeira, 2004), words (Kanske & Kotz, 2007; Kissler et al., 2006; Schapkin, Gusev, & Kuhl, 2000) and facial expressions (Ashley, Vuilleumier, & Swick, 2004; Eimer et al., 2003; Schulz, Kaufmann, Kurt, & Schweinberger, 2012), relative to respective neutral or non-affective controls. As such, the P200 might reflect the enhanced activation of the amygdala (Vuilleumier & Pourtois, 2007) - sensitive to distinct features associated with emotion portrayal (Begleiter et al., 1979; Carretié & Iglesias, 1995). Specifically, in response to faces, the P200 likely reflects processing of second order, configural facial features (Itier & Taylor, 2004; Latinus & Taylor, 2006).

3.2.3 N200

The N200 is a negative deflection in the ERP that is maximal over right anterior sites between 175 and 300 ms post stimuli onset (Bokura, Yamaguchi, & Kobayashi, 2001; Bruin & Wijers, 2002; Heinze, Munte, Gobiet, Niemann, & Ruff, 1992; Strik, Fallgatter, Brandeis, & Pascual-Marqui, 1998). Among other processes associated with executive functioning (*see section 3.4.1 [page 50]*), the N200 is thought to index automatic alerting, stimulus classification, and template matching (Botvinick et al., 2001; Heinze et al., 1992; Luck, 2005; Sumich et al., 2013; van Veen & Carter, 2002; Yeung et al., 2004, *but see Folstein & van Petten, 2008 for review*).

In response to faces, the N200 likely reflects involvement of the vACC when integrating amygdala-projected emotionally salient information (Carretié et al., 2004; Kanske & Kotz, 2010b). Additionally, N200 generation has been localised to other frontal-temporal brain regions including the lateral OFC, dorsolateral prefrontal cortex (dIPFC), hippocampus, and superior temporal cortex (Bekker, Kenemans, & Verbaten, 2005; Bokura et al., 2001; Nieuwenhuis, Yeung, & Cohen, 2004; van Veen & Carter, 2002). During maturation of the brain, connections within these networks are pruned in order to increase inter-cellular communication (Chandrasekaran, Plas, Gonzalez, & Crair, 2005). As evidenced in community and ADHD samples, this pattern of maturation, indexed by N200 attenuation, is inversely correlated with age (Sumich et al., 2012).

3.2.4 P300

The final component of interest, the P300, is a positive deflection in the ERP occurring around 300 ms post stimulus onset (Bauer & Hesselbrock, 1999). The P300 is thought to reflect processes involved during information processing (Hillyard & Kutas, 1983; Polich & Herbst, 2000), such as the effortful/automatic orientation of attention and evaluation of incoming stimuli against pre-existing working memory representations (Donchin & Coles, 1988; Ito, Larsen, Smith, & Cacioppo, 1998; Nieuwenhuis, Aston-Jones, & Cohen, 2005; Polich & Kok, 1995). Although largest at posterior sites overall (Baumeister et al., 2014), task

condition modulates the P300, such that responses are maximal at central posterior sites in response to target/go stimuli (Luck, 2005) and central anterior sites in response to nogo stimuli (Baumeister et al., 2014; Bokura et al., 2001; Fallgatter et al., 2005).

During face processing, the P300 is thought to index voluntary evaluation of facial emotion (Halgren et al., 1994) and cognitive processing capacity (Donchin & Coles, 1988; Polich, 2003). Whilst P300 augmentation suggests that fewer neural resources are required to attend to environmentally/motivationally relevant stimuli (Bauer & Hesselbrock, 1999; Branchey, Buydens-Branchey, & Liever, 1988; Harmon-Jones, Barratt, & Wigg, 1997; Polich, 2007), P300 attenuation likely reflects unorganised resource allocation, poor attention to facial stimuli, and deficient cognitive functioning (Meier, Perrig, & Koenig, 2012).

The P300 is generated by a large neuronal network, which spans several cortical and subcortical regions (Duncan et al., 2009). On account of their roles in attention, perception, and memory, the cingulate, as well as bilateral frontal, parietal, and limbic cortices have been associated with P300 generation (Volpe et al., 2007). Although source localisation of the P300 to the ACC is considered highly validated (Pizzagalli, Oakes, & Davidson, 2003), more recent evidence suggests additional generation within the insula, OFC, as well as the frontal gyrus, hippocampus, and dIPFC of the right hemisphere (Bocquillon et al., 2011; Campanella et al., 2013; O'Connell et al., 2012).

3.3 Facial Threat and Distress Processing

In section 3.2 (page 39) the N170, P200, N200, and P300, were discussed both in terms of how they might index stages of face processing, and how they might map onto associated neural networks described in chapter two. With facial affect thought to alter the posterior EEG signal as early as 80 to 100 ms poststimulus onset (Hillyard, Mangun, Woldorff, & Luck, 1995), the following section explores how ERP responses to faces might be moderated as a function of facial threat and distress.

3.3.1 N170

It has been suggested that electrophysiological findings supporting N170 insensitivity to facial affect are relatively concurrent (Rellecke, Sommer, & Schacht, 2013). Previously, although not measured at optimal temporo-occipital sites in some investigations (e.g. Chai et al., 2002), amplitude variation has not been evidenced when comparing the N170 response to neutral, happy, fearful, and disgusted facial stimuli in community (Ashley et al., 2004; Bobes, Martin, Olivares, & Valdes-Sosa, 2000; Chai et al., 2012; Eimer et al., 2003; Eimer & Holmes, 2002, 2007; Halgren et al., 2000; Herrmann et al., 2002; Krolak-Salmon, Fischer, Vighetto, & Mauguiere, 2001; Stockdale et al., 2015) or psychopathic (Eisenbarth et al., 2013; Flor, Birbaumer, Hermann, Ziegler, & Patrick, 2002) samples. Such findings lead some to consider the N170 as a neural index of early structural and configural face encoding - preceding the identification of emotion (Ashley et al., 2004; Eimer, 2000; Eimer & Holmes, 2002, 2007; Holmes, Kiss, & Eimer, 2006; Ioannides, Liu, Kwapien, Drozdz, & Streit, 2000; Polich & Herbst, 2000).

Recently, debate surrounding the influence of facial affect on the N170 has been expanded via a meta-analysis of fifty-seven investigations (Hinojosa et al., 2015). This analysis sought to delineate whether facial affect moderated the N170 response, and if so, whether this moderation was emotion-specific or uniform across facial expressions. Results suggested that whilst N170 responses to faces depicting sadness and disgust did not differ from those evoked to neutral facial stimuli, N170 responses to angry, fearful, and happy faces were augmented. This finding conflicts the independent face processing model discussed in Bruce & Young (1986, *see chapter one, section 1.3.1 [page 7]*) in that rather than identity and expression being processed over separate routes, the N170 might better index an integrated face processing mechanism. Although the emotion-specific hierarchy of N170 activation observed in Hinojosa et al. (2015) might be partially explained by a dearth of investigations using sad and disgusted facial stimuli, the authors explained these findings in

terms of fearful, angry, and happy facial expressions having increased biological significance.

Angry and fearful facial expressions are negatively valenced, which, relative to neutral and positively valenced stimuli, have been shown to augment the N170 (Smith, Cacioppo, Larsen, & Chartrand, 2003). Angry faces have elicited N170 augmentation across three different visual paradigms. First, in an emotion classification task, though not replicated in their schizophrenic sample, Ibáñez et al. (2012) concluded angry, relative to happy facial expressions, evoked larger N170 amplitudes in a non-clinical cohort. Second, similar findings were observed using schematic facial stimuli, comparing negatively- (e.g. angry), to positively- (e.g. happy) valenced and neutral face drawings (Eger et al., 2003). Third, when priming schematic faces with affect-based words, larger N170 amplitudes were evoked to angry, compared to happy faces, irrespective of the valence of the prime word (Krombholz et al., 2007).

Similarly, N170 augmentation (Ashley et al., 2004; Batty & Taylor, 2003; Campanella, Quinet, Bruyer, Crommelinck, & Guérit, 2002; Ramos-Loyo, Gonzalez-Garrido, Sanchez-Loyo, Medina, & Basar-Eroglu, 2009; Rotschtein et al., 2010; Shannon, Patrick, Venables, & He, 2013; Williams et al., 2006) and shorter latencies (Dawson, Webb, Carver, Panagiotides, & McPartland, 2004; Taylor et al., 2004) have been observed to fearful, relative to neutral, angry, and happy facial expressions. These findings suggest that whilst both angry and fearful facial stimuli augment the N170, possibly via similar threat-specific networks discussed in chapter two, sections 2.3 (page 24) and 2.5 (page 29), distress might elicit an additive response.

Some investigators propose affect-driven N170 responses might be a function of voluntary attention (Holmes et al., 2003) and/or visual priming (Krombholz et al., 2007). Previously, though not observed in paradigms requiring participants to classify faces from objects, or males from females (Boucsein, Schaefer, Sokolov, Schroder, & Furedy, 2001), N170 augmentation to affective, relative to neutral expressions, has been observed in paradigms requiring emotion classification (Eimer, 2000c; Ibánez et al., 2012). In support of this, N170 responses to facial stimuli were augmented when primed with affectively valenced words (Krombholz et al., 2007). In such cases, priming is thought to orientate voluntary attention to the affective elements of stimuli, resulting in augmentation of the N170 response.

3.3.2 P200

When contrasting P200 responses between emotionally valenced (i.e. pleasant and unpleasant), and neutral stimuli, P200 augmentation has been observed to unpleasant stimuli, across task modalities (Carretié et al., 2004; Delplanque et al., 2004; Gerdes et al., 2013; Olofsson & Polich, 2007). Whilst the P200 reported in Carretié et al. (2004) was measured at frontal sites during a target detection task, P200 augmentation has been observed at occipital-parietal sites during a repetition task (Olofsson & Polich, 2007), and when the affective stimuli were either accompanied by a concurrent auditory stimulus (Gerdes et al., 2013), or characterised by low arousal levels (Delplanque et al., 2004).

When evoked to facial expressions, although P200 augmentation has been observed in response to neutral faces in one investigation (Tortosa, Lupiáñez, & Ruz, 2013), P200 augmentation is typically considered a function of threat intensity (Gonzalez-Roldan et al., 2011). Previously, angry facial expressions have evoked larger P200 amplitudes, and shorter peak latencies, in a large community sample during an emotional oddball paradigm (Chai et al., 2012). As P200 augmentation is also observed in videogame settings in response to pistol-laden targets, compared to those without weapons (Correll, Urland, & Ito, 2006), it is likely that the P200 is responsive to biological relevant social cues (Carretié et al., 2004).

Personality and affective priming might modulate enhanced positivity around 200 ms. Previously, individuals reporting anxious personality traits, characterised by hyper-attention to threat (Zhang, Li, & Zhou, 2008) and

47

individuals with depression, known to orientate their attention to distress (Dai & Feng, 2012; Gotlib & Joormann, 2010), have shown larger P200 amplitudes to facial threat and sadness, respectively. Furthermore, as observed for the N170 (*see section 3.3.1 [page 45]*), P200 amplitude might function as an interaction between facial affect and priming. In a paradigm whereby facial expressions were preceded by affective words, acting to orientate attention towards facial affect, P200 responses to fearful facial expressions were significantly larger than for neutral faces (Shannon et al., 2013). Together, P200 augmentation to biologically salient stimuli, such as threat and distress, suggests the P200 might index a threat attention bias and preferential processing of biologically significant stimuli.

3.3.3 N200

In respect to the N200, the perception of angry, relative to neutral and happy facial expressions has been associated with shorter N200 latencies (Feldmann-Wüstefeld, Schmidt-Daffy, & Schübo, 2011; Weymar, Low, & Hamm, 2011) and augmented N200 amplitudes in child (Lewis, Todd, & Honsberger, 2007; Nelson & Nugent, 1990; Todd, Lewis, Meusel, & Zelazo, 2008) and adult (Holmes et al., 2014) samples. Similarly, N200 augmentation has been observed in response to fearful faces (Dennis & Chen, 2007). Although not observed in Thomas, Johnstone, & Gonsalvez (2007), larger N200 amplitudes have been elicited in response to negatively valenced, relative to neutral words (Kanske & Kotz, 2010). The null finding in Thomas et al. (2007) might be explained by the N200 being captured over a long epoch that possibly encompassed multiple positively and negatively deflected components, in addition to the N200 - thus attenuating the effect of valence.

Threat-driven N200 augmentation likely reflects threatening social cues capturing attention and commanding greater neural resources in order to redirect attention away from threat during conflict resolution. Recently, Holmes et al. (2014) evidenced enhanced N200 responses to angry facial expressions after depleting cognitive resources using a complex number recall paradigm. In

this investigation, depletion of cognitive resources was thought to detriment the ability to divert attention away from angry facial expressions and so additional resources, indicative of an enhanced N200 response, were required to resolve this process.

3.3.4 P300

Though predominant in response to negative stimuli (Yee & Miller, 1987), P300 augmentation has been observed to positively and negatively valenced, relative to neutral, non-facial stimuli (Delplanque, Silvert, Hot, & Sequeira, 2005; Keil et al., 2002; Mardaga & Hansenne, 2009). When evoked to faces, angry, relative to happy faces are thought to augment the amplitude (Lang et al., 1990) and elongate the latency (Morita, Morita, Yamamoto, Waseda, & Maeda, 2001) of the P300 response. These patterns are indicative of the biological significance of threat demanding a stronger neuronal response.

Similarly, P300 amplitude augmentation is observed to facial distress. Relative to neutral facial stimuli, sad (Ehlers, Walls, Garcia-Andrade, & Phillips, 2001) and fearful (Campanella et al., 2013) faces have elicited P300 augmentation in samples of Mission Indians and individuals from the community, respectively. Interestingly, whilst Ehlers et al. reported P300 amplitude augmentation to sad, relative to happy faces, the P300 evoked to happy faces reported in Campanella et al. was equal to that evoked to fearful faces. Therefore, although fear, relative to sadness might augment N170/P200 responses (*see sections 3.3.1 [page 39] and 3.3.2 [page 41]*), it is possible that sadness better moderates the P300.

3.3.5 Threat/Distress Processing Summary

Taken together, as indexed on an electrophysiological level, there is strong evidence to suggest that facial threat (e.g. anger) and distress (e.g. fear, and to some extent sadness) might moderate face processing across several distinct stages. As a function of their biological and social significance (*see chapter* one, section 1.3.1 [page 7]), facial threat and distress appear to [1] command greater neural resources during early stages of structural and affect processing (i.e. N170, P200), [2] capture attention (i.e. N200), and [3] yield a stronger neural response during stimulus evaluation (i.e. P300).

3.4 Executive Function

As discussed in chapters one (*see section 1.3.3 [page 11]*) and two (*see section 2.6 [page 31]*), humans are typically able to moderate their thoughts and behaviour through executive functioning. This section describes the N200 and P300 responses evoked during experimental conditions whereby the inhibition, or extinction of a motor response is required. Motor extinction, in particular, is a core aspect of the VIM and so associated ERP responses likely serve as an index of motor extinction efficiency in response to stop stimuli.

3.4.1 N200

Evoked to stimuli denoting the need to inhibit (nogo-) or extinguish (stop-) a motor response, the N200 is thought to reflect pre-motor cognitive control (Huster et al., 2013; Kopp, Rist, & Mattler, 1996; Patel & Azzam, 2005), and is associated with response selection (Jodo & Kayama, 1992; Kok, 1986; Schmajuk et al., 2006). During inhibitory conditions, competing go and nogo/stop responses create conflict and result in the recruitment of compensatory attentional mechanisms in order to assist response selection (Botvinick, 2007; Botvinick, Braver, Barch, Carter, & Cohen, 2001; Kok, 1986; Nieuwenhuis, Yeung, van den Wildenberg, & Ridderinkhof, 2003). This mechanism is evidenced on an electrophysiological level by augmentation of the N200 response during inhibitory, relative to go conditions (Baumeister et al., 2014; Eimer, 1993; Enriquez-Geppert et al., 2010; Falkenstein, Hoormann, & Hohnsbein, 1999; Folstein & van Petten, 2008; Liu et al., 2015; van Boxtel, van der Molen, Jennings, & Brunia, 2001). Conflict monitoring efficiency during motor inhibition tasks might be moderated by stimulus frequency. With the exception of Pfefferbaum, Ford, Weller, & Kopell (1985), decreasing the frequency of task-relevant go and nogo stimuli is thought to augment the N200 response (Nieuwenhuis et al., 2003), irrespective of whether an overt (e.g. button press) or covert (e.g. mental counting) response is required. The null finding reported in Pfefferbaum et al. (1985) might be explained by [1] the use of word strings as go/nogo stimuli, instead of letters, symbols, or auditory tones used elsewhere (*see Nieuwenhuis et al., 2003*), and/or [2] the observation of a short P300 latency. Pfefferbaum et al. argued that the short P300 latency, which was inversely correlated with N200 amplitude, restricted full development of the N200 response.

On a neurobiological level, the inhibitory N200 is thought to reflect activation within the superior- and medial-frontal gyri, OFC, and ACC (Amodio, Master, Yee, & Taylor, 2008; Enriquez-Geppert et al., 2010) - an inhibitory network discussed throughout chapter two, section 2.6 (from page 31). This generation is further supported by EEG source localisation of a letter-based go/nogo oddball paradigm (Nieuwenhuis et al., 2003). Though based on a small sample, Nieuwenhuis et al. explained most of the variance in anterior N200 generation by a dipole based in the ACC during nogo trials. When inhibition is successful, activation of frontal executive networks manifests in large, anterior scalp distributions (Falkenstein et al., 1999; Jodo & Kayama, 1992). However, although limited to the investigation of ADHD samples, when inhibition is unsuccessful, the inhibitory N200 manifests over a bilateral, inferior-posterior scalp distribution (Dimoska et al., 2003; Pliszka, et al., 2000). This latter distribution is thought to represent a preparatory response to the presentation of stop stimuli in the absence of concurrent motor extinction (Pliszka, et al., 2000).

3.4.2 P300

The P300 is thought to index aspects of executive function (Miyake, Friedman, Emerson, Witzki, & Howerter, 2000), such as response inhibition when evoked during nogo and stop conditions (Bruin & Wijers, 2002; Bruin, Wijers, &

Staveren, 2001; Burle et al., 2004; Fallgatter et al., 2005; Pfefferbaum et al., 1985; Rubia et al., 2001). Maximal at anterior midline sites (De Jong et al., 1990; Dimoska, Johnstone, & Barry, 2006; Kok, Ramautar, de Ruiter, Band, & Ridderinkhof, 2004), the P300 response is larger during inhibitory, relative to go trials (Bekker et al., 2004; Polich, 2007) and augments further following successful, relative to unsuccessful response inhibition (Bekker et al., 2005; De Jong et al., 1990; Hughes et al., 2012; Liotti, Pliszka, Perez, Kothmann, & Woldorff, 2005; Overtoom et al., 2002). Whereas P300 augmentation during successful inhibition is indicative of an efficient motor extinction mechanism (Dimoska et al., 2006), the P300 evoked during unsuccessful stops might better index error related, and/or conflict processing (Bekker et al., 2005).

P300 responses to nogo and stop trials are thought to recruit similar underlying networks (van Boxtel et al., 2001), involving the IFG, insula, and ACC (Bae, Kim, Im, & Lee, 2011). As delineated in chapter two, section 2.6 (page 31), the IFG is thought to play a stronger role in motor extinction than motor inhibition tasks (Enriquez-Geppert et al., 2010; Mattia et al., 2012). Enriquez-Geppert et al. tasked participants with categorising coloured squares with infrequent stop signals. Results suggested the stop-P300 was driven by bilateral IFG activation.

3.5 Negative Affect

Throughout chapter two, the generation of negative affect was localised to subcortical/limbic networks. However, as EEG is restricted to the study of cortical generators (Luck, 2005), investigation into the associations between anger and ERPs is scarce (O'Connell et al., 2012). This section delineates the existing anger/ERP literature in terms of both trait and state anger.

Trait anger, the disposition to experience anger, has been associated with augmentation of the P300 response in some (Liu et al., 2015; Stewart et al., 2010), but not all (Bond & Surguy, 2000; Harmon-Jones et al., 1997) investigations. Using an emotional-word Stroop paradigm (Stroop, 1992), Stewart et al. (2010) evidenced a positive association between the parietal P300 evoked to negative stimuli and trait anger. Similar results have been obtained using a visual go/nogo paradigm (Liu et al., 2015). In a contrast between high- and low-scoring respondents on a psychometric assessment of outward anger, Liu et al. (2015) evidenced P300 augmentation and faster reaction times in the high-anger group, during the go condition. However, the inverse was observed in the nogo condition, such that lower P300 amplitudes were reported in the high-, relative to low-anger group. Together, these results suggest that individuals with high trait anger show hyperarousal to task-relevant stimuli and are less efficient at recruiting inhibitory resources. Conversely, when evoked to auditory stimuli, no correlation was observed between trait anger and P300 amplitude in child (Harmon-Jones et al., 1997) and adult (Bond & Surguy, 2000) samples.

ERPs have also been modulated through state anger. Although not a direct measure of anger, Stockdale, Morrison, Kmiecik, Garbarino, & Silton (2015) evoked an increase in negative mood and a decrease in positive mood, using a violent film. Subsequently, Stockdale et al. reported a bilateral decrease in N170 amplitude, and a fronto-midline decrease in P200 amplitude, in response to both fearful and happy facial expressions. This finding suggests exposure to violence results in acute decrement to face processing overall. Conversely, the P200, recorded over midline-parietal sites in Bertsch, Böhnke, Kruk, & Naumann (2009), was augmented in response to fearful and angry faces following the induction of anger by way of aversive blasts of white noise. This investigation suggests anger induction evokes hypervigilance to facial threat. However, as anger was induced using varying methodologies, and the P200 was evoked and measured using different tasks and scalp sites, these two investigations are not comparable.

Anger-induced moderation of the N200 and P300 has been observed in investigations where adolescents were tasked with a visual go/nogo paradigm that induced anger via a sudden loss of 'points' (Lewis et al., 2006; Stieben et al., 2007). In Lewis et al., following anger induction, both inhibitory N200 and

P300 responses measured over fronto-midline sites were augmented. Similarly, in a comparison between adolescent samples characterised by externalising, or both externalising and internalising traits, relative to non-clinical controls, Stieben et al. evidenced nogo-N200 augmentation in the mixed trait - and in turn externalising - cohorts following anger induction. This investigation did not report findings for the P300. Furthermore, augmentation of inhibitory N200 and P300 responses were reported in Stockdale et al. (2015), after viewing a negative mood-inducing film (*see above for paradigm specifications*).

Taken together, anger, both at state and trait level, might moderate electrophysiological indices of face processing and motor inhibition. Specifically, it appears that anger is associated with fewer cognitive resources being spent on processing target stimuli, which in the context of simple behavioural paradigms, might reduce conflict and facilitate inhibition. However, as observed in chapter one, section 1.2.1.1 (page 2), more intense episodes of anger can result in an unrestrained behavioural response. Further investigation, using clear quantifications of anger, should be carried out in order to better understand associations between anger induction and ERPs.

3.6 CU and Aggressive Traits

EEG provides a useful tool for investigating how CU and aggressive traits moderate distinct cognitive processes associated with facial affect recognition and motor extinction. The use of electrical activity when investigating aggressive traits can be traced back to Hill & Sargant (1943), who evidenced electrical abnormalities in a male who murdered his mother for no identifiable reason. This section considers how the N170, P200, N200, and P300 might manifest in samples characterised by aggressive and CU traits. Where appropriate, these findings are discussed in the context of the VIM.

3.6.1 N170

Individuals characterised by aggressive and CU traits evidence disruption to the fusiform-STS-amygdala network (*see chapter two, section 2.3 [page 24]*), which in turn, might reflect on an electrophysiological level through N170 attenuation (*see section 3.2.1 [page 39]*). Therefore, the N170 represents a useful index for investigating the roles of aggression and CU traits on early stages of face processing and emotion classification. However, this association has only been investigated in young adults with poor emotional expression (Meaux et al., 2014), and clinical samples with schizophrenia (Ibáñez et al., 2012), bipolar disorder (Degabriele, Lagopoulos, & Malhi, 2011), and high, relative to low psychopathic traits (Eisenbarth et al., 2013). Whilst these samples are characterised by emotional bluntness and aggression, to date no investigation has specifically associated the N170 with aggressive or CU traits.

Meaux et al. (2014) presented young adults with facial expressions depicting sadness, fear, anger, disgust, surprise, and happiness. Though not a function of emotion - possibly resulting from referencing to bilateral mastoid sites (*see section 3.2.1 [page 39]*) - the average N170 response was inversely correlated with emotional expressivity, a qualitatively similar concept to CU traits (Kimonis et al., 2008). Furthermore, although not supported by behavioural data, N170 responses were lower in samples of schizophrenia (Ibáñez et al., 2012) and bipolar (Degabriele et al., 2011) patients, relative to controls. In these investigations, the N170 was evoked to happy, angry, and sad facial expressions, and was not modulated by affect.

In contrast to the findings above, differences in N170 amplitude were not observed between groups of female psychopaths scoring high and low on the Psychopathy Checklist - Revised (PCL-R; Hare, 1991) in response to fearful, angry, and happy facial expressions (Eisenbarth et al., 2013). This null result could be explained through task parameters. Although all paradigms required face processing, Ibáñez et al. and Degabriele et al. asked participants to view and respond to target faces irrespective of emotion, and Eisenbarth tasked
participants with empathising with the target stimuli. The latter might have implicated an alternative processing mechanism.

3.6.2 P200

Similar to the N170 (*see section 3.6.1 [page 55]*), to date, the P200 has not been investigated in association with aggressive or CU traits, specifically. Previously, P200 responses to neutral facial expressions presented in parallel with an aversive odour have been observed to be lower in small samples of psychopathic, relative to non-psychopathic males (Flor et al., 2002). Furthermore, in young adults derived from the community, P200 amplitudes were inversely correlated with emotional control, a key factor in regulating the emotional urges that precede an aggressive response (Meaux et al., 2014, *see chapter one, section 1.3.2 [page 9]*).

Whilst similar findings have not been replicated in a comparison between individuals with, and without bipolar disorder (Degabriele et al., 2011), this null finding might be an artefact of the P200 being measured from a composite of left and right occipito-parietal sites. This method might mask subtle hemispheric effects. Nevertheless, these findings suggest P200 attenuation to facial stimuli might be observed in samples characterised by aggressive/CU traits, and warrants more specific, hypothesis-driven investigation.

3.6.3 N200

In contrast to the limited number of investigations associating aggressive/CU traits with the N170 (*see section 3.6.1 [page 55]*) and P200 (*see section 3.6.2 [page 56]*), these personality traits have been investigated more frequently, using a variety of paradigms and stimuli, in association with the N200 and P300.

Although not observed in two samples of male offenders, relative to nonoffender controls (Barratt et al., 1997; Munro et al., 2007), inhibitory N200 responses have shown attenuation in forensic samples and samples characterised by externalising traits. Over a series of oddball and Flanker paradigms, which used simple keyboard characters/arrows as stimuli, male offenders characterised by high impulsivity and violent tendencies, relative to age-matched controls, evidenced smaller anterior midline inhibitory N200 responses (Chen et al., 2005, 2008, 2014). Furthermore, in a sample of juvenile male offenders, incarcerated for extreme and violent crimes, relative to controls, exhibited attenuated N200 responses during motor inhibition trials (Vilà-Balló et al., 2014). Such findings suggest that externalising and aggressive traits might detriment preparation of an inhibitory response. Similarly, in non-forensic samples, the inhibitory N200 has been attenuated in cohorts of children with, relative to without, ADHD (Dimoska et al., 2003; Pliszka et al., 2000). Whilst Pliszka et al. (2000) observed right anterior N200 attenuation in response to visual stop signals, Dimoska et al. (2003) observed midline anterior N200 attenuation to auditory stop signals.

Variation in N200 amplitude has also been observed as a function of CU traits (Kiehl, Bates, Laurens, Hare, & Liddle, 2006; Sumich et al., 2012). In Kiehl et al., the N200 evoked to auditory tones at midline sites, was larger in male inmates scoring high, relative to low, on measures of psychopathy. Furthermore, although an inverse relationship between age and N200 amplitude has been observed in young males with, and without ADHD, this association was not replicated in a cohort of males with CD (Sumich et al., 2012). In this study, the N200 was evoked to letters and was measured in response to background stimuli in order to better index cortical maturation without drawing on networks involved in executive functioning. Together, these findings suggest that CU traits might be associated with delayed cortical remodelling and poorly developed brain function as described in section 3.2.3 (page 43) (de Brito et al., 2009; Sumich et al., 2012).

In contrast to these findings, one investigation has reported N200 attenuation in offenders with psychopathy, relative to offenders with and without schizophrenia

(Kiehl, Smith, Hare, & Liddle, 2000). Although N200 augmentation was expected, this result could be an artefact of either aggressive traits not being controlled for, a lack of comparison to a non-offender cohort, and/or the use of an equal ratio of go/nogo stimuli. Stimulus frequency has been shown to moderate N200 amplitude (*see section 3.4.1 ([page 50]*), and a 50:50 ratio of stimuli has been thought to attenuate the inhibitory N200 response (Kok, 1986). Additionally, associations between CU traits and N200 amplitudes were not observed in samples of comorbid ADHD and CD outpatients (Du et al., 2006) or aggressive offenders with psychopathic traits (Munro et al., 2007), relative to controls. However, these findings might be explained by the N200 being measured over a large 200 ms epoch in Du et al. and baseline aggressive traits not being controlled for in Munro et al.

3.6.4 P300

First reported in alcohol dependency (Porjesz, Begleiter, & Garozzo, 1980; *see Polich et al., 1994 for review*), early P300 investigations found reduced P300 amplitudes in alcohol- (Bauer, O'Connor, & Hesselbrock, 1994), cocaine-(Bauer, 1997), and substance- (Iacono, Carlson, Malone, & McGue, 2002) dependent patients with, relative to without, antisocial personality traits. In these investigations, the P300 was evoked using visual (e.g. letter, light, schematic faces) oddball paradigms, and showed specific attenuation over anterior midline sites.

More recently, variation in P300 amplitude has been shown to index a wide spectrum of externalising disorders characterised by poor inhibitory control (Hicks et al., 2007; Iacono, Malone, & McGue, 2003; Pandey et al., 2012). For example, using variants of the visual oddball paradigm, parietal P300 amplitude attenuation has been observed in large cohorts of individuals with severe ADHD symptoms (Brandeis et al., 2002), and has been confirmed through metaanalysis (Szuromi, Czobor, Komlósi, & Bitter, 2011). Similarly, the auditory P300 is attenuated in schizophrenic patients, relative to individuals presenting schizophrenic-like symptoms, and in turn, controls (Trestman et al., 1996), with the visual inhibitory P300 attenuated in patients with BPD, relative to age- and sex-matched controls (Ruchsow et al., 2008). However, these results might be better explained by a single externalising factor. Patrick et al. (2006) observed an inverse correlation between parietal P300 amplitude, evoked using a visual oddball paradigm, and the breadth and severity of externalising symptoms.

This point being made, there exists a large literature documenting associations between aggressive/CU traits and the P300. In one three-way comparison of two inmate groups characterised by reactive/instrumental aggression and a cohort of non-forensic controls, visual P300 attenuation was observed in both inmate cohorts, at frontal sites (Barratt et al., 1997). Furthermore, after correcting for age, intelligence, response accuracy, and reaction time, P300 responses to schematic faces were attenuated in a large forensic cohort comprising reactive/non-reactive aggressors and non-violent offenders (Bernat, Hall, Steffen, & Patrick, 2007). However, whilst this finding suggests an effect of aggression on the P300 within a forensic setting, unlike findings reported in Barratt et al., this finding is restricted to offenders due to the absence of non-incarcerated controls.

Similar attenuation of the P300 has been observed in community samples characterised by impulsive aggressive tendencies, relative to non-aggressive controls (Gerstle et al., 1998). Gerstle et al. evidenced frontal P300 attenuation during an auditory oddball paradigm, but noted that half of their aggressive cohort reported extensive brain damage, which may have accounted for some variation in data. An inverse relationship between the visual P300 and aggression has also been evidenced in students (Bartholow, Bushman, & Sestir, 2006). Bartholow et al. suggested that P300 amplitude was inversely correlated with both experience of violent video games and laboratory-measured aggression.

Though some variance in P300 amplitude might be attributed to trait impulsivity, this position has been clarified in non-forensic samples. Using a three-stimulus

rotating-head paradigm, whereby the third factor was affective stimuli, Venables et al. (2011) reported an inverse relationship between impulsivity and anteriormidline P300 amplitude. However, as this finding only held when controlling for self-reported aggression, impulsivity might only moderate the P300 via trait aggression.

Conflicting findings have been reported in response to nogo stimuli. An absence of visual inhibitory P300 amplitude variation has been observed between samples of reactive- and non-aggressive offenders (Chen et al., 2005), alcohol dependant individuals with and without comorbid antisocial behaviour (Costa et al., 2000), and violent and non-violent juvenile offenders (Vilà-Balló et al., 2014). Similar findings have been reported between samples of aggressive and non-aggressive ADHD patients and controls using a facial affect paradigm whereby a motor response was required to a successive train of angry faces, but not if succeeded by a happy face (Meier et al., 2012). In this investigation, although nogo-P300 attenuation was evident in both delinquent and non-delinquent patients, relative to controls, P300 amplitude did not differ between aggressive and non-aggressive patients. However, it is possible that P300 differences might have been masked by ADHD symptoms, medication, and/or the samples including offenders characterised by both reactive and instrumental aggression.

With few exceptions (Bauer & Hesselbrock, 1999; Du et al., 2006), individuals characterised by instrumental aggression exhibit P300 amplitudes similar to (Patrick, 2008; Raine & Venables, 1987, 1988), or greater than (Bernat et al., 2007) the general population. Psychopaths (Flor et al., 2002; Raine, 1989), and individuals presenting CU traits such as fearlessness (Carlson & Thái, 2010) also evidence P300 augmentation. However, as the P300 of individuals with psychopathic traits, relative to controls, shows attenuation when evoked to nogo stimuli (Kim & Jung, 2014), CU traits might differentially moderate electrophysiological responses as a function of task. P300 attenuation during

motor inhibition might reflect individuals characterised by CU traits requiring greater neuronal resources to inhibit motor responses.

Compounding the inverse association between P300 amplitude and reactive aggression, P300 augmentation is absent in psychopaths during negative-affect-driven interpersonal encounters (Hicks et al., 2007; Kiehl, Hare, Liddle, & McDonald, 1999) and individuals reporting high self-centred impulsivity (Carlson, Zayas, & Guthormsen, 2009). Furthermore, P300 attenuation is only found to associate with antisocial behaviour when controlling for psychopathic traits (Gao & Raine, 2009). However, these findings fail to explain why P300 attenuation was observed in instrumental aggressors in Barratt et al. (1997).

Evidently, the P300 has utility for indexing CU traits, antisocial behaviour, and aggression (Krueger, 1999; Patrick, 2008). Nonetheless, whilst the stop-P300 is a key index of the effects of facial distress on motor inhibition in regards to the VIM, although extensively investigated during go and nogo trials, the stop-P300 has not been associated with aggressive or CU traits. Absence of such knowledge warrants further investigation. Once established, it would be possible to assess how aggressive and CU traits might moderate resulting electrophysiological responses.

3.7 Chapter Summary

In chapters one and two, the VIM was framed as a cognitive mechanism reliant on the ability to process facial distress in order to extinguish antisocial and aggressive behaviour. Chapter three used electrophysiological literature to index the distinct neurobiological stages of face processing and motor extinction described in chapter two, on an electrophysiological level. Specifically, structural, task salient, categorical, and evaluative stages of face processing were teased apart using N170, P200, N200, and P300 ERP responses, prior to discussing how threat- and distress-specific facial expressions might command greater mobilisation of neuronal resources. Moreover, N200 and P300 responses were discussed in terms of motor inhibition efficiency. To date, no paradigm has sufficiently indexed the VIM, namely indexing electrophysiological response to distress-induced motor extinction. As mirrored on a cognitive and neurobiological level, electrophysiological indices of distress processing and motor extinction are deficient in populations characterised by CU and aggressive traits, and so the use of a distress-induced motor extinction paradigm might have utility for indexing aggressive and CU traits on an electrophysiological level. Chapter four accumulates these concepts, and discusses how the VIM, as indexed on a cognitive, neurobiological, and electrophysiological level, might be moderated by diet, specifically dietary intake of omega-3 polyunsaturated fatty acids.

Chapter Four: Dietary impact on aggression

4.1 Introduction

Thus far, this thesis has discussed the possible cognitive, social, and biological underpinnings of reactive aggression, their associations with aggressive and CU traits, and how they might be indexed on an electrophysiological level. Chapter four delineates how dietary intake, more specifically omega-3 polyunsaturated fatty acids (PUFAs) intake, might moderate personality traits (i.e. aggression, CU) and their underpinning neurobiological networks.

4.2 Dietary Modulators of Aggression

A nutritious diet, including daily intake of micronutrients such as zinc, iron, and lithium, is essential for healthy development of the central nervous system (Portillo-Reyes, Pérez-García, Loya-Méndez, & Puente, 2014; Raine, 2013; Serfaty & de Velasco, 2014). Malnourished diets, defined by a lack of protein, fatty acids, vitamins, and nutrients (Tofail et al., 2008), might contribute to the emergence of psychopathologies, characterised by poor cognition and aggression (Georgieff, 2011; Grantham-McGregor et al., 2007; Pollitt, 2000).

For example, zinc deprived rodents (Takeda et al., 2008), and their offspring (Halas, Reynolds, & Sandstead, 1977) have evidenced increased aggression, relative to non-zinc deprived controls. In human samples, lower blood concentrations of zinc and copper have been reported in violent, relative to non-violent patients with schizophrenia (Tokdemir, Plota, Acik, Gursu, & Cikim, 2003), with increased ratios of copper to zinc reported in individuals presenting frequent incidents of assaultive behaviour, relative to a small cohort of non-violent controls (Walsh, Isaacson, Rehman, & Hall, 1997). Additionally, supplementation of lithium has been shown to decrease the duration and intensity of aggression in community (Campbell et al., 1995; Malone, Delaney, Luebbert, Cater, & Campbell, 2000; Malone, Luebbert, Pena-Ariet, Biesecker, &

Delaney, 1994) and forensic (Sheard, 1971; Sheard, Marini, Bridges, & Wagner, 1976) populations.

Fluctuations in dietary glucose metabolism have also been associated with impaired cognition and aggression (Gailliot & Baumeister, 2007). Previously, increased self-reported trait aggression has been reported in two student samples presenting comorbid hypoglycaemia - a medical condition associated with diminished glucose content (Benton, Kumari, & Brain, 1982; Donohoe & Benton, 1999). However, although assessed in a small sample, differences in blood glucose have not been observed when contrasting aggressive and nonaggressive patients with brain injury after controlling for weight, age, and time since injury (Stanislav, Crismon, & Childs, 1998). Elsewhere, this glucose/aggression association has only been observed to manifest under mild provocation (Denson, von Hipple, Kemp, & Teo, 2010).

4.3 Polyunsaturated Fatty Acids

Essential fatty acids (EFAs) benefit brain function during childhood and early adulthood (Bhatia et al., 2011, Luchtman & Song, 2012; Yehuda, 2003) and are thought to play a key role in the development of cognitive and emotional function (Marszalek & Lodish, 2005). Between .25 and 3 g EFA intake is recommended daily for healthy development (Food and Agriculture Organisation, 2010).

4.3.1 Composition, Intake, and Cell-level Importance

The two main families of EFAs - omega-3s and -6s - comprise 30 to 35% of total brain fatty acids in humans (Luchtman & Sog, 2012; SanGiovanni & Chew, 2005). Following dietary consumption, EFAs are desaturated and elongated via the addition of double bonds and [*two*] carbon units (SanGiovanni & Chew, 2005). This procedure produces long-chain polyunsaturated fatty acids (LCPUFA), which are structurally classified by [1] the number of carbon units,

[2] the number of double bonds, and [3] the location of the first double bond relative to the methyl omega terminal of the chain (SanGiovanni & Chew, 2005).

Around 50% of the dry brain weight of adults is made up of lipids, 35% of which are LCPUFAs (Serfaty & de Velasco, 2014). Parent LCPUFAs include the omega-6 linoleic acid (LA, 18:2n-6) and the omega-3 alpha-linoleic acid (ALA; 18:3n-3). LA contributes to the synthesis of the second order omega-6 fatty acids arachidonic acid (AA, 20:4n-6) and adrenic acid (22:4n-6), which are prominent in grey and white matter, respectively. ALA is synthesised into eicosapentaenoic acid (EPA, 20:5n-3) and docosahexaenoic acid (DHA, 22:6n-3). EPA is the precursor to DHA metabolism (SanGiovanni & Chew, 2005), and accounts for between 15% (Aid, Vancassel, Poumes-Ballihaut, Chalon, Guesnet, & Lavialle, 2003; Benton, 2007) and 30% (Innis, 2007; Salem Jr., Litman, Kim, & Gawrisch, 2001) of fatty acids in grey matter.

Conversion of ALA to EPA and DHA is thought to be less than 5% in humans (Brenna, 2002; Holub, 2002), and may be decremented further by, among other variables, alcohol consumption (Serfaty & de Velasco, 2014). During metabolism, omega-3s and -6s undergo cyclooxygenase enzyme competition (Arbuckle, MacKinnon, & Innis, 1994; Hrboticky, MacKinnon, & Innis, 1990), whereby EPA and DHA act to reduce AA levels (Endres et al., 1989; SanGiovanni & Chew, 2005). Omega-6 augmentation follows omega-3 deficiency (Calder & Grimble, 2002; Salem et al., 2001).

Through the absence of delta-15 and -12 desaturase enzymes, vertebrates are unable to synthesize LCPUFA *de novo* (Gadoth, 2008; SanGiovanni & Chew, 2005; Sinn, Milte, & Howe, 2010) and so LCPUFAs must be obtained through dietary omega-3 and -6 (McNamara et al., 2007; Owen, Rees, & Parker, 2008; Smith, 2008; Su et al., 1999). Omega-6s are predominantly derived from poultry, eggs, and vegetable oils (Luchtman & Sog, 2013), which are found in processed foods that often undergoing hydrogenation and the removal of omega-3s in order to prolong food preservation (James, Montgomery, & Williams, 2011; Singh, 2005). Omega-3s are derived from seeds (e.g. flaxseed and rapeseed), grains, nuts (e.g. walnuts), green leafy vegetables, fatty fish, and shellfish (Innis, 2008; James, et al., 2011; Kris-Etherton et al., 2000; Marszalek & Lodish, 2005). Independent contributions of DHA and EPA from fish are measured as a function of the age, diet, species, and farming location of the fish, as well as the season they were caught (Kolakowska, Zienkowicz, Domiszewski, & Bienkiewicz, 2006).

For optimal psychological and physical well-being, humans are expected to consume a 4:1 diet of omega-3 and -6 (Gomez-Pinilla, 2008; Simpoulos, 2002). Recent decades have marked a reduction in omega-3 consumption in Western diets, which is now thought to comprise 5 to 20 times higher omega-6s than -3s (Ailhaud et al., 2006; Appleton et al., 2007; Bazan, 2005). In particular, Hallahan & Garland (2004) suggest that the modern Western diet mainly consists of meat and saturated fats, with comparatively little fresh vegetables and fish consumed.

Heightened omega-6 intake, at the expense of omega-3, is associated with health deficiencies such as cancer (Anderson & Ma, 2009; Schram et al., 2007) and cardiovascular diseases (Hibbeln Jr., Ferguson, & Blasbalg, 2006; Kris-Etherton, Harris, & Appel, 2002; Mozaffarian & Wu, 2011; Ryan et al., 2010; Sinclair, Begg, Mathai, & Weisinger, 2007), as well as the development of psychiatric disorders characterised by emotional deregulation (Conklin et al., 2007; Freeman et al., 2006; Peet & Horrobin, 2002; Simopoulos, 2002). Additionally, increased dietary intake of omega-6 is comorbid with zinc deficiency (Bourre, 2006), highlighted in section 4.2 (page 63) to associate with aggression.

Omega-3 selectively accumulates in the foetal central nervous system between the final trimester (Green, Glozman, Kamensky, & Yavin, 1999) and up to two years after birth (Martinez, 1992). Omega-3s are essential for cell membrane development (Riemer, Maes, Chrisophe, & Rief, 2010). In particular, whilst EPA, via eicosanoids production, is important for anti-inflammation, blood clotting, and neuroimmunology (Sinclair et al., 2007; Sinn & Howe, 2008), DHA moderates membrane properties such as fluidity, thickness, plasticity, and cell permeability (Jump, 2002; Jumpsen & Clandinin, 1997; Tanabe et al., 2004) and protects cells from death (Carlson, 2001; Crawford, 2006; Innis, 2000; Itomura et al., 2005; Marszalek & Lodish, 2005; Michael-Titus & Priestley, 2014; Uauy & Dangour, 2006). Membrane changes influence how cells incorporate proteins (Bazan, 2005; Luchtman & Sog, 2013), and so DHA-rich membranes help shape brain circuitry by increasing molecular information processing (Michael-Titus & Priestley, 2014; Serfaty & de Velasco, 2014). In essence, DHA promotes inter- and intra-cellular communication (Bazan, 2005; Litman, Niu, Polozova, & Mitchell, 2001; Uauy & Dangour, 2006).

Foetal accruement of fatty acid coincides with synaptic development during both prenatal and postnatal growth (Carlson, 2001; Crawford et al., 2003; Green et al., 1999). DHA availability decreases with age (Soderberg, Edlund, Kristensson, & Dallner, 1991) and contributes to cognitive decline and the breakdown of cell communication (Danthiir, Burns, Nettelbeck, Wilson, & Wittert, 2011). Fatty acids are vital for brain growth and development (Conklin et al., 2007), and benefit early stages of brain modelling (Chen & Su, 2012; de Velasco et al., 2012). In particular, DHA facilitates neuronal growth through dendritic branching, synaptic formation, and myelination (Calderon & Kim, 2004; Sinn & Howe, 2008; Yehuda, Rabinovitz, Carasso, & Mostofsky, 2002). Early omega-3-deficient cell membranes contribute to synaptic degradation (Suzuki, Manabe, Wada, & Crawford, 1997) and dysfunction in synaptic pruning (Carlson, 2001). As described in chapter three, section 3.6.3 (page 57), synaptic pruning is essential for cell specialisation. However, at least in rodents, effects of neonatal omega-3 deprivation on cell growth appear reversible through DHA and EPA supplementation during infanthood (de Velasco et al., 2012).

4.3.2 Polyunsaturated Fatty Acids and Aggression

Throughout the last quarter of a century, omega-3 deficiency and low omega-3:-6 ratios have been associated with suboptimal child development and behavioural pathologies (Crawford, Hassam, & Williams, 1976; Fiennes, Sinclair, & Crawford, 1973). More recently, review publications have not only linked low omega 3 intake/blood measures to impulsive, aggressive behaviours, but have suggested these behaviours may be resolved through dietary supplementation (Garland & Hallahan, 2006). This section highlights how omega-3 might moderate aggression at baseline and through supplementation, in animal and human samples.

4.3.2.1 Correlational Studies

Whilst not establishing cause and effect, correlating two variables together within, or across a given population(s) indicates the relationship strength between those variables. Previously, a cross-national analysis of 26 countries has suggested an association between low fish consumption (omega-3s), and high rates of homicide mortality (Hibbeln, 2001; The World Health Organisation, 1995). Although this result survived after correcting for population, as some of the countries included in the sample reported both high omega-3 intake and high homicide rates, it is important to consider the importance of omega-3 intake in the context of co-variates.

Elsewhere, outward aggression (Meyer et al., 2015), and self-harm (Garland et al., 2007) have been inversely correlated with blood levels of omega-3. In Meyer et al. (2015), though only significant at trend level, omega-3 was inversely correlated with self-reported physical aggression in a large cohort of males from an offender correctional facility. In Garland et al. (2007), plasma concentrations of omega-3, as well as individual levels of EPA and DHA, were inversely associated with incidents of self-harm in patients presenting at an accident and emergency unit. These correlations held after correcting for age, sex, substance use (i.e. alcohol and nicotine), and economic status.

In particular, DHA has been inversely associated with both psychometricassessed aggression in a sample of patients derived from a forensic mental health clinic (Zaalberg et al., 2015), and hostility in a larger sample of young males taking part in a longitudinal health assessment (Iribarren et al., 2004). However, whilst additional associations were reported between aggression and omega-3:-6 ratios in Zaalberg et al. (2015), a comparative association was not observed after correcting for age, sex, race, education, employment, BMI, and substance use in Iribarren et al. (2004).

In contrast, very few investigations have assessed the relationship between omega-3 intake and CU traits. One such investigation, Gow, Vallee-Tourangeau et al. (2013), reported an inverse association between psychometricassessments of callousness, uncaring, and unemotional traits, and total blood measures of omega-3, EPA, and DHA at trend-level significance. However, in this investigation, [1] the CU/omega-3 association was only observed in children with, but not without ADHD, and [2] unlike the other investigations noted above, omega-3 was not associated with psychometric-based trait aggression in either ADHD or control group. Elsewhere, although not a direct measure of CU traits, omega-3 intake has been shown to be inversely associated with unfair offers of remuneration in a large student sample (Emanuele, Brondino, Bertona, & Gerald, 2009).

4.3.2.2 Group Comparison Studies

Group comparison analysis has been used to explore relationships between LCPUFA and aggression in both animal and human samples. In rodents, reduced omega-3 (DeMar et al., 2006) and increased omega-6 (Raygada, Cho, & Hilakivi-Clarke, 1998) has been associated with more frequent episodes of blocking behaviours, over longer durations, during resident-intruder paradigms. Whereas DeMar et al. compared Long-Evans rat pups fed an omega-3 deficient, relative to adequate, diet for 15 weeks, Raygada et al. contrasted offspring of mothers fed either a standard, or omega-6-rich diet during pregnancy. Similarly, in a canine sample, aggressive, relative to non-aggressive German Shepherd dogs presented low omega-3 levels, and sub optimal omega-3:-6 ratios (Re, Zanoletti, & Emanuele, 2008). However, this association was specific to DHA, not EPA.

In humans, omega-3 deficiency has been associated with externalising behaviours. For example, in samples of cocaine addicts, recruited from a substance abuse clinic, patients with, relative to those without, histories of aggression were found to have lower serum levels of total omega-3 and DHA (Buydens-Branchey, Branchey, McMakin, & Hibbeln Jr., 2003). Although this finding suggests omega-3 deficiency might facilitate aggression, this result should be discussed in light of the aggressive group consisting of only 6 patients, and the nonaggressive group containing some patients who evidenced low omega-3 serum levels. Moreover, after separating a cohort of adolescent males on the basis of their omega-3 intake, those reporting low omega-3 evidenced increased parent- and teacher-rated behavioural and learning problems, including hyperactivity, impulsivity, and poor conduct control (Stevens, Zentall, Abate, Kuczek, & Burgess, 1996). Elsewhere, although omega-3 has been measured in males derived from the community both with, and without histories of inflicting physical harm (Umhau et al., 2006), this investigation only assessed relationships between DHA and folic acid, not intergroup variation.

In a forensic sample, low levels of omega-3, and high levels of omega-6 were found in violent offenders characterised by frequent, alcohol-fuelled episodes of aggression, relative to non-aggressive controls without prior alcohol use problems (Virkkunen, Horrobin, Jenkins, & Manku, 1987). Furthermore, EPA deficiency has been associated with trait aggression in individuals with major depressive disorder with, but not without, comorbid substance abuse (Beier et al., 2014). However, as alcohol consumption is thought to compromise the structural composition of fatty acids (Pawlosky, Bacher, & Salem, 2001), controlling for alcohol use when assessing the relationship between fatty acids and aggression has importance.

4.3.2.3 Supplementation Studies

Whilst correlational and group studies are useful for identifying associations between omega-3 intake/blood measures and aggression at baseline and in contrast to nonaggressive controls, supplementation studies have utility for assessing the direct impact of omega-3 on behaviour, traits, and underlying neurobiological networks. This section critically evaluates the effectiveness of omega-3 supplementation on aggression in human samples.

Recently, the synergistic effects of mineral and DHA supplementation on aggression have been assessed using a randomised-controlled design (Long & Benton, 2013). In adult males without previous histories of aggression, daily supplementation of approximately 670 mg DHA over a twelve-week period was observed to reduce the use of aggressive dialogue in a picture-frustration paradigm, as well as impulsivity as indexed by a visual number-based stop-go task. However, there was no effect of DHA supplementation on either self-reported verbal or physical aggression, and DHA did not show additive benefit when supplemented alongside minerals. In samples with ADHD, DHA-rich supplementation over a two- (Hamazaki & Hirayama, 2004 [510 mg DHA; 120 mg EPA]) and four- (Stevens et al., 2003 [480 mg DHA; 80 mg EPA]) month period has been shown to improve parent- and teacher-rated aggression and oppositional/defiant behaviour, respectively. However, as ADHD diagnosis was not verified in Stevens et al., this particular result might better reflect DHA-driven alleviation of ADHD symptoms.

Interestingly, though twice the dose delivered in Hamazaki & Hirayama (2004) and Stevens et al. (2003), neither a DHA- (1032 mg DHA; 264 mg EPA), nor EPA-rich (108 mg DHA; 1109 mg EPA) supplement, in contrast to a sunflower oil placebo, improved oppositional behaviour in an ADHD sample over a four-month period (Milte et al., 2012). An absence of improvement to ADHD symptoms was also reported in a large sample of children, following daily supplementation of 345 mg DHA, over a four-month period (Voigt et al., 2001). Moreover, DHA-rich supplementation (510 mg DHA; 120 mg EPA) to the food of

166 children was in fact observed to *increase* task-based measures of aggression towards others, as measured by a picture-frustration task (Itomura et al., 2005). This effect was only observed in male, not female participants, and could possibly be an artefact of low aggression scores at baseline.

Null effects of DHA supplementation have been reported elsewhere. In Hamazaki et al. (1996), a small sample of students were administered either a large daily dose of DHA (1500-1800 mg) or a soybean placebo over a threemonth period. Whilst no effect of DHA was observed, students administered the soybean placebo evidenced increased rates of aggression, as indexed by aggressive dialogue in a picture-frustration task. As this experiment was conducted over an exam period, Hamazaki et al. explained the results in terms of DHA possibly having protective properties during stressful situations. As discussed in section 1.3.1 (page 7), environmental triggers such as stressors may precede aggression.

In respect to the effect of EPA supplementation on aggression, there exists very little literature. One such example, Zanarini & Frankenburg (2003), reported that relative to a placebo, daily supplementation of 1000 mg EPA over a two-month period reduced both verbal and physical aggression in female, BPD patients. Further support of the effect of omega-3 on aggression comes from investigations using a mixed EPA and DHA supplement.

Incidents of aggression have been reduced in two clinical trials using a composite omega-3 supplement; one based in the United Kingdom (Gesch et al., 2002) and one based in the Netherlands (Zaalberg, Nijman, Bulten, Stroosma, & van der Staak, 2010). Although only a small dose of daily supplementation was administered (44 mg DHA; 80 mg EPA) between 142 to 172 days, Gesch et al. (2002) observed a 26.3% reduction in violent incidents in a large sample of male offenders. However, as the active supplement also included 13 vitamins and 12 minerals, it is unclear how much variance in behaviour was explained by omega-3 alone. Previously, there have been mixed

reports as to the effectiveness of mineral and vitamin supplementation on aggression. Whereas Long & Benton (2013) found no effect of mineral supplementation, Schoenthaler et al. (1997) elicited a 28% decrease of incidents of violence in imprisoned juveniles following supplementation of 12 vitamins and 11 minerals, relative to placebo. In the Dutch trial, Zaalberg et al. (2010) found a 34% reduction in aggression and rule-breaking behaviour using a much larger omega-3 supplement than Gesch et al. (400 mg DHA; 400 mg EPA).

Recently, a randomised-controlled trial assessed the impact of omega-3 supplementation on aggressive and CU traits (Raine, Portnoy, Liu, Mahoomed, & Hibbeln, 2015). In this trial, a large sample of children aged between 8 and 16 years received a daily fruit juice drink (1000 mg), supplemented with either omega-3 (300 mg DHA; 200 mg EPA) or a placebo matched for weight, colour, and taste. In the active group, child-reported reactive aggression, parent-reported antisocial behaviour (e.g. CU traits, narcissism, and impulsivity), and psychopathic traits (e.g. cold-heartedness and fearlessness) were reduced over a six-month period.

Although clinical trials seem to indicate an omega-3-evoked reduction in aggression assessed using psychometric measures, simple behavioural paradigms and both parent and teacher ratings, there are some inconsistent findings. For example, in a sample of 47 self-harmers, although a large, composite omega-3 supplement (900 mg DHA; 1200 mg EPA) improved symptoms of depression and suicidality over a four-month period, there was no reduction in aggression or impulsivity (Hallahan, Hibbeln, Davis, & Garland, 2007). Moreover, there was no difference in parent- and teacher-rated aggression in a sample of 40 children with ADHD following a DHA-rich (500 mg DHA; 100 mg EPA), relative to olive oil-rich dietary plan matched for taste and smell, over a two-month period (Hirayama, Hamazaki, & Terasawa, 2004). However, this investigation assessed aggression using only two questions requiring congruent answers from parents and teachers and so likely limited the variance in data.

4.3.2.4 Fatty Acids/Aggression Summary

Though not conclusive, baseline aggression in animals and humans might be a function of omega-3 blood content, as evidenced in correlational and group comparison studies. Furthermore, *de novo* omega-3 supplementation might benefit the reduction of aggression under certain circumstances, such as presence of a stressor. Although stressors were not implicitly outlined in studies conducted in forensic settings (Gesch et al., 2002; Zaalberg et al., 2010), stress might result from unfavourable physical and environmental conditions. Whilst omega-3 might reduce aggression, possibly through moderation of neurotransmission (Parletta, Milte, & Meyer, 2013), it remains unclear whether this effect is global, or influential at a specific point during the reactive aggressive processing route outlined throughout this thesis. The following sections suggest how omega-3 might modulate stages of '*face processing*', '*negative affect*', and '*executive functioning*'.

4.4 Face Processing

DHA is concentrated in sensory and vascular retinal cells and comprises around 20% of retinal fatty acids, which enrich membrane disks in the outer photoreceptors (Innis, 2008; SanGiovanni & Chew, 2005). Prenatal omega-3 deficiency has been shown to detriment long-term primate retinal function (Anderson, Neuringer, Lin, & Connor, 2005), with specific DHA deficit in humans associated with visual system dysfunction (SanGiovanni & Chew, 2005; Uauy et al., 2001) and impaired visual attention (Fedorova & Salem, 2006). Postpartum, omega-3 supplementation of infant formula benefits developmental needs such as visual acuity (Birch et al., 2010; Uauy et al., 2003).

On a biological level, although omega-3 has been shown to be rich in brain regions involved in face processing, such as the temporal lobe (McNamara, 2010), to date, no studies have associated omega-3 concentrations in specific brain networks (e.g. fusiform gyrus-STS-amygdala) in regards to face processing efficacy. However, on an electrophysiological level, EPA and DHA have been shown to moderate distinct components of the face-evoked ERP, in samples of children and adolescents with ADHD symptoms.

Though not generalisable to non-ADHD, Gow, Matsudaira, et al. (2009) evidenced differential effects of EPA and DHA on the N170 in a sample of ADHD children. Here, red blood cell lipid levels of DHA were positively correlated with the right temporal N170 evoked to a 10 ms presentation of facial fear. This finding was argued to reflect activation of underlying neural networks associated with empathy (*see chapter two, section 2.5 [page 29]*), and suggests a benefit of DHA for improving neural transmission of these networks in response to facial distress. Conversely, EPA, but not DHA, positively associated with N170 and P300 responses to happy and happy minus fearful/sad facial expressions, respectively, at anterior midline sites. This finding suggests orientation of attention towards happy faces over several face processing stages. Additionally, whilst not concurrent with the notion that fatty acids benefit face processing, EPA was inversely associated with N170 amplitude over left temporal and bilateral occipital sites to happy and sad faces, respectively, and so warrants further investigation.

Whilst effects of omega-3 on face-evoked ERPs have also been observed in Gow, Sumich et al. (2013), this investigation did not report correlations between blood fatty acid levels and N170, P200, and N200 responses, in either ADHD or control samples. However, had this association been made, results would have been limited due to ERPs being measured over midline, and not bilateral, temporal sites as suggested in chapter three, sections 3.2.1 (page 39) and 3.2.2 (page 41). No correlations between omega-3 fatty acid levels and the

P300 responses to facial expressions reached statistical significance in either ADHD or control sample.

4.5 Negative Affect

As discussed in section 1.3.2 (page 9), negative affect plays a role in the antecedence of an aggressive response. This section describes the associations between omega-3 and mood in general, before noting correlational and supplementation studies that have investigated how similar underlying processes might moderate anger, specifically.

The role of omega-3 in the moderation of affective processes of rodents, children, adolescents, and adults is well documented (see Freeman et al., 2006 for review), and is likely a function of omega-3-specific inflammatory effects (Raison & Miller, 2013). Omega-3 intake is thought to benefit affective processes, such as depression (Nemets, Nemets, Apter, Bracha, & Belmaker, 2006), and is inversely related to the severity of depressive symptoms across several age ranges (Freeman et al., 2006; Owen et al., 2008; Montgomery & Richardson, 2008; Tiemeier, Breteler, van Popele, Hofman, & Witteman, 2003). Furthermore, DHA (Edwards, Peet, Shay, & Horrobin, 1998; Fedorova & Salem, 2006) and EPA (Zanarini & Frankenburg, 2003) are reduced at baseline in individuals reporting depressive traits. Although daily supplementation of 1000 mg EPA (Peet & Horrobin, 2002) and 9600 mg DHA (Su et al., 2003) has been shown to reduce depression symptoms, Su (2009) suggests EPA, relative to DHA, might have greater treatment benefit. Omega-3 supplementation did not reduce depression in Marangell et al. (2003) or Silvers, Woolley, Hamilton, Watts, & Watson (2005).

4.5.1 Correlational Studies

Similar to aggression (*see section 4.3 [page 64]*), levels of blood-measured omega-3 have been inversely associated with anger-related traits. Though not a function of specific metabolites, increased composite omega-3 has been

positively associated with self-reported anger and hostility in male offenders (Meyer et al., 2015). Furthermore, in drug naive patients with schizophrenia, red blood cell lipid levels of DHA and EPA were inversely correlated with hostility (Watari, Hamazaki, Hirata, Hamazaki, & Okubo, 2010). However, whilst this investigation sampled patients without histories of malnutrition or drug/alcohol abuse, correlations did not control for current dietary intake and lacked a nonclinical control sample.

In community samples, DHA, but not EPA, serum levels have been positively associated with psychometric-measured agreeableness, after adjusting for age, sex, and race (Conklin et al., 2007). Additionally, though specific for black respondents, Iribarren et al. (2004) evidenced inverse correlations between hostility and retrospective self-reported levels of total omega-3, DHA, and EPA, after controlling for alcohol use. However, as [1] the self-report assessment of omega-3 only took into account intake over the previous month, and [2] due to the nature of the investigation, hostility and omega-3 were reported two years apart, this relationship assumes stability of both omega-3 consumption and trait hostility.

4.5.2 Supplementation Studies

Compared to investigations testing the effect of omega-3 supplementation on self-reported, task-induced, and historical incidents of aggression (*see section 4.3 [page 64]*), there is a paucity of research into the effect of omega-3 on anger. Over two decades ago, reduced hostility was reported in individuals fed a low fat/high carbohydrate and fish diet (high omega-3:-6 ratio), relative to those fed a '*standard Western diet*' (low omega-3:-6 ratio) over a five-year assessment (Weidner, Connor, Hollis, & Connor, 1992). However as this investigation lacked an accurate assay of omega-3 intake, direct group comparisons are limited, with some variance possibly accounted for by other dietary factors.

Elsewhere, following a three-month period of daily EPA-rich supplementation (500 mg DHA; 2250 mg EPA; 250 mg other LCPUFA), relative to a soybean oil placebo, self-reported feelings of anger and anxiety were reduced in a small sample of males recruited from substance abuse clinics (Buydens-Branchey, Branchey, & Hibbeln, 2008). This finding was also replicated using a comparable design, supplement, and sample elsewhere (Buydens-Branchey & Branchey, 2008). In this latter investigation, following the three-month supplementation period, anger scores in a subsample of patients remained attenuated for a further three months post supplementation.

Anger has also been reduced as a function of daily EPA-rich omega-3 supplementation (800 mg DHA; 1600 mg EPA; 400 mg other LCPUFA) over a period as short as 35 days (Fontani et al., 2005a, 2005b). These investigations found no effect of sex, but also evidenced benefit for self-reported anxiety and depression - compounding the possible role of omega-3 in mood. However, in both cases, investigations were limited to small sample sizes. Additionally, although containing a large dose of EPA, DHA-rich (6000 mg DHA; 1200 mg EPA) supplementation of omega-3 over twelve weeks was found to decrease self-reported irritation in a small sample of violent schizophrenic patients (Légaré et al., 2007).

4.5.3 Underlying Mechanisms

Moderation of anger as a function of omega-3 intake suggests an effect of omega-3 on affect-serving regions such as the amygdala (Conklin et al., 2007). Although to date there have been no investigations into amygdaloidal omega-3 concentrations and anger, specifically, chronic omega-3 supplementation has been shown to increase amygdaloidal cell concentrations of EPA and DHA in rodents (Taha et al., 2013). In humans, baseline amygdala omega-3 concentrations have been shown to be equivocal in patients with schizophrenia, bipolar disorder, and MDD evidencing affect and behavioural problems, relative to controls (Hamazaki et al., 2010). In chapter two, section 2.4 (page 26), reactive aggression was described through amygdala projections to the hypothalamus alongside a disruption to the hippocampal-mediated HPA axis. Omega-3 intake might moderate HPA efficacy by altering oxidant membrane levels (Conklin et al., 2007; Songur et al., 2004). In rodents, hypothalamic cells of rodents fed an omega-3-restricted, relative to a control diet, comprised lower DHA and increased omega-6 concentrations (Chen & Su, 2013; Li et al., 2006). However, it remains unclear as to whether hypothalamic DHA deficiency can be replenished. Whilst omega-3 supplementation over a twenty-four-week period did not fully replenish DHA in Li et al. (2006), full omega-3 replenishment was achieved in Chen & Su (2013) after seven weeks.

Hypothalamic activation moderates the pituitary gland through expression of CRF (Ressler & Nemeroff, 2000), a neuropeptide hyperexpressed in omega-3 deficient domestic abusers (Hibbeln, Bissette, Umhau, & George, 2004). The mechanism underpinning fatty acid-induced increases in CRF expression is thought to be AA metabolites prostaglandin E2 and F2. These metabolites control the release of CRF messenger ribonucleic acid (RNA) (Cambronero, Rivas, Borrell, & Guaza, 1992; Hibbeln et al., 2004), and share a positive correlation with omega-6:-3 ratio (Cambronero et al., 1992; Hoffmann et al., 1986; Lands et al., 1992). Specifically, EPA, relative to DHA, has been shown to suppress greater RNA expression (Mickleborough et al., 2009).

Although DHA and EPA have been shown to moderate the hypothalamus and pituitary gland respectively, it is DHA that seemingly moderates dendritic branching within hippocampal neurons - the final node of the HPA axis (Wurtman, 2008, *see chapter two, section 2.4 [page 26]*). The hippocampus responds to adrenal gland cortisol release, which is increased in primates with greater omega-3 blood concentrations (Laugero et al., 2011). Functionally, DHA deficiency disrupts synaptic fluidity in the hippocampus at a cellular level (Aid et al., 2003; He et al., 2009), and is evident through long-lasting synaptic dysfunction in omega-3 deficient rodents (Patten et al., 2013). There is some

evidence to support DHA supplementation playing a preventative role in stressinduced hippocampal dysfunction in rat pups (Feng et al., 2012).

Structurally, omega-3 deficiency attenuates hippocampal cell body size (Ahmad et al., 2002) and impairs hippocampal cell migration in rat pups fed an omega-3 deficient diet (Yavin, Himovichi, & Eilam, 2009). During foetal development, this reduction in hippocampal outgrowth might predominantly relate to DHA, but not EPA deficiency (Cao et al., 2009). Conversely, chronic supplementation of omega-3 might augment hippocampal volume in rodents fed omega-3 deficient (Venna et al., 2009) and non-omega-3 deficient diets (Calderon & Kim, 2004). Additionally, DHA supplementation might restore cellular growth impairment (Cao et al., 2009; Yavin et al., 2009), though some studies argue only for females (Patten et al., 2013).

4.6 Executive Functioning

Omega-3 has been shown to moderate processes relating to executive functioning, including attention, working memory, and motor inhibition (Branca, 2006; Johnson, Ostlund, Fransson, Kadesjö, & Gillberg, 2009; Karr, Alexander, & Winningham, 2011; Sinn & Bryan, 2007). As discussed in chapter one, section 1.3.3 (page 11), executive functioning typically moderates reactive aggressive behaviour in humans.

Attention, the ability to orient towards target stimuli is poor in ADHD, a disorder first related to EPA deficiency in Colquhoun & Bunday (1981). Subsequently low omega-3 levels have been reported in both child (Bekoroglu et al., 1996) and adult (Young, Conquer, & Thomas, 2005) samples with ADHD. EPA-rich supplementation has been found to improve attention over a four-week period in a sample derived from the community (Fontani et al., 2005, 2009 [800 mg DHA; 1600 mg EPA; 400 mg other LCPUFA]) and over a four-month period in two samples of children evidencing ADHD symptoms with and without oppositional traits (Gustafsson et al., 2010 [500 mg EPA]). Elsewhere, some (McNamara et al., 2010b; Milte et al., 2012), but not all (Voigt et al., 2001)

investigations suggest DHA might benefit attention as well. However, as ADHD patients, relative to non-clinical controls, show reduced blood DHA concentration at baseline (Burgess et al., 2000), it is possible that this disparity is an artefact of individuals characterised by poor executive functioning having a decremented ability to metabolise EPA into DHA.

On a neurobiological level, EPA-rich (600 mg DHA; 1800 mg EPA), relative to DHA-rich (1650 mg DHA; 550 mg EPA) supplementation over a four-week period has been shown to better improve cognitive efficiency, as evidenced by lower activation of the left ACC during a spatial working memory paradigm (Bauer et al., 2014). Previously, working memory, as indexed by digit span and letter number sequencing, has been shown to benefit from EPA-rich supplementation (600 mg EPA; 120 mg DHA), relative to an olive oil placebo, over a sixteen-week period (Widenhorn-Müller, Schwanda, Scholz, Spitzer, & Bode, 2014). However, this finding is limited to children diagnosed with ADHD.

4.7 Chapter Summary

This chapter outlined the possible benefit of omega-3 on health in general, before highlighting investigations that have used correlational analysis, group comparisons, and supplementation to indicate relationships between omega-3 and reactive aggression/CU traits. Although some conflicting findings have been presented, it is apparent that nutrition, specifically omega-3 dietary intake, might benefit individuals characterised by aggressive and antisocial behaviour. In relation to the VIM, the role of omega-3 was discussed in association with threat and distress processing, negative affect, and executive control. However, whilst omega-3 might facilitate face processing and its associated neurobiological underpinnings, to date, there is no documented investigation into the role of omega-3 fatty acids on distress-induced motor extinction, and very little understanding of how these processes might reflect on an electrophysiological level. As EEG is a useful tool for indexing efficacy of neuronal networks underpinning aggression and CU traits, expansion of this literature has both theoretical and practical importance.

Chapter Five: Thesis aims and the delineation and justification of methods

5.1 Introduction

This chapter is split into two sections. The first delineates the aims of this thesis and the second details the self-report measures used in conjunction with EEG methodology to fulfil these aims.

5.2 Thesis Aims

This thesis aims to:

- 1. Explore associations between physical aggression and CU traits in an adult community sample
- Develop a sensitive experimental paradigm for characterising electrophysiological indices of different processing components of the VIM
- Explore associations between physical aggression/CU traits and electrophysiological responses to face processing (i.e. N170, P200), and distress-induced motor extinction (i.e. stop-N200, stop–P300)
- 4. Explore associations between omega-3 intake, physical aggression, and CU traits
- Explore associations between omega-3 intake and electrophysiological responses to face processing (i.e. N170, P200), and distress-induced motor extinction (i.e. stop-N200, stop–P300)

To date, though there exists a relatively comprehensive literature associating psychopathy-related personality traits (e.g. aggressive, CU traits) with atypical activation of the VIM - and their possible benefit from an omega-3-rich diet – investigation of these associations on an electrophysiological level appears to have been overlooked. Development of a novel EEG paradigm that allows for the simultaneous assessment of the distinct processing components of the VIM

not only facilitates investigation into how such traits might differentially moderate particular stages of the VIM, but also allows for delineation of more temporally accurate effects that might otherwise be difficult to identify on a behavioural level. Moreover, with increasing interest into the effect of omega-3 dietary intake on both aggressive and CU traits (Buydens-Branchey & Branchey, 2008; Gow, Vallee-Tourangeau et al., 2013; Raine et al., 2015), questions asked within this thesis will benefit understanding of how omega-3 might not only associate with VIM-related traits, but also distinct processing components of the VIM as measured using EEG.

5.3 Psychometric and Quantitative Measures

This section describes the scales used throughout this thesis to index physical aggression, CU traits, omega-3 intake, and alcohol/drug use. The design, history, and validation of each measure are discussed, alongside their respective limitations. Development and piloting of the FAST used in conjunction with these measures is described in chapter seven (from page 102).

5.3.1 Aggression Questionnaire v.2

The Aggression Questionnaire (AQ) Version 2 (Buss & Warren, 2000) measures self-perceived aggression and anger - the propensity to aggress (e.g. "I have threatened people I know"). The AQ is written using accessible language, can be completed by children and adults, and is suitable for use in several populations (e.g. community, clinical, forensic, military). The 34-item scale is scored on a five-point Likert scale ranging from 1 ('*Not at all like me*') to 5 ('*Completely like me*'), and can be completed both on paper and online. Both mediums are used in this thesis. Total scores range from 34 to 170, with higher scores indicative of greater aggression. Twelve item pairs index inconsistent responding, with responses containing five or more pairs differing by more than one point removed. The AQ is split into the five subscales: physical aggression, verbal aggression, anger, hostility, and indirect aggression. Whereas physical and verbal aggression account for the manifestation of an aggressive response, anger and hostility index the affective and cognitive components of aggression, respectively (Buss & Perry, 1992; Buss & Warren, 2000). Indirect aggression represents the instrumental component of an aggressive response. Total scale reliability is high (r = .94), with subscale reliabilities as follows: physical aggression (r = .88), verbal aggression (r = .76), anger (r = .78), hostility (r = .82), and indirect aggression (r = .71) (Buss & Warren, 2000).

Responses are standardised across age (9 to 18, 19 to 39, and 40 to 88), with physical and verbal aggression scores standardised across sex. Standardisation levels are based on a sample of 2,138 Americans (1,252 female) aged 9 to 88, recruited from schools, churches, and community centres (Buss & Warren, 2000). Racial backgrounds of the baseline sample consisted of Asian (1%), Black (15%), Hispanic (8%), Native American (<1%), White (72%), and Other (3%). Validity-wise, the AQ is comparable to existing measures of anger, hostility, and pro-aggressive attitudes (*'Children's Inventory of Anger'*, Nelson & Finch, 2000; *'Novaco Anger Scale and Provocation Inventory'*, Novaco, 2003; *'Attitudes Toward Guns and Violence Questionnaire'*, Shapiro, 2000) with correlation coefficients ranging between .37 and .74.

The AQ is a revision of the Buss and Perry Aggression Questionnaire Version 1 (BPAQ; Buss & Perry, 1992), which itself is the successor to the Buss-Durkee Hostility Inventory (Buss & Durkee, 1957). The 29-item BPAQ is similar to the AQ in that it taps into cognitive (8 items) and affective (7 items) motivational components of aggression, as well as physical (9 items) and verbal (5 items) behavioural manifestations, but does not measure indirect aggression. Similar to the AQ, items are scored on a five-point Likert scale ranging from 1 (*'Extremely unlike me'*) to 5 (*'Extremely like me'*) and show internal consistency ranging from .70 to .88 for physical aggression, hostility and the total scores, and from .53 to .67 for verbal aggression and anger.

5.3.2 Inventory of Callous-Unemotional Traits

The Inventory of Callous–Unemotional Traits (ICU; Frick, 2003) measures the occurrence and intensity of CU traits, and consists of 24-items (e.g. "I do not show my emotions to others") measured on a four-point Likert scale from 0 ('*Not at all*') to 3 ('*Definitely true*'). The ICU can be completed using self-report, interview, or using parent-/teacher-ratings. For this thesis, self-reported values are used. Total scores are calculated by reverse scoring positively worded items prior to summation.

Confirmatory factor analysis on a large (n = 1,443) sample of German adolescents aged 12 to 18 suggested a three-way factor structure consisting of callousness, uncaring, and unemotional traits (Essau et al., 2006). Confirmation of this structure is provided in Fanti, Frick, & Georgiou (2009), Kimonis et al. (2008), and Roose, Bijttebier, Claes, Decoene, & Frick (2010). Eleven items load onto the callousness subscale (e.g. "The feelings of others are unimportant to me"), denoting callous attitudes towards others, and a further eight items (e.g. "I hide my feelings from others") load onto the uncaring subscale, characterised by a lack of concern for one's own performance and the wellbeing of others. The final five items load onto the unemotional subscale (e.g. "I try not to hurt others' feelings") and denote a lack of emotion expression. ICU stability has been confirmed over one year in Australian cohorts aged between 4 to 9 (r = .55; Dadds, Fraser, Frost, & Hawes, 2005) and 17 to 24 years old (r = .60; Blonigen, Hicks, Krueger, Patrick, & Iacono, 2006).

The ICU has three key limitations. First, though internal consistency is high in total ICU (.74 to .85) and callous and uncaring (.53 to .81) subscales, the unemotional subscale has low internal consistency (Essau et al., 2006; Frick, 2006; Kimonis et al., 2008). Low internal consistency might be an artefact of only five items loading onto the unemotional subscale (Essau et al., 2006; Fanti et al., 2009; Kimonis et al., 2008). Therefore, conclusions drawn from this scale should be considered with caution (Berg et al., 2013; Guelker, Barry, Barry, & Malkin, 2014). Second, though the ICU contains positively- and negatively-

worded items (Essau et al., 2006), these are not distributed evenly across subscales. The callousness scale consists mainly of negatively-worded items and the uncaring scale consists mainly of positively-worded items (Feilhauer, Cima, & Arntz, 2012). Third, although the ICU is thought to remain valid in adults (P. Frick, personal communication, December 12, 2011), it has predominantly been tested in adolescents (Essau et al., 2006; Fanti et al., 2009; Kimonis et al., 2008; Roose, et al., 2010).

The use of the ICU in this thesis is justified over other measures of CU traits such as the PCL-R (Hare, 2003) and the Psychopathic Personality Inventory – Revised (PPI–R; Lilienfeld & Widows, 2005). The PCL-R encompasses two factors - emotional detachment and antisocial deviance, which can be further divided into affective, interpersonal, lifestyle, and behavioural components (Hare, 2003; Hare & Neumann, 2005). However, this assay is time consuming and requires a combination of structured interview and historical information. Furthermore, the PCL-R only contains four items assessing CU traits, specifically, and so might not be an accurate conceptualisation (Feilhauer et al., 2012).

The PPI-R is an assay of personality characteristics as opposed to behaviours associated with psychopathy and can be used in forensic, clinical, and community samples (Gonsalves, McLawsen, Huss, & Scalora, 2013). However, the PPI-R is both time-consuming with 154 self-report items, and fails to tap into distinct CU traits, with the two key subscales indexing self-centred impulsivity and fearless dominance.

The ICU emerged from the Antisocial Process Screening Device (APSD; Frick & Hare, 2001), and is a 20-item assay of psychopathic traits in children and adolescents, which can be completed by parent-, teacher-, or self-report. The APSD has a three-factor structure comprising a narcissism (7 items; e.g. "You brag a lot about your abilities, accomplishments, or possessions"), impulsivity (5 items; e.g. "You act without thinking of the consequences"), and CU dimension

(6 items; e.g. "You are concerned about the feelings of others"). Clinic and community samples of preadolescent boys and girls (Frick et al., 2000) and a detained sample of adolescent boys and girls (Vitacco, Rogers, & Neumann, 2003) confirm this three-way factor structure.

Although the APSD is a common measure of CU traits, having been used solely in twenty-six out of thirty studies, and in combination in a further two studies between 1997 and 2013 (Waller, Gardner, & Hyde, 2013), the internal consistency of the CU subscale is only moderate (*alpha* = .63; Bijttebier & Decoene, 2009). Poor internal consistency likely results from [1] only six items belong to the CU subscale, to which five are positively worded, and [2] each item being rated on a three-point Likert scale ranging from 0 (*'Not at all'*) to 2 (*'Definitely true'*), effectively prompting midline responses (Essau et al., 2006; Kimonis et al., 2008). Additionally, measures of CU traits do not delineate between subtypes of CU traits. The ICU was developed to overcome these limitations - providing a more extended assessment of CU traits.

5.3.3 Food Frequency Questionnaire

The Food Frequency Questionnaire (FFQ; Sublette et al., 2011) was developed as a means of rapidly assessing omega-3 intake in mood disorder research. Consisting of 21 items assessing intake of commonly accessible fish and seafood over a six-month period, as well as intake of sushi, walnuts, flaxseed, flaxseed oil, cod liver oil, and canola oil, the FFQ allows the calculation of independent DHA, ALA, and EPA consumption. Omega-3 contributions are calculated as a function of fish type, portion sizes (i.e. small, medium, large), consumption frequency (i.e. monthly, weekly, daily), and sex of responder. Dietary supplements are also assessed.

The FFQ was validated with omega-3 blood content in adults aged 18 to 73 years old with (n = 34) and without (n = 27) major depressive disorder (Sublette et al., 2011). After controlling for participants reporting no omega-3 intake, DHA (r = .50, p < .001) and EPA (r = .47, p < .001) correlated significantly with their

87

respective plasma levels. Furthermore, although no significant association was reported between FFQ- and plasma-measured ALA (r = .22, p = .09), ALA is not assessed in this thesis.

Although the briefness of this measure limits the distinction of wild versus farmed fish and excludes measurement of omega-3-rich sources such as eggs, its use is justifiable. Whilst there are other assays of omega-3 intake available (Fawzi, Rifas-Shiman, Rich-Edwards, Willett, & Gillman, 2004; Garland et al., 1998; Hunter et al., 1992; Parra et al., 2002), these consist of more than 100 items and are time-consuming to complete (Sublette et al., 2011). Additionally, although brief 14- and 15-item measures are available (Laviolle et al., 2005; Svilaas et al., 2002) these do not solely assess omega-3 intake and so do not provide an accurate assessment of omega-3 intake.

5.3.4 Alcohol Use Disorders Identification Test

The AUDIT (Saunders, Aasland, Babor, de la Fuente, & Grant, 1993) is a 10item measure designed to help practitioners identify individuals who present alcohol dependence and drinking behaviours that are detrimental to themselves or others. This measure can be administered via interview or self-assessment and can be completed either online or offline. Items are scored between 0 and 4, and are summated to produce a total between 0 and 40. Scores of 0-7 suggest a possible benefit for "*alcohol education*", and scores of 20-40 indicate a need for "*referral to specialists for diagnostic evaluation and treatment*"). Questions 1 to 3 assess alcohol consumption, questions 4 to 6 assess alcohol dependence, and questions 7 to 10 pertain to alcohol-related problems (Babor, Higgins-Biddle, Saunders, & Monteiro, 2006). Internal reliability is high in control (.80) and alcohol-dependent (.80) samples (Moussas et al., 2009).

5.3.5 Cannabis Use Disorders Identification Test

The CUDIT (Adamson & Sellman, 2003) is a 10-item assay of cannabis use, formulated from the AUDIT (*see section 5.3.4 [page 88]*) by replacing the word

"alcohol" with "cannabis". The CUDIT assesses the subdomains of cannabis consumption, using behaviours, adverse psychological reactions, and problems. The measure can be administered via interviewer or self-assessment and can be completed either on- or offline. Each item is scored from 0 to 4, giving a minimum possible score of 0 and a maximum possible score of 40, which can be used to inform clinical opinion (i.e. whilst scores of 0-7 suggests benefit for "*cannabis education*", scores of 20-40 indicate a need for "*referral to specialists for diagnostic evaluation and treatment*"). Internal reliability and predictive value was found to be high (.84) in a moderately large sample of alcohol dependent outpatients (Adamson & Sellman, 2003).

Chapter Six: Correlations between physical aggression & callous-unemotional traits

6.1 Introduction

Physical aggression and CU traits manifest in disorders characterised by affective, interpersonal, and behavioural dysfunction (American Psychological Association, 2013), and as such are commonly used as proxy measures of psychopathy (Dawel et al., 2012). Discussed throughout chapter one, section 1.3.4 (page 12), the association between aggression and CU traits is well established (*see Frick & Dickens, 2006 for review*), with CU traits thought predict long-lasting patterns of antisocial behaviour (Frick & White, 2008), increased aggression severity (Frick & Dickens, 2006), and poor treatment outcomes (Frick et al., 2014). When presented together, CU traits facilitate the use of aggression as a means of acquiring resources and social-status through decreased social inhibition and the reorientation of attention away from the negative consequences of aggression towards the social rewards there of (Guelker et al., 2014; Pardini & Byrd, 2012; Raine, 2013).

CU traits comprise callousness (i.e. the poor affective response towards the feelings of others), uncaring (i.e. a lack of concern for one's own performance and the wellbeing of others), and unemotional (i.e. a lack of emotion expression) components (Fanti et al., 2009; Frick, 2003; Kimonis et al., 2008; Roose et al., 2010), which might differentially associate with aggressive traits and related behaviours. Whilst both callousness and uncaring traits have been positively associated with self-reported aggression and offending behaviour (Guelker et al., 2014; Kimonis et al., 2008), uncaring traits, specifically, have been found to better predict delinquency (Essau et al., 2006; Guelker et al., 2014; Kimonis et al., 2010). Conversely, unemotional traits are considered poor predictors of behavioural problems (Guelker et al., 2014).

Although a key medium for the manifestation of aggressive behaviour (Buss & Perry, 1992), to date, only one investigation has assessed the relationship between CU traits and measures of *physical* aggression in particular (Feilhauer et al., 2012). Sampling adolescents within community, clinical, and forensic settings, Feilhauer et al. documented positive, albeit weak, correlations between physical aggression, and all three components of the ICU. However, such inferences are solely derived from child and adolescent samples, and so their relationship in adults remains unclear. As considerable developmental changes occur between adolescence and adulthood (Bryd, Kahn, & Pardini, 2013), delineating relationships between CU traits and physical aggression within an adult cohort would provide valuable insight into how such relationships vary as a function of age.

Scores on psychometric measures of physical aggression, as measured by the AQ (Buss & Perry, 1992), and CU traits, as measured by the ICU (Frick, 2003), have shown to vary as a function of sex. In regards to the AQ, robust sex differences have been observed in the physical aggression subscale, with males consistently reporting higher physical aggression than females in adult samples (Bernstein & Gesn, 1997; Buss & Perry, 1992; Williams, Boyd, Cascardi, & Poythress, 1996). This pattern is also observed in the elderly (Morales-Vives & Vigil-Colet, 2010).

In regards to the ICU, although sex differences have not been observed in some analyses (Guelker et al., 2014; Roose et al., 2010), others have documented higher scores for males, relative to females, both uniformly across the three CU components (Essau et al., 2006) and for the unemotional dimension, specifically (Kimonis et al., 2008). Additionally, although not a true index of physical aggression, callousness has been shown to predict problematic behaviour in both male and female children, with uncaring traits only predictive of problematic behaviour in males (Essau et al., 2006). Possibly indicative of a more inclusive relationship between aggressive and CU traits in males, the authors explain this finding through males, relative to females,
reporting consistently higher scores on composite measures of psychopathy as indexed by the PCL-R, *see chapter five, section 5.3.2 [page 84]*) - during both adolescence and adulthood (Vitale & Newman, 2001).

The current study sought to delineate the relationships between self-reported physical aggression, callousness, uncaring, and unemotional traits, in an adult cohort derived from the community. Physical aggression was hypothesised to correlate positively with callousness, uncaring, and to a lesser extent, unemotional traits in males, but only with callousness in females. Furthermore, it tested whether these traits, and their bivariate relationships, would vary as a function of sex. Informed by literature targeting these associations in child/adolescent samples, physical aggression and all three CU traits were hypothesised to be higher in males than females.

6.2 Methods

6.2.1 Participants

A total of 184 participants (136 female; M age = 20.46; SD = 2.42) were recruited via online, snowball sampling. Participants consented to take part in a brief online survey after responding to an advertisement disseminated via social media.

6.2.2 Measures

Participants completed a battery of demographics and online versions of the AQ and ICU (*see chapter five, sections 5.3.1 [page 83] and 5.3.2 [page 84], respectively*). For this investigation, only the physical aggression subscale of the AQ was used.

6.2.3 Procedure

Participants were briefed and provided informed consent. On average, the questionnaire battery took less than 10 minutes to complete, and respondents

were provided with means of redistributing the survey through social media, if they wished. The final section of the online battery debriefed respondents and provided them with a means of contacting the lead investigator (DF).

6.2.4 Statistical Analysis

Incomplete responses were removed. Scores for the physical aggression subscale of the AQ, and the callous, uncaring, and unemotional subscales of the ICU were reverse scored and calculated into an R Statistics v.2.15 dataset (R Core Team, 2012). Physical aggression scores underwent *t*-transformations as a function of age and sex. Four *t*-tests were conducted to outline baseline sex differences between psychometric responses. Six Pearson's partial-correlations were performed, holding age (demeaned), sex (demeaned), and scores on the psychometric measures *not* included in the specific analysis, constant. For example, a correlation between physical aggression and callousness controlled for scores on uncaring and unemotional measures, in order to find the unique variance shared by physical aggression and callousness. *Post hoc*, a further twelve Pearson's partial-correlations were performed, in addition to six *z*-difference comparisons in order to test differences in correlation effect sizes as a function of sex. Significance values were corrected for multiple comparisons (*see Benjamini & Hochberg, 1995*).

6.3 Results

Twenty-nine incomplete responses were removed, leaving 155 participants (M age = 20.28; SD = 2.36). Of the complete responses, 123 were female (M age = 20.25; SD = 2.79) and 32 were male (M age = 20.29; SD = 2.25).

6.3.1 Psychometric Measures

Means and standard deviations for scores on physical aggression, callousness, and uncaring and unemotional traits, for males, females, and the total sample are exhibited in Table 2. Relative to females, males responded significantly higher on measures of *callousness* (t = 2.73, df = 153, p = .01, *padj* = .03), but

not *physical aggression* (t = 1.04, df = 153, p = .86, *padj* = .86), *unemotional* traits (t = 1.99, df = 153, p = .05, *padj* = .14), or *uncaring* traits (t = 1.12, df = 153, p = .27, *padj* = .53). Comparisons of psychometric responses between sexes are presented in Figure 1.

Table 2.

weals and standard deviations of psychometric measures	Means and standard	deviations of	psychometric	measures
--	--------------------	---------------	--------------	----------

	P. Aggression	Callousness	Uncaring	Unemotional
Male	52.84 (8.40)	6.22 (3.33)	8.28 (3.50)	8.31 (3.01)
Female	51.04 (8.82)	4.46 (3.23)	7.49 (3.60)	7.08 (3.14)
Total	51.41 (8.73)	4.82 (3.32)	7.65 (3.58)	7.34 (3.14)

Note. P. Aggression = Physical Aggression



Figure 1. Means and standard deviations of psychometric responses (A = physical aggression, B = callousness, C = uncaring, D = unemotional) for males and females. *Note.* * p < .05

6.3.2 Correlations

As presented in Table 3, bivariate partial correlations are reported for physical aggression, callousness, uncaring, and unemotional traits in the total sample, as well as for males and females separately. Correlations controlled for age, sex, and scores on psychometric measures *not* included in the specific analysis (*see section 6.2.4 [page 93] for an example*). All correlations underwent multiple comparison correction.

Table 3.

		P. Aggression	Callousness	Uncaring	Unemotional
Male	P. Aggression	1	.29**	.03	.10
	Callousness	-	1	.10	.11
	Uncaring	-	-	1	.01
	Unemotional	-	-	-	1
Female	P. Aggression	1	.25**	.12	.17
	Callousness	-	1	.32***	.18
	Uncaring	-	-	1	.05
	Unemotional	-	-	-	1
Total	P. Aggression	1	.28**	.08	.15
	Callousness	-	1	.33***	.12
	Uncaring	-	-	1	.06
	Unemotional	-	-	-	1

Bivariate correlation coefficients between psychometric measures and sex

Note. * p < .05, ** p < .005, *** p < .001, P. Aggression = Physical Aggression

6.3.2.1 Total Sample

In the total sample, *physical aggression* was significantly correlated with *callousness* (r = .28, p < .001, *padj* < .05), but not *uncaring* (r = .08, p = .33, *padj* = .45) or *unemotional* (r = .15, p = .07, *padj* = .26) traits. *Physical aggression* accounted for 7.6% of the variance in *callousness* in the total sample. Between the components of the ICU, *callousness* was significantly correlated with *uncaring* traits (r = .33, p < .001, *padj* < .001), explaining 10.9% of variance in the total sample, but was not significantly correlated with *uncaring* traits (r = .12, p = .14, *padj* = .43). There was no statistically significant correlation between *uncaring* and *unemotional* traits (r = .06, p = .45, *padj* = .45). Correlations between physical aggression and callousness (subplot C), and callousness and uncaring traits (subplot G) in the total sample are displayed in Figure 2.

6.3.2.2 Males

In males, *physical aggression* significantly correlated with *callousness* (r = .29, p < .001, padj < .05), but not *uncaring* (r = .03, p = .70, padj = .88) or *unemotional* (r = .10, p = .22, padj = .66) traits. *Physical aggression* accounted for 8.4% of the variance in *callousness* in males. Between the components of the ICU, *callousness* was not significantly correlated with either *uncaring* (r = .10, p = .22, padj = .66), or *unemotional* (r = .11, p = .17, padj = .66) traits. There was no significant correlation between *uncaring* and *unemotional* traits (r = .01, p = .88, padj = .88). Correlations between physical aggression and callousness (subplot A), and callousness and uncaring traits (subplot D) in males are displayed in Figure 2.

6.3.2.3 Females

In females, *physical aggression* significantly correlated with *callousness* (r = .25, p < .05, *padj* < .05), but not *uncaring* (r = .12, p = .13, *padj* = .26) or *unemotional* (r = .17, p = .04, *padj* = .12) traits. *Physical aggression* accounted for 6.3% of the variance in *callousness* in females. Between the components of

the ICU, *callousness* was significantly correlated with *uncaring* traits (r = .32, p < .001, *padj* < .001), explaining 10.5% of variance in females, but was not significantly correlated with *unemotional* traits (r = .18, p = .03, *padj* = .11). There was no significant correlation between *uncaring* and *unemotional* traits (r = .05, p = .51, *padj* = .51). Correlations between physical aggression and callousness (subplot B) and callousness and uncaring traits (subplot E) in females are displayed in Figure 2.



Figure 2. Scatter plots of partial correlations between physical aggression vs. callousness (subplots A - C), and callousness vs. uncaring traits (subplots D - F), for males, females, and total sample. Males - physical aggression vs. callousness (subplot A, r = .29), callousness vs. uncaring (subplot D, r = .10). Female - physical aggression vs. callousness (subplot B, r = .25), callousness vs. uncaring (subplot C, r = .32). Total sample - physical aggression vs. callousness (subplot F, r = .32). Total sample - physical aggression vs. callousness (subplot C, r = .28), callousness vs. uncaring (plot F, r = .33). Shaded areas denote 95% confidence interval areas along linear associations.

6.3.2.4 Sex Differences

A comparison of each bivariate partial correlation between males and females revealed no statistically significant difference (*z*-difference) in effect size after controlling for sample size.

6.4 Discussion

This investigation is the first to delineate the relationships between self-reported physical aggression, callousness, uncaring, and unemotional traits, in an adult cohort. It was hypothesised that [1] psychometric measures of physical aggression and CU traits would be uniformly higher in males than females, and [2] physical aggression would be positively correlated with callousness, uncaring, and unemotional traits in males, but only with callousness in females.

In contrast to expectations, males only scored higher than females on scores of callousness, with measures of physical aggression, uncaring, and emotional traits yielding similar scores across sexes. Relating to physical aggression, the current findings diverge from those observed elsewhere that sampled from adolescent (Reyna, Ivacevich, Sanchez, & Brussino, 2011), adult (Bernstein & Gesn, 1997; Buss & Perry, 1992; Williams et al., 1996), and elderly populations (Morales-Vives & Vigil-Colet, 2010). In these investigations, males reported higher scores on comparative measures than females. Possibly an artefact of the current sample consisting predominantly of students, a population considered less prone to generate pro-aggressive ideologies (Finkel, et al., 2009, *see chapter one, section 1.3.3 [page 11]*), males, relative to females, have reported greater scores on physical aggression traits in a sample of Spanish students (García-León et al., 2002). However, it should be noted that akin to this investigation, the sample reported García-León et al. also comprised predominantly of females.

In respect to CU traits, male adolescents have been shown to score higher across all three subscales of the ICU (i.e. callousness, uncaring, unemotional), compared to female counterparts (Essau et al., 2006). Whilst not specific to the ICU, males are thought to score higher than females during both adolescence and adulthood, on a wide array of psychopathic traits measured across behavioural, interpersonal, and affective dimensions (Vitale & Newman, 2001). In the current study, only scores on callousness (i.e. the indifference to the feelings of others) were greater for males than females. It is possible that a low rate of male respondents contributed to the lack of statistical difference across all psychometric traits between sexes in this investigation. Although the variance in data for each psychometric was comparable between sexes, the male cohort made up only a quarter of the overall sample, and after removal of incomplete responses, yielded just 32 datasets.

The second key finding of this investigation was that regardless of sex, physical aggression correlated with callousness, but not uncaring or unemotional traits, after controlling for age and CU/aggressive traits not included in the specific analysis. This finding contrasts previously reported relationships between psychometric-based measures of physical aggression and CU traits (Feilhauer et al., 2012). Although established in adolescents, not adults, Feilhauer et al. documented positive associations between AQ-measured physical aggression, and all three components of the ICU - compounding the well-established association between CU traits and behaviours relating to aggression (Frick & Dickens, 2006).

In this investigation, an association between physical aggression and callousness, but not uncaring or unemotional traits, might signal differential relationships between CU traits and physical aggression across age. Whilst useful for identifying the possibility of age-specific relationships, further validation is needed in the form of direct group comparisons and/or longitudinal research. From these findings, it is possible that the relationships between physical aggression and unemotional and uncaring traits, indicative of low emotional reactivity and caring towards others (Ciucci & Baroncelli, 2014), might represent differently in adult and adolescent cohorts.

Bivariate analysis of the ICU suggested that in line with some (Berg et al., 2013; Essau et al., 2006; Fanti et al., 2009) but not all (Guelker et al., 2014) factor analyses of the ICU, callousness was positively associated with uncaring traits in the composite sample. As a function of sex, although callousness and uncaring traits were positively correlated in females, but not males, effect sizes were comparable after controlling for sample size. Therefore, it is possible that a statistically significant association between callousness and uncaring traits in males might emerge in a larger cohort. However, in contrast to results presented here, positive correlations between unemotional traits and both callousness (Essau et al., 2006; Fanti et al., 2009) and uncaring traits (Berg et al., 2013; Essau et al., 2006; Fanti et al., 2009; Guelker et al., 2014) have been reported in adolescent samples elsewhere. Though potentially an artefact of the sample reported here being derived from an adult cohort, because of the low internal consistency of the unemotional subscale of the ICU (Essau et al., 2006; Fanti et al., 2009; Kimonis et al., 2008), conclusions should be drawn with caution (Guelker et al., 2014).

Results of this investigation should be considered in light of two limitations. First, only associations between CU traits and physical aggression were reported leaving associations with other subscales of the AQ (i.e. verbal and indirect aggression, anger, hostility) unknown. This decision was taken with the understanding that the assessment of other aggressive traits would require a far larger sample size in order to produce sufficient statistical power. Additionally, as the key aims of this thesis involve delineating the relationships between physical aggression/CU traits and the electrophysiological indices of the VIM (see chapter five, section 5.1 [page 82]), these specific analyses were computed with the prospect of better understanding results reported elsewhere in this thesis. However, as males and females have long been shown to aggress in qualitatively different manners (White & Kowalski, 1994), there is clear importance of capturing this distinction in relation to CU traits. Second, as noted on page 93, although this analysis controlled for age within the sample, it does not allow for a direct comparison between adolescent and adult cohorts. This is an important avenue of exploration given previous declarations of aggressive and CU traits evidencing the capacity to alter over time, as a function of maturation and intervention (Moffitt, 1993).

In summary, although CU traits are thought to remain stable throughout adolescence and early adulthood (Frick, et al., 2003; Frick & White, 2008; Kruh et al., 2005), current findings suggest their association with physical aggression might not. In this adult cohort, across both males and females, physical aggression was shown to correlate with callousness, but not uncaring or unemotional traits. Discussed in light of study limitations, these results suggest trait-specific relationships between physical aggression and CU traits. Future investigation should observe these associations as a function of age, using group comparisons or longitudinal measures, whilst exploring similar associations with other manifestations and motivational underpinnings of aggression.

Chapter Seven: Baseline assessment of the Facial Affect Stop-Go Task

7.1 General Introduction

The VIM is a social-developmental model thought to underpin the extinction of a behavioural response, as a function of the successful perception of distress (Blair et al., 1995, 1999; Blair, 1995, 2001, *see chapter one, section 1.3.4 [page 12]*). The high temporal resolution of EEG makes it a useful tool for disentangling the time course of facial affect processing and motor extinction stages of the VIM. However, to date, electrophysiological signatures of the VIM have not been identified. In order to achieve this, a novel FAST is required. This chapter is split into two sections. The first documents the development of a FAST through three pilot studies, each accompanied by discussion of paradigm utility and shortcomings. The second uses the final iteration of the FAST to characterise ERPs related to facial affect processing and distress-induced motor extinction.

7.2 Development of the Facial Affect Stop-Go Task

7.2.1 Task Development

In laboratory settings, motor inhibition is typically investigated using go-nogo and SSTs (Aron, 2007, *see chapter one, section 1.3.5 [page 19]*). Whereas gonogo paradigms allow investigation of pre-motor inhibition, SSTs better reflect motor extinction after the fact (Verbruggen & Logan, 2008). Previously, nogo and stop stimuli have consisted of auditory (e.g. tones, white noise) and visual (e.g. geometric shapes, frames) stimuli, as well as colour changes there of (Huster et al., 2014; Johnstone et al., 2007; Rubia et al., 2003). However, as argued in chapter one, section 1.3.5 (page 19), the use of emotion, specifically distress, as a stop stimulus is absent in existing literature. The use of distress as a stopping mechanism is imperative to the exploration the VIM on an electrophysiological level.

7.2.1.1 Stimuli

Investigation of the VIM using EEG requires a paradigm whereby one stimulus cues the initiation of a motor response, and a second commands the need to halt said response. As the VIM underpins inhibition of aggression (Blair, 1995), which itself might manifest in response to threat (Blanchette, 2006; Fox et al., 2007, *see chapter one, section 1.2.1.1 [page 2]*), angry facial expressions are fitting stimuli to signal a motor response. Likewise, as VIM motor extinction is thought to result from an empathic response to facial distress (Blair, 2001; Decety & Jackson, 2004, *see chapter one, section 1.3.4 [page 11]*), fearful and sad facial expressions are appropriate signals for motor extinction.

The Hillyard Principle (*see Luck, 2005*) dictates that in order to account for as much variance in the data as possible, where appropriate, stimuli should consist of similar physical properties. In the context of the FAST, although necessary to use several facial identities as to not attenuate the N170 response (*see chapter three, section 3.2.1 [page 39]*), trial-specific stimuli should be of the same identity. This approach has additional benefit for increasing task validity. For this paradigm, male and female facial expressions were chosen from the MacBrain NimStim Face Stimulus Set (Tottenham et al., 2009). Facial stimuli with open mouths were chosen in order to increase affect valence.

7.2.1.2 Trials

As discussed in chapter three, section 3.1 (page 37), accurate delineation of ERPs from resting state EEG requires high SNR (Luck, 2005). There is no definitive rule as to the optimum number of trials within a given EEG experiment, and so trial quantity largely depends on the sample, stimulus, environment, and ERP of interest (Luck, 2004, 2005). Although Luck (2004) suggests that as few as 30 trials per condition, per stimulus might be enough to accurately model the P300, as many as 200 trials might be required for comparable accuracy of earlier components. As the FAST seeks to model both early (e.g. N170, P200) and late (e.g. N200, P300) ERP components within a single paradigm, a compromise must be reached. Furthermore, trial quantity

must be considered in light of practicality and participant engagement. For example, whilst the inclusion of a 100 trials per condition might be ideal, responses from dis-engaged/tired participants would likely detriment accurate interpretation of the EEG signal.

7.2.1.3 Stimulus Presentation and Filtering

As ERPs are capable of exceeding the initiation of a subsequent ERP, temporal overlap of ERP components should be minimised (*see chapter three, section 3.1 [page 37]*). Overlapping ERPs detriment waveform interpretation, and are particularly problematic for stop paradigms where shorter inter-stimulus intervals (ISI) are required (Woldorff, 1993). In these instances, stop stimuli need to be presented prior to the completion of a motor response. During a FAST, ERPs evoked to go stimuli (i.e. angry facial expressions) would likely overlap with those evoked to subsequent stop stimuli (i.e. fearful/sad facial expressions).

A case has been made for the partial removal of temporal overlap using highpass filters (Woldorff, 1993). Whilst ideal for investigation of high-frequency components, such as those used to index early auditory brainstem responses, high-pass filtering distorts long latency, slow-wave components such as the P300, which become attenuated (Luck, 2005).

A second method of offline ERP deconvolution, the two-stage adjacent response technique (ADJAR), reduces temporal overlap through jittering the ISI around the mean (Luck, 2005; Woldorff, 1993). After averaging across trials, stage one of the ADJAR acts as a low-pass filter in order to preserve the timelocked, stimulus-specific ERP signal by averaging out trial-by-trial interactions. For greater effect (i.e. less temporal overlap), the range of the ISI must be sufficiently wide in order to encapsulate overlapping ERP components. Woldorff (1993) suggests a duration that surpasses that of the slowest dominant waveform likely to interact with the subsequent ERP. The second stage of the ADJAR is applied when temporal overlap is too severe to be attenuated by jittering the ISI alone. In these instances, estimation of the exact overlap between the first and second stimulus is derived over a series of iterations. Iterations are varied across the jittered ISIs with the final estimation of overlap being defined when the model can no longer be manipulated. Subtraction of this estimation from the ERP response to the second stimulus removes any residual EEG signal relating to stimulus one, resulting in a waveform depicting the ERP response to the second stimulus alone. However, the ADJAR is limited to instances whereby a substantial ISI range can be used. In Bekker et al. (2005), a range of 240 ms was not sufficient to completely remove overlapping ERP responses. The following sections outline three pilot studies documenting the initial iterations of the FAST.

7.2.2 FAST v.1

7.2.2.1 Participants

Six undergraduate students (3 females; M age = 19.66; SD = .82) were recruited by opportunity sampling from Nottingham Trent University.

7.2.2.2 Procedure

Data were recorded in a sound-attenuated room. After consenting, participants were sat 60 cm away from a 19" CRT monitor (1024 x 768 resolution; 85 Hz refresh rate). EEG apparatus was shown and attached to participants and stability checks were made. Participants completed version one of the FAST (*see Figure 3*), which consisted of 210 trials (90 background; 60 go; 60 stop). Background stimuli consisted of neutral, surprised, and happy facial expressions, go stimuli consisted of angry facial expressions, and stop stimuli consisted of fearful and sad facial expressions. Stimuli comprised twenty identities (10 females; African- and Caucasian-American; IDs 01, 03, 05, 06, 07, 08, 09, 10, 20, 27, 28, 29, 30, 31, 32, 33, 34, 35, 36).

Participants completed twenty practice trials using two identities not in the experimental trials (1 female; African-American; IDs 13, 38). Following a lead in period of 4000 ms, participants were randomly presented with a background or go stimulus. Participants were asked to respond to go stimuli (200 ms, 60 ms jitter) via a button press. On 50% of go trials, participants were presented with a stop stimulus (200 ms), to which they were asked to not complete their motor response. Trials were separated by a 900 ms inter-trial-interval (100 ms jitter).



Figure 3. FAST version 1. Participants respond to infrequent go (i.e. angry) stimuli among frequent background (i.e. neutral, surprised, happy) stimuli. Following a response period of between 140 and 260 ms, a stop stimulus (i.e. fear, sadness) denoted the requirement of motor response extinction.

7.2.2.3 Physiological Data Acquisition

EEG activity was recorded using an active-electrode, 64-channel Active-Two acquisition system (BioSemi, Amsterdam, Netherlands), sampled at 2048 Hz and digitised at 24-bits. Impedances were kept below 5 kohms as recommended in Luck (2005). Data were collected using ActiView v.6.05 (BioSemi, Amsterdam, Netherlands). All EEG signals were average referenced on-line. The N170, P200, N200, and P300 were measured in response to background stimuli, and the stop-N200 and stop-P300 were measured in response to stop stimuli.

7.2.2.4 Signal Processing

Vertical and horizontal electrooculography artefacts were identified and corrected (Jung et al., 2000). A high-pass filter of .01 Hz (~ 6 Hz, forward shift) was used to remove slow drifts as described in section 7.2.1.3 (page 104) and a low-pass filter of 35 Hz (~ 24 dB/oct, 0-phase shift) was applied to remove high frequency waves. A 50 Hz notch filter was used to remove electrical noise. Trials were baseline corrected prior to averaging (-200 ms).

The N170 and P200, evoked to both go and background trials, were referenced to the average of all the electrodes in order to avoid attenuating ERPs measured over temporo-parietal sites (*see Joyce & Rossion, 2005 for review of reference-specific attenuation of the N170, as well as chapter three, section 3.2.1 [page 39]*). The N200 and P300, evoked to both go and stop trials, were re-referenced to the linked mastoids (i.e. the bone behind the ear) in order to [1] minimise any spatial distortion resulting from the use of average referencing (Luck, 2005), and [2] increase the generalisability of results as a function of the wide-use of linked mastoids when referencing later perceptual components (Luck, 2005; Woodman, 2010).

7.2.2.5 Results and Discussion

Data inspection and participant feedback revealed two confounding problems. First, participants periodically reported an inability to accurately interpret the emotion depicted by stimulus one. This deficit is likely a function of the relatively brief 140 to 260 ms display period being insufficient to accurately categorise facial affect. Previously, Fox et al. (2000) noted that participants in their investigation reported facial expressions - including those depicting anger thought to command a rapid response (*see chapter one, section 1.3.1 [page 7]*) - were difficult to interpret when presented for 300 ms. This was supported by behavioural data evidencing increased reaction times and error rates for neutral, happy, and angry schematic faces. Second, ERPs evoked to stimulus one and two still presented extreme convolutions after averaging across jittered trials. Such convolution prevented the accurate interpretation of the P200, N200, and P300 evoked to stimulus one, and also the N170 and P200 evoked to stimulus two. As fearful and sad facial expressions were not used as background stimuli, distortion of N170 and P200 responses to stimulus two means that the FAST v.1 does not enable accurate assessment of N170 and P200 responses to fearful and sad facial expressions.

Although a jitter period of over 300 ms is recommended, such that Woldorff (1993) refers to a 'short jitter' epoch of between 150 to 350 ms, this pilot only used a jitter period of 120 ms around a 200 ms mean (140 to 260 ms). In this paradigm, this jitter period was justified by the need for the stop stimulus to be temporally close to the go stimulus in order to allow successful motor extinction. Elongation of this jitter period would have likely resulted in participants completing go responses prior to the presentation of stop stimuli.

7.2.3 FAST v.2

The second iteration of the FAST (v.2) addressed the problems highlighted in section 7.2.2.5 (page 107), that of [1] participant-reported inability to accurately categorise facial expressions represented in stimulus one, and [2] temporal overlap distorting the P200, N200, and P300 responses to stimulus one and the N170 and P200 responses to stimulus two.

7.2.3.1 Participants

Six undergraduate students (2 females; M age = 19.66, SD = .52) were recruited by opportunity sampling from Nottingham Trent University.

7.2.3.2 Procedure

Data were recorded in a sound-attenuated room. After consenting, participants were sat 60 cm away from a 19" CRT monitor (1024 x 768 resolution; 85 Hz

refresh rate). EEG equipment was shown and attached to participants before stability checks were made. Participants completed the FAST v.2 (*see Figure 4*). Background stimuli consisted of neutral, fearful, and sad facial expressions, go stimuli consisted of angry facial expressions, and stop stimuli consisted of fearful and sad facial expressions of seventeen identities (10 females; African-and Caucasian-American; IDs 01, 03, 05, 06, 07, 08, 09, 10, 20, 21, 23, 25, 26, 32, 34, 35, 36).

Twenty practice trials were given using two identities not in the experimental trials (1 female; African-American; IDs 13, 38). The paradigm consisted of 272 trials, split into 2 counterbalanced blocks. Each block began with a 4000 ms lead-in, with each trial consisting of the random presentation of a background or go stimulus for 250 ms. A black display (500 ms) followed stimulus one, prior to presentation of a go cue (red cross; 200 ms). Participants were asked to respond via a button box press when the go cue followed a go stimulus (i.e. angry face), but not when it followed a background stimulus (i.e. neutral, fearful, or sad face). On 50% of the go trials, participants were presented with a stop stimulus (i.e. fearful or sad faces; 250 ms), upon which they were instructed to halt their motor responses. Trials were separated by a 900 ms (100 ms jitter), inter-trial-interval.

7.2.3.3 Physiological Data Acquisition

Physiological data acquisition was identical to that reported in section 7.2.2.3 (page 106).

7.2.3.4 Signal Processing

Signal processing was identical to that reported in section 7.2.2.4 (page 107).



Figure 4. FAST version 2. Participants observe infrequent go (i.e. angry) stimuli amongst frequent background (i.e. neutral, fearful, sad) stimuli for a duration of 250 ms. Following a blank display lasting 500 ms, if stimulus one was a go stimulus, participants were asked to respond via button box when they observed a red cross. A stop stimulus (i.e. fearful, sad) denoted the requirement of motor response extinction.

7.2.3.5 Results and Discussion

The second pilot investigation sought to address the limitations of categorisation and temporal overlap outlined in section 7.2.2.5 (page 107). In the FAST v.1, participants reported the duration of 140 to 260 ms was not sufficient enough to accurately categorise facial expressions. In the second iteration of the FAST, stimulus presentation times were increased to a fixed 250 ms in order to better facilitate stimulus processing. Furthermore, as a means of moderating ERP convolution and distortion of the early face-related ERP responses to stop stimuli, the FAST v.2 included sad and fearful facial expressions as background stimuli. This inclusion provided a means of measuring N170 and P200 responses to distress, without interference from preceding ERP waveforms. As to not elongate the paradigm, the outcome of which is highlighted in section 7.2.1.2 (page 103), surprised and happy facial

expressions were removed as background stimuli. Although the use of fearful and sad facial expressions as both background and stop stimuli had potential to habituate ERP responses to distress (*see chapter three, section 3.2.1 [page 45]*), visual inspection of data recorded during the first and final third of the paradigm did not reveal such habituation to either fearful or sad facial expressions.

Piloting of the FAST v.2 yielded three limitations. First, although the duration of stimulus one was elongated to 250 ms, participants yet again reported low confidence in their responses. To assure validity of this paradigm as a means of investigating the VIM on an electrophysiological level, it is vital that participants have awareness of the stimuli that they are initiating, and extinguishing motor response to. Furthermore, frequent presentation of stimuli within a given trial (i.e. stimulus one, black screen, go cue, stimulus two), over successive trials, led to participants reporting disorientation. Disorientation likely hinders the ability to efficiently categorise facial affect - ultimately manifesting in poorer response accuracy and reduced task validity.

Second, the go cue (i.e. red cross) evoked a slow, low frequency wave between 300 and 700 ms post stimulus onset, which did not return to baseline before presentation of stimulus two. This ERP overlapped with, and distorted, N170 and P200 responses to stimulus two. Although still possible to measure N170, P200, N200, and P300 responses to stimulus one, residual temporal overlap of the slow wave evoked to the red cross would likely distort ERP responses evoked during stop trials. Undistorted measurement of stop-ERPs is key to understanding the affect-specific electrophysiological distinctions during motor extinction.

Third, from a behavioural perspective, the 200 ms ISI between the go cue and stop stimulus was not short enough to prevent completion of a go response. With feedback from the first iteration of the FAST suggesting 200 ms was sufficient to successfully initiate motor extinction, disparity between these two pilot investigations might be explained through priming. Whilst participants were instructed to initiate a go response directly following stimulus one onset in the FAST v.1, participants were asked to withhold this response until after the presentation of the go cue - some 750 ms post stimulus onset - in the FAST v.2. Priming could decrease motor response times as a function of resolved processing conflict (Kristjánsson & Jóhannesson, 2014).

7.2.4 FAST v.3 and Experimental Methods

The third iteration of the FAST (v.3) addressed the limitations of the FAST v.2 highlighted in section 7.2.3.5 (page 110), that of [1] disorientation to frequent stimulus presentation, [2] a slow wave component evoked to the go cue overlapping N170 and P200 responses to stimulus two, and [3] response priming.

7.2.4.1 Participants

Six undergraduate students (2 females, M age = 19.66, SD = .82) were recruited by opportunity sampling from Nottingham Trent University.

7.2.4.2 Procedure

Data were recorded in a sound-attenuated room. After consenting, participants were sat 60 cm away from a 19" CRT monitor (1024 x 768 resolution; 85 Hz refresh rate). EEG equipment was shown and attached to participants before stability checks were made. Participants completed the FAST v.3 (*see Figure 5*). Go and stop stimuli consisted of neutral, angry, sad, fearful, and surprised expressions from 17 identities (8 females; African- and Caucasian-American; IDs 01, 03, 05, 06, 07, 08, 09, 10, 20, 21, 23, 25, 26, 32, 34, 35, 36). The paradigm consisted of 272 trials, split into 2 counterbalanced blocks. Each block contained 8 affect-pair combinations (Background Pairs: *Fear-Fear; Sad-Sad; Neutral-Sad; Neutral-Fear*, Go Response Pairs: *Anger-Anger; Anger-Surprise*, Stop Response Pairs: *Anger-Fear; Anger-Sad*).

Each block had a 4000 ms lead-in followed by presentation of stimulus one (800 ms; 100 ms jitter), a black screen acting as an ISI (160 ms; 40 ms jitter) and finally presentation of stimulus two (800 ms; 100 ms jitter). A red fixation cross accompanied an inter-trial-interval of 1800 ms (200 ms jitter). Participants were instructed to place their right index finger on a red button box key, moving it to an adjacent green key following the extinction of a go stimulus (i.e. angry face). Participants were instructed to interrupt go responses by returning their finger to the red key, if stop stimuli subsequently appeared. If stimulus one was not an angry facial expression no response was required at all during the trial. Twenty practice trials were given using two identities not in the experimental trials (1 female; African-American; IDs 13, 38).



Figure 5. FAST version 3. Participants responded to infrequent go (i.e. angry) stimuli among frequent background (i.e. neutral, fearful, sad) stimuli, presented between 700 and 900 ms, when the go stimulus dissipated. Following a black screen lasting between 120 and 200 ms, a stop stimulus (i.e. fear or sadness) was presented, denoting the requirement of motor response extinction.

7.2.4.3 Physiological Data Acquisition

Physiological data acquisition was identical to that reported in section 7.2.2.3 (page 106).

7.2.4.4 Signal Processing

Signal processing was identical to that reported in section 7.2.2.4 (page 107).

7.2.4.5 Results and Discussion

The aim of the third pilot investigation was to address the limitations highlighted in response to the FAST v.2 as outlined in section 7.2.3.5 (page 110), namely that of disorientation, go cue related slow wave activity, and response priming. First, participant-reported deficits in categorisation efficiency and feelings of disorientation were resolved by extending the presentation duration uniformly across stimuli from 250 ms to 800 ms. In Fox et al. (2000), greater accuracy of facial affect categorisation was observed after increasing the stimulus duration from 300 ms to 800 ms. In this investigation, elongation of stimulus duration facilitated [1] face perception accuracy, and [2] the measurement of deconvoluted ERP waveforms in response to both stimulus one and two. In the main, participants reported accurate affect categorisation.

Second, removal of the go cue, observed in the FAST v.2 to evoke a slow waveform that became convoluted with the ERP evoked to stimulus two, facilitated the accurate measurement of ERP responses to stimulus two. Third, contraction of the ISI between the go and stop stimuli was reduced to 160 ms (40 ms jitter) in order to combat priming effects. Participants reported that this ISI provided a suitable period in which to initiate and extinguish motor responses.

7.3 Testing of the Facial Affect Stop-Go Task

7.3.1 Introduction

Electrophysiological indices of facial affect processing can be split into early (e.g. N170; P200) and late (e.g. N200; P300) components. Evoked to faces, the N170 and P200 are thought to index the integrated processing of structural (i.e. identity) and second-order (e.g. task-salient, face components) facial features, respectively (Apicella et al., 2013; Hinojosa et al., 2015; Itier & Taylor, 2004) and show temporo-parietal dominance over the right hemisphere (Bentin et al., 1996; Hinojosa et al., 2015; Rossion et al., 2003).

Although conflicting results have been reported elsewhere (*see chapter three, sections 3.3.1 [page 45] and 3.3.2 [page 47]*), the N170 and P200 have shown a bias towards negative, over positive or neutral imagery (Chai et al., 2012; Smith et al., 2003). A negativity bias likely reflects reallocation of neuronal resources when processing biologically significant social stimuli. Specifically, the N170 shows augmentation in response to angry, relative to happy facial expressions (Eger et al., 2003; Ibánez et al., 2012; Krombholz, 2007), and in turn shows additive augmentation to fearful expressions (Batty & Taylor, 2003; Blau et al., 2007; Taylor et al., 2004; Ramos-Loyo et al., 2009; Shannon et al., 2013). Similarly, P200 augmentation is observed to facial threat (Chai et al., 2012; Gonzalez-Roldan et al., 2011) and distress (Shannon et al., 2013). Though not previously investigated, as sad facial expressions are indicative of distress, N170 and P200 augmentation resembling that evoked to fear would be expected.

N200 and P300 ERP components are typically evoked prior to motor responses and appear maximal over anterior and parietal midline sites, respectively (Patel & Azzam, 2005; Polich, 2007). Whilst both ERP components index higher order, cognitive control processes (Hillyard & Kutas, 1983), the N200 best encapsulates stimulus classification and conflict monitoring (Folstein & van Petten, 2008; van Veen & Carter, 2002), and the P300 better indexes working memory and orientation of neuronal resources to motivationally relevant stimuli (Polich, 2007).

During face processing, relative to happy and neutral facial expressions, fearful faces evoke N200 and P300 augmentation (Campanella et al., 2013; Dennis & Chen, 2007), with P300 augmentation also evident to angry (Lang et al., 1990) and sad facial expressions (Ehlers et al., 2001). Such amplitude variation likely denotes threat, in the form of anger and distress, creating processing conflicts that require additional neuronal resources to resolve.

The N200 and P300 can also be evoked to stop stimuli demanding the halting of a motor response, and are maximal over frontal midline sites (Bruin & Wijers, 2002; Dimoska et al., 2006; Kok et al., 2004; Pliszka et al., 2000). Anterior scalp topographies likely index neuronal generators associated with motor extinction such as the superior- and medial-frontal gyri, OFC, and ACC (Amodio et al., 2008; Enriquez-Geppert et al., 2010). Whilst facial affect has previously been used in motor inhibition paradigms (*see chapter three, section 3.4 [page 50]*), neither nogo paradigms nor SSTs have used facial affect as a stopping mechanism. Therefore, the role of facial distress (i.e. fear and sadness) on stop-N200 and stop-P300 responses remains unknown.

This investigation sought to delineate the effect of angry, sad, and fearful facial stimuli on ERPs associated with face processing (N170 and P200) during a FAST. Additionally, it aimed to outline how fearful and sad facial expressions moderate the N200 and P300 during both background and stop conditions. The N170 and P200 were hypothesised to [1] exhibit right hemispheric dominance and [2] augment in response to fearful and sad, and in turn angry facial expressions, relative to neutral faces. The N200 was hypothesised to show [3] an anterior scalp distribution to both background and stop stimuli with the P300 being maximal at [4] parietal sites to background stimuli and [5] anterior sites to stop stimuli. Both the N200 and P300 were expected to [6] show greater augmentation to fearful, relative to sad facial expressions.

7.3.2 Methods

7.3.2.1 Participants

Fifty-seven psychology students (35 females, M age = 19.25, SD = 1.89) were recruited from Nottingham Trent University after replying to an online advertisement. Inclusion criteria required participants to be right-handed, 18 to 65 years old, and without current diagnosis of psychiatric or neurological disorders, or medication that might affect electrophysiological responses. Participants refrained from consuming alcohol for twenty-four hours, and nicotine and caffeine for three hours prior to experimentation. Participants were compensated for their time with research participation credits.

7.3.2.2 Procedure

After written consent was obtained, participants completed the FAST v.3 as described in section 7.2.4.2 (page 112).

7.3.2.3 Physiological Data Acquisition

Physiological data acquisition was identical to that reported in section 7.2.2.3 (page 106). Mean peak amplitudes were measured using BESA Research v.5.3.7 (BESA GmbH, Gräfelfing, Germany) for N170 (150 to 220 ms interval), P200 (180 to 250 ms interval), N200 (210 to 310 ms interval), and P300 (300 to 480 ms interval) responses. On go/background trials, ERP responses were time-locked to the presentation of stimulus one, with ERP responses time-locked to presentation of stimulus two on stop trials. The measurement of mean, over peak, amplitudes mediated ERP attenuation thought to result from averaging multiple trials over various latencies (Luck, 2005).

7.3.2.4 Signal Processing

Signal processing was identical to that reported in section 7.2.2.4 (page 107).

7.3.2.5 Statistics

Average temporo-parietal N170 and P200 responses for each emotion were created using left (P7, PO7, O1, PO3) and right (P8, PO8, O2, PO4) electrodes. The N200 and P300 for both background and stop conditions were measured bilaterally and at midline electrodes at both anterior and parietal sites (F5, Fz, F6, P5, Pz, P6). Mean ERP amplitudes were entered into an R Statistics v.2.15 database (R Core Team, 2012).

For the N170 and P200, analyses of variance (ANOVA) were computed using the within-subjects factors *emotion* (i.e. anger, fear, sadness, neutral) and *hemisphere* (i.e. left, right). For the N200 and P300, ANOVA were computed using the within-subjects factors *emotion* (i.e. fear, sadness), *hemisphere* (i.e. left, midline, right), and *task* (i.e. background, stop). Interaction effects were computed for each ERP. *Post hoc* second order ANOVAs and *t*-tests were used to delineate statistically significant interactions. All *post hoc* tests were corrected for multiple comparisons (*see Benjamini & Hochberg, 1995*). Partial Eta squared is reported for each main effect and interaction as an index of effect size. As the N170 and N200 are negative deflections of the ERP, correlations are inverted. For example, in the context of the N170 and N200, inverse correlations between psychometrics and negative amplitudes suggest a reduction in amplitude, and are presented as positive associations. *T*-tests were used to identify differences in successful motor extinction and associated reaction times between fearful and sad stop stimuli.

7.3.3 Results

7.3.3.1 Behavioural

There was no significant difference in motor extinction success between fearful and sad facial stimuli (fearful [81.58%, SE = 2.48], sad [82.97%, SE = 2.31]; *t* [56] = 1.37, p = .18, padj = .31). However, motor extinction reaction time were shorter to sad (762.29 ms, SE = 26.90) than fearful (822.13 ms, SE = 30.48) stop stimuli (*t* [56] = 4.20, p < .001, padj < .001).

7.3.3.2 N170

Means and standard deviations of N170 amplitudes evoked to fearful, sad, angry, and neutral facial expressions are summarised in Table 4.

There was no statistically significant main effect of *hemisphere* (left [2.00 μ v, SE = .41]; right [2.04 μ v, SE = .38]; *F* [1, 56] = .02, *p* = .90, η_p^2 = .00) on N170 amplitude. Neither was there a statistically significant interaction between *hemisphere* and *emotion* (*F* [3, 54] = 2.43, *p* = .08, η_p^2 = .09).

Table 4.

•			
Emotion	Hemisphere	Μ [μν]	SD
Fear	Left	1.77	3.21
	Right	2.23	3.41
Sad	Left	1.69	3.30
	Right	2.31	3.04
Anger	Left	2.24	3.03
-	Right	2.08	3.44
Neutral	Left	1.56	3.10
	Right	2.32	2.64

Descriptions of N170 amplitudes across emotion and hemisphere to background stimuli

Note. Left site comprised an average of P07, P7, P03, and O1 electrodes. Right site comprised an average of P08, P8, P04, and O2 electrodes.

A main effect of *emotion* on the N170 was identified (*F* [3, 54] = 12.58, *p* < .001, η_p^2 = .41). Although Mauchly's test indicated violation of the assumption of sphericity (X^2 [5] = 29.65, *p* < .001), the effect held after correcting degrees of freedom using Greenhouse-Geisser sphericity estimates (ε = .74, *F* [2.23, 124.64] = 6.37, *p* < .05, η_p^2 = .10). The effect of *emotion* on the N170 was explored using six paired-sample *t*-tests. As evidenced in Figure 6, the effect of *emotion* was driven by larger N170 amplitudes to anger (1.61 µv, SE = .37) than fearful (2.00 µv, SE = .38; *t* [56] = 2.42, *p* = .02, *padj* = .04), sad (2.16 µv, SE = .39; *t* [56] = 3.29, *p* < .05, *padj* < .05), or neutral facial expressions (2.31 µv, SE = .34; *t* [56] = 6.02, *p* < .001, *padj* < .001). There was no statistically significant difference in N170 amplitude between responses to fearful and sad (t [56] = -.71, p = .48, padj = .48), fearful and neutral (t [56] = -1.67, p = .10, padj = .15) or sad and neutral facial expressions (t [56] = -1.10, p = .28, padj = .33). Main effects are summarised in Table 8 (page 140).

7.3.3.3 P200

Means and standard deviations of P200 amplitudes evoked to fearful, sad, angry, and neutral facial expressions are summarised in Table 5 (page 121).

There was a statistically significant main effect of *hemisphere* on the P200 (*F* [1, 56] = 12.38, p < .001, $\eta_p^2 = .18$) such that the right, relative to left hemispheric P200 was consistently larger across all emotions (fear left [5.03 µv, SE = .37] vs. fear right [6.42 µv, SE = .40]; t [56] = -4.38, p < .001, padj < .001; sad left [5.21 µv, SE = .39] vs. sad right [6.29 µv, SE = .47]; t [56] = -2.62, p = .01, padj = .01; anger left [5.01 µv, SE = .39] vs. anger right [6.09 µv, SE = .48]; t [56] = -2.83, p = .01, padj = .01; neutral left [5.56 µv, SE = .37] vs. neutral right [6.62 µv, SE = .40]; t [56] = -3.46, p < .001 padj < .05). This effect is illustrated in Figure 6.

There was also a statistically significant main effect of *emotion* on the P200 (F [3, 54] = 7.66, p < .001, $\eta_p^2 = .30$), see Figure 6. Although violating Mauchly's test of sphericity (X^2 [5] = 22.63, p < .001), the effect remained significant after correcting degrees of freedom using Greenhouse-Geisser estimates of sphericity ($\varepsilon = .79$, F [2.36, 132.09] = 3.79, p = .02, $\eta_p^2 = .06$). Paired-sample *t*-tests suggested this effect was driven by larger P200 amplitudes to neutral (6.09 µv, SE = .35), relative to fearful (5.73 µv, SE = 35; t [56] = -2.13, p = .04, padj = .04), sad (5.75 µv, SE = .38; t [56] = -2.47, p = .02, padj = .05), and angry facial expressions (5.55 µv, SE = .39; t [56] = 4.68, p < .001, padj < .001). P200 amplitude did not differ in response to fearful and sad (t [56] = -.09, p = .93, padj = .93), fearful and angry (t [56] = .97, p = .34, padj = .40), and sad and angry facial expressions (t [56] = 1.28, p = .21, padj = .31). There was no statistically

interaction between *hemisphere* and *emotion* for the P200 (F [1, 54] = 1.06, p = .37, η_p^2 = .06). Main effects are summarised in Table 8 (page 140).

Table 5.

Descriptions of P200 amplitudes across emotion and hemisphere to background stimuli

Emotion	Hemisphere	Μ [μν]	SD
Fear	Left	5.03	2.75
	Right	5.21	2.90
Sad	Left	5.01	2.90
	Right	5.56	2.78
Anger	Left	6.42	3.04
	Right	6.29	3.58
Neutral	Left	6.09	3.70
	Right	6.62	2.98

Note. Left site comprised an average of P07, P7, P03, and O1 electrodes. Right site comprised an average of P08, P8, P04, and O2 electrodes.



Figure 6. ERP waveforms to fearful, sad, angry, and neutral facial expression. Left ERP is an average of P07, P7, O1, and T7 electrodes. Right ERP is an average of P08, P8, O2, and T8 electrodes. ERP waveform is representative of EEG signal referenced to averaged electrodes.

7.3.3.4 N200

Means and standard deviations of N200 amplitudes evoked to fearful and sad facial stimuli across hemisphere, site, and task are displayed in Table 6 (page 123). Main effects, interactions, and *post hoc* analyses are summarised in Table 8 (page 140).

7.3.3.4.1 Main Effects

There was no statistically significant main effect of *emotion* (F[1, 56] = 2.01, p = .16, $\eta_p^2 = .04$) such that the N200 amplitude evoked to fearful facial expressions (.11 µv, SE = .12) did not differ significantly to that evoked to sad facial expressions (.17 µv, SE = .07). Similarly, there was no main effect of *Task* (F[1, 56] = .53, p = .47, $\eta_p^2 = .01$) with the N200 in background conditions (.17 µv, SE = .10) yielding similar amplitudes to the N200 in stop conditions (.11 µv, SE = .07).

There were two statistically significant main effects. First, a main effect of *site* $(F[1, 56] = 113.06, p < .001, \eta_p^2 = .67)$ such that on average, N200 amplitudes were larger over anterior (-1.32 µv, SE = .13) than parietal sites (1.60 µv, SE = .18). Second, there was a main effect of *hemisphere* ($F[2, 55] = 13.18, p < .001, \eta_p^2 = .32$). Paired-sample *t*-tests suggested that on average, the N200 measured over midline sites (-.40 µv, SE = .18) was larger than when measured over both left (.19 µv, SE = .12; t[56] = 2.32, p = .02, padj = .04) and right hemispheres (.63 µv, SE = .15; t[56] = -5.14, p < .001, padj < .001). Although the left, relative to right hemisphere initially indicated larger N200 amplitudes, this did not survive Hochberg correction (t[56] = -2.00, p = .05, padj = .08). These main effects are illustrated in Figure 7 (page 130).

Table 6.

Descriptions of N200 amplitudes across emotion, hemisphere, site location, and task

Emotion	Stimuli	Site	Hemisphere	Μ [μν]	SD
Fear	Background	Anterior	Left	-2.15	2.16
			Midline	-3.36	1.65
			Right	-2.08	1.84
		Posterior	Left	2.79	2.52
			Midline	1.66	2.78
			Right	4.03	2.89
	Stop	Anterior	Left	42	1.80
			Midline	.27	2.16
			Right	-1.02	1.72
		Posterior	Left	.38	1.94
			Midline	01	2.33
			Right	1.19	2.46
Sad	Background	Anterior	Left	-2.23	1.84
			Midline	-3.37	1.88
			Right	-1.64	1.71
		Posterior	Left	2.64	2.11
			Midline	1.61	2.83
			Right	4.16	2.99
	Stop	Anterior	Left	.18	1.91
			Midline	.24	1.83
			Right	25	1.51
		Posterior	Left	.34	1.77
			Midline	23	2.15
			Right	.64	2.25

Note. Anterior and parietal sites are measured at F and P electrodes, respectively with left, midline, and right hemispheric sites indexed by corresponding electrodes at sites 5, z, and 6, respectively.

7.3.3.4.2 Interactions

The variables of *hemisphere, task, site,* and *emotion* did not yield a statistically significant four-way interaction (F[2, 55] = 2.79, p = .07, $\eta_p^2 = .09$). Furthermore, *emotion* did not interact with *hemisphere* (F[2, 55] = 1.96, p = .15, $\eta_p^2 = .07$), *task* (F[1,56] = .13, p = .72, $\eta_p^2 = .01$), *hemisphere* * *task* (F[2,55] = 2.00, p = .15, $\eta_p^2 = .07$), or *site* * *task* (F[1, 56] = 2.74, p = .10, $\eta_p^2 = .05$). An initial significant interaction between *hemisphere* and *site* (F[2, 55] = 6.89, p < .05, $\eta_p^2 = .20$) violated Mauchly's test of sphericity (X^2 [2] = 9.78, p = .01) and after adjusting degrees of freedom using Greenhouse-Geisser estimates of sphericity ($\epsilon = .86$), did not survive (F[1.72, 96.33] = 2.70, p = .08, $\eta_p^2 = .05$).

There was a significant interaction between *emotion* and *site* (*F* [1, 56] = 7.51, *p* = .01, η_p^2 = .12). Larger N200 amplitudes were recorded over anterior, relative to parietal sites, to both sad (anterior [-1.18 µv, SE = .12], parietal [1.53 µv, SE = .17]; *t* [56] = -10.70, *p* < .001, *padj* < .001) and fearful facial expressions (anterior [-1.46 µv, SE = .16], parietal [1.68 µv, SE = .20]; *t* [56] = 9.95, *p* < .001, *padj* < .001). Though not present at parietal sites (*t* [56] = 1.75, *p* = .09, *padj* = .11), fearful, relative to sad, facial expressions augmented N200 amplitudes at anterior sites (*t* [56] = -2.90, *p* = .01, *padj* = .01).

A statistically significant three-way interaction between *hemisphere, site* and *task* (*F* [2, 56] = 3.94, *p* = .03, η_p^2 = .13) violated Mauchly's test of sphericity (X^2 [2] = 7.22, *p* = .03) but maintained statistical significance following correction of degrees of freedom using Greenhouse-Geisser estimates of sphericity (ϵ = .89; *F* [1.78, 99.73] = 11.23, *p* < .001, η_p^2 = .17). This interaction was explored using *post hoc* lower order ANOVAs.

7.3.3.4.2.1 Hemisphere * Site | Task

Evoked to background stimuli, there was a statistically significant main effect of *hemisphere* (*F* [2, 55] = 28.45, *p* < .001, η_p^2 = .51) such that on average, N200 amplitude was larger over midline sites (-.87 µv, SE = .21), relative to both left (.26 µv, SE = .14; *t* [56] = 4.25, *p* < .001, *padj* < .001), and right hemispheres (1.12 µv, SE = .18; *t* [56] = 7.61, *p* < .001, *padj* < .001). N200 amplitude was greater in the left, than right hemisphere (*t* [56] = 3.48, *p* < .001, *padj* < .001). There was also a significant main effect of *site* (*F* [1, 56] = 209.58, *p* < .001, η_p^2 = .79) such than on average, N200 amplitude was larger over anterior, relative to parietal sites (anterior [-2.47 µv, SE = .17], parietal [2.82 µv, SE = .24]). Paired-sample *t*-tests suggested N200 amplitudes were larger over anterior, than parietal sites at left (anterior [-2.19 µv, SE = .24], parietal [2.72 µv, SE = .28]; *t* [56] = 11.19, *p* < .001, *padj* < .001), right (anterior [-1.86 µv, SE = .21], parietal [4.10 µv, SE = .38]; *t* [56] = -12.15, *p* < .001, *padj* < .001), and midline sites (anterior [-3.37 µv, SE = 1.63], parietal [1.63 µv, SE = .36]; *t* [56] = 12.12, *p* < .001, *padj* < .001).

After correcting degrees of freedom using estimates of sphericity ($\varepsilon = .90$) following a violation of the assumption of sphericity using Mauchly's test (X^2 [2] = 6.73, p = .04), there was a statistically significant interaction between *hemisphere* and *site* (F [1.79, 100.43] = 3.32, p = .05, $\eta_p^2 = .06$). Paired-sample *t*-tests suggested N200 amplitudes evoked to background stimuli over anterior sites were larger at midline sites (-3.37 µv, SE = .21) than both left (-2.19 µv, SE = .24; t [56] = 4.47, p < .001, *padj* < .001) and right hemispheres (-1.86 µv, SE = .21; t [56] = 6.46, p < .001, *padj* < .001). There was no significant difference between N200 amplitude over left and right hemispheres (t [56] = 1.26, p = .21, *padj* = .31). Over parietal sites, N200 amplitudes evoked to background stimuli were larger at midline sites (1.63 µv, SE = .36) than over the right hemisphere (4.10 µv, SE = .38; t [56] = 6.39, p < .001, *padj* < .001). A similar effect was observed between midline sites and the left hemisphere but this did not survive multiple comparison correction (2.72 µv, SE = .28; t [56] = 2.50, p = .02, padj = .27). Furthermore, N200 amplitudes were larger over the left, than right hemisphere (*t* [56] = 3.34, p = .002, padj = .042).

When evoked to stop stimuli, whilst there was no statistically significant main effect of *hemisphere* (*F* [2, 55] = .96, p = .96, $\eta_p^2 = .01$), there was a significant main effect of *site* (*F* [1, 56] = 4.18, p = .050, $\eta_p^2 = .07$) such that on average, N200 amplitude was larger over anterior, than parietal sites (anterior [-.17 µv, SE = .13], parietal [.39 µv, SE = .17]). Paired-sample *t*-tests suggested this effect was driven by larger N200 amplitudes at anterior, than parietal sites in the right (anterior [-.64 µv, SE = .19], parietal [.91 µv, SE = .29]; *t* [56] = 4.19, *p* < .001, *padj* < .001), but not left hemisphere (anterior [-.12 µv, SE = .22], parietal [.36 µv, SE = .22]; *t* [56] = 1.59, *p* = .12, *padj* = .31), or at midline sites (anterior [.26 µv, SE = .25], parietal [-.12 µv, SE = .28]; *t* [56] = 1.19, *p* = .24, *padj* = .31).

There was a statistically significant interaction between *hemisphere* and *site* (*F* [2, 55] = 22.64, p < .001, $\eta_p^2 = .45$). Paired-sample *t*-tests suggested N200 amplitudes evoked to stop stimuli at anterior sites were statistically larger over the right hemisphere (-.64 µv, SE = .19) than midline sites (.26 µv, SE = .25; *t* [56] = 3.92, p < .001, *padj* < .001) but not the left hemispheres (-.12 µv, SE = .22; *t* [56] = 1.04, p = .31, *padj* = .31). There was no statistically significant difference in N200 amplitude between left hemispheres and midline sites (*t* [56] = 1.61, p = .11, *padj* = .31). After correcting for multiple comparisons, N200 amplitudes over parietal sites did not differ between the midline (-.12 µv, SE = .28) and either the right (.91 µv, SE = .29; *t* [56] = 3.07, p < .05, *padj* = .06) or left hemisphere (.36 µv, SE = .22; *t* [56] = 1.26, p = .21, *padj* = .31). There was no statistically significant difference in N200 amplitude between item midline (-.12 µv, SE = .28) and either the right (.91 µv, SE = .29; *t* [56] = 3.07, p < .05, *padj* = .06) or left hemisphere (.36 µv, SE = .22; *t* [56] = 1.26, p = .21, *padj* = .31). There was no statistically significant difference in N200 amplitude between left and right hemispheres (*t* [56] = 1.64, p = .11, *padj* = .31).

7.3.3.4.2.2 Hemisphere * Task | Site

At anterior sites, whilst there was no statistically significant main effect of *hemisphere* (*F* [2, 55] = 2.26, *p* = .11, η_p^2 = .08), there was a significant main effect of *task* (*F* [1, 56] = 241.01, *p* < .001, η_p^2 = .81) such that on average, N200 amplitude was larger in response to background than stop stimuli (background [-2.47 µv, SE = .17], stop [-.17 µv, SE = .13]). Paired-sample *t*-tests suggested this effect was congruent over both left (background [-2.19 µv, SE = .24], stop [-.12 µv, SE = .22]; *t* [56] = 7.41, *p* < .001, *padj* < .001), and right hemispheres (background [-1.86 µv, SE = .21], stop [-.64 µv, SE = .19]; *t* [56] = 5.19, *p* < .001, *padj* < .001), and at midline sites (background [-3.37 µv, SE = .21], stop [.26 µv, SE = .25]; *t* [56] = 13.35, *p* < .001, *padj* < .001).

There was a statistically significant interaction between *hemisphere* and *task* (*F* [2, 55] = 25.06, p < .001, $\eta_p^2 = .48$). Paired-sample *t*-tests suggested N200 amplitudes evoked to background stimuli over anterior sites were statistically larger at midline sites (-3.37 µv, SE = .21) than over both left (-2.19 µv, SE = .24; *t* [56] = 4.47, p < .001, *padj* < .001) and right hemispheres (-1.86 µv, SE = .21; *t* [56] = 6.46, p < .001, *padj* < .001). There was no statistically significant difference in N200 amplitude between left and right hemispheres (*t* [56] = 1.26, p = .21, *padj* = .31). Evoked to stop stimuli, N200 amplitudes were significantly larger over the right hemisphere (-.64 µv, SE = .19) than midline sites (.26 µv, SE = .25; *t* [56] = 3.92, p < .001, *padj* < .001) but not the left hemispheres (-.121 µv, SE = .22; *t* [56] = 1.04, p = .31, *padj* = .31). There was no statistically significant difference in N200 amplitude between left and right hemispheres (*t* [56] = 1.26 µv, SE = .22; *t* [56] = 1.04, p = .31, *padj* = .31). There was no statistically significant difference in N200 amplitude between left and right hemispheres (*t* [56] = 1.61, p = .11, *padj* = .31).

At parietal sites, there was a statistically significant main effect of *hemisphere* (*F* [2, 55] = 17.76, p < .001, $\eta_p^2 = .39$), such than on average N200 amplitude was larger at midline sites (.76 µv, SE = .27) than over the right hemisphere (2.51 µv, SE = .29; *t* [56] = 5.82, p < .001, *padj* < .001). Additionally, larger N200 amplitudes were observed at midline sites relative to the left hemisphere (1.54 µv, SE = .22; *t* [56] = 2.14, p = .04, *padj* = .31) as well as over the left

127
relative to right hemisphere (*t* [56] = 3.11, *p* < .05, *padj* = .06). However, neither finding survived multiple comparison correction. A second statistically significant main effect of *task* (*F* [1, 56] = 122.40, *p* < .001, η_p^2 = .69), suggested that on average, N200 amplitude was larger in response to stop than background stimuli (stop [.39 µv, SE = .17], background [2.82 µv, SE = .24]).

There was a statistically significant interaction between *hemisphere* and *task* $(F[2, 55] = 6.30, p < .05, \eta_p^2 = .19)$. Paired-sample *t*-tests suggested N200 amplitudes evoked to background stimuli over parietal sites were statistically larger at midline sites (1.63 μ v, SE = .36) than over the right hemispheres (4.10 μ v, SE = .38; t [56] = 6.39, p < .001, padj < .001). A similar difference was observed between midline sites and the left hemisphere but this did not survive multiple comparison correction (2.72 μ v, SE = .28; t [56] = 2.50, p = .02, padj = .27). Furthermore, the N200 to background stimuli was larger over the left than right hemisphere (t [56] = 3.34, p < .05, padj = .04). Evoked to stop stimuli, N200 amplitudes were statistically larger over midline sites (-.12 μ v, SE = .28) than the left hemisphere (.36 μ v, SE = .22; *t* [56] = 1.26, *p* = .21, *padj* = .31). Again, a similar difference was observed between midline sites and the right hemisphere but this did not survive multiple comparison correction (.91 μ v, SE = .29; t [56] = 3.07, p < .05, padj = .06). There was no statistically significant difference in N200 amplitude between left and right hemispheres (t [56] = 1.64, p = .11, padi = .31).

7.3.3.4.2.3 Task * Site | Hemisphere

Over the left hemisphere, whilst there was no statistically significant main effect of *task* (*F*[1, 56] = .65, p = .42, η_p^2 = .01), there was a significant main effect of *site* (*F*[1, 56] = 68.43, p < .001, η_p^2 = .55) such that on average, N200 amplitude was larger over anterior than parietal sites (anterior [-1.16 µv, SE = .19], parietal [1.54 µv, SE = .22]). There was a statistically significant interaction between *task* and *site* (*F*[1, 56] = 135.97, p < .001, η_p^2 = .71). Paired-sample *t*tests suggested N200 amplitudes evoked to background stimuli over the left hemisphere were statistically larger over anterior (-2.19 μ v, SE = .24) than parietal sites (2.72 μ v, SE = .28; t [56] = 11.19, p < .001, padj < .001). No statistical difference in N200 amplitude was observed over the left hemisphere to stop stimuli (anterior [-.12 μ v, SE = .24], parietal [.36 μ v, SE = .22]; t [56] = 1.59, p = .12, padj = .31). Furthermore, whilst N200 amplitude was statistically larger for background, relative to stop stimuli over anterior sites (t [56] = 7.41, p< .001, padj < .001), the inverse was observed over parietal sites (t [56] = 10.09, p < .001, padj < .001).

Over midline sites, there was a statistically significant main effect of task (F[1, 56] = 16.05, p < .001, η_p^2 = .22), such that on average, N200 amplitude was larger when evoked to background than stop stimuli (background [-.87 μ v, SE = .21], stop [.07 μ v, SE = .21]). There was a second significant main effect of *site* $(F[1, 56] = 58.07, p < .001, \eta_p^2 = .51)$ such that on average, N200 amplitude was larger over anterior than parietal sites (anterior [-1.16 μ v, SE = .19], parietal $[.76 \,\mu v, SE = .27]$). There was a statistically significant interaction between *task* and site (F[1, 56] = 171.18, p < .001, $\eta_p^2 = .75$). Paired-sample *t*-tests suggested N200 amplitudes evoked to background stimuli over midline sites were statistically larger over anterior (-3.37 μ v, SE = .21) than parietal sites (1.63 µv, SE = .36; t [56] = 12.12, p < .001, padj < .001). No statistical difference in N200 amplitude was observed over the left hemisphere to stop stimuli (anterior [.26 μ v, SE = .25], parietal [-.12 μ v, SE = .28]; t [56] = 1.19, p = .24, padj = .31). Furthermore, whilst N200 amplitude was statistically larger for background, relative to stop stimuli over anterior sites (t [56] = 13.35, p < .001, padj < .001), the inverse was observed over parietal sites (t [56] = 5.06, p < .001, *padj* < .001).

Over the right hemisphere, there was a statistically significant main effect of *task* (*F* [1, 56] = 31.17, *p* < .001, η_p^2 = .36), such that on average, N200 amplitude was larger when evoked to stop than background stimuli (stop [.14 µv, SE = .16], background [1.12 µv, SE = .18]). There was a second significant main effect of *site* (*F* [1, 56] = 111.10, *p* < .001, η_p^2 = .67) such that on average,

N200 amplitude was larger over anterior than parietal sites (anterior [-1.25 μ v, SE = .16], parietal [2.51 μ v, SE = .29]). There was a statistically significant interaction between *task* and *site* (*F* [1, 56] = 78.75, *p* < .001, η_p^2 = .58). Paired-sample *t*-tests suggested N200 amplitudes evoked to background stimuli over the right hemisphere were statistically larger over anterior (-1.86 μ v, SE = .21) than parietal sites (4.10 μ v, SE = .38; *t* [56] = 12.15, *p* < .001, *padj* < .001). A similar relationship was observed to stop stimuli (anterior [-.64 μ v, SE = .19], parietal [.91 μ v, SE = .29]; *t* [56] = 4.19, *p* < .001, *padj* < .001). Furthermore, whilst N200 amplitude was statistically larger for background, relative to stop stimuli over anterior sites (*t* [56] = 5.19, *p* < .001, *padj* < .001), the inverse was observed over parietal sites (*t* [56] = 8.85, *p* < .001, *padj* < .001).

7.3.3.5 P300

Means and standard deviations of P300 amplitudes to fearful and sad facial expressions collapsed over their respective trials are exhibited in Table 7 (page 133), with associated waveforms exhibited in Figure 7 (page 131). Main effects, interactions, and *post hoc* analyses are summarised in Table 8 (page 140).

7.3.3.5.1 Main Effects

Although there was no main effect of *task* (*F* [1,56] = .40, *p* = .53, η_p^2 = .01) or *emotion* (*F* [1, 56] = 2.15, *p* = .15, η_p^2 = .04), there was a main effect of *site* on the P300 (*F* [1, 56] = 156.99, *p* < .001, η_p^2 = .74). On average, larger P300 amplitudes were recorded over parietal, relative to anterior electrode sites (parietal [1.94 µv, SE = .15], anterior [-.89 µv, SE = .11)]. A main effect of *hemisphere* (*F* [2, 55] = 6.44, *p* < .05, η_p^2 = .19), suggested midline P300 amplitudes (.91 µv, SE = .14) were larger than over left (.13 µv, SE = .11; *t* [56] = -3.61, *p* < .001, *padj* < .05) and right hemispheres (.54 µv, SE = .13; *t* [56] = 2.22, *p* = .03, *padj* = .04). Additionally, P300 amplitude was larger in the right, relative to left hemisphere (*t* [56] = -2.15, *p* = .04, *padj* = .05).

7.3.3.5.2 Interactions

Emotion did not significantly interact with *hemisphere, site,* and *task* (*F* [2,55] = .17, p = .84, $\eta_p^2 = .01$) or *hemisphere* and *site* (*F* [2,55] = 2.29, p = .11, $\eta_p^2 = .08$). However, e*motion* formed three-way interactions with *site* and *task* (*F* [1, 56] = 10.49, p < .05, $\eta_p^2 = .16$) and *hemisphere* and *task* (*F* [2,55] = 5.79, p = .01, $\eta_p^2 = .17$). These interactions were explored using *post hoc* lower order ANOVAs.



Figure 7. ERP waveforms to fearful and sad stop facial expressions during background and stop task conditions, at bilateral frontal, midline, and parietal electrode sites. ERP waveform is representative of EEG signal referenced to linked mastoid electrodes.

7.3.3.5.2.1 Emotion * Task | Site

Over anterior sites, there was a significant main effect of *emotion* (*F* [1, 56] = $5.13, p = .03, \eta_p^2 = .08$), such that on average, larger P300 amplitudes were evoked to fearful, relative to sad faces (fearful [-.78 µv, SE = .10], sad [-1.01 µv, SE = .14]). A second main effect of *task* (*F* [1, 56] = 121.06, *p* < .001, η_p^2 = .68), suggested that on average, larger P300 amplitudes were evoked to stop, relative to background stimuli (stop [.06 µv, SE = .12], background [-1.84 µv, SE = .16]). Additionally, there was a significant interaction between *emotion* and *task* (*F* [1, 56] = 11.75, *p* < .001, η_p^2 = .17), with paired-sample *t*-tests suggesting this interaction was driven by larger responses to sad, relative to fearful, stop stimuli (sad [.35 µv, SE = .12], fearful [-.23 µv, SE = .17]; *t* [56] = - 3.92, *p* < .001, *padj* < .001), with no amplitude difference when evoked to background stimuli (sad [-1.90 µv, SE = .16], fearful [-1.78 µv, SE = .18]; *t* [56] = .85, *p* = .40, *padj* = .93).

Over parietal sites, there was a significant main effect of *emotion* (*F* [1, 56] = 155.55, p < .001, $\eta_p^2 = .74$), such that on average, larger P300 amplitudes were evoked to sad, relative to fearful faces (sad [1.90 µv, SE = .15], fearful [-1.01 µv, SE = .14]). A second main effect of *task* (*F* [1, 56] = 7.48, p = .01, $\eta_p^2 = .12$), suggested that on average, larger P300 amplitudes were evoked to background, relative to stop stimuli (background [.63 µv, SE = .12], stop [.27 µv, SE = .10]). Additionally, there was a significant interaction between *emotion* and *task* (*F* [1, 56] = 118.43, p < .001, $\eta_p^2 = .68$). Although not surviving multiple comparison correction, paired-sample *t*-tests suggested this interaction might be driven by larger responses to fearful, relative to sad, stop stimuli (fearful [1.11 µv, SE = .19], sad [.77 µv, SE = .16]; *t* [56] = 2.66, p = .01, padj = .31). No amplitude difference was observed when evoked to background stimuli (fearful [2.87 µv, SE = .21], sad [3.04 µv, SE = .22]; *t* [56] = 1.13, p = .26, padj = .93).

132

Table 7.

Descriptions of P300 amplitudes across emotion, hemisphere, site location, and task

Emotion	Stimuli	Site	Hemisphere	Μ [μν]	SD
Fear	Background	Anterior	Left	-1.66	1.82
			Midline	-1.91	2.01
			Right	-1.77	1.82
		Posterior	Left	2.240	2.35
			Midline	3.47	2.22
			Right	2.89	2.19
	Stop	Anterior	Left	68	1.93
			Midline	1.00	1.98
			Right	-1.00	1.72
		Posterior	Left	.27	2.01
			Midline	1.39	2.16
			Right	1.67	1.98
Sad	Background	Anterior	Left	-1.94	1.64
			Midline	-2.32	1.75
			Right	-1.44	1.89
		Posterior	Left	2.25	1.89
			Midline	3.60	2.44
			Right	3.25	2.54
	Stop	Anterior	Left	.19	1.64
			Midline	1.21	1.69
			Right	35	1.57
		Posterior	Left	.40	1.88
			Midline	.85	1.95
			Right	1.05	1.89

Note. Anterior and parietal sites are measured at F and P electrodes, respectively with left, midline, and right hemispheric sites indexed by corresponding electrodes at sites 5, z, and 6, respectively.

7.3.3.5.2.2 Emotion * Site | Task

When evoked to background stimuli, whilst there was no main effect of *emotion* (F [1, 56] = .09, p = .76, η_p^2 = .01), there was a significant main effect of *site* (F [1, 56] = 226.76, p < .001, η_p^2 = .80), such that on average, larger P300 amplitudes were evoked over parietal, relative to anterior sites (parietal [2.95 μ v, SE = .20], anterior [-1.84 μ v, SE = .16]). There was no significant interaction between *emotion* and *site*.

When evoked to stop stimuli, whilst there was no main effect of *emotion* (*F* [1, 56] = 3.15, p = .08, $\eta_p^2 = .05$), there was a significant main effect of *site* (*F* [1, 56] = 12.79, p < .001, $\eta_p^2 = .19$), such that on average, larger P300 amplitudes were evoked over parietal, relative to anterior sites (parietal [.94 µv, SE = .16], anterior [.06 µv, SE = .09]). Additionally, there was a significant interaction between *emotion* and *site* (*F* [1, 56] = 14.34, p < .001, $\eta_p^2 = .20$). Paired-sample *t*-tests suggested that whilst P300 amplitudes were larger when evoked to sad, relative to fearful facial stimuli at anterior sites (sad [.35 µv, SE = .11], fear [-.23 µv, SE = .17]; *t* [56] = 3.92, p < .001, padj < .001), the inverse was observed at parietal sites (fearful [1.11 µv, SE = .19], sad [.77 µv, SE = .16]; *t* [56] = 2.66, p = .01, padj = .31). However, this latter result did not survive multiple comparison correction.

7.3.3.5.2.3 Site * Task | Emotion

When evoked to fearful facial stimuli, whilst there was no main effect of *task* (*F* [1, 56] = .85, p = .36, $\eta_p^2 = .02$), there was a significant main effect of *site* (*F* [1, 56] = 133.48, p < .001, $\eta_p^2 = .70$), such that on average, larger P300 amplitudes were evoked over parietal, relative to anterior sites (parietal [1.99 μ v, SE = .16], anterior [-1.01 μ v, SE = .14]). Additionally, there was a significant interaction between *site* and *task* (*F* [1, 56] = 70.72, p < .001, $\eta_p^2 = .56$). Paired-sample *t*-tests suggested that whilst P300 amplitudes were larger when evoked to fearful stop, relative to background stimuli at anterior sites (stop [-.23 μ v, SE = .17], background [-1.78 μ v, SE = .18]; *t* [56] = 7.74, p < .001, padj < .001), the

inverse was observed at parietal sites (background [2.87 μ v, SE = .21], stop [1.11 μ v, SE = .19]; *t* [56] = 7.03, *p* < .001, *padj* < .001).

When evoked to sad facial stimuli, whilst there was no main effect of *task* (*F* [1, 56] = .01, *p* = .91, η_p^2 = .00), there was a significant main effect of *site* (*F* [1, 56] = 152.71, *p* < .001, η_p^2 = .73), such that on average, larger P300 amplitudes were evoked over parietal, relative to anterior sites (parietal [1.90 µv, SE = .15], anterior [-.78 µv, SE = .10]). Additionally, there was a significant interaction between *site* and *task* (*F* [1, 56] = 135.39, *p* < .001, η_p^2 = .71). Paired-sample *t*-tests suggested that whilst P300 amplitudes were larger when evoked to sad stop, relative to background stimuli at anterior sites (stop [-.35 µv, SE = .11], background [-1.90 µv, SE = .16]; *t* [56] = 11.24, *p* < .001, *padj* < .001), the inverse was observed at parietal sites (background [3.04 µv, SE = .22], stop [.77 µv, SE = .16]; *t* [56] = 9.60, *p* < .001, *padj* < .001).

7.3.3.5.2.4 Emotion * Hemisphere | Task

When evoked to background stimuli, whilst there was no main effect of *emotion* $(F[1, 56] = .09, p = .76, \eta_p^2 = .01)$, there was a significant main effect of *hemisphere* $(F[1, 56] = 3.62, p = .03, \eta_p^2 = .12)$. Though not surviving multiple comparison correction, paired-sample *t*-tests suggested that P300 amplitudes were larger in the right hemisphere $(.73 \ \mu v, SE = .15; t [56] = 2.48, p = .02, padj = .43)$ and at midline sites $(.71 \ \mu v, SE = .16; t [56] = 2.26, p = .03, padj = .67)$, relative to the left hemisphere $(.22 \ \mu v, SE = .13)$. There was no statistical difference between the P300 measured at midline sites and those over the right hemisphere (t [56] = -.09, p = .93, padj = .93). There was no significant interaction between *emotion* and *hemisphere*.

When evoked to stop stimuli, whilst there was no main effect of *emotion* (*F* [1, 56] = 3.15, p = .08, $\eta_p^2 = .05$), there was a significant main effect of *hemisphere* (*F* [1, 56] = 7.44, p < .001, $\eta_p^2 = .21$). Paired-sample *t*-tests suggested that P300 amplitudes were larger at midline sites (1.10 µv, SE = .20) than over the

left hemisphere (.05 μ v, SE = .15; *t* [56] = 3.67, *p* < .001, *padj* < .001). Whilst a similar finding was observed between midline sites and the right hemisphere (.34 μ v, SE = .15; *t* [56] = 3.18, *p* < .05, *padj* = .07), this finding did not survive multiple comparison correction There was no statistical difference between the P300 measured over the left or right hemispheres (*t* [56] = 1.18, *p* = .24, *padj* = .93).

Additionally, there was also a significant interaction between *emotion* and *hemisphere* (F[1, 56] = 8.31, p < .001, $\eta_p^2 = .23$). Paired-sample *t*-tests suggested that P300 amplitudes were larger when evoked to fearful, relative to sad facial stimuli over the left (fearful [-.21 μ v, SE = .16], sad [.30 μ v, SE = .17]; t [56] = 4.12, p < .001, padj < .001), but not right hemisphere (fearful [.33 μ v, SE = .16], sad [.35 μ v, SE = .17]; t [56] = .11, p = .91, padj = .93) or midlines sites (fearful [1.19 μν, SE = .22], sad [1.03 μν, SE = .20]; t [56] = 1.39, p = .17, padj = .93). Moreover, when evoked to fearful faces, P300 amplitudes were larger over midline sites (1.19 μ v, SE = .22), relative to the left hemisphere (-.21 μ v, SE = .16; t [56] = 4.55, p < .001, padj < .001). Although P300 amplitudes were also found to be larger over midline, relative to the right hemispheres (.33 μ v, SE = .16; t [56] = 3.32, p < .05, padj = .07) and over the right, relative to left hemisphere (t [56] = 2.17, p = .03, padj = .75), these findings did not survive multiple comparison correction. When evoked to sad faces, whilst P300 amplitudes were larger over midline sites (1.03 μ v, SE = .20) than both left (.30 μv , SE = .17; t [56] = 2.45, p = .02, padj = .43) and right hemispheres (.35 μv , SE = .17; t [56] = 2.55, p = .01, padj = .36), these findings did not survive multiple comparison correction. There was no statistical difference between the P300 measured over the left or right hemispheres (t [56] = .19, p = .85, padj = .93).

7.3.3.5.2.5 Emotion * Task | Hemisphere

In the left hemisphere, there was no statistically significant main effect of either *emotion* (*F* [1, 56] = 2.89, *p* = .10, η_p^2 = .05) or *task* (*F* [1, 56] = 1.16, *p* = .29, η_p^2 = .02). However, there was a significant interaction between *emotion* and

task (*F* [1, 56] = 8.90, p < .05, $\eta_p^2 = .14$) such that whilst P300 amplitudes were larger when evoked to sad, relative to fearful stop stimuli (sad [.30 µv, SE = .17], fearful [-.21 µv, SE = .16]; *t* [56] = 4.12, p < .001, *padj* < .001), there was no amplitude difference for background stimuli (sad [.16 µv, SE = .15], fearful [.29 µv, SE = .17]; *t* [56] = .75, p = .46, *padj* = .93).

At midline sites, there were no main effects of either *emotion* (F [1, 56] = 2.25, p = .14, η_p^2 = .04) or *task* (F [1, 56] = 3.06, p = .09, η_p^2 = .05). Additionally, there was no significant interaction between *emotion* and *task* (F [1, 56] = .01, p = .92, η_p^2 = .00).

In the right hemisphere, whilst there was no statistically significant main effect of *emotion* (*F*[1, 56] = 2.22, *p* = .14, η_p^2 = .04), there was a statistically significant main effect of *task* (*F*[1, 56] = 6.58, *p* = .01, η_p^2 = .10), such that on average the P300 response to background stimuli was larger than that to stop stimuli (background [.73 µv, SE = .15], stop [.34 µv, SE = .15]). There was no statistically significant interaction between *emotion* and *task* (*F*[1, 56] = 3.02, *p* = .09, η_p^2 = .05).

7.3.3.5.2.6 Hemisphere * Task | Emotion

When evoked to fearful facial stimuli, there was a significant main effect of *hemisphere* (*F* [2, 55] = 8.20, p < .001, $\eta_p^2 = .23$) but not *task* (*F* [1, 56] = .85, p = .36, $\eta_p^2 = .02$), Paired-sample *t*-tests suggested that P300 amplitudes were larger at midline sites (.99 µv, SE = .16) than in both the left (.04 µv, SE = .13; *t* [56] = 3.94, p < .001, *padj* < .001) and right hemispheres (.45 µv, SE = .14; *t* [56] = 2.94, p < .001, *padj* < .001). There was no statistical difference between the P300 measured over left and right hemispheres (t [56] = 1.88, p = .06, *padj* = .93).

Additionally, there was a significant interaction between *hemisphere* and *task* (*F* [2, 55] = 4.18, p = .02, $\eta_p^2 = .13$). Although not surviving multiple comparison

correction, paired-sample *t*-tests suggested that P300 amplitudes evoked to fearful faces were significantly larger when used as background, relative to stop stimuli in the left hemisphere (background [.29 μ v, SE = .17], stop [-.21 μ v, SE = .16]; t [56] = 2.56, p = .01, padj = .36). There were no statistically significant differences in the right hemisphere (background [.56 μ v, SE = .17], stop [.33 μ v, SE = .16]; t [56] = 1.36, p = .18, padi = .93) or at midline sites (background [.78] μ v, SE = .19], stop [1.19 μ v, SE = .22]; t [56] = -1.59, p = .12, padj = .93). Moreover, when used as background stimuli, there was no statistical difference in P300 amplitudes evoked to fearful faces over midline sites (.78 μ v, SE = .19), relative to the left (.29 μ v, SE = .17; t [56] = 1.89, p = .06, padj = .93), or right hemisphere (.56 μ v, SE = .17; t [56] = 1.03, p = .31, padj = .93). There was also no statistical difference between the left and right hemisphere (t [56] = .98, p = .33, *padj* = .93). When used as stop stimuli, P300 responses to fearful faces were larger at midline sites (1.19 μ v, SE = .22) than over the left hemisphere (-.21 μv, SE = .17; t [56] = 4.55, p < .001, padj < .001). Although P300 differences were observed between midline sites and the right hemisphere (.33 μ v, SE = .17; t [56] = 3.32, p < .05, padj = .07) and over the right, relative to left hemisphere (t [56] = 2.17, p = .03, padj = .75), these findings did not survive multiple comparison correction.

When evoked to sad facial stimuli there was a significant main effect of *hemisphere* (*F* [2, 55] = 3.86, p = .03, $\eta_p^2 = .12$), but not *task* (*F* [1, 56] = .01, p = .91, $\eta_p^2 = .00$), Although not surviving multiple comparison correction, paired-sample *t*-tests suggested that P300 amplitudes were significantly larger at midline sites (.84 µv, SE = .15) than in the left (.23 µv, SE = .12; *t* [56] = 2.77, p = .01, *padj* = .26). No P300 amplitude differences were observed between the right hemisphere (.63 µv, SE = .14) and either midline sites (*t* [56] = 1.08, p = .28, *padj* = .93) or the left hemisphere (*t* [56] = 1.97, p = .05, *padj* = .93).

Additionally, there was a significant interaction between *hemisphere* and *task* (*F* [2, 55] = 4.42, p = .02, $\eta_p^2 = .14$). Paired-sample *t*-tests suggested that P300 amplitudes evoked to sad faces were significantly larger when used as

background, relative to stop stimuli in the right (background [.90 µv, SE = .17], stop [.35 µv, SE = .17]; *t* [56] = 2.89, *p* = .01, *padj* = .20), but not left hemisphere (background [.16 µv, SE = .17], stop [.30 µv, SE = .16]; *t* [56] = -.69, *p* = .49, *padj* = .93) or at midline sites (background [.64 µv, SE = .19], stop [1.03 µv, SE = .20]; *t* [56] = -1.56, *p* = .12, *padj* = .93). Moreover, when used as background stimuli, although P300 amplitude was significantly larger in the right (.90 µv, SE = .17), than left hemisphere (.30 µv, SE = .17; *t* [56] = 3.26, *p* < .05, *padj* = .07), this did not survive multiple comparison correction. No P300 amplitude differences were observed between midline sites (.64 µv, SE = .19) and either the right (*t* [56] = 1.07, *p* = .29, *padj* = .93) or left hemisphere (*t* [56] = 1.94, *p* = .06, *padj* = .93). When used as stop stimuli, P300 responses to sad faces were larger at midline sites (1.03 µv, SE = .20) than both left (.30 µv, SE = .17; *t* [56] = 2.45, *p* = .02, *padj* = .43), and right hemispheres (.35 µv, SE = .17; *t* [56] = 2.55, *p* = .01, *padj* = .36). There was no significant difference in P300 amplitude between left and right hemispheres (*t* [56] = .19, *p* = .85, *padj* = .93).

Table 8.

ERP	Effect	F	df	η_{p}^{2}
N170	Emotion	6.37**	2.23, 124.64	.10
P200	Hemisphere	12.38***	1, 56	.18
	Emotion	3.79*	2.36, 132.09	.06
N200	Site	113.06***	1, 56	.67
	Hemisphere	13.18***	2, 55	.32
	Emotion*Site	7.51**	1, 56	.12
	Hemisphere*Site*Task	11.23***	1.78, 99.73	.17
	Hemisphere {Background}	28.45***	2, 55	.51
	Site {Background}	209.58***	1, 56	.79
	Hemisphere*Site {Background}	3.32*	1.79, 100.43	.06
	Site {Stop}	4.18*	1, 56	.07
	Hemisphere*Site {Stop}	22.64***	2, 55	.45
	Task {Anterior}	241.01***	1, 56	.81
	Hemisphere*Task {Anterior}	25.06***	2, 55	.48
	Hemisphere {Parietal}	17.76***	2, 55	.39
	Task {Parietal}	122.40***	1, 56	.69
	Hemisphere*Task {Parietal}	6.30**	2, 55	.19
	Site {Left}	68.43***	1, 56	.55
	Task*Site {Left}	135.97***	1, 56	.71
	Task {Midline}	16.05***	1, 56	.22
	Site {Midline}	58.07***	1, 56	.51
	Task*Site {Midline}	171.18***	1, 56	.75
	Task {Right}	31.17***	1, 56	.36
	Site {Right}	111.10***	1, 56	.67
_	Task*Site {Right}	78.75***	1, 56	.58
P300	Site	156.99***	1, 56	.74
	Emotion*Site*Task	10.49**	1, 56	.16
	Emotion*Hemisphere*Task	5.79**	2, 55	.17
	Emotion {Anterior}	5.13*	1, 56	.08
	Task {Anterior}	121.06***	1, 56	.68
	Emotion* I ask {Anterior}	11.75***	1, 56	.17
	Emotion {Parietal}	155.55***	1, 56	.74
	Task {Parietal}	7.48*	1, 56	.12
	Emotion* I ask {Parietal}	118.43***	1,56	.68
	Site {Background}	226.76^^^	1, 56	.80
	Site (Stop)	12.79^^^	1, 56	.19
	Emotion*Site {Stop}	14.34***	1,56	.20
	Site {Fearful}	133.48^^^	1,56	.70
	Site (Deal)	/0./2***	1,56	.56
		152.71***	1,56	./3
	Site ask (Sad)	135.39	1,56	./1
	Hemisphere (Background)	3.62"	1,56	.03
	Hemisphere (Stop)	7.44	1, 56	.21
	Emotion Hemisphere (Stop)	8.31	1,56	.23
	Emotion [®] Lask {Lett}	8.90***	1,56	.14
	I ask {Rigiil}	0.00	1,50	.11
	Hemisphere (Fearful)	8.20	1,56	.23
	Hemiophere (Cad)	4.19	∠, 55 0, 55	.13
	Hemisphere (Sad)	3.86	2,55	.12
	Hemisphere [®] Lask {Sad}	4.42	2,55	.14

Statistically significant ANOVA summary table for main effects and interactions across ERPs

Note. * p < .05, ** p < .005, *** p < .001. All statistics are reported after adjusting degrees of freedom using Greenhouse-Geisser estimates of sphericity where necessary

7.4 Discussion

This investigation is the first to characterise the electrophysiological responses to face processing and distress-induced motor extinction as an index of the VIM, using a novel FAST. Key findings of this investigation are:

- 1. N170 augmentation was observed in response to angry, but not fearful, sad, and neutral facial expressions.
- P200 augmentation was observed in response to neutral, but not angry, fearful, and sad facial expressions. P200 responses were larger in the left, relative to right, hemisphere uniformly across emotion.
- N200 responses were maximal over anterior midline sites when evoked to background stimuli and over right hemispheric anterior sites when evoked to stop stimuli. There was no effect of emotion on either background or stop stimuli.
- 4. P300 responses to both fearful and sad stop stimuli were larger over anterior, relative to parietal sites, with the inverse observed for background stimuli. Fearful, relative to sad facial expressions evoked larger P300 amplitudes overall. However, whilst P300 amplitudes were larger in response to fearful stimuli over anterior sites, P300 amplitudes were larger in response to sad stimuli over parietal sites.

7.4.1 N170

The N170 was hypothesised to show right hemispheric dominance; augmenting in response to distress (i.e. fear, sad) and in turn to angry, relative to neutral, facial expressions. First, although N170 dominance over the right hemisphere has been reported elsewhere (Bentin et al., 1996; Rossion et al., 2003, Stockdale et al., 2015, *but see Hinojosa et al., 2015 for review*), a main effect of hemisphere was not found in this investigation. It is possible that this null result might be explained through the use of averaged electrode referencing, which was used in order to mitigate attenuation of the temporo-parietal N170/P200 signal (*see Kappenman & Luck, 2012 for review*). Previously, whilst a nasal reference was used in Bentin et al., Rossion et al. referenced EEG signal to the

left mastoid, which might have accounted for right hemispheric dominance as a function of left hemispheric signal attenuation. However as the meta-analysis reported in Hinojosa et al. (2015) highlights both the common use of averaged referencing when measuring the N170 and a predominant right hemispheric N170 distribution, this current finding is atypical.

Second, whilst N170 augmentation was expected in response to both threat (angry) and distress (fear; sad), relative to neutral facial expressions, *post hoc* analysis revealed the main effect of emotion was driven by sensitivity to threat. No amplitude variation was observed between N170 responses to fearful, sad, or neutral expressions. This finding is in line with the threat-superiority effect (Gilbert, 2005; Gilbert et al., 1995), which proposes a self-protection system that directs attention to potential danger. Furthermore, it is supported by investigations suggesting a N170 bias towards negative stimuli (Chai et al., 2012), including angry facial expressions (Eger et al., 2003; Ibánez et al., 2012; Krombholz et al., 2007; Taylor et al., 2004). However, anger-specific augmentation of the N170 contrasts investigations reporting a negativity bias associated with N170 augmentation to fearful facial expressions (Batty & Taylor, 2003; Blau et al., 2007; Taylor et al., 2004; Ramos-Loyo et al., 2009; Shannon et al., 2013). In particular, Taylor et al. (2004) reported fear-evoked N170 augmentation, relative to angry facial expressions.

Conflicting findings might be explained by methodological variation. As discussed in chapter three, section 3.2.1 (page 39), some, but not all (Hinojosa et al., 2015), believe the N170 to reflect early structural face encoding, preceding emotion categorisation (Boucsein et al., 2001; Chai et al., 2012; Dawson et al., 2004; Eimer, 2000; Holmes, 2002). As the FAST requires a motor response to angry facial expressions, anger-specific N170 augmentation reported here might be a function of participations being primed to respond to angry facial stimuli, which unlike fearful, sad, and neutral expressions, were target stimuli. Previously, N170 augmentation has been evoked to actual (Ibánez et al., 2012) and schematic (Eger et al., 2003; Krombholz, 2007) angry

faces when primed with negatively or positively valenced words, but not during un-primed trials (Boucsein et al., 2001; Holmes, 2002) and so supports previous claims that the face-evoked N170 functions through voluntary attention (Holmes et al., 2003). Whilst the FAST does not use affective words to prime responses, task instructions ask participants to attend to the affective elements of stimulus one in order to respond to angry facial expressions.

Measuring the N170 from stimulus two instead of stimulus one might elicit N170 augmentation to fearful and sad facial expressions as participants are directed to respond to these emotions, specifically, during this part of the trial. However, this hypothesis cannot reliably be explored using the current paradigm because of [1] possible convolution of early ERP components and [2] stop stimuli only consisting of fearful and sad expressions. Future investigation should seek to clarify this effect by investigating the role of emotion-specific task instructions on ERPs. Using a FAST, this could be achieved by varying the facial expression used as go and stop stimuli. Although such modifications would alter the paradigm in terms of ecological validity, they are important in order to better understand the role of emotion on the N170 within a FAST.

7.4.2 P200

In contrast to findings relating to the N170, the P200 evoked in this investigation demonstrated right hemispheric dominance. This finding supports classic characterisations of the P200 demonstrating larger P200 amplitudes over right, relative to left, occipito-temporal sites (Farah, 1990; Watanabe et al., 1999).

As an index of second-order processing of facial features such as emotion (Itier & Taylor, 2004; Latinud & Taylor 2006), the P200 was hypothesised to augment in response to facial distress. In contrast to expectations, the current investigation found P200 amplitude was, on average, larger in response to neutral, not affective facial expressions. Additionally, there was no statistically significant amplitude difference in P200 responses to angry, fearful, or sad faces. Previously, although neutral-specific P200 augmentation has been

reported in one investigation (Tortosa et al., 2013), P200 augmentation to affective facial expressions is more commonly observed (Ashley et al., 2004; Chai et al., 2012; Eimer et al., 2003; Schulz et al., 2012) including augmentation to threat and distress, specifically (Gonzalez-Roldan et al., 2011; Shannon et al., 2013).

In this investigation, P200 augmentation to neutral facial stimuli might be an artefact of participants being primed to respond to facial affect. As participants were primed to expect facial affect, P200 augmentation to neutral facial expressions might reflect the recruitment of additional neuronal resources when determining the threat potential of more ambiguous neutral expressions. This theory might help explain the P200 augmentation to neutral expressions reported in Tortosa et al. (2013), where participants were asked to assign reward based on facial expressions. Such conflict might mask subtle effects of emotion and so warrants further investigation in respect to the FAST.

7.4.3 N200

The N200, an electrophysiological index of stimulus classification and conflict monitoring (Fostein & van Petten, 2008; van Veen & Carter, 2002), was hypothesised to present an anterior scalp distribution. As expected, irrespective of facial expression, the N200 was maximal over anterior electrode sites in response to both background and stop stimuli, with stop stimuli showing a slight bias over the right hemisphere. Previously, N200 anterior scalp distributions have been observed in response to background, nogo, and stop stimuli (Bruin & Wijers, 2002; Patel & Azzan, 2005), thought to reflect activation of frontal networks comprising the dIPFC, superior- and medial-frontal gyri, lateral OFC, and ACC (Amodio et al., 2008; Anderer et al., 2004; Enriquez-Geppert et al., 2010; van Veen & Carter, 2002).

Furthermore, findings of this investigation suggested that N200 responses to fearful and sad facial expressions were statistically similar, irrespective of stimulus type (i.e. background, stop). Although N200 augmentation to fearful

facial expressions has been reported in Dennis & Chen (2007), this finding was relative to happy and neutral, but not sad facial expressions, and so cannot be used as a direct comparison to observations reported here. However, as Dennis & Chen interpret their findings as reflecting a distress-driven response, with fear and sadness thought to share similar neural underpinnings (Adolphs & Tranel, 2004, *see chapter two, section 2.5 [page 29]*), it is possible that comparable N200 augmentation might have be evidenced to sad facial expressions.

7.4.4 P300

The P300 was hypothesised to present maximally over parietal and anterior electrode sites when evoked to background and stop stimuli, respectively. Although P300 responses were larger over parietal electrode sites on average, higher-order interaction effects supported this stimulus-/site-specific hypothesis. This pattern of activation supports an extensive literature documenting both the parietal distribution of the P300 response to background stimuli (Baumeister et al., 2014; Luck, 2005, *but see Polich, 2007 for review*) and the anterior distribution of the P300 response to no-go/stop stimuli (Baumeister et al., 2014; Dimosaka et al., 2006; Fallgatter et al., 2005; Hillyard & Kutas, 1983; Kok et al., 2004).

Though maximal at anterior midline sites, subsequent analysis suggested a right hemispheric bias of the stop-P300 response, relative to left and midline sites. Right hemispheric dominance might reflect activation of the right IFG, thought to underpin the generation of the stop-P300 (Bae et al., 2011; Enriquez-Geppert et al., 2010; Kim et al., 2011; Mattia et al., 2012, *but see chapter three, section 3.4.2 [page 52]*). This investigation offers a novel finding such that it is the first to characterise the stop-P300 to facial stimuli, in particular facial-distress. It appears that anterior stop-P300 distributions are uniform across stimulus modality.

The second hypothesis relating to the P300, that of larger P300 responses to fearful, relative to sad stop stimuli, was not supported. In this characterisation of

the electrophysiological indices of the FAST, there were no amplitude differences between the P300 evoked to fearful and sad faces during stop conditions. On a neurobiological level, fearful and sad facial expressions are thought to share similar processing networks (Adolphs & Tranel, 2004), and so moderate the P300 as a function of their biological significance (Halgren et al., 1994; Rossignol et al., 2005). However, when P300 responses to background and stop stimuli were combined, P300 amplitudes were, on average, larger in response to fearful, relative to sad, facial expressions. In relation to the VIM, though not specific to stop stimuli, this finding might reflect the predominant influence of fear thought to trigger distress-induced motor extinction (Blair, 2000; Blair et al., 2004; Dadds et al., 2008; Fairchild et al., 2009; Iria & Barbosa, 2009). As this is the first investigation to use facial distress as a stop stimulus, future replication is required in order to test this effect.

7.4.5 Limitations

Though the FAST provides a useful tool for investigated the electrophysiological indices of face processing and distress-induced motor extinction, is important to delineate its constraints. First, fearful and sad facial expressions were selected as stopping stimuli based on their theoretical relevance to avoidance behaviour (Berkowitz, 1993; Blair, 1995). However, it remains unknown how ERP responses reported here compare to non-distress facial expressions, such as neutral, happy, and angry expressions. Evoking stop ERPs to non-distress emotions is a logical and necessary advancement in validating the FAST. If the FAST is to be regarded as a valid index of the VIM, stop-P300 responses to non-distress facial expressions should be attenuated; indicative of non-distressed faces being less biologically significant.

Similarly, although the FAST is useful for delineating electrophysiological responses to sad and fearful facial expressions, currently, it has not been used to index electrophysiological responses to facial distress in the form of pain. Akin to fear and sadness, pain processing recruits a network that includes the amygdala, ACC, and insula (Akitsuki & Decety, 2009; Decety et al., 2008, 2009;

Sewards & Sewards, 2003). Dysfunction of this network has been associated with poor empathic responses (Raine, 2013), which, as described throughout this thesis might detriment activation of the VIM (Blair, 1995). Currently, knowledge of ERP responses to facial pain is fragmentary, with only a single study, sampling fibromyalgia patients, observing an absence of N170 amplitude variation between painful, neutral and angry facial expressions (González-Roldán, Muñoz, Cifre, Sitges, & Montoya, 2013). Painful faces would be expected to evoke similar stop-ERPs to sad or fearful facial expressions.

Second, in its current incarnation, the FAST can only assess the role of *facial* distress on motor extinction. In addition to facial expressions, distress can be imparted through auditory signals such as crying, whimpers, and screaming, and through body posture such as presentation of open palms and cowering (Blair et al., 2001; Dadds et al., 2006; Stevens et al., 2001). In accordance with body posture, other physical characteristics such as height, weight, muscle mass, and body size would likely moderate stopping, and associated electrophysiological responses. Inclusion of stimuli encapsulating physical characteristics might further moderate motor extinction. Although the importance of facial expressions has been delineated in chapter one, section 1.3.4 (page 12), distress presented through other modalities might have an additive effect on distress-induced motor extinction, which in turn might be quantified on an electrophysiological level. Delineating VIM-relevant electrophysiological responses, as a function of stimulus modality, would add a further dimension to the FAST; increasing its utility.

Finally, although the FAST provides an index of distress-induced motor extinction on an electrophysiological level, source localisation of EEG signal and simultaneous fMRI would benefit understanding of the neurobiological underpinnings of this paradigm. FMRI allows for the localisation of haemodynamic changes within the brain and so is useful for attributing taskrelated regional changes. However, as haemodynamic change can take several seconds to return to baseline, stopping paradigms such as the FAST might not readily transfer into an fMRI environment.

7.4.6 Summary

In summary, the FAST provides a useful tool for investigating the VIM on an electrophysiological level, and in the most part, evokes comparable ERP responses to those reported elsewhere. However, unlike other paradigms, the FAST has utility for evoking face- and motor extinction-related ERPs within a single paradigm, and for the first time, characterises electrophysiological indices of distress-induced motor extinction. Furthermore, this investigation provides grounding for future manipulations of the FAST in order to better [1] index the VIM in terms of ecological validity and [2] understand the neurobiological underpinnings of distress-induced motor extinction. As the VIM is thought to be dysfunctional in individuals characterised by aggressive and CU traits (Blair, 1995, 2001), it is important to understand how ERPs elicited using the FAST might associate with such traits.

Chapter Eight: Correlations between physical aggression/callousness-unemotional traits and faceevoked ERPs

8.1 Introduction

Observed in both adolescent (Dadds et al., 2008; Sylvers et al., 2011) and adult (Blair, 2000; Blair et al., 2004; Fairchild et al., 2009) samples, aggressive and CU traits are thought to predict poor categorisation of, and atypical responses to, facial distress (Marsh & Blair, 2008; Wilson et al., 2011, *see chapter one, section 1.3.4 [page 11]*). In such samples where distress recognition is reported preserved, results likely reflect variation in methodology. For example, Glass & Newman (2006) presented offenders with multiple response options in a face/affect matching task, and Eisenbarth et al. (2008) allowed females with psychopathy an unlimited amount of time when categorising facial expressions. Such paradigms likely facilitated stimulus processing and contributed to increased face recognition accuracy. Compounding this point, a second task reported in Eisenbarth et al. reported sadness processing deficits when stimuli were presented for a much shorter duration of 33 ms.

A recent case has been made for disentangling aggressive/CU traits when investigating facial affect recognition deficits (Dawel et al., 2012). Although sampling from investigations with small sample sizes using only a few trials per condition, the meta-analysis reported in Dawel et al. identified prevalent face processing deficits across facial expressions. This analysis was constrained to individuals scoring high on the affective (i.e. CU traits), but not antisocial (i.e. aggressive traits), factor of psychopathy and contrasts earlier meta-analyses documenting distress-specific processing deficits in samples characterised by both affective and antisocial traits (Marsh & Blair, 2008; Wilson et al., 2011). As suggested in chapter six of this thesis (from page 90), relationships between physical aggression and distinct CU traits may be nonlinear in adults, and so each trait might differentially moderate face processing.

Discussed in chapter one, section 1.3.1 (page 7), aggressive traits detriment facial affect recognition by enhancing threat-specific processing and contributing to the misattribution of threat in ambiguous facial expressions. Previously, threat-biased misattribution has been observed in individuals with psychopathy (Eisenbarth et al., 2008), but as this investigation did not tease apart affective and antisocial traits, it is possible that this threat-bias was driven by the antisocial factor of psychopathy. An amplified response to threat combined with deficient processing of distress might result in a failure to trigger the VIM (*see chapter one, section 1.3.4 [page 12]*), and in turn, facilitate prolonged, uninhibited acts of aggression (Blair, 1995; Marsh & Ambady, 2007).

Due to its high temporal resolution, EEG provides a useful tool for observing the neuronal response to faces (Kappenman & Luck, 2012, *but see chapter three, section 3.2 [page 39]*), and so might aid investigation into how aggressive/CU traits moderate distinct temporal stages of face processing. Faces elicit a well-documented electrophysiological response comprising the N170, an index of early structural face encoding, and the P200, an index of second-order feature integration (Eimer & Holmes, 2002; Rossion et al., 2003; Tortosa et al., 2013).

N170 and P200 responses are associated with individual differences in emotional skill (Meaux et al., 2014). Specifically, the N170 has been shown to correlate with a cluster of traits associated with callous and uncaring behaviours. N170 attenuation has been observed in schizophrenic samples reporting blunted affect (Ibáñez et al., 2012), as well as healthy cohorts presenting low emotional sensitivity/expressivity (Meaux et al., 2014) and high fearless dominance (Almeida et al., 2014). Fearless dominance is thought to relate to uncaring traits (Kimonis et al., 2013). Conversely, N170 *augmentation* has been related to cold-heartedness (Almeida et al., 2014), a factor associated with callous and unemotional traits (Patrick, 2010), and so suggests qualitatively different CU traits might differentially modulate the N170.

Elsewhere, a lack of N170 amplitude variation has been reported in small samples of males (Flor et al., 2002) and females (Eisenbarth et al., 2013) with psychopathy. However, Flor et al. measured a slightly earlier negative component, at non-temporo-parietal sites, during an odour/facial conditioning task, and Eisenbarth et al. measured the N170 using an average of frontal, central, and parietal electrode sites during a mood induction task. In such investigations, inconsistent findings might relate to the use of paradigms that did not solely assess face processing ability and/or the non-optimal measurement of the N170 (*see chapter three, section 3.2.1 [page 39]*)

In contrast, there is only a small literature relating the P200 to emotional skills such as the decoding, understanding, and knowledge-based reasoning of emotion. Not only do such skills have importance for the perception of emotion, but they also facilitate the management of emotion-induced thoughts and behaviours (Meaux et al., 2014). Whereas males with psychopathy have evidenced P200 *augmentation* to neutral facial expressions (Flor et al., 2002), individuals drawn from the community with poor emotional control (i.e. the ability to regulate outward displays of experienced emotion) have shown P200 *attenuation* (Meaux et al., 2014). Whilst not a direct assay of physical aggression, poor control over negative affect is key to the manifestation of physical aggression (*see chapter two, section 2.4 [page 26]*). The apparent differentiation between the effects of CU traits and physical aggression on the P200 warrants further investigation.

Taken together, although the N170 and P200 have been moderated by traits associated with CU behaviours and poor emotional control, a risk factor for physical aggression, associations between N170 and P200 responses with distinct CU and aggressive traits remain unclear - as does the effect of emotion. This investigation delineates these associations in a community sample. CU traits, particularly uncaring traits, were hypothesised to associate with N170 attenuation (poor emotional sensitivity) and physical aggression was expected to associate with P200 attenuation (poor emotional control).

8.2 Methodology

Core methodology has been documented in chapter seven, section 7.3.2 (page 117). For this investigation, only N170 and P200 responses to stimulus one (comprising neutral, fearful, sad, and angry faces) were reported.

8.2.1 Measures

Participants completed the AQ and ICU (*see chapter five, sections 5.3.1 [page 83] and 5.3.2 [page 84], respectively).* For this investigation, only the physical aggression subscale of the AQ was used.

8.2.2 Statistical Analysis

Psychometric responses and mean ERP amplitudes for each subscale and trial type were calculated and entered into an R Statistics v.2.15 dataset (R Core Team, 2012). Average temporo-parietal N170 and P200 responses for each emotion were created using left (P7, PO7, O1, PO3) and right (P8, PO8, O2, PO4) electrodes. Pearson's partial-correlations were calculated for *trait* (i.e. callousness, uncaring, unemotional, physical aggression), *ERP* (i.e. N170, P200), *emotion* (i.e. anger, fear, sadness neutral), and *hemisphere* (i.e. left, right), and controlled for age and sex (demeaned). As temporal overlap of ERP components has been reported up to 200 ms post stimulus onset (*see Kappenman & Luck, 2012*), partial correlations involving the P200 also controlled for preceding N170 amplitude. Furthermore, bilateral N170 and P200 responses were correlated with task behavioural results (i.e. stopping reaction time, stopping success). All comparisons were corrected for multiple comparisons (*see Benjamini & Hochberg, 1995*).

8.3 Results

Means and standard deviations for psychometric measures were as follows: *physical aggression* (M = 52.98 [*t*-transformed], SD = 8.61), *callousness* (M = 5.98, SD = 2.97), *uncaring* (M = 8.00, SD = 3.33), and *unemotional* (M = 8.16, SD = 3.06). Mean amplitudes for N170 and P200 responses are reported in Table 4 (page 119) and Table 5 (page 121), respectively.



Figure 8. Scatter plots of correlations between ICU-measured uncaring traits and ERPs evoked over left temporo-parietal sites (i.e. PO7, P7, PO4, and O1). Subplots A to D indicate the N170 evoked to fearful (r = .41), angry (r = .41), sad (r = .56), and neutral (r = .47) faces. Subplots E to H indicate the P200 evoked to fearful (r = .37), angry (r = .41), sad (r = .51), and neutral (r = .44) faces. Shaded areas denote 95% confidence interval areas along linear associations.

8.3.1 Behavioural

Neither N170 nor P200 responses correlated with stopping reaction times or percentage of correct stopping trials for any emotion.

8.3.2 Correlations

8.3.2.1 N170

In the left, but not right hemisphere, the temporo-parietal N170 response to sad (r = -.56, p < .001, padj < .001) and neutral (r = -.47, p < .001, padj < .001) facial expressions exhibited statistically significant, negative, and moderately strong correlations with *uncaring* traits after controlling for age and sex. Similar, albeit weaker correlations were observed between *uncaring* traits and N170 responses to fearful (r = -.41, p < .05, padj = .07) and angry facial expressions (r = -.41, p < .05, padj = .06) at trend-level significance after correcting for multiple comparisons. These results suggest *uncaring* traits are inversely associated with the ability to identify facial structure regardless of the expression depicted. Correlations between *uncaring* traits and left N170 responses are depicted in subplots A to D of Figure 8.

8.3.2.2 P200

In the left, but not right hemisphere, the temporo-parietal P200 evoked to sad (r = .51, p < .001, padj = .01) and neutral (r = .44, p < .001, padj = .03) facial expressions exhibited statistically significant, positive, and moderately strong correlations with *uncaring* traits after controlling for age, sex, and stimulus-matched N170 amplitude. Although *uncaring* traits correlated in a similar, albeit weaker fashion to the left P200 evoked to angry (r = .41, p < .05, padj = .06) and fearful facial expressions (r = .37, p = .01, padj = .17), these did not survive multiple comparisons correction. Results suggest *uncaring* traits are positively associated with the neuronal response to task salient features depicted in sad and neutral facial expressions, as indexed by the P200. Correlations between

uncaring traits and left hemispheric P200 responses are depicted in subplots E to H of Figure 8.

No statistically significant correlations between either ERP evoked at bilateral temporo-parietal sites to sad, fearful, angry, or neutral facial expressions, and *physical aggression*, *callousness*, or *unemotional* traits survived multiple comparisons correction.

8.4 Discussion

This investigation sought to assess the distinct relationships between physical aggression and CU traits on early electrophysiological indices of face processing. After controlling for age, sex - and in the case of the P200 - N170 amplitude, physical aggression, callousness, and unemotional traits were not correlated with either N170 or P200 response to faces. In contrast, uncaring traits were negatively correlated with N170 responses uniformly across facial expressions, and positively correlated with P200 responses to neutral and sad facial expressions, specifically. Significant correlations were limited to the left hemisphere.

8.4.1 N170

CU traits were hypothesised to correlate with N170 attenuation. In this investigation, uncaring traits, but not callousness or unemotional traits, were inversely correlated with the N170 response to sad and neutral facial expressions, with similar trends observed in response to fearful and angry facial expressions. Though controlling stringently for multiple comparisons, this investigation used a large number of correlations and so the inverse relationship between uncaring traits and the N170 is likely pervasive across facial affect. This finding is supported, to some degree, by the meta-analysis reported in Dawel et al. (2012) that found individuals scoring high on the affective factor of psychopathy evidenced deficient face processing uniformly

across facial affect. However, the current results suggest a specific association with uncaring traits.

Previously, N170 attenuation has been associated with poor emotional sensitivity/expressivity (Meaux et al., 2014) and fearless dominance (Almeida et al., 2014) in cohorts derived from the community, as well as clinical samples characterised by blunted affect (Ibáñez et al., 2012). Whilst fearless dominance is thought to strongly relate to the uncaring subscale of the ICU (Kimonis et al., 2013), this is the first investigation to correlate distinct subscales of the ICU with N170 responses to facial affect. Therefore, although not measured in this investigation specifically, the current results partly support associations between fearless dominance and the N170 observed elsewhere (Almeida et al., 2014).

Neither callousness, the poor affective response to the feelings of others, nor unemotional traits, the lack of emotional expression (Frick, 2003) associated with N170 amplitude. Previously, only one investigation has measured the N170 in association with callous-related traits (Almeida et al., 2014). Here, Almeida et al. found positive associations between cold-heartedness scores and N170 responses to angry facial expressions in a community sample. Although a related measure to cold-heartedness (Patrick, 2010), callousness was not correlated with the N170 in this investigation when evoked to comparable stimuli. It is possible this disparity is attributed to task effects. Whilst the FAST orientates attention to facial affect, the paradigm used in Almeida et al. tasked participants with responding to non-face targets. Discussed in chapter three, section 3.2.1 (page 39), N170 amplitude might be associated with task-specific voluntary attention (Holmes et al., 2003).

8.4.2 P200

This is the first investigation to correlate psychometric measures of physical aggression with P200 responses to facial affect. One previous investigation observed P200 attenuation uniformly across the six core facial expressions (*see chapter one, section 1.3.1 [page 7]*) in relation to poor emotional control, an

index of the likelihood to aggress (Meaux et al., 2014). In contrast to expectations, physical aggression did not correlate with P200 amplitude in this investigation. This result supports previous findings using auditory stimuli, which concluded no correlation between auditory P200 amplitude and either antisocial behaviour or reactive aggression (Barratt et al., 1997; Lijffijt et al., 2012). In the wider context of distress-induced motor extinction, although physical aggression does not appear to associate with stages of face processing, its effect on later downstream mechanisms of cognitive and behavioural control remains unknown.

Prior to this investigation, scores on distinct subscales of the ICU have not been investigated in relation to the P200. In this investigation, uncaring, but not callousness or unemotional traits were found to positively correlate with P200 amplitude, measured over the left hemisphere, in response to sad and neutral facial expressions. Positive trends were also observed between uncaring traits and P200 responses to angry and fearful expressions. Somewhat counterintuitive, although these correlations cannot dictate causations, they indicate uncaring traits might benefit the processing of second-order facial features. As N170 amplitudes were included in this analysis as a confound regressor, temporally overlapping ERP responses cannot be used to explain these results. It is possible that the P200 augmentation observed here might reflect a compensatory measure for socially functioning individuals with uncaring traits during face processing. Whilst this explanation is not concurrent with the P200 augmentation observed in the psychopathic sample reported in Flor et al. (2002), as stated on page 149, the negative components in each study are not comparable.

8.4.3 Implications, Alternative Explanations, and Future Research

Efficient processing of facial affect is a core requirement of the VIM with the ability to accurately process facial distress considered key when triggering distress-induced behavioural inhibition (Blair, 2001). On an electrophysiological level, results of this investigation suggest uncaring traits in individuals drawn

from the community might contribute to a deficient ability to process both structural and configurable aspects of faces, as indexed by the N170 and P200, respectively. Ultimately, in the absence of sufficient executive control (Blair, 2001) dysfunctional processing of faces - particularly those depicting distress might contribute to a failure to trigger the VIM, and in turn the manifestation of prolonged and uninhibited acts of aggression (Blair, 1995; Marsh & Ambady, 2007).

Dysfunction to the VIM has been evidenced in individuals with psychopathic traits (Blair, 1995, 2001; Blair et al., 2004) - characterised by affective, interpersonal, and behavioural malfunction (Hare, 2003). Recent meta-analytic work suggests an association between the affective factor of psychopathy (i.e. CU traits) and pervasive face processing impairment across a constellation of positive and negative emotions after controlling for aggressive traits (Dawel et al., 2012). This current investigation not only supports the view that distinguishing between CU and aggressive traits is important when investigating VIM malfunction but also extends this knowledge by showing specific associations between poor face processing efficacy and uncaring, above callousness, unemotional, and physically aggressive traits. Though limited to a community sample, such findings have implications for our understanding of how distinct psychopathy-related traits might moderate the face processing stage of the VIM and the methodological gain from distinguishing between such traits.

Results of this investigation should be discussed in light of alternative explanations and future avenues of investigation. First, it cannot be ruled out that the absence of correlations between ERP responses and physical aggression, callousness, and uncaring traits were an artefact of the duration of time that stimuli were presented for in the FAST. Previously, female psychopaths, characterised by a constellation of aggressive and CU traits, evidenced distress-processing deficits when presented with stimuli for a short, but not an *ad libitum* duration (Eisenbarth et al., 2008). As stimuli are presented for 800 ms in the FAST, this duration might have masked trait-related facial affect processing deficits indexed on both electrophysiological and behavioural levels. Replication of this investigation using shorter stimulus durations might act to tease apart these trait associations in community samples thought to be proficient at facial affect recognition (Calder et al., 1996 Sawada et al., 2014).

Second, although nuisance regressors were controlled for where appropriate, conclusions in this investigation are drawn from correlations and so causation cannot be inferred. In order to verify these associations, it is necessary to replicate this investigation by comparing electrophysiological responses from groups of high and low trait responders. However, in order to investigate associations between ERP responses and all four psychometric traits observed in this investigation, large age and sex matched cohorts would be required to account for these groups.

8.4.4 Summary

Although wider associations might have been masked as a function of task design, this investigation concluded uncaring traits, but not physical aggression, callousness, and unemotional traits, were associated with both structural and second-order stages of face processing during a FAST. The ability to accurately process facial expressions is a key component of cognitive models of aggression inhibition, such as the VIM. Using EEG to investigate how aggressive/CU traits might moderate distinct stages of face processing is important for gaining a better understanding of how such traits might contribute to the manifestation, and disinhibition of an aggressive response. In addition to face processing, it is important to understand how physical aggression and CU traits might relate to stress-induced motor extinction.

Chapter Nine: Correlations between physical aggression/callous-unemotional traits and motor extinction ERPs

9.1 Introduction

The ability to extinguish a motor response as a function of social stimuli is key for adaptive behaviour (Huster et al., 2014; Sagaspe et al., 2011). As modelled by the VIM, facial distress is thought to halt aggressive/antisocial behaviour in typically developing humans - possibly reflecting the alleviation of a distress-induced empathic response (Blair, 1995, *see chapter one, section 1.3.4 [page 12]*). As such, the VIM is considered dysfunctional in individuals characterised by affective, interpersonal, and behavioural malfunction, such as those exhibiting CU and aggressive traits (Blair, 2001; Hare, 2003). In chapter eight, uncaring traits, but not physical aggression, callousness, or unemotional traits, were shown to correlate with electrophysiological indices of face processing (N170 and P200). The current investigation extends this knowledge by further delineating the relationships between physical aggression/CU traits and electrophysiological indices of distress-induced motor extinction (stop-N200 and stop-P300).

In response to stimuli denoting the need to inhibit or extinguish a motor response, the N200 and P300 are thought to index the recognition of inhibitory need and inhibitory control efficiency, respectively (Bruin & Wijers, 2002; Hughes et al., 2012; Kok et al., 2004). To date, whilst not previously investigated in response to stop stimuli, physical aggression and CU traits have been shown to moderate N200 and P300 responses evoked to paradigms requiring the passive viewing of stimuli and/or premotor inhibitory control (*see chapter three, sections 3.4.1 [page 50] and 3.4.2 [page 52] for review*).

There is disparity as to how physical aggression and CU traits might moderate the N200 response. In both adolescents with CD (Sumich et al., 2012) and adults with psychopathy (Kiehl et al., 2006), N200 augmentation has been reported in response to background (visual letters) and nogo (auditory tones) stimuli, respectively. This positive association is thought to reflect the emergence of CU traits alongside disrupted cortical remodelling (de Brito et al., 2009; Sumich et al., 2012). Elsewhere, N200 *attenuation* has been reported in offenders with psychopathy, relative to offenders with and without schizophrenia (Kiehl et al., 2000). However, this result might be an artefact of either methodological variation or co-varying aggressive traits (*see chapter three, section 3.6.3 [page 57]*). As anterior midline nogo-N200 attenuation has been observed in male offenders characterised by impulsive and violent behaviour, relative to age-matched controls (Chen et al., 2005, 2008), aggressive traits might be associated with poor mobilisation of neuronal resources to inhibitory stimuli and a deficit for the recognition of inhibitory need.

Similarly, auditory and visual P300 responses, across both target and nogo stimuli, have shown attenuation in aggressive samples derived from forensic and community settings (see chapter three, section 3.6.4 [page 58] for review). In contrast, P300 amplitude is thought to remain stable (Patrick, 2008; Raine & Venables, 1987, 1988) or even augment (Carlson & Thái, 2010; Flor et al., 2002; Raine, 1989) in individuals characterised by CU traits. This interaction is further supported by findings of P300 attenuation in individuals with psychopathy during, but not outside of, interpersonal-affective encounters (Hicks et al., 2007). However, as P300 attenuation has been observed in individuals with psychopathy during inhibitory, but not target detection trials (Kim & Jung, 2014), the relationship between CU traits and the P300 might be dependent on task – possibly reflecting specific moderation of inhibitory networks. This finding warrants further investigation.

Taken together, although N200 and P300 responses to background, target, and nogo stimuli have been investigated in samples characterised by aggressive and CU traits, such ERP/trait associations have not been investigated during stop tasks. This investigation delineates these associations in the context of the

VIM, using a FAST, whereby stop ERPs were evoked to fearful and sad facial expressions. It was hypothesised that [1] physical aggression would be inversely correlated with N200 and P300 responses, to both background and stop stimuli. CU traits were hypothesised to [2] positively correlate with N200 responses to both background and stop stimuli, but [3] inversely correlate with P300 responses to stop stimuli only.

9.2 Methods

Core methodology has been documented in chapter seven, section 7.3.2 (page 117). For this investigation, only N200 and P300 responses to fearful/sad background and stop stimuli were reported.

9.2.1 Measures

Participants completed the AQ and ICU (*see chapter five, sections 5.3.1 [page 83] and 5.3.2 [page 84], respectively).* For this investigation, only the physical aggression subscale of the AQ was used.

9.2.2 Statistical Analysis

Questionnaire responses and mean ERP amplitudes for each component (i.e. N200, P300, stop-N200, stop-P300) across each facial expression (i.e. fearful, sad) were calculated and entered into an R Statistics v.2.15 dataset (R Core Team, 2012). Averaged N200 and P300 responses to background and stop stimuli were measured at anterior- (Fz) and parietal- (Pz) midline sites. Pearson's partial-correlations were calculated for *trait* (i.e. callousness, uncaring, unemotional, physical aggression), *trial type* (i.e. background, stop), *ERP* (i.e. N200, P300), *emotion* (i.e. fearful, sad), and *site* (i.e. anterior, parietal). Further associations were explored between *trait* (i.e. callousness, uncaring, unemotional, physical aggression) and *behavioural measures* (i.e. stopping time, accuracy) for each *emotion* (i.e. fearful, sad). Correlations

controlled for age and sex (demeaned), and were corrected for multiple comparisons (*see Benjamini & Hochberg, 1995*).

9.3 Results

9.3.1 Psychometrics

Means and standard deviations for psychometric measures were as follows: physical aggression (M = 52.98 [t-transformed], SD = 8.61), callousness (M = 5.98, SD = 2.97), uncaring (M = 8.00, SD = 3.33), and unemotional traits (M = 8.16, SD = 3.06). Mean amplitudes for N200 and P300 responses are reported in Tables 6 (page 123) and 7 (page 133), respectively.

9.3.2 Behavioural data

Correlations between *trait* (i.e. callousness, uncaring, unemotional, physical aggression) and *behaviour* (i.e. stopping time, accuracy) across *emotion* (i.e. fearful, sad) are displayed in Table 9. Callousness shared a moderate, positive correlation with reaction times to both fearful (r = .27, p = .05, padj = .62) and sad facial stimuli (r = .32, p = .02, padj = .27), as did uncaring traits (fearful [r = .26, p = .05, padj = .62], sad [r = .31, p = .02, padj = .29]). However, these correlations did not survive multiple comparison correction.

Table 9.

Correlations between psychometric measures and behavioural data

	Fearful		Sad	
	Stopping	DT	Stopping	DT
	Accuracy	ΠI	Accuracy	111
P. Aggression	03	22	02	16
Callousness	08	.27	09	.32
Uncaring	.08	.26	.02	.31
Unemotional	.04	.04	.01	.02

Note. * p < .05, ** p < .005, *** p < .001, P.Aggression = Physical Aggression
9.3.3 Correlations

9.3.3.1 N200

At anterior midline sites, physical aggression was strongly and positively correlated with N200 responses to facial distress during stop conditions (fear [r = .51, p < .001, padj < .001], sad [r = .42, p < .001, padj = .03]). Physical aggression did not correlate with N200 responses evoked to background stimuli. At parietal midline sites, uncaring traits were positively correlated with N200 responses to background facial distress (fear [r = .29, p = .03, padj = .19], sad [r = .32, p = .02, padj = .16]), but did not survive multiple comparison correction. No other CU traits correlated with N200 responses evoked to either background or stop stimuli. Significant correlations are displayed in subplots A and B of Figure 9.

9.3.3.2 P300

At anterior midline sites, physical aggression shared a moderately strong, negative correlation with P300 responses to facial distress during stop conditions (fear [r = -.40, p < .05, padj = .04], sad [r = -.32, p = .02, padj = .16]). However, the latter did not survive multiple comparison correction. Physical aggression did not correlate with P300 responses evoked to background stimuli. Uncaring and unemotional traits were inversely correlated with stop-P300 responses to sad facial expressions at anterior and parietal midline sites, respectively (uncaring [r = -.30, p = .02, padj = .18], unemotional [r = -.25, p = .01, padj = .12]). However, neither correlation survived multiple comparison correction. No other CU traits correlated with P300 responses to either background or stop stimuli. Significant correlations are displayed in subplots C and D of Figure 9.



Figure 9. Correlations between physical aggression and stop-N200 (subplots A [r = .51] and B [r = .42]) and stop-P300 amplitudes (subplots C [r = -.40] and D [r = -.32]) evoked to fearful and sad facial expressions, measured at Fz. Shaded areas denote 95% confidence interval areas along linear associations.

9.4 Discussion

This investigation delineated the relationships between physical aggression/CU traits and electrophysiological indices of distress-induced motor extinction. Although no associations were observed between psychometric measures and behavioural responses, physical aggression positively correlated with N200 amplitude, and negatively correlated with P300 amplitude to stop, but not background stimuli. CU traits did not exhibit statistically significant correlations with electrophysiological responses to either stop or background stimuli. Results are discussed in the context of the VIM, in light of limitations and future investigation.

9.4.1 N200

Irrespective of stimulus type, N200 amplitude was hypothesised to inversely correlate with physical aggression and positively correlate with CU traits. Unexpectedly, the current investigation suggested that physical aggression was both [1] statistically unrelated to N200 responses to *background* stimuli, and [2] positively, not negatively, correlated with N200 responses to fearful and sad *stop* stimuli.

The N200 response to stop stimuli is thought to index the preparatory response preceding motor extinction (Kok et al., 2004). Previously, N200 responses to nogo stimuli have shown attenuation over anterior midline sites in samples of male offenders characterised by impulsive and violent behaviour, relative to age-matched controls (Chen et al., 2005, 2008). The observation of a positive correlation between the anterior N200 response to stop stimuli and physical aggression in this investigation might be explained in two ways. First, N200 augmentation in community-based individuals with high trait scores for physical aggression might reflect the need for increased recruitment of neuronal resources when preparing to extinguish a motor response. Second, stop-N200 augmentation might be an artefact of co-varying CU traits previously shown to associate with N200 augmentation (Kiehl et al., 2006; Sumich et al., 2012). However, as no CU traits evidenced statistically significant correlations with N200 responses in this investigation (*see page 164*), this explanation is unlikely.

The relationship between the N200 response to facial expressions and physical aggression was limited to stop stimuli. This finding echoes previous investigation identifying an absence of auditory N200 amplitude variation between male offenders, characterised by behavioural control problems, and non-offender controls (Barratt et al., 1997; Munro et al., 2007). However, as this is the first investigation to associate self-reported physical aggression with N200 amplitude in a community sample, further replication of this null finding is required to verify this association.

In contrast to expectations, no subscale of the ICU was significantly correlated with N200 responses to either fearful or sad facial expressions, across background or stop stimuli. Previously, N200 augmentation, indicative of disrupted cortical remodelling (de Brito et al., 2009; Sumich et al., 2012), has been evidenced in adults with psychopathy and adolescents with CD (Kiehl et al., 2006; Sumich et al., 2012). The null effect reported in this investigation suggests callousness, uncaring, and unemotional traits do not correlate with electrophysiological indices of automatic alerting/stimulus classification (background) or the preparation of motor extinction (stop). However, as this community sample is unlikely to display CU traits to the extent of those in forensic or clinical populations, a group comparison between low- and high-extreme ICU scorers might better answer this question. Here, larger N200 responses to background and stop stimuli would be expected in samples reporting high, relative to low, CU traits.

9.4.2 P300

As P300 attenuation has been repeatedly observed in forensic and community cohorts characterised by aggressive traits (see chapter three, section 3.6.4 [page 50] for review), physical aggression scores were hypothesised to negatively correlate with FAST-evoked P300 responses to both background and stop stimuli. However, in this investigation, whilst physical aggression scores negatively correlated with the P300 response to stop stimuli, comparable results were not observed to background stimuli. Previously, aggression-related P300 attenuation has been observed in community samples in response to both auditory and visual stimuli (Bartholow et al., 2006; Gerstle et al., 1998). Results of this investigation might be explained by methodology. First, P300 responses were measured in response to background, but not target stimuli, and second, said stimuli consisted of facial expressions.

In this investigation, whilst not elicited in response to background stimuli, the observation of an inverse correlation between physical aggression and P300 responses to stop stimuli is novel. However, as this relationship is limited to [1]

self-reported physical aggression, in [2] a community sample, and [3] in response to facial expressions depicting distress, further replication is required. Results of this investigation suggest individuals reporting higher trait physical aggression might be less efficient at engaging motor extinction mechanisms. Although not supported by behavioural data, it remains to be seen whether variation in behavioural responses would emerge from sampling individuals with higher degrees of trait aggression. Previously, increased reaction times have been positively associated with AQ-measured aggressive traits during an affect categorisation paradigm (Pawliczek et al., 2013). However, Pawliczek et al. defined high and low aggression groups based on non-transformed, total AQ scores, and not the physical aggression subscale, specifically.

In light of previous investigations documenting [1] comparable P300 responses to target/background stimuli between individuals with and without CU traits (Patrick, 2008; Raine & Venables, 1987, 1988), but [2] P300 attenuation during inhibitory tasks (Kim & Jung, 2014), CU traits were hypothesised to negatively correlate with P300 responses to stop, but not background facial stimuli. After correcting for multiple comparisons, no CU trait was associated with P300 responses to either stop or background stimuli in this investigation. Disparity between results reported here and those in Kim & Jung (2014) could be explained by variation in task (i.e. nogo vs. stop), stimuli (i.e. geometric shapes vs. faces), or assay of CU traits (i.e. PPI-R vs. ICU), and so warrants further investigation.

9.4.3 Chapter Summary

Findings of this investigation have implications for better understanding how CU traits and physical aggression associate with distress-induced motor extinction on an electrophysiological level. In contrast to associations reported in chapter eight (*see page 147*), whereby uncaring traits, but not physical aggression, were correlated with electrophysiological indices of face processing, this investigation suggests that individuals with high trait scores of physical aggression might require greater recruitment of neuronal resources when

preparing a motor extinction response (N200). Moreover, such individuals appear less able to mobilise neuronal resources during actual response extinction as a function of facial distress (P300). Although findings in this investigation are novel and contribute to theoretical knowledge relating to the distinct stages of the VIM, it remains to be seen whether such findings are comparable to those observed in populations characterised by extreme physical aggression/CU traits.

Chapter Ten: Relationships between omega-3 intake, callous-unemotional traits, and physical aggression

10.1 Introduction

Malnourishment and dietary deficiency contribute to the development of psychopathologies characterised by poor cognition, impulsivity, and aggression (Freeman et al., 2006; Grantham-McGregor et al., 2007; Pollitt, 2000, *but see chapter four for review [from page 63]*). With a growing interest in using diet, specifically omega-3, as a moderator of aggressive behaviour (Buydens-Branchey & Branchey, 2008) and recent investigation into the effects of omega-3 on CU traits (Gow, Vallee-Tourangeau et al., 2013; Raine et al., 2015), there is reason to expect omega-3 might moderate the VIM response. Focusing solely on omega-3s EPA and DHA, this investigation explores relationships between omega-3 dietary intake, physical aggression, and CU traits. Results are discussed in light of methodological and practical utility, limitations, and future avenues of investigation.

Low plasma levels of composite omega-3 and sub optimal ratios of omega-3:-6 have been associated with increased aggressive behaviour in rodent (DeMar et al., 2006; Raygada et al., 1998) and canine (Re et al., 2008) samples. In humans, an inverse association between fatty fish consumption and homicide rates has been observed in a large cross national analysis (Hibbeln, 2001), with dietary deficiency of both EPA and DHA being associated with increased historical incidents of violence (Buydens-Branchey et al., 2003; Umhau et al., 2006). Elsewhere, whilst DHA has been inversely correlated with psychometric measures of aggression (Zaalberg et al., 2015), similar associations with EPA have only been observed in the absence of comorbid substance abuse (Beier et al., 2014). Alcohol and drug use is thought to moderate the structural composition of fatty acids (Pawlosky et al., 2001), and so substance use might be a possible confound when associating fatty acids and aggression.

Further support for the relationship between omega-3 and aggression comes from supplementation research. Although not conclusive (*see chapter four, section 4.3.2.3 [page 71]*), DHA and EPA supplementation has been found to reduce incidents of violence and rule breaking behaviour in forensic settings (Gesch et al., 2002 [44 mg DHA; 80 mg EPA]; Zaalberg et al., 2010 [400 mg DHA; 400 mg EPA]). Moreover DHA-rich supplementation (510 mg DHA; 120 mg EPA) has benefited parent- and teacher-rated aggression in adolescents with ADHD (Hamazaki & Hirayama, 2004; Stevens et al., 2003). However, as such studies vary in terms of target population, qualification of aggression, and supplementation type/dose, direct comparisons are difficult to draw.

Throughout this thesis, focus has been placed on trait measures of physical aggression. Composite levels of omega-3 intake have been inversely correlated with self-reported physical aggression in a large cohort of male offenders, but only at trend-level significance (Meyer et al., 2015). Decreased physical aggression has also been observed in a small sample of female patients with BPD following two months of daily EPA, relative to placebo, supplementation (Zanarini & Frankenburg, 2003 [1000 mg EPA]). However, whilst a three-month period of daily DHA supplementation (670 mg DHA) has been shown to reduce aggressive dialogue in adults – as measured by a picture-frustration paradigm - no effect on physical aggression was observed (Long & Benton, 2013). It is possibly that whilst omega-3 might benefit aggression as a whole, there might be variation in the effect of DHA/EPA on physical aggression, specifically.

In contrast to trait aggression, very few studies have investigated the relationship between omega-3 intake and CU traits, with none to date controlling for the co-variation of aggressive traits. Because of the intricate relationships between aggressive and CU traits explored in chapter six (*see from page 90*), incorporating these relationships into a statistical model is key. Although limited to an adolescent sample with ADHD symptomology, total blood measures of omega-3, EPA, and DHA have been shown to inversely relate to psychometric measures of callousness, uncaring, and unemotional traits (Gow,

Vallee-Tourangeau et al., 2013). However, these results only reached trendlevel significance, and in contrast to results reported above, omega-3 did not correlate with trait aggression in either ADHD or control sample. Furthermore, supplementation of 300 mg DHA and 200 mg EPA resulted in reduced parentreported cold-heartedness, fearlessness, and CU traits, which persisted over a six-month period (Raine et al., 2015).

This investigation delineates associations between EPA and DHA intake on self-reported physical aggression and CU traits in a community sample. Driven by the limited literature documenting the specific associations between omega-3 intake and physical aggression/CU traits, it was hypothesised that [1] physical aggression would be negatively correlated with EPA, but not DHA intake, and [2] CU traits would be negatively correlated with both EPA and DHA intake.

10.2 Methods

10.2.1 Participants

A total of 98 participants (88 female; M age = 21.47; SD = 3.07) were recruited via online, snowball sampling. Participants consented to take part in a brief online questionnaire after responding to an advertisement disseminated via social media.

10.2.2 Measures

Participants completed a battery of questionnaires consisting of demographics and online versions of the AQ, ICU, and FFQ (*see chapter five, sections 5.3.1 [page 83], 5.3.2 [page 84], and 5.3.3 [page 87]*). For this investigation, only the physical aggression subscale of the AQ, and EPA and DHA subscales of the FFQ were used.

10.2.3 Procedure

Participants were briefed and provided informed consent. On average, the questionnaire battery took less that 20 minutes to complete, and respondents were provided with means of redistributing the survey through social media, if they so wished. Respondents were debriefed and provided a means of contacting the lead researcher (DF).

10.2.4 Statistical Analysis

Scores for the physical aggression subscale of the AQ and the callous, uncaring, and unemotional subscales of the ICU were reverse scored and calculated into an R Statistics v.2.15 dataset (R Core Team, 2012). Physical aggression scores were *t*-transformed as a function of age and sex. Guided by Sublette et al. (2011), EPA and DHA intake were calculated as a function of fish type, portion size, consumption frequency, and sex of responder. Due to low self-reported omega-3 consumption, EPA and DHA consumption data were not normally distributed (skew [EPA (3.04, SE = .13), DHA (3.61, SE = .13)], kurtosis [EPA (11.13, SE = .25), DHA (15.87, SE = .25)]), and so underwent *In*transformation. Precedent for *In*-transformation of FFQ data is set in Beier et al. (2014).

A total of eight Pearson's partial correlations were used to delineate relationships between psychometric traits (i.e. physical aggression, callousness, uncaring, unemotional) and daily intake of EPA and DHA. Correlations controlled for demeaned age, sex, and scores on psychometric measures *not* included in the specific analysis (*see chapter six, section 6.2.4 [page 93] for further explanation*). Partial correlations were corrected for multiple comparisons (*see Benjamini & Hochberg, 1995*).

10.3 Results

Means and standard deviations for each self-report measure were as follows: *physical aggression* (M = 51.16, SD = 7.48), *callousness* (M = 4.14, SD = 3.34),

uncaring (M = 7.45, SD = 3.73), *unemotional* (M = 6.86, SD = 3.10). Mean daily EPA consumption was .01 g (SD = .02) and mean daily DHA consumption was .13 g (SD = .28). Following *In*-transformation the mean EPA score was -2.70 (SD = 1.02) and the mean DHA score was -1.78 (SD = 1.17).

Table 10.

Partial correlations between fatty acid intake (EPA, DHA) and psychometric scores (physical aggression, callousness, uncaring, unemotional)

	P. Aggression	Callousness	Uncaring	Unemotional
EPA	39***	13	.07	14
DHA	24	14	.16	.09





Fig 10. Correlations between physical aggression and *In*-transformed EPA (r = -.39, subplot A) and DHA intake (r = -.24, subplot B). Prior to *In*-transformation, EPA and DHA intake were measured in g/day. Shaded areas denote 95% confidence interval areas along linear associations.

Partial correlations between *In*-transformed EPA and DHA intake and psychometric measures are reported in Table 10. Neither EPA nor DHA intake correlated with any subscale of the ICU. EPA shared a moderately strong, inverse correlation with self-reported physical aggression (r = -.39, p < .001, *padj* < .001, *see subplot A of Figure 10*), and contributed to 27.11% of explained variance. Although a similar association was observed between DHA and physical aggression (r = -.24, p = .02, *padj* = .16, *see subplot B of Figure 10*), this correlation did not survive multiple comparison testing.

10.4 Discussion

Whilst relationships between aggression and CU traits are well established (Frick et al., 2003; Frick & White, 2008, *but see chapter one, section 1.3.4 [page 12]*), their independent associations with omega-3 are unclear. This investigation explored correlations between self-reported EPA and DHA intake with psychometric measures of physical aggression, callousness, and uncaring and unemotional traits. Results suggested a single inverse association between physical aggression and intake of EPA, but not DHA. Neither DHA nor EPA correlated with scores on any subscale of the ICU.

To date, investigation between omega-3 and physical aggression has suggested an inverse association between composite dietary omega-3 intake and scores on self-reported measures of physical aggression, in male offenders (Meyer et al., 2015). Although restricted to EPA, and not DHA, this investigation extends these findings into a community sample, using the FFQ to measure omega-3 intake. Results reported here are further supported by investigations using supplementation. Whilst Zanarini & Frankenburg (2003) observed decreased self-reported physical aggression in female patients with BPD following EPA supplementation, no effect on physical aggression was observed in the sample of non-forensic adults reported in Long & Benton (2013) following DHA supplementation. Thought present prior to multiple comparison correction, self-reported intake of DHA and physical aggression were not statistically associated in this investigation. However, as blood assays of DHA have previously shown negative associations with the antecedence to aggress (Fontani et al., 2005; Légaré et al., 2007), and observed aggression in both canine (Re et al., 2008) and human samples (Buydens-Branchey et al., 2003; Hamazaki et al., 2004; Hibbeln et al., 1998; Stevens et al., 2003; Virkkunen, et al., 1987), this result warrants further replication.

This is the first investigation to document associations between self-reported omega-3 intake and scores on subscales of the ICU. Opposing the study hypothesis - bore out of, albeit, a limited literature - this investigation indicated that neither EPA nor DHA intake was significantly correlated with any subscale of the ICU. Previously, low EPA and total omega-3 blood measures have been associated with increased CU traits in male children with ADHD (Gow, Vallee-Tourangeau et al., 2013). Although worth noting that results in Gow, Vallee-Tourangeau et al. only reached trend-level significance, disparity of results could be explained by differences in blood-/self-assessed omega-3 intake and sample populations.

Findings reported in this investigation should be considered in light of three limitations. First, the FFQ has only been validated in patients with major depressive disorder (Sublette et al., 2011) and so its validity across cognitive function, race, and culture remains unknown. However, as noted in chapter five, section 5.3.3 [page 87], the use of the FFQ is justified in this investigation. Nevertheless, results here would benefit from a blood assay of omega-3. Blood measurements would allow for both the accurate quantification of omega-3, and provide a means of assessing omega-6 without the need for additional questionnaire measures. Omega-6 intake has been positively associated with aggressive behaviour (Hibbeln, 2001; Virkkunen et al., 1987; Zaalberg et al., 2015), and would likely moderate the relationship between physical aggression and omega-3 observed here.

Second, this investigation did not control for variation in the intake of substance use (e.g. alcohol, drug). Alcohol consumption and substance abuse have been shown to compromise omega-3 structural composition (Pawlosky et al., 2001), and contribute to the relationship between EPA deficiency and trait aggression (Beier et al., 2014). Beier et al. observed an inverse association between EPA blood measures and aggression in patients with major depressive disorder, but only in the presence of comorbid substance abuse. Although not ethically viable in this online cohort, future investigation would be expected to show a greater effect between low omega-3 intake and increased physical aggression in individuals with high substance consumption.

Third, though not detrimental to statistical evaluation, self-reported measures of EPA and DHA required logarithmic transformation due to a large proportion of respondents reporting only minimal, if any, omega-3 consumption. This data distribution might reflect the sample comprising mostly of students, who might consume fewer, more expensive omega-3-containing foods. Future investigation should seek to delineate direct group comparisons between high and low omega-3 consuming samples, in regards to the presence of aggressive/CU traits. If such results were true to those reported here, high, relative to low, omega-3 consumers would be expected to evidence increased trait physical aggression.

In summary, whilst not associated with CU traits, self-assessed EPA, but not DHA intake was shown to inversely correlate with psychometric-based physical aggression. Such findings are novel and contribute to a growing literature delineating the role of omega-3 in aggressive/CU traits. Future investigation should seek to extend these results using biological assays and demographic-matched group comparisons.

Chapter Eleven: Correlations between omega-3 intake and the electrophysiological indices of the Violence Inhibition Mechanism

11.1 Introduction

The FAST is a useful tool for investigating the VIM as it allows for the simultaneous measurement of electrophysiological responses to facial affect and distress-induced motor extinction (*see chapter seven [from page 102]*). When investigated alongside personality traits associated with VIM dysfunction, distinct correlations have been observed between uncaring traits/physical aggression, and electrophysiological indices of [1] face processing (*see chapter eight [from page 149]*) and [2] distress-induced motor extinction (*see chapter nine [from page 159]*), respectively. From a preventative/treatment perspective, there is a growing interest into how omega-3 might serve to moderate aggressive and CU traits (Raine et al., 2015). Specifically, omega-3 has been shown to negatively correlate with both physical aggression (Meyer et al., 2015; Zanarini & Frankenburg, 2003, *but see chapter 10 [from page 169*) and ICU-measured CU traits (Gow, Vallee-Tourangeau et al., 2013). However, little is known about how omega-3 intake might relate to the VIM, and how this might reflect on an electrophysiological level.

In response to faces, the N170 and P200 are thought to reflect activation of medial temporal networks comprising the fusiform gyrus and STS (Halgren et al., 1994; Nummenmaa et al., 2010; Sadeh et al., 2010). Although an abundance of omega-3 has been identified in the medial temporal lobe (McNamara, 2010), omega-3 localised to the fusiform gyrus/STS has not been used to predict face processing efficacy or electrophysiological indices there of. To date, the single investigation to measure both ERP responses to faces and omega-3 blood concentrations did not report correlations between omega-3 and ERPs prior to 250 ms (Gow, Sumich et al., 2013). However, had this data been presented, conclusions would have been limited due to ERPs being recorded

over midline, and not optimum temporo-parietal sites (Luck & Kappenman, 2013, *see chapter three, section 3.2.1 [page 41]*). In this investigation, there were no statistically significant correlations between omega-3 fatty acid levels and the face-evoked P300.

Motor extinction can be indexed on an electrophysiological level by stop-N200 and stop-P300 ERPs, representative of the recognition of inhibitory need and motor extinction efficacy, respectively (Hughes et al., 2012; Kok et al., 2004). In this thesis, both N200 and P300 responses to stop stimuli have been correlated with trait physical aggression (*see chapter nine [from page 159]*), which in turn, has shown an inverse relationship with self-reported EPA intake (*see chapter ten*). However, previously, no investigation has delineated the relationship between omega-3 intake and electrophysiological indices of motor extinction.

Although a function of alcohol and drug use (Pawlosky et al., 2001), omega-3 might benefit neuronal networks associated with motor extinction through its role in synaptic development (Carlson, 2001; Serfaty & de Velasco, 2014; Suzuki et al., 1997). For example, efficiency of synaptic pruning, a process key for cell specialisation and inter-cellular communication (Bazan, 2005; Kitajka et al., 2004; Litman et al., 2001; Michael-Titus & Priestley, 2014; Mitchell et al., 2001; Uauy & Dangour, 2006), has been associated with greater cell concentrations of DHA (Carlson, 2001). On an electrophysiological level, though not previously investigated alongside omega-3 intake, dysfunctional synaptic pruning has been associated with delayed N200 attenuation, as a function of age, in males with CD (Sumich et al., 2012).

This investigation aimed to delineate associations between EPA and DHA intake on electrophysiological indices of face processing and distress-induced motor extinction. Previously, physical aggression and CU traits have been shown to moderate face processing and motor extinction (*see chapters seven and eight*), as well as being correlated with omega-3 intake (*see chapter ten*). In this investigation, these covariates were held constant. EPA and DHA intake

were hypothesised to positively correlate with [1] N170 and P200 responses, with [2] no prior literature to suggest discriminate effects of emotion. Furthermore, [3] DHA intake was hypothesised to inversely correlate with stop-N200 amplitude and [4] EPA intake was predicted to positively correlate with stop-P300 amplitude.

11.2 Methods

Core methodology has been documented in chapter seven, section 7.3.2 (page 117). In the interest of working in a parsimonious manner, only ERPs of direct interest, such as those evidencing statistically significantly associations with physical aggression or CU traits in chapters eight (*see from page 149*) and nine (*see from page 159*) of this thesis, were co-opted. For this investigation, N170 and P200 responses measured over the left hemisphere, and stop-N200 and stop-P300 responses measured over anterior midline sites were reported.

11.2.1 Measures

Participants completed the FFQ (*see section 5.3.3 [page 87]*), AUDIT (*see section 5.3.4 [page 88]*), and CUDIT (*see section 5.3.5 [page 88]*).

11.2.2 Statistical Analysis

FFQ, AUDIT, and CUDIT responses, as well as mean ERP amplitudes for each trial type were calculated and entered into an R Statistics v.2.15 dataset (R Core Team, 2012). ERPs consisted of averaged left (P7, PO7, O1, PO3), but not right, temporo-parietal N170 and P200 responses evoked to fearful, sad, neutral, and angry facial expressions, as well as anterior midline stop-N200 and stop-P300 responses evoked to fearful and sad facial expressions. Correlations between omega-3 intake and ERPs were investigated using two sets of Pearson's partial correlations. This first associated *omega-3* (i.e. DHA, EPA) and *ERPs* (i.e. N170, P200, stop-N200, stop-P300) across *emotion* (i.e. anger, fear, sadness, neutral). The second associated *omega-3* (i.e. DHA, EPA) and *behavioural responses* (i.e. reaction time, percentage correct). For reasons

documented in chapter ten, section 10.4 (page 174), partial correlations controlled for the effects of demeaned alcohol and cannabis use, age, and sex. All correlations were corrected for multiple comparisons (*see Benjamini & Hochberg, 1995*).

11.3 Results

Mean daily EPA consumption was .01 g (SD = .02) and mean daily DHA consumption was .15 g (SD = .23). Both EPA and DHA data violated assumptions of skew (EPA [2.21, SE = .32], DHA [1.59, SE = .32]) and kurtosis (EPA [5.54, SE = .62], DHA [1.18, SE = .62]) and so underwent *In*-transformation prior to analysis (*In*-EPA [M = -6.29, SD = 2.63], *In*-DHA [M = -3.96, SD = 2.87]).

11.3.1 Behavioural Data

Neither EPA nor DHA correlated with stopping times or accuracy.

11.3.2 N170 and P200

Neither EPA nor DHA were correlated with left temporo-parietal N170 or P200 responses to fearful, sad, neutral, or angry facial expressions. Although EPA initially correlated with the N170 response to angry facial expressions (r = .32, p = .03, padj = .51), such that N170 amplitude increased alongside EPA intake, this finding did not survive multiple comparison testing.

11.3.3 Stop-N200 and Stop-P300

EPA intake shared moderately strong, positive correlations with both stop-N200 and stop-P300 responses to fearful (stop-N200 [r = .43, p < .05, padj = .05], stop-P300 [r = .49, p < .001, padj = .01]) and sad (stop-N200 [r = .32, p = .03, padj = .50], stop-P300 [r = .34, p = .02, padj = .38]) facial expressions. However only responses to fearful expression survived multiple comparison correction.



Figure 11. Partial correlations between central-midline stop-N200 and stop-P300 responses to sad and fearful facial expressions, and *In*-EPA (subplots A to D) and *In*-DHA intake (subplots E to H). Shaded areas denote 95% confidence interval areas along linear associations.

DHA intake was positively and moderately correlated with stop-N200 and stop-P300 responses to fearful (stop-N200 [r = .30, p = .04, padj = .72), stop-P300 [r = .33, p = .02, padj = .51)], but not sad (stop-N200 [r = .26, p = .07, padj = .87], stop-P300 [r = .21, p = .15, padj = .87]) facial expressions. However no correlations survived multiple comparison correction. Partial correlations describing EPA and DHA intake are evidenced in subplots A to D and E to H of Figure 11, respectively.

11.4 Discussion

Omega-3s EPA and DHA are considered important for healthy development and brain function (Carlson, 2001; Chen & Su, 2012; Conklin et al., 2007, de Velasco et al., 2012). The primary aim of this investigation was to delineate the associations between EPA/DHA consumption and the electrophysiological indices of face processing and distress-induced motor extinction. Results suggest that whilst omega-3 intake did not correlate with either N170 or P200 responses to faces, DHA and EPA were correlated with electrophysiological indices of the preparatory response preceding motor extinction and motor extinction efficiency, respectively. In light of limitations, results are discussed in the context of implications for cognitive and neurobiological models of distressinduced motor extinction, and future investigation.

The primary aim of this investigation was achieved through exploring four key hypotheses. First, because of its role in visual acuity and attention (Birch et al., 2010; Fedorova & Salem, 2006; Uauy et al., 2000), self-reported intake of omega-3 was hypothesised to correlate positively with electrophysiological indices of face processing. However, in this investigation neither N170 nor P200 responses to faces correlated with omega-3 intake. Whilst a single correlation suggested a relationship between EPA intake and the N170 response to angry faces, this did not survive correction for multiple comparisons. Although the second hypothesis proposed omega-3 intake would interact with face processing efficiency equally across facial expressions, the overall absence of

associations between omega-3 intake and N170/P200 responses was unexpected.

Findings presented here suggest that self-reported omega-3 intake over a sixmonth period does not correlate with ability to process structural and featural aspects of faces. In the absence of previous literature, it remains to be seen whether this effect is restricted to FAST-evoked ERPs, or whether similar results are elicited using other paradigms. As noted in chapter seven, section 7.3.3 (page 118), FAST-evoked N170/P200 responses might be atypical due to their susceptibility to instruction-led voluntary attention (Holmes et al., 2003).

In regards to the electrophysiological indices of motor extinction, DHA was hypothesised to inversely correlate with stop-N200 amplitude, indicative of disrupted synaptic reorganisation (Carlson, 2001; Sumich et al., 2012). The stop-N200 is thought to represent a negatively distributed, preparatory response preceding motor extinction (Hughes et al., 2012). As DHA did not associate with either stop-N200 or stop-P300 amplitudes, results did not support this hypothesis. Unexpectedly, *increased* EPA intake was associated with stop-N200 augmentation. Although this finding suggests EPA intake might be associated with poorer recognition of the need to extinguish a motor response, there is no supporting evidence to reason this association.

Whilst further replication of these results is required in order to assess their validity, it is possible these finding might represent an artefact of the FFQ. Disrupted synaptic reorganisation is thought to occur during child development (Crawford et al., 2003), yet the FFQ only assesses omega-3 intake for the sixmonths prior to questionnaire completion. As diet can differ throughout life, it is possible that the relationship between omega-3 consumption and stop-N200 amplitude might be better answered through an historical index of omega-3 consumption, or via a longitudinal assay.

As hypothesised, there was a positive correlation between EPA intake and stop-P300 responses to both fearful and sad facial stimuli. Similar associations were observed in regards to DHA, but these did not survive multiple comparison testing. Building on evidence derived from patients with ADHD indicating stop-P300 attenuation (Overtoom et al., 2002) and baseline EPA deficiency (Bekoroglu et al., 1996; Colquhoun & Bunday, 1981; Young et al., 2004, 2005), this is the first investigation to directly correlate omega-3 intake with the stop-P300. Supported by recent investigations suggesting a role of omega-3 in the fluidity of molecular information transmission (Michael-Titus & Priestley, 2014; Serfaty & de Velasco, 2014), the current results suggest that omega-3s might facilitate the mobilisation of neuronal responses resulting in motor extinction.

A clear limitation of this investigation is the lack of analysis between omega-3 intake and [1] the N170/P200 over the right hemisphere, [2] the stop-N200/stop-P300 over parietal sites, and [3] the background N200/P300 over both parietal and anterior sites. These ERPs were omitted from the analysis in order to focus statistical power on ERPs associated with physical aggression/CU traits within this thesis. However, with omega-3 found to associate with P300 responses (at trend-level significance) elsewhere (Gow, Sumich et al., 2013), there is scope for future investigation to fully delineate associations between omega-3 intake and face-evoked ERPs.

In summary, whilst neither self-reported EPA nor DHA related with electrophysiological indices of face processing, EPA, but not DHA, was found to differentially moderate electrophysiological indices of the recognition of inhibitory need and motor extinction efficiency. Discussed in the absence of supporting literature and methodological (i.e. omega-3 assay) limitations, these findings ground future investigation into the role of omega-3 as a treatment intervention for individuals with VIM deficits. However, current results prove this knowledge to be fragmentary, warranting further investigation.

Chapter Twelve: Conclusion

12.1 Introduction

This thesis aimed to [1] explore associations between physical aggression and CU traits in an adult community sample, [2] develop a sensitive experimental paradigm for characterising electrophysiological indices of different processing components of the VIM, [3] explore associations between physical aggression/CU traits and electrophysiological responses to face processing (i.e. N170, P200), and distress-induced motor extinction (i.e. stop-N200, stop–P300), [4] explore associations between omega-3 intake, physical aggression, and CU traits, and [5] explore associations between omega-3 intake and electrophysiological responses to face processing (i.e. N170, P200), and distress to face processing (i.e. N170, P200), and distress between omega-3 intake, physical aggression, and CU traits, and [5] explore associations between omega-3 intake and electrophysiological responses to face processing (i.e. N170, P200), and distress-induced motor extinction (i.e. stop-N200, stop–P300).

ERPs have been used to index face processing (N170 [Bentin et al., 1996; Rossion, 2014; Rossion & Jacques, 2011]; P200 [Itier & Taylor, 2004; Latinus & Taylor, 2006]) and motor extinction (stop-N200 [Albrecht et al., 2005; Huster et al., 2013]; stop-P300 [Bae et al., 2011; Bekker et al., 2005; Liotti et al., 2005]). However, to date, this knowledge has not been used to better the understanding of the VIM on an electrophysiological level. Whilst used in some motor extinction paradigms (Meier et al., 2012; Pawliczek et al., 2013), facial affect has not been used as a stopping stimulus, nor has its neural response been indexed via EEG. Prior to this thesis, both the electrophysiological response to distress-induced motor extinction and its relationships with personality traits associated with poor face processing and motor extinction were unknown. Furthermore, although omega-3 has been shown to benefit aggressive behaviour in forensic (Gesch et al., 2002; Zaalberg, 2010), clinical (Hamazaki & Hirayama, 2004; Stevens et al., 2003), and community (Long & Benton, 2013; Raine et al., 2015) samples, this knowledge was previously unexplored in relation to the VIM. This thesis serves as a necessary progression in resolving these gaps in knowledge.

The following sections draw together the key empirical findings of this thesis and show how they systematically combine in order to answer the research questions outlined above. Theoretical implications and the place of these findings within the current literature are discussed, before outlining how these results might advise future policy. Finally, future avenues of research are outlined in order to overcome wider limitations and extend knowledge presented within this thesis.

12.2 Empirical Findings and Theoretical Implication

Empirical findings are discussed within their respective chapters: "*Chapter Six: Correlations between physical aggression & callous-unemotional traits*", "*Chapter Seven: Baseline assessment of the Facial Affect Stop-go Task*", "*Chapter Eight: Correlations between physical aggression/callousnessunemotional traits and face-evoked ERPs*", "*Chapter Nine: Correlations between physical aggression/callous-unemotional traits and motor extinction ERPs*", "*Chapter Ten: Relationships between omega-3 intake, callousunemotional traits, and physical aggression*", and "*Chapter Eleven: Correlations between omega-3 intake and the electrophysiological indices of the Violence Inhibition Mechanism*". This section draws together key empirical findings and discusses [1] their relevance to answering the main research questions, and [2] their theoretical contributions.

12.2.1 Research Question One

The first research question asked how personality traits associated with poor face processing and motor extinction related to one another within an adult cohort. CU traits were broken down into callousness, uncaring, and unemotional traits, with physical aggression chosen as the key index of aggression due to its greater relevance for reactive aggression and physical harm (Buss & Perry, 1992). Other motivations for and manifestations of aggression were deferred for parsimonious value.

In contrast to previous investigations evidencing greater scores on both physical aggression (Bernstein & Gesn, 1997; Buss & Perry, 1992; Morales-Vives & Vigil-Colet, 2010; Reyna et al., 2011; Williams et al., 1996), and CU traits for males than females, findings of this investigation suggested that males only out-scored females on callous traits. As this finding does not compare to other samples comprising predominantly of students (García-León et al., 2002), it was explained by the low yield of male respondents and so requires further investigation. Furthermore, partial correlations, controlling for age and sex suggested that the relationship between aggression and CU traits might be driven by callousness in adults. This finding differs from those in adolescent samples, where aggression was observed to correlate with all three subscales of the ICU (Fanti et al., 2009; Feilhauer et al., 2012; Guelker et al., 2014; Kimonis et al., 2008). Whilst warranting future age-specific, group-contrasts, this finding has theoretical implication as to how uncaring and unemotional traits might manifest in adulthood and how they relate to aggressive traits and behaviours. It is possible that the ICU is not ideal for measuring these traits in an adult population.

12.2.2 Research Question Two

The second research question asked whether a FAST could be used to accurately index the VIM on an electrophysiological level, and whether associated ERP responses would be comparable to those evoked by typical face presentation and motor inhibition/extinction paradigms. Following three pilot investigations (*see chapter seven, sections 7.2.2 [page 105] and 7.2.3 [page 108] and 7.2.4 [page 112]*), a FAST was developed that succeeded in teasing apart the distinct stages of the VIM. Specifically, indices of structural face encoding (N170) and second-order face processing (P200), as well as motor preparation (stop-N200) and extinction efficiency (stop-P300). For the first time, FAST-evoked ERP responses were delineated.

In line with previous characterisations, electrophysiological indices of face processing were maximal at temporo-parietal sites (N170 [Hinojosa, et al.,

2015; Luck & Kappenman, 2013]; P200 [Farah, 1990; Watanabe, Kakigi, Koyama, & Kirino, 1999]), likely reflecting their neuronal origins in the OFA, STS, and fusiform gyrus (Adolphs, 2002; Itier & Taylor, 2004; Nummenmaa, Hyönä, et al., 2010; Rossion et al., 2003; Sadeh et al., 2010). However, in contrast to expectations, the N170 was augmented to angry facial expressions, and the P200 evidenced augmentation to neutral facial stimuli, but not those depicting facial affect. Previously, both the P200 (Ashley et al., 2004; Eimer et al., 2003; Schulz, 2012), and the N170 in some (Ibáñez et al., 2012; Hinojosa et al., 2015; Smith et al., 2003), but not all (Ashley et al., 2004; Eimer, 2000; Eimer & Holmes, 2002) investigations have evidenced augmentation in response to facial affect.

Findings of this investigation were explained by task instructions orientating the attention of participants to facial affect. Specifically, participants were tasked with evoking a motor response to angry facial expressions (go stimuli) and extinguishing said response to fearful or sad facial expressions (stop stimuli). The observation of anger-specific N170 augmentation is likely an artefact of angry, as opposed to neutral, fearful, or sad facial expressions being target stimuli, and so might reflect a readiness to respond to anger. Previously, affectdriven N170 responses have been observed in tasks where participants were required to attend to the affective elements of facial stimuli (Holmes et al., 2003; Ibánez et al., 2012). It remains to be seen if comparative findings would be observed using other facial expressions as go stimuli. Similarly, the neutralspecific P200 augmentation might reflect participants recruiting greater neuronal resources in an attempt to resolve the ambiguity of neutrality when tasked with attending/responding to facial affect. P200 augmentation to neutral facial expressions has been observed once before in participants tasked with assigning reward, based on facial affect (Tortosa et al., 2013).

In regards to N200 and P300 responses, in line with previous nogo and stop investigations using non-affect stimuli, both the N200 (Enriquez-Geppert et al., 2010; Huster et al., 2013; Liu et al., 2015; Patel & Azzam, 2005) and P300 (De

Jong et al., 1990; Dimoska et al., 2006; Kok et al., 2004) were larger over anterior than parietal sites in response to stop facial-stimuli. Furthermore, whilst neither stop-N200 nor stop-P300 responses varied as a function of emotion, the composite P300 (recorded over anterior-midline sites) was larger in response to fearful than sad facial distress. In accordance with the VIM (Blair, 1995, 2000), this finding likely reflects a predominant effect of the fear due to its biological and social importance (Blair, 2000; Blair et al., 2004; Dadds et al., 2008; Fairchild et al., 2009).

Taken together, the FAST appears a useful tool for indexing the VIM on an electrophysiological level and provides research utility that is likely to remain current as it forms a benchmark for future experimentation and manipulation. Previously, simple face presentation paradigms have facilitated investigation of the N170 and P200 to neutral and affective facial expressions. However, facial expressions, particularly those depicting distress have been overlooked in motor inhibition and motor extinction paradigms. Development of the FAST contributes experimental knowledge to the motor extinction literature and provides a means of measuring electrophysiological responses during distress-induced stopping.

12.2.3 Research Question Three

The third research question asked how physical aggression and CU traits moderated electrophysiological indices of face processing (N170, P200) and distress-induced motor extinction (stop-N200, stop-P300). Building on knowledge obtained in chapters six (from page 90) and seven (from page 102), results suggested that whilst uncaring, but not physical aggression, callousness, or unemotional traits, was associated with left hemispheric N170 (attenuation) and P200 (augmentation) responses during face processing, only physical aggression was associated with stop-N200 (augmentation) and stop-P300 (attenuation) responses during motor extinction.

The inverse association between scores on the uncaring scale of the ICU and the N170 is supported by previous investigation in community samples. Here, structural face processing, as indexed by the N170, was associated with poor emotional expression (Meaux et al., 2014) and fearless dominance (Almeida et al., 2014). In particular, fearless dominance is strongly related to the uncaring subscale of the ICU (Kimonis et al., 2013). Furthermore, in contrast to expectations, after controlling for N170 amplitude, positive correlations were observed between uncaring traits and the P200 response. This finding was explained as reflecting a compensatory mechanism for socially functioning individuals presenting uncaring traits. However, it remains to be seen whether P200 augmentation is abolished in samples presenting higher levels of uncaring traits. The absence of an association between N170/P200 amplitude and physical aggression is also supported by the meta-analysis reported in Dawel et al. (2012) where face processing was associated with affective, but not antisocial components of psychopathy. This is the first investigation to associate electrophysiological indices of face processing with subscales of the ICU and physical aggression.

In regards to electrophysiological indices of motor extinction, the positive correlation between physical aggression and the stop-N200 was thought to reflect increased neuronal activity during motor extinction preparation. In a similar manner to the P200, noted above, as nogo-N200 attenuation has been observed in forensic samples characterised by extreme aggression, it remains to be seen whether comparative stop-N200 attenuation, in response to facial distress, is elicited in samples presenting higher levels of aggressive traits. Moreover, the inverse association between physical aggression and the stop-P300 both supports prior investigations reporting P300 attenuation as a function of aggression (Chen et al., 2005; Costa et al., 2000; Vilà-Balló et al., 2014) and extends these findings into the context of facial-distress stop stimuli.

12.2.4 Research Questions Four and Five

Using the findings from research questions one to three as a benchmark, research questions four and five asked how physical aggression, CU traits, and FAST-evoked ERPs might relate to dietary intake of omega-3. With a growing interest into the benefit of omega-3 for aggressive behaviour (Buydens-Branchey & Branchey, 2008), and recent support for the reduction of aggressive/CU traits as a function of omega-3 supplementation (Raine et al., 2015), answering these research questions has important theoretical and practical utility.

Neither self-reported EPA nor DHA intake correlated with any subscale of the ICU. Previously, CU traits have been shown to inversely correlate with EPA blood content in children with ADHD (Gow, Vallee-Tourangeau et al., 2013). Whilst this disparity might reflect choice of omega-3 assay (i.e. blood content vs. self-report), it is possible that the null finding in the current investigation is explained by statistical analysis controlling for physical aggression. Additionally, physical aggression was negatively correlated with EPA, but not DHA intake over a six-month period. Previously, EPA deficiency has been related to increased aggression in animal (DeMar et al., 2006; Raygada et al., 1998; Re et al., 2008) and human samples (Buydens-Branchey et al., 2006; Umhau et al., 2006), with supplementation of EPA shown to benefit physical/verbal aggression (Zanarini & Frankenburg, 2003), anger (Buydens-Branchey & Branchey, 2008), and violent incidents (Gesch et al., 2002). Although not supplemented by biological measures, these findings have theoretical implications for [1] the use of the FFQ as a quick and inexpensive measurement of omega-3 intake and [2] the importance of disentangling CU and aggressive traits.

On an electrophysiological level, although not correlated with N170 or P200 components, self-reported EPA and DHA intake was positively correlated with both stop-N200 and stop-P300 responses. Combined with evidence presented in chapters ten (from page 169) and nine (from page 159), which delineated

inverse associations between physical aggression and omega-3 intake and electrophysiological indices of motor extinction efficiency, respectively, these latter findings were expected. Understanding how omega-3 might moderate personality traits associated with dysfunctional VIM responses and associated ERPs at baseline provides a theoretical foothold, and so warrants further investigation into the practical utility of omega-3 supplementation on these indices.

12.3 Scientific Implications

Results documented within this thesis pose several scientific implications. First, though contradictory to evidence derived from adolescent samples (Fanti et al., 2009; Feilhauer et al., 2012; Guelker et al., 2014; Kimonis et al., 2008), physical aggressive traits in adults were solely associated with callous, but not uncaring or unemotional traits using comparable measures. Though potentially indicating poor utility of the ICU as a tool for distinguishing CU traits in adult samples, it cannot be ruled out that uncaring and unemotional traits manifest differently in adulthood. With such knowledge seemingly overlooked in the current literature, results reported here warrant further investigation.

Second, although individuals with psychopathic traits exhibit notable dysfunction in activating the VIM (Blair, 1995, 2001; Blair et al., 2004), there is importance, at least on an electrophysiological level in adults drawn from the community, to distinguish between CU and aggressive traits when investigating the different processing stages of the VIM. Such findings build on previous suggestions of trait specificity when investigating face processing proficiency (Dawel et al., 2012). Specifically, results documented within this thesis evidence associations between uncaring traits and a deficient ability to process both structural and configurable aspects of faces, and physically aggressive traits and deficient facial distress-induced motor extinction. In the context of the VIM, whilst CU traits might relate to vulnerability for poor responses to distressrelated social cues, and a possible failure to trigger the VIM (Blair, 2005), aggressive traits might ultimately associate with deficits in distress-induced motor extinction.

Third, this thesis implicates the use of the FFQ as both a time- and cost-efficient means of assessing omega-3 intake. Though limited to an historical assessment of omega-3 intake over a six-month period, results of this thesis suggest specific associations between omega-3 intake and both decreased self-reported physical aggression and increased efficiency of distress-induced motor extinction – a trait/ERP association identified earlier in this thesis. Relationships between omega-3 intake and CU traits/face processing efficacy were not observed. Such findings have importance for understanding the mechanisms relating omega-3 intake to psychopathy-related traits and distinct processing stages of the VIM. Furthermore, findings supplement the growing interest into how omega-3 might be used to benefit and/or alleviate psychopathy-related traits.

12.4 Recommendation for Future Research and Limitations

Although specific experiment-by-experiment limitations have been discussed throughout this thesis, it is important to consider the experimental results in light of wider limitations and future avenues of research. This section discusses *'sample'*, *'diet'*, and *'statistical'* limitations.

12.4.1 Sample

Experiments reported in this thesis predominantly sampled from the student population. Whilst important to understand the effect of physical aggression and CU traits on distress-induced motor extinction, student-biased samples might limit the spectrum of psychometric responses. As documented in chapters six (from page 90) and ten (from page 169), although psychometric measures were normally distributed, they were centred on a low mean, and so would benefit from an increased number of data points across the response spectrum. Such sampling methods would have additional value for facilitating the ability to contrast high- and low- scoring subgroups in order to negate the limitations of correlational analysis (i.e. the inability to infer causation). In this thesis, data predominantly underwent partial correlational analysis. Moreover, collapsing individuals into subgroups would reduce between-subject variation - further benefiting group comparisons (Luck, 2005).

Furthermore, sex was not equally distributed within the samples and so, in the most part, sex-based comparisons could not reliably be observed. Although bivariate correlations as a function of sex were observed in chapter six (from page 90), this decision was made because [1] it was an initial baseline investigation, [2] the overall sample size was sufficiently large enough to warrant this comparison, and [3] unlike other investigations in this thesis, this particular investigation was not saturated in terms of multiple comparisons. However, in acknowledgment of the importance of sex differences in the manifestation of aggression and CU traits, where possible, investigations within this thesis sought to control for the variation in sex. Clearly, it is important that future investigations into this area acknowledge sex differences.

12.4.2 Reported Diet

This thesis is also limited to the assumptions formed around diet. First, associations between omega-3 intake and psychometric/electrophysiological measures are formed from self-reports that index omega-3 over a six-month period. Whereas blood assays of omega-3 have been associated with aggressive (Meyer et al., 2015) and CU (Gow, Vallee-Tourangeau et al., 2013) traits, which might vary over time (Barratt, 1991; Raine, 2013), it is difficult to accurately compare current levels of omega-3 using questionnaire methods. Furthermore, as reported in chapter four (from page 63), omega-3 plays a vital role in brain maturation during embryonic growth and early years of life (Carlson, 2001; Crawford et al., 2003; Riemer et al., 2010). It is possible that associations between omega-3 intake and electrophysiological indices of face processing, not observed in chapter eleven (from page 177), might be better indexed through historical/longitudinal assessments.

Second, the FFQ has not previously been validated in non-major depressive populations (Sublette et al., 2011), or as a function of cognitive ability and other social/developmental variables, and so might not be stable in samples of young adults derived from the community. In light of this limitation, although a quick and cost-effective assay of omega-3 intake, which has been justified in chapter five, section 5.3.3 (page 87), blood assays would likely provide a more reliable and robust measure of omega-3. However, it is important to note that the collection of blood is time consuming and expensive to analyse, especially in the context of investigations requiring large sample sizes.

12.4.3 Statistical Considerations

Finally, it is important to contextualise the results of this thesis in light of statistical considerations. First, as documented in chapters ten (from page 169) and eleven (from page 177), EPA and DHA consumption were not normally distributed. Although mediated through logarithmic transformation, skewed distributions suggest homogeneous fish consumption. Whilst homogeneity might ultimately prevent the accurate delineation of subtle relationships, the survival of an observed inverse association between EPA intake and physical aggression suggests enough variation within the data was present to capture this association. However, in order to more accurately explore associations between omega-3 intake and psychometric traits/ERP responses, future investigation would benefit from sampling individuals with a wider variety of fish consumption.

Second, as mentioned in section 12.4.1 (page 193), conclusions of this thesis are predominantly drawn form correlational analyses, which prevent the inference of causation. In relation to dietary intake of omega-3, specifically, group analysis has benefit for understanding differences in VIM-related personality traits and electrophysiological responses between high and low omega-3 consumers. Furthermore, experimental manipulation in the form of omega-3 supplementation would provide an effective measure of the direct

impact of omega-3 on both psychometric measures of physical aggression/CU traits and FAST-evoked ERPs. With omega-3 supplementation previously shown to reduce parent-, teacher-, and self-reported aggression, as well as incidents of aggressive behaviour (Gesch et al., 2002; Hamazaki & Hirayama, 2004; Stevens et al., 2003; Zaalberg et al., 2010), omega-3/ERP associations observed in this thesis suggest that supplementation of omega-3 would likely augment stop-N200 and stop-P300 responses to facial distress.

Third, although this thesis has focused on trait scores of physical aggression as a function of its relationship to reactive aggression and physical harm (Buss & Perry, 1992), this decision partly reflects consideration of statistical power. Future investigations should seek to investigate other manifestations of aggression (e.g. verbal and indirect aggression) and associated traits (e.g. anger and hostility) in a similar manner, or in contrast to one another using a larger sample. As aggression can be characterised in a multitude of ways (Barratt et al., 1997, *but see chapter one for review*), it is important to understand how such characterisations might differentially relate to CU traits, omega-3 intake, and electrophysiological indices of the VIM.

12.5 Policy Implication

Whilst expanding theoretical and scientific knowledge, this thesis has additional relevance for policy consideration. Currently, in England and Wales, each of the forty-two criminal justice areas are required to establish a Multi-Agency Public Protection Arrangement (MAPPA) in order to better protect the public from violent offenders (Criminal Justice Act 2003, s. 325). Specifically, whilst violent behaviour is managed, and so reduced, as a function of social, financial, and educational teams working alongside criminal justice agencies, distinct contributions from food agencies are absent (MAPPA Guidelines 2012, s. 3.7). Though correlational in nature, findings reported in this thesis, alongside others reviewed in chapter four, section 4.3.2 [*from page 68*]) suggest a potential benefit of omega-3 for scores on psychometric measures of physical aggression and aggressive-related behaviours.

Within forensic settings, whilst diets are required to be healthy, balanced, low in saturated fat, sugar, and salt, and of sufficient fruit and fibre (Catering: Meals for Prisoners [England] Prison Service Instruction 2010 No. 44) they place only little importance on omega-3 consumption. Currently, the Catering: Meals for Prisoners instruction only requires two portions of fish, with only one comprising omega-3-rich oily fish, to be served per week. Disregarding other dietary sources of omega-3 (e.g. seeds, grains, or leafy vegetables, *see chapter four, section 4.3.1 [page 64]*) or menu fluctuations across institutions and seasons, this level of omega-3 consumption is far lower than the daily recommended 0.25 to 3 g (Food and Agriculture Organisation, 2010).

Taken together, these results suggest possible practical implication for the use of omega-3 as a preventative or treatment measure for physical aggression, but perhaps not CU traits. However, such utility requires more rigorous testing, over additional assays of aggression, and biologically robust measures of omega-3. Although omega-3 intake has recently been explored as having important public health ramifications (Beier et al., 2014), since this and other investigations only support a reduction, and not an abolition of aggression, it is important that future investigation of the benefit of omega-3 considers it alongside wider treatment strategies.

12.6 Thesis Conclusion

In spite of the wealth of investigation outlining traits associated with atypical activation of the VIM - and their benefit from an omega-3-rich diet - empirical research has overlooked the investigation of these associations on an electrophysiological level. The VIM comprises distinct stages of face processing and distress-induced motor extinction. For the first time, this thesis provides a means to tease apart the various stages of the VIM using EEG. Key findings suggest that whilst uncaring traits associate with electrophysiological indices of structural/featural face processing, physical aggression associates with electrophysiological indices of distress-induced motor extinction. Moreover, in light of a growing interest in using omega-3 as a moderator of aggressive and

CU traits, self-reported omega-3 intake was found to correlate with physical aggression/distress-induced motor extinction, but not CU traits/face processing.

Together, these results signify the importance for not only distinguishing between aggressive and CU traits, but for discussing VIM dysfunction in terms of distinct processing stages. Discussed in the context of its limitations and wider application to both theoretical knowledge and policy, this thesis provides a strong foothold for future experimentation in order to better establish its position within the psychological and forensic literature.
References

- Adamson, S., & Sellman, J. D. (2003). A prototype screening instrument for cannabis use disorder: The Cannabis Use Disorders Identification Test (CUDIT) in an alcohol dependent clinical sample. *Drug and Alcohol Review*, *22*, 309-331.
- Adolphs, R. (2001). The neurobiology of social cognition. *Current Opinion in Neurobiology, 11,* 231-239.
- Adolphs, R. (2002). Neural systems for recognizing emotion. *Current Opinion in Neurobiology, 12,* 169-177.
- Adolphs, R. (2010). What does the amygdala contribute to social cognition? Annals of the New York Academy of Sciences, 1191, 42-61.
- Adolphs, R., Damasio, H., & Tranel, D. (2002). Neural Systems for Recognition of Emotional Prosody: A 3-D Lesion Study. *Emotion, 2,* 23-51.
- Adolphs, R., Damasio, H., Tranel, D., & Damasio, A. R. (1996). Cortical systems for the recognition of emotion in facial expressions. *Journal of Neuroscience*, *16*, 7678-7687.
- Adolphs, R., & Tranel, D. (2003). Amygdala damage impairs emotion recognition from scenes only when they contain facial expressions. *Neuropsychologia, 41,* 1281-1289.
- Adolphs, R., Tranel, D., Damasio, H., & Damasio, A. R. (1994). Impaired recognition of emotion in facial expressions following bilateral damage to the human amygdala. *Nature, 372,* 669-672.
- Adolphs, R., Tranel, D., Damasio, H., & Damasio, A. R. (1995). Fear and the human amygdala. *The Journal of Neuroscience*, *15*, 5879-5892.
- Adolphs, R., Tranel, D., Young, A. W., Calder, A. J., Phelps, E. A., Anderson, A. ... Damasio, A. R. (1999). Recognition of facial emotion in nine individuals with bilateral amygdala damage. *Neuropsychologia*, *37*, 1111-1117.
- Ahmad, A., Moriguchi, T., & Salem, N. (2002). Decrease in neuron size in docosahexaenoic acid-deficient brain. *Paediatric Neurology, 26*, 210-218.

- Aid, S., Vancassel, S., Poumes-Ballihaut, C., Chalon, S., Guest, P., & Lavialle, M. (2003). Effect of a diet-induced n-3 PUFA depletion on cholinergic parameters in the rat hippocampus. *Journal of Lipid Research, 44*, 1545-1551.
- Ailhaud, G., Massiera, F., Weill, P., Legrand, P., Alessandri, J. M., & Guesnet,
 P. (2006). Temporal changes in dietary fats: role of n-6 polyunsaturated
 fatty acids in excessive adipose tissue development and relationship to
 obesity. *Progress in Lipid Research*, 45, 203-236.
- Albrecht, B., Banaschewski, T., Brandeis, D., Heinrich, H., & Rothenberger, A. (2005). Response inhibition deficits in externalizing child psychiatric disorders: An ERP-study with the Stop-task. *Behavioral and Brain Functions*, *9*, 1-22.
- Alderman, N., Knight, C., & Morgan, C. (1997). Use of a modified version of the Overt Aggression Scale in the measurement and assessment of aggressive behaviors following brain injury. *Brain Injury*, *11*(7), 503-523.
- Alegre, M., Alvarez-Gerriko, I., Valencia, M., Iriarte, J., & Artieda, J. (2008).
 Oscillatory changes related to the forced termination of a movement.
 Clinical Neurophysiology, *119*, 290-300.
- Allen-Walker, L., & Beaton, A. A. (2015). Empathy and perception of emotion in eyes from the FEEST/Ekman and Friesen faces. *Personality and Individual Differences, 72*, 150-154.
- Alink, L. R., van Ijzendoorn, M. H., Bakermans-Kranenburg, M. J., Mesman, J., Juffer, F., & Koot, H. M. (2008). Cortisol and externalizing behavior in children and adolescents: mixed meta-analytic evidence for the inverse relation of basal cortisol and cortisol reactivity with externalizing behavior. *Developmental Psychobiology*, *50*, 427-450.
- Allison, T., Puce, A., & McCarthy, G. (2000). Social perception from visual cues: role of the STS region. *Trends in Cognitive Science*, *4*(7), 267-278.
- Allison, T., Puce, A., Spencer, D. D., & McCarthy, G. (1999).
 Electrophysiological studies of human face perception. I: Potentials generated in occipitotemporal cortex by face and non-face stimuli. *Cerebral Cortex*, *9*, 415-430.

- Almeida, P. R., Ferreira-Santos, F., Vieira, J. B., Moreira, P. S., Barbosa, F., & Marques-Teixeira, J. (2014). Dissociable effects of psychopathic traits on cortical and subcortical visual pathways during facial emotion processing: An ERP study on the N170. *Psychophysiology*, *51*, 645-657.
- Amaral, D. G., Price, J. L., Pitkanen, A., & Carmichael, S. T. (1992). Anatomical organization of the primate amygdaloid complex. In J. P. Aggleton (Ed.), *The amygdala: Neurobiological aspects of emotion, memory, and mental dysfunction* (pp. 1-66). New York, NY: Wiley-Liss.
- American Psychiatric Association. (2000). *Diagnostic and Statistical Manual of Mental Disorders* (4th ed.). Washington, DC: American Psychiatric Association.
- American Psychiatric Association. (2013). *Diagnostic and statistical manual of mental disorders* (5th ed.). Arlington, VA: American Psychiatric Publishing.
- Amodio, D. M., Master, S. L., Yee, C. M., & Taylor, S. E. (2008).
 Neurocognitive components of behavioral inhibition and activation systems: Implications for theories of self-regulation. *Psychophysiology*, 45, 11-19.
- Anastassiou-Hadjicharalambous, X., & Warden, D. (2008). Can children's heart rate be used as a marker of differential responsiveness to others' varying emotional states? *International Journal of Psychology*, 43(3-4), 422-422.
- Anckarsater, H. (2006). Central nervous changes in social dysfunction: Autism, aggression, and psychopathy. *Brain Research Bulletin, 69,* 259-265.
- Anderson, S. W., Bechara, A., Damasio, H., Tranel, D., & Damasio, A. R.
 (1999). Impairment of social and moral behavior related to early
 damage in human prefrontal cortex. *Nature Neuroscience, 2,* 469-479.
- Anderson, C. A., & Bushman, B. J. (2002). Human aggression. *Annual Review* of *Psychology*, *53*, 27-51.

- Anderson, C. A., & Carnagey, N. L. (2004). Violent evil and the general aggression model. In A. Miller (Ed.), *The social psychology of good and evil* (pp. 168-192). New York, NY: Guilford Publications.
- Anderson, A. K., Christoff, K., Panitz, D., de Rosa, E., & Gabrieli, J. D. E. (2003). Neural correlates of the automatic processing of threat facial signals. *Journal of Neuroscience*, *23*(13), 5627-5633.
- Anderson, C. A., & Huesmann, L. R. (2003). Human aggression: A socialcognitive view. In M. A. Hogg, & J. Cooper (Eds.), *Handbook of social psychology* (pp. 296-323). London, UK: Sage.
- Anderson, N. E., & Kiehl, K. A. (2012). The psychopath magnetized: insights from brain imaging. *Trends in Cognitive Sciences, 16*(1), 52-60.
- Anderson, B. M., & Ma, D. W. (2009). Are all n-3 polyunsaturated fatty acids created equal? *Lipids Health Discussion, 8*, 33.
- Anderson, G. J., Neuringer, M., Lin, D. S., & Connor, W. E. (2005). Can prenatal N-3 fatty acid deficiency be completely reversed after birth?
 Effects on retinal and brain biochemistry and visual function in rhesus monkeys. *Paediatric Research, 58,* 865-872.
- Antonucci, A. S., Gansler, D. A., Tan, S., Bhadelia, R., Patz, S., & Fulwiler, C.
 (2006). Orbitofrontal correlates of aggression and impulsivity in psychiatric patients. *Psychiatry Research*, *147*, 213-220.
- Apicella, F., Sicca, F., Federico, R. R., Campatelli, G., & Muratori, F. (2013).
 Fusiform gyrus responses to neutral and emotional faces in children with autism spectrum disorders: a high density ERP study. *Behavioral Brain Research, 251,* 155-162.
- Apostolou, M. (2013). The evolution of rape: The fitness benefits and costs of a forced-sex mating strategy in an evolutionary context. *Aggression and Violent Behavior, 18*(5), 484-490.
- Appleton, K. M., Woodside, J. V., Yarnell, J. W. G., Arveiler, D., Haas, B., Amouyel, P., ... Evans, A. (2007). Depressed mood and dietary fish intake: Direct relationship or indirect relationship as a result of diet and lifestyle? *Journal of Affective Disorders, 104*(1-3), 217-223.

- Arab-Tehrany, E., Jacquot, M., Caiani, C., Imran, M., Desobry, S., & Linder, M. (2012). Beneficial effects of oxidative stability of omega-3 long-chain polyunsaturated fatty acids. *Trends in Food Science & Technology, 25,* 24-33.
- Arbuckle, L. D., MacKinnon, M. J., & Innis, S. M. (1994). Formula 18:2(n-6) and 18:3(n-3) content and ratio influence long-chain polyunsaturated fatty acids in the developing piglet liver and central nervous system. *Journal* of Nutrition, 124, 289-298.
- Archer, J. (2001). A strategic approach to aggression. *Social Development, 10*(2), 267-271.
- Aron, A. R. (2007). The neural basis of inhibition in cognitive control. The *Neuroscientist, 13*, 214-228.
- Aron, A. R. (2011). From reactive to proactive and selective control: developing a richer model for stopping inappropriate responses. *Biological Psychiatry, 69,* 55-68.
- Aron, A. R., Behrens, T. E., Smith, S., Frank, M. J., & Poldrack, R. A. (2007). Triangulating a cognitive control network using diffusion-weighted magnetic resonance imaging (MRI) and functional MRI. *Journal of Neuroscience, 27*, 3743-3752.
- Aron, A. R., Fletcher, P. C., Bullmore, E. T., Sahakian, B. J., & Robbins, T. W. (2003). Stop-signal inhibition disrupted by damage to right inferior frontal gyrus in humans. *Nature Neuroscience*, *6*, 115-116.
- Aron, A. R., & Poldrack, R. A. (2006). Cortical and subcortical contributions to stop signal response inhibition: role of the subthalamic nucleus. *Journal* of Neuroscience, 26, 2424-2433.
- Aron, A. R., Robbins, T. W., & Poldrack, R. A. (2004). Inhibition and the right inferior frontal cortex. *Trends in Cognitive Science*, *8*, 170-177.
- Ashley, V., Vuilleumier, P., & Swick, D. (2004). Time course and specificity of event-related potentials to emotional expressions. *NeuroReport*, 15, 211-216.

- Assary, E., Salekin, R. T., & Barker, E. D. (2014). Big-Five and Callousunemotional Traits in Preschoolers. *Journal of Psychopathology and Behavioral Assessment, 36*(4), 1-9.
- Atkinson, A. P., Heberlein, A. S., & Adolphs, R. (2007). Spared ability to recognise fear from static and moving whole-body cues following bilateral amygdala damage. *Neuropsychologia*, 45(12), 2772-2782.
- Auestad, N., Scott, D. T., Janowsky, J. S., Jacobsen, C., Carroll, R. E., Montalto, M. B., ... Hall, R. T. (2003). Visual, cognitive, and language assessments at 39 months: a follow-up study of children fed formulas containing long-chain polyunsaturated fatty acids to 1 year of age. *Pediatrics, 112*(1), 177-183.
- Azar, R., Zoccolillo, M., Paquette, D., Quiros, E., Baltzer, F., & Tremblay, R. E. (2004). Cortisol levels and conduct disorder in adolescent mothers. Journal of the American Academy of Child and Adolescent Psychiatry, 43, 461-468.
- Babor, T. F., Higgins-Biddle, J. C., Saunders, J. B., & Monteiro, M. G. (2006).
 AUDIT: The Alcohol Use Disorders Identification Test: guidelines for use in primary care (2nd Ed). Geneva: World Health
 Organization: Department of Mental Health and Substance
 Dependence.
- Bae, K. Y., Kim, D. W., Im, C. H., & Lee, S. H. (2011). Source imaging of P300 auditory evoked potentials and clinical correlations in patients with posttraumatic stress disorder. *Progress in Neuro-Psychopharmacology* and Biological Psychiatry, 35(8), 1908-1917.
- Balconi, M., & Lucchiari, C. (2005). Event-related potentials related to normal and morphed emotional faces. *The Journal of Psychology*, *139*(2), 176-192.
- Band, G. P. H., & van Boxtel, G. J. M. (1999). Inhibitory motor control in stop paradigms: review and reinterpretation of neural mechanisms. *Acta Psychologia*, 101, 179-211.
- Barkley, R. (1998). Attention-deficit hyperactivity disorder: A handbook for diagnosis and treatments. New York, NY: Guildford.

- Baron, R. B., & Richardson, D. (1994). *Human aggression*. New York, NY: Plenum.
- Baron, K. G., Smith, T. W., Butner, J., Nealey-Moore, J., Hawkins, M. W., & Uchino, B. N. (2007). Hostility, anger, and marital adjustment:
 Concurrent and prospective associations with psychosocial vulnerability. *Journal of Behavioral Medicine*, *30*, 1-10.
- Barratt, E. S., Stanford, M. S., Kent, T. A., & Felthous, A. (1997).
 Neuropsychological and cognitive psychophysiological substrates of impulsive aggression. *Biological Psychiatry*, *41*, 1045-1061.
- Barry, R. J., Clarke, A. R., McCarthy, R., Selikowitz, M., Brown, C. R., & Heaven, P. C. L. (2009). Event-related potentials in adults with Attention-Deficit/Hyperactivity Disorder: an investigation using an intermodal auditory/visual oddball task. *International Journal of Psychophysiology*, *71*, 124-131.
- Barry, C. T., Frick, P. J., DeShazo, T. M., McCoy, M. G., Ellis, M., & Loney, B.
 R. (2000). The importance of callous—unemotional traits for extending the concept of psychopathy to children. *Journal of Abnormal Psychology*, *109*, 335-340.
- Barry, T. D., Thompson, A., Barry, C. T., Lochman, J. E., Adler, K., & Hill, K. (2007). The importance of narcissism in predicting proactive and reactive aggression in moderately to highly aggressive children. *Aggressive Behavior, 33*(3), 185-197.
- Bartholow, B. D., Bushman, B. J., & Sestir, M. A. (2006). Chronic violent video game exposure and desensitization to violence: Behavioral and eventrelated brain potential data. *Journal of Experimental Social Psychology*, 42, 532-539.
- Bassarath, L. (2001). Neuroimaging studies of antisocial behavior. *Canadian Journal of Psychiatry, 46,* 728-732.
- Batson, C. D. (1991). *The altruism question: Toward a social-psychological answer*. Hillsdale, NJ: Erlbaum Associates.

- Batson, C. D., Fultz, J., & Schoenrade, P. A. (1987). Distress and empathy:
 Two qualitatively distinct vicarious emotions with different motivational consequences. *Journal of Personality*, *55*, 19-39.
- Batty, M., & Taylor, M. J. (2003). Early processing of the six basic facial emotional expressions. *Cognitive Brain Research*, *17*, 613-620.
- Bauer, L. O. (1997). Frontal P300 decrements, childhood conduct disorder, family history, and the prediction of relapse among abstinent cocaine abusers. *Drug and Alcohol Dependence, 44,* 1-10.
- Bauer, L. O., O'Connor, S., & Hesselbrock, V. M. (1994). Frontal P300 decrements in antisocial personality disorder. *Alcoholism: Clinical and Experimental Research, 18,* 1300-1305.
- Bauer, L. O., & Hesselbrock, V.M. (1999). P300 decrements in teenagers with conduct problems: implications for substance abuse risk and brain development. *Biological Psychiatry*, 46, 264-272.
- Bauer, L. O., & Hesselbrock, V. M. (2003). Brain maturation and subtypes of conduct disorder: Interactive effects of P300 amplitude and topography in male adolescents. *Journal of the American Academy of Child and Adolescent Psychiatry*, 42, 106-115.
- Bauer, I., Hughes, M., Rowsell, R., Cockerell, R., Pipingas, A., Crewther, S., & Crewther, D. (2014). Omega-3 supplementation improves cognition and modifies brain activation in young adults. *Human Psychopharmacology*, 29(2), 133-144.
- Baumeister, S., Hohnmann, S., Wolf, I., Plichta, M. M., Rechtsteiner, S., Zangl,
 M., ... Brandeis, D. (2014). Sequential inhibitory control processes
 assessed through simultaneous EEG-fMRI. *NeuroImage, 94*, 349-359.
- Baxter, M. H., & Murray, E. A. (2002). The amygdala and reward. *Nature Reviews Neuroscience, 3,* 563-573.
- Bazan, N. G. (2005). Lipid signalling in neural plasticity, brain repair, and neuroprotection. *Molecular Neurobiology, 32*, 89-103.
- Bechara, A., Damasio, H., & Damasio, A. R. (2000). Emotion, decision-making, and the orbitofrontal cortex. *Cerebral Cortex, 10,* 295-307.

- Begleiter, H., Porjesz, B., & Garozzo, R. (1979). Visual evoked potentials and affective rating of semantic stimuli. In E. H. Begleiter (Ed.), *Evoked brain potentials and behavior* (pp. 127-141). New York, NY: Plenum Press.
- Beier, A. M., Lauritzen, L., Galfalvy, H. C., Cooper. T. B., Oquendo, M. A.,
 Grunebaum, M. F., ... Sublette, M. E. (2014). Low plasma
 eicosapentaenoic acid levels are associated with elevated trait
 aggression and impulsivity in major depressive disorder with a history of
 comorbid substance use disorder. *Journal of Psychiatric Research, 57*, 133-140.
- Bekaroğlu, M., Aslan, Y., Gedik, Y., Değer, O., Mocan, H., Erduran, E., &
 Karahan, C. (1996). Relationships between serum free fatty acids and
 zinc, and attention-deficit hyperactivity disorder: A research note. *Journal of Child Psychology and Psychiatry*, *37*, 225-227.
- Bekker, E. M., Kenemans, J. L., Hoeksma, M. R., Talsma, D., & Verbaten, M. N. (2005). The pure electrophysiology of stopping. *International Journal of Psychophysiology*, 55, 191-198.
- Bekker, E. M., Kenemans, J. L., & Verbaten, M. N. (2004). Electrophysiological correlates of attention, inhibition, sensitivity and bias in a continuous performance task. *Clinical Neurophysiology*, *115*, 2001-2013.
- Bekker, E. M., Kenemans, J. L., & Verbaten, M. N. (2005). Source analysis of the N2 in a cued go/nogo task. *Cognitive Brain Research, 22*, 221-231.
- Bellmore, A. D., Witkow, M. R., Graham, S., & Juvonen, J. (2005). From beliefs to behavior: The mediating role of hostile response selection in predicting aggression. *Aggressive Behavior*, *33*, 453-472.
- Bendig, A. W. (1962). Factor analytic scales of covert and overt hostility. *Journal of Consulting Psychology, 26*, 200.
- Benjamini, Y., & Hochberg, Y. (1995). Controlling the False Discovery Rate: A practical and powerful approach to multiple testing. *Journal of the Royal Statistical Society B, 57*(1), 289-300.

- Bentin, S., Allison, T., Puce, A., Perez, E., & McCarthy, G. (1996).
 Electrophysiological studies of face perception in humans. *Journal of Cognitive Neuroscience*, *8*, 551-565.
- Benton, D. (2007). The impact of diet on antisocial, violent, and criminal behaviour. *Neuroscience and Biobehavioral Reviews, 31,* 752-774.
- Benton, D., Kumari, N., & Brain, P. F. (1982). Mild hypoglycaemia and questionnaire measures of aggression. *Biological Psychology*, 14, 129-135.
- Berardi, N., Pizzorusso, T., & Maffei, L. (2000). Critical periods during sensory development. *Current Opinions in Neurobiology, 10*, 138-145.
- Berg, J. M., Lilienfeld, S. O., Reddy, S. D., Latzman, R. D., Roose, A., Craighead, L. W., ... Raison, C. L. (2013). The Inventory of Callous and Unemotional Traits: A construct-validation analysis in an at-risk sample. *Assessment*, 20, 532-544.
- Berger, H. (1929). Uber das elektroenzephalogramm des menschen. Archiv für Psychiatrie und Nervenkrankheiten, 87, 527-570.
- Berkowitz, L. (1990). On the formation and regulation of anger and aggression:
 A cognitive-neoassociationistic analysis. *American Psychologist, 45,* 494-503.
- Berkowitz, L. (1993). *Aggression: Its causes, consequences, and control.* New York, NY: McGraw-Hill.
- Berkowitz, L., & Harmon-Jones, E. (2004). Toward an understanding of the determinants of anger. *Emotion, 4,* 107-130.
- Bernat, E. M., Hall, J. R., Steffen, B. V., & Patrick, C. J. (2007). Violent offending predicts P300 amplitude. *International Journal of Psychophysiology, 66,* 161-167.
- Bernstein, I. H., & Gesn, P. R. (1997). On the dimensionality of the Buss/Perry aggression questionnaire. *Behavioral Research Therapy, 35,* 563-568.
- Bertsch, K., Böhnke, R., Kruk, M. R., & Naumann, E. (2009). Influence of aggression on information processing in the emotional stroop task-an event-related potential study. *Frontiers in Behavioral Neuroscience, 3,* 28-31.

- Best, M., Williams, J. M., & Coccaro, E. F. (2002). Evidence for a dysfunctional prefrontal circuit in patients with an impulsive aggressive disorder. *Proceedings of the National Academy of Science, USA, 99*(12), 8448-8453.
- Bettencourt, B., Talley, A., Benjamin, A., & Valentine, J. (2006). Personality and Aggressive Behavior Under Provoking and Neutral Conditions: A Meta-Analytic Review. *Psychological Bulletin*, *132*, 751-777.
- Beyer, F., Münte, T. F., & Krämer, U. M. (2014). Increased neural reactivity to socio-emotional stimuli links social exclusion and aggression. *Biological Psychology*, *96*, 102-110.
- Bhatia, H. S., Agrawal, R., Sharma, S., Huo, Y. X., Ying, Z., & Gomez-Pinilla, F.
 (2011). Omega-3 fatty acid deficiency during brain maturation reduces neuronal and behavioral plasticity in adulthood. *PLoS One, 6*, e28451.
- Bijttebier, P., & Decoene, S. (2009). Assessment of psychopathic traits in children and adolescents further validation of the antisocial process screening device and the childhood psychopathy scale. *European Journal of Psychological Assessment, 25*(3), 157-163.
- Birbaumer, N., Veit, R., Lotze, M., Erb, M., Hermann, C., & Grodd, W. (2005).
 Deficient Fear Conditioning in Psychopathy: A Functional Magnetic
 Resonance Imaging Study. *Archive of General Psychiatry*, *62*, 799-805.
- Birch, E. E., Khoury, J. C., Berseth, C. L., Castaneda, Y. S., Couch, J. M., Bean, ... Scalabrin, D. M. (2010). The impact of early nutrition on incidence of allergic manifestations and common respiratory illnesses in children. *Journal of Pediatrics*, *156*, 869-871.
- Bird, G., & Viding, E. (2014). The self to other model of empathy: Providing a new framework for understanding empathy impairments in psychopathy, autism, and alexithymia. *Neuroscience and Biobehavioral Reviews, 47*, 520-532.
- Bishop, S., Duncan, J., Brett, M., & Lawrence, A. D. (2004). Prefrontal cortical function and anxiety: controlling attention to threat-related stimuli. *Nature Neuroscience*, *7*, 184-188.

- Bjork, J. M., Dougherty, D. M., & Moeller, F. G. (1997). A positive correlation between self-ratings of depression and laboratory measured aggression. *Psychiatry Research*, 69, 33-38.
- Björkqvist, K., Österman, K., & Kaukiainen, A. (1992). The development of direct and indirect aggressive strategies in males and females. In K.
 Bjorkqvist & P. Niemelä (Eds.), *Of mice and women: Aspects of female aggression* (pp. 51-64). San Diego, CA: Academic Press.
- Björkqvist, K., Österman, K., & Lagerspetz, K. M. J. (1994). Sex differences in covert aggression among adults. *Aggressive Behavior, 20*(1), 27-33.
- Blair, K. S., Morton, J., Leonard, A., & Blair, R. J. R. (2006). Impaired decisionmaking on the basis of both reward and punishment information in individuals with psychopathy. *Personality and Individual Differences, 41*, 155-165.
- Blair, R. J. R. (1995). A cognitive developmental approach to morality: Investigating the psychopath. *Cognition, 57,* 1-29.
- Blair, R. J. R. (1997). Moral reasoning in the child with psychopathic tendencies. *Personality and Individual Differences*, *22*, 731-739.
- Blair, R. J. R. (1999). Responsiveness to distress cues in the child with psychopathic tendencies. *Personality and Individual Differences*, *27*, 135-145.
- Blair, R. J. R. (2001). Neurocognitive models of aggression, the Antisocial Personality Disorders and Psychopathy. *Journal of Neurology, Neurosurgery and Psychiatry, 71*(6), 727-731.
- Blair, R. J. R. (2003). Neurobiological basis of psychopathy. *The British Journal* of Psychiatry: The Journal of Mental Science, 182, 5-7.
- Blair, R. J. R. (2005). Responding to the emotions of others: dissociating forms of empathy through the study of typical and psychiatric populations. *Consciousness and Cognition, 14*(4), 698-718.
- Blair, R. J. R. (2013). Psychopathy: cognitive and neural dysfunction. *Dialogues in Clinical Neuroscience, 15*(2), 181-190.
- Blair, R. J. R., & Cipolotti, L. (2000). Impaired social response reversal: A case of "acquired sociopathy". *Brain*, *123*, 1122-1141.

- Blair, R. J. R., & Coles, M. (2000). Expression recognition and behavioural problems in early adolescence. *Cognitive Development*, *15*, 421-434.
- Blair, R. J. R., Colledge, E., Murray, L., & Mitchell, D. G. V. (2001). A selective Impairment in the Processing of Sad and Fearful Expressions in Children with Psychopathic Tendencies. *Journal of Abnormal Child Psychology, 29*(6), 491-498.
- Blair, R. J. R., & Frith, U. (2000). Neurocognitive explanations of the antisocial personality disorders. *Criminal Behaviour and Mental Health*, *10*, 66-82.
- Blair, R. J. R., Jones, L., Clark, F., & Smith, M. (1997). The psychopathic individual: A lack of responsiveness to distress cues? *Psychophysiology*, *34*, 192-198.
- Blair, R. J. R., Mitchell, D. G. V., & Colledge, E. (2004). Reduced sensitivity to other's fearful expressions in psychopathic individuals. *Personality and Individual Differences*, *37*, 1111-1122.
- Blair, R. J. R., Morris, J. S., Frith, C. D., Perrett, D. I., & Dolan, R. (1999).Dissociable neural responses to facial expressions of sadness and anger. *Brain*, *122*, 883-893.
- Blair, R. J. R., Peschardt, K. S., Budhani, S., Mitchell, D. G., & Pine, D. S. (2006). The development of psychopathy. *Journal of Child Psychology* and Psychiatry, 47, 262-276.
- Blair, R. J. R., Sellars, C., Strickland, I., Clark, F., Williams, A. O., Smith, M., & Jones, L. (1995). Emotion attributions in the psychopath. *Personality* and Individual Differences, 19, 431-437.
- Blanchard, D. C., Hynd, A. L., Minke, K. A., Minemoto, T., & Blanchard, R. J.
 (2001). Human defensive behaviors to threat scenarios show parallels to fear- and anxiety-related defence patterns of non-human mammals. *Neuroscience and Biobehavioral Reviews, 25,* 761-770.
- Blanchette, I. (2006). Snakes, spiders, guns, and syringes: How specific are evolutionary constraints on the detection of threatening stimuli? *Quarterly Journal of Experimental Psychology*, *59*, 1484-1504.
- Blonigen, D. M., Hicks, B. M., Krueger, R. F., Patrick, C. J., & Iacono, W. (2005). Psychopathic personality traits: Heritability and genetic overlap

with internalizing and externalizing psychopathology. *Psychological Medicine*, *35*, 1-12.

- Bobes, M. A., Martin, M., Olivares, E., & Valdes-Sosa, M. (2000). Different scalp topography of brain potentials related to expression and identity matching of faces. *Cognitive Brain Research*, 9, 249-260.
- Bobes, M. A., Ostrosky, F., Diaz, K., Romero, C., Borja, K., Santos, Y., &
 Valdés-Sosa, M. (2013). Linkage of functional and structural anomalies in the left amygdala of reactive-aggressive men. *Social, Cognitive, and Affective Neuroscience, 8*(8), 928-936.
- Bocquillon, P., Bourriez, J. L., Palmero-Soler, E., Betrouni, N., Houdayer, E., Derambure, P., & Dujardin, K. (2011). Use of swLORETA to localize the cortical sources of target- and distracter-elicited P300 components. *Clinical Neurophysiology*, *122*, 1991-2002.
- Boddaert, N., Chabane, N., Gervais, H., Good, C. D., Bourgeois, M., Plumet, M.
 H., ... Zilbovicius, M. (2004). Superior temporal sulcus anatomical abnormalities in childhood autism: a voxel-based morphometry MRI study. *Neuroimage, 23*, 364-369.
- Böhnke, R. Bertsch, K., Kruk, M. R., & Naumann, E. (2010). The relationship between basal and acute HPA axis activity and aggressive behavior in adults. *Journal of Neural Transmission*, *117*, 629-637.
- Bokura, H., Yamaguchi, S., & Kobayashi, S. (2001). Electrophysiological correlates for response inhibition in a go/nogo task. *Clinical Neurophysiology, 112,* 224-232.
- Bond, A. J., & Surguy, S. M. (2000). Relationship between attitudinal hostility and P300 latencies. *Progress in Neuropsychopharmacology*, 24, 1277-1288.
- Botvinick, M. M. (2007). Conflict monitoring and decision making: reconciling two perspectives on anterior cingulate function. *Cognitive Affective and Behavioral Neuroscience*, *7*, 356-366.
- Botvinick, M. M., Braver, T. S., Barch, D. M., Carter, C. S., & Cohen, J. D. (2001). Conflict monitoring and cognitive control. *Psychology Review, 108*, 624-665.

- Botvinick, M. M., Cohen, J. D., & Carter, C. S. (2004). Conflict monitoring and anterior cingulate cortex: an update. *Trends in Cognitive Science, 8,* 539-546.
- Bötzel, K., & Grüsser, O. J. (1989). Electric brain potentials evoked by pictures of faces and non-faces: a search for face-specific EEG-Potentials. *Experimental Brain Research*, 77, 349-360.
- Boucher, L,. Palmeri, T. J., Logan, G. D., & Schall, J. D. (2007). Inhibitory control in mind and brain: an interactive race model of countermanding saccades. *Psychological Review*, 114, 376-397.
- Boucsein, W., Schaefer, F., Sokolov, E. N., Schroder, C., & Furedy, J. (2001).
 The color-vision approach to emotional space: cortical evoked potential data. *Integrative Physiological and Behavioral Science, 36*, 137-153.
- Bourre, J. M. (2006). Effects of nutrients (in food) on the structure and function of the nervous system: update on dietary requirements for brain. Part 2: macronutrients. *Journal of Nutritional Health and Aging*, *10*, 386-399.
- Branca, F. (2006). Nutritional solutions to major health problems of preschool children: How to optimise growth and development. *Journal of Pediatric Gastroenterology and Nutrition, 3*(43), 4-7.
- Branchey, M. H., Buydens-Branchey, L., & Lieber, C. S. (1988). P3 in alcoholics with disordered regulation of aggression. *Psychiatry Research*, 25, 49-58.
- Brandeis, D., Banaschewski, T., Baving, L., Georgiewa, P., Blanz, B., Warnke,
 A., ... Scheuerpflug, P. (2002a). Multicenter P300 brain mapping of
 impaired attention to cues in hyperkinetic children. *Journal of the American Academy of Child and Adolescent Psychiatry*, *41*, 990-998.
- Brass, M., Derrfuss, J., Forstmann, B., & von Cramon, D. Y. (2005). The role of the inferior frontal junction area in cognitive control. *Trends in Cognitive Science, 9,* 314-316.
- Braver, T. S., Barch, D. M., Gray, J. R., Molfese, D. L., & Snyder, A. (2001). Anterior cingulate cortex and response conflict: effects of frequency, inhibition and errors. *Cerebral Cortex*, 11, 825-836.

- Brenna, J. T. (2002). Efficiency of conversion of alpha-linolenic acid to long chain n-3 fatty acids in man. *Current Opinion in Clinical Nutrition & Metabolic Care*, *5*(2), 127-132.
- Breiter, H. C., Etcoff, N. L., Whalen, P. J., Kennedy, W. A., Rauch, S. L., Buckner, R. L., ... Rose, B. R. (1996). Response and habituation of the human amygdala during visual processing of facial expression. *Neuron*, *17*, 875-887.
- Brower, M., & Price, B. (2001). Neuropsychiatry of frontal lobe dysfunction in violent criminal behavior: a critical review. *Journal of Neurology, Neurosurgery, and Psychiatry, 71,* 720-726.
- Brown, J. D., L'Engle, K. L, Pardun, C. J, Guo, G., Kenneavy, K., & Jackson, C. (2006). Sexy media matter: Exposure to sexual content in music, movies, television, and magazines predicts black and white adolescents' sexual behavior. *Pediatrics*, *117*(4), 1018-1027.
- Brown, S., & Schäfer, E. (1888). An investigation into the functions of the occipital and temporal lobes of the monkey's brain. *Philosophical Transactions of the Royal Society B, 179*, 303-327.
- Bruce, V., & Young, A. (1986). Understanding face recognition. *British Journal of Psychology*, *77*(3), 305-327.
- Bruin, K. J., & Wijers, A. A. (2002). Inhibition, response mode, and stimulus probability: a comparative event-related potential study. *Clinical Neurophysiology*, *113*, 1172-1182.
- Bruin, K. J., Wijers, A. A., & van Staveren, A. S. (2001). Response priming in a go/nogo task: Do we have to explain the go/nogo N2 effect in terms of response activation instead of inhibition? *Clinical Neurophysiology*, *112*, 1660-1671.
- Bryant, B. K. (1982). An index of empathy for children and adolescents. *Child Development, 53*, 413-425.
- Bryd, A. L., Kahn, R. E., & Pardini, D. A. (2013). A validation of the Inventory f Callous-Unemotional Traits in a community sample of young adult males. *Journal of Psychopathology and Behavioral Assessment, 35*(1).

- Buck, R. (1984). *The communication of emotion*. New York, NY: The Guilford Press.
- Burgess, J. R., Stevens, L., Zhang, W., & Peck, L. (2000). Long-chain polyunsaturated fatty acids in children with attention-deficit hyperactivity disorder. *American Journal of Clinical Nutrition*, 71(1), 327-330.
- Burke, J. D., Loeber, R., & Lahey, B. B. (2007). Adolescent conduct disorder and interpersonal callousness as predictors of psychopathy in young adults. *Journal of Clinical Child and Adolescent Psychology, 36*(3), 334-346.
- Burle, B., Vidal, F., & Bonnet, M. (2004). Electroencephalographic nogo potentials in a no-movement context: The case of motor imagery in humans. *Neuroscience Letters*, *360*, 77-80.
- Bush, G., Frazier, J. A., Rauch, S. L., Seidman, L. J., Whalen, P. J., Jenike, M.
 A., ... Biederman, J. (1999). Anterior Cingulate Cortex Dysfunction in
 Attention- Deficit/Hyperactivity Disorder Revealed by fMRI and the
 Counting Stroop. *Biological Psychiatry*, 45, 1542-1552.
- Bush, G., Luu, P., & Posner, M. I. (2000). Cognitive and emotional influences in anterior cingulate cortex. *Trends in Cognitive Science, 4,* 215-222.
- Buss, A. H. (1961). *The psychology of aggression.* New York, NY: John Wiley & Sons.
- Buss, A. H., & Durke, A. (1957). An inventory for assessing different kinds of hostility. *Journal of Consulting Psychology, 21,* 343-349.
- Buss, A. H., & Perry, M. (1992). The Aggression Questionnaire. *Journal of Personality and Social Psychology*, *63*, 452-459.
- Buss, A. H., & Warren, W. L. (2000). *Aggression Questionnaire: Manual.* Los Angeles, CA: Western Psychological Services.
- Buss, D. M., & Shackelford, T. K. (1997). From vigilance to violence: Mate retention tactics in married couples. *Journal of Personality and Social Psychology*, 72, 346-361.
- Butter, C. M., & Snyder, D. R. (1972). Alterations in aversive and aggressive behaviours following orbital frontal lesions in rhesus monkeys. *Acta Neurobiological Experiments, 32,* 525-565.

- Buydens-Branchey, L., & Branchey, M. (2008). Long-chain n-3 polyunsaturated fatty acids decrease feelings of anger in substance abusers. *Psychiatry Research, 157,* 95-104.
- Buydens-Branchey, L., Branchey, M., McMakin, D. L., & Hibbeln, J. R. (2003).
 Polyunsaturated fatty acid status and aggression in cocaine addicts.
 Drug and Alcohol Dependence, 71(3), 319-323.
- Cacioppo, J. T., & Berntson, G. G. (1994). Relationship between attitudes and evaluative space: a critical review, with emphasis on the separability of positive and negative substrates. *Psychological Bulletin, 115,* 401-423.
- Caharel, S., Leleu, A., Bernard, C., Viggiano, M. P., Lalonde, R., & Rebaï, M.
 (2013). Early holistic face-like processing of Arcimboldo paintings in the right occipito-temporal cortex: evidence from the N170 ERP component. *International Journal of Psychophysiology, 90*(2), 157-164.
- Calder, P. C. (2012). Mechanisms of Action of (n-3) Fatty Acids. *Journal of Nutrition, 142,* 592-599.
- Calder, A., & Grimble, R. F. (2002). Polyunsaturated fatty acids, inflammation and immunity. *European Journal of Clinical Nutrition, 56*, 14-19.
- Calder, A., Lawrence, A., & Young, A. W. (2001). Neuropsychology of fear and loathing. *Nature Reviews Neuroscience, 2,* 352-363.
- Calder, A. J., Young, A. W., Rowland, D., Perrett, D. I., Hodges, J. R., & Etcoff, N. L. (1996). Facial emotion recognition after bilateral amygdala damage: differentially severe impairment of fear. *Cognitive Neuropsychology*, *13*, 699-745.
- Calderon, F., & Kim, H. Y. (2004). Docosahexaenoic acid promotes neurite growth in hippocampal neurons. *Journal of Neurochemistry, 90*, 979-988.
- Cambronero, J. C., Rivas, F. J., Borrell, J., & Guaza, C. (1992). Role of arachidonic acid metabolism on corticotropin-releasing factor (CRF)release induced by interleukin-1 from superfused rat hypothalami. *Journal Neuroimmunology, 39,* 57-66.
- Campanella, S., Hanoteau, C., Dépy, D., Rossion, B., Bruyer, R., Crommelinck, M., & Guerit, J. M. (2000). Right N170 modulation in a face

discrimination task: An account for categorical perception of familiar faces. *Psychophysiology, 37*(6), 796-806.

- Campanella, S., Peigneux, P., Petit, G., Lallemand, F., Saeremans, M., Noël,
 X., ... Verbanck, P. (2013). Increased cortical activity in binge drinkers
 during working memory task: a preliminary assessment through a
 functional magnetic resonance imaging study. *PLoS One*, *8*, e62260.
- Campanella, S., Quinet, P., Bruyer, R., Crommelinck, M., & Guérit, J. M. (2002).
 Categorical perception of happiness and fear facial expressions: an
 ERP study. *Journal of Cognitive Neuroscience*, *14*, 210-227.
- Campbell, S. B. (1995), Behaviour problems in preschool children: A review of recent research, *Journal of Child Psychology and Psychiatry, 36*(1), 115-119.
- Campbell, A. (1999). Staying alive: Evolution, culture, and women's intrasexual aggression. *Behavioral and Brain Sciences, 22,* 203-252.
- Camras, L. (1977). Facial expressions used by children in a conflict situation. *Child Development, 48,* 1421-1435.
- Cao, D., Kevala, K., Kim, J., Moon, H. S., Jun, S. B., Lovinger, D., & Kim, H. Y. (2009). Docosahexaenoic acid promotes hippocampal neuronal development and synaptic function. *Journal of Neurochemistry*, *111*, 510-521.
- Caputo, A. A., Frick, P. J., & Brodsky, S. L. (1999). Family violence and juvenile sex offending: Potential mediating roles of psychopathic traits and negative attitudes toward women. *Criminal Justice and Behavior, 26,* 338-356.
- Cardinal, R. N., Parkinson, J. A., Hall, J., & Everitt, B. J. (2002). Emotion and motivation: the role of the amygdala, ventral striatum, and prefrontal cortex. *Neuroscience and Biobehavioral Review, 26*, 321-352.
- Carlson, S. E. (2001). Docosahexaenoic acid and arachidonic acid in infant development. *Seminars in Neonatology, 6,* 437-449.
- Carlson, S. R., & Thai, S. (2010). ERPs on a continuous performance task and self-reported psychopathic traits: P3 and CNV augmentation are

associated with Fearless Dominance. *Biological Psychology, 85*, 318-330.

- Carlson, S. M., Zayas, V., & Guthormsen, A. (2009). Neural correlates of decision making on a gambling task. *Child Development*, *80*, 1076-1096.
- Carmel, D., & Bentin, S. (2002). Domain specificity versus expertise: factors influencing distinct processing of faces. *Cognition, 83,* 1-29.
- Carr, L., Iacoboni, M., Dubeau, M. C., Mazziotta, J. C., & Lenzi, G. L. (2003). Neural mechanisms of empathy in humans: A relay from neural systems for imitation to limbic areas. *Proceedings of the National Academy of Science of the USA, 100,* 5497-5502.
- Carr, M. B., & Lutjemeier, J. A. (2005). The relation of facial affect recognition and empathy to delinquency in youth offenders. *Adolescence, 40,* 601-619.
- Carretié, L., & Iglesias, J. (1995). An ERP study on the specificity of facial expression processing. *International Journal of Psychophysiology, 19,* 183-192.
- Carretié, L., Martín-Loeches, M., Hinojosa, J. A., & Mercado, F. (2001). Emotion and attention interaction studied through event-related potentials. *Journal of Cognitive Neuroscience, 13*, 1109-1128.
- Carretié, L., Hinojosa, J. A., Martín-Loeches, M., Mercado, F., & Tapia, M. (2004). Automatic Attention to Emotional Stimuli: Neural Correlates. *Human Brain Mapping 22*, 290-299.
- Carter, C. S., & van Veen, V. (2007). Anterior cingulate cortex and conflict detection: an update of theory and data. *Cognitive Affective and Behavioral Neuroscience*, 7, 367-379.
- Catania, A. (1999). Thorndike's legacy: learning, selection, and the law of effect. *Journal of the Experimental Analysis of Behavior, 72*, 425-428.
- Catmur, C., Walsh, V., & Heyes, C. (2007). Sensorimotor learning configures the human mirror system. *Current Biology*, *17*, 1527-1531.
- Catmur, C., Walsh, V., & Heyes, C. (2009). Associative sequence learning: the role of experience in the development of imitation and the mirror

system. *Philosophical Transactions of the Royal Society B, 364,* 2369-2380.

- Ceponiene, R., Rinne, T., & Naatanen, R. (2002). Maturation of cortical sound processing as indexed by event-related potentials. *Clinical Neurophysiology, 113,* 870-882.
- Chabrol, H., Valls, M., van Leeuwen, N., & Bui E. (2012). Callous-unemotional and borderline traits in nonclinical adolescents: Personality profiles and relations to antisocial behaviors. *Personality and Individual Differences,* 53, 969-973.
- Chai, X. J., Castañón, A. N., Ongür, D., & Whitfield-Gabrieli, S. (2012). Anticorrelations in resting state networks without global signal regression. *Neuroimage*, 59(2), 1420-1428.
- Chamberlain, S. R., Menzies, L., Hampshire, A., Suckling, J., Fineberg, N. A., del Campo, N. ... Sahakian, B. J. (2008). Orbitofrontal dysfunction in patients with obsessive-compulsive disorder and their unaffected relatives. *Science*, *321*, 421-422.
- Chambers, C. D., Bellgrove, M. A., Stokes, M. G., Henderson, T. R., Garavan,
 H., Robertson, I. H., ... Mattingley, J. B. (2006). Executive "brake failure" following deactivation of human frontal lobe. *Journal of Cognitive Neuroscience*, *18*, 444-455.
- Chandrasekaran, A. R., Plas, D. T., Gonzalez, E., & Crair, M. C. (2005). Evidence for an instructive role of retinal activity in retinotopic map refinement in the superior colliculus of the mouse. *Journal of Neuroscience, 25,* 6929-6938.
- Chen, C. Y., Muggleton, N. G., & Chang, J. R. (2014). Inefficiency of post-error adjustment in impulsive violent offenders. *Neuroreport, 25*(13), 1024-1029.
- Chen, C. Y., Muggleton, N. G., Juan, C. H., Tzeng, O. J., & Hung, D. L. (2008).
 Time pressure leads to inhibitory control deficits in impulsive violent offenders. *Behavioural Brain Research*, *187*, 483-488
- Chen, H., & Su, H. M. (2013). Exposure to a maternal n-3 fatty acid-deficient diet during brain development provokes excessive hypothalamic-

pituitary-adrenal axis responses to stress and behavioural indices of depression and anxiety in male rat offspring later in life. *Journal of Nutritional Biochemistry*, *24*, 70-80.

- Chen, C. Y., Tien, Y. M., Juan, C. H., Tzeng, O. J., & Hung, D. L. (2005). Neural correlates of impulsive-violent behavior: an event-related potential study. *Neuroreport, 16,* 1213-1216.
- Chikazoe, J., Jimura, K., Asari, T., Yamashita, K., Morimoto, H., Hirose, S., ...
 Konishi, S. (2009). Functional dissociation in right inferior frontal cortex
 during performance of go/nogo-go task. *Cerebral Cortex, 19,* 146-152.
- Christian, R., Frick, P. J., Hill, N., Tyler, L. A., & Frazer, D. (1997). Psychopathy and conduct problems in children: II. Subtyping children with conduct problems based on their interpersonal and affective style. *Journal of the American Academy of Child and Adolescent Psychiatry, 36,* 233-241.
- Cima, M., Smeets, T., & Jelicic, M. (2008). Self reported trauma, cortisol levels, and aggression in psychopathic and non psychopathic prison inmates. *Biological Psychology, 78,* 75-86.
- Ciucci, E., & Baroncelli, A. (2014). The emotional core of bullying: Further evidences of the role of callous-unemotional traits and empathy. *Personality and Individual Differences, 67,* 69-74.

Cleckley H. (1941). The Mask of Sanity. St. Louis, MO: Mosby

- Cleckley H. (1976). *The Mask of Sanity* (5th Ed). St. Louis, MO: Mosby.
- Coccaro, E. F. (1992). Impulsive aggression and central serotonergic system function in humans: An example of a dimensional brain-behavior relationship. *International Clinical Psychopharmacology, 7,* 3-12.
- Coccaro, E. F. (2003). Intermittent explosive disorder: Taming temper in the volatile, impulsive adult. *Current Psychiatry, 2,* 42-60.
- Coccaro, E. F., Kavoussi, R. J., Cooper, T. B., & Hauger, R. L. (1997). Central serotonin activity and aggression: Inverse relationship with prolactin response to d-fenfluramine, but not CSF 5-HIAA concentration, in human subjects. *American Journal of Psychiatry*, *154*, 1430-1435.
- Coccaro, E. F., Lee, R., & Coussons-Read, M. (2014). Elevated plasma inflammatory markers in individuals with intermittent explosive disorder

and correlation with aggression in humans. *JAMA Psychiatry*, *71*, 158-165.

- Coccaro, E. F., McCloskey, M. S., Fitzgerald, D. A., & Phan, K. L. (2007). Amygdala and orbitofrontal reactivity to social threat in individuals with impulsive aggression. *Biological Psychiatry, 62,* 168-178.
- Coccaro, E. F., Siever, L. J., Klar, H. M., Maurer, G., Cochrane K., Cooper T.
 B., ... Davis, K. L. (1989). Serotonergic studies in patients with affective and personality disorders correlates with suicidal and impulsive aggressive-behavior. *Archives of General Psychiatry*, *46*, 587-599.
- Coccaro, E. F., Sripada, C. S., Yanowitch, R. N., & Phan, K. L. (2011). Corticolimbic function in impulsive aggressive behavior. *Biological Psychiatry, 69,* 1153-1159.
- Coccaro, E. F., Solis, O., Fanning, J., & Lee, R. (2015). Emotional intelligence and impulsive aggression in Intermittent Explosive Disorder. *Journal of Psychiatric Research, 61,* 135-140.
- Cohen, D. J., Eckhardt, C. I., & Schagat, K. D. (1998). Attention allocation and habituation to anger-related stimuli during a visual search task. *Aggressive Behavior, 24*, 399-409.
- Coid, J., Yang, M., Ullrich, S., Roberts, A., Moran, P., Bebbington, P.,... Hare,
 R. (2009). Psychopathy among prisoners in England and Wales.
 International Journal of Law and Psychiatry, 32(3), 134-141.
- Coles, M. G. H., Gratton, G., & Fabiani, M. (1990). In Cacioppo, J. T. (ed.) Event-related potentials. *Principles of psychophysiology. Physical, social, and inferential elements*. Cambridge, UK: Cambridge University Press.
- Conklin, S. M., Gianaros, P. J., Brown, S. M., Yao, J. K., Hariri, A. R., Manuck, S. B., & Muldoon, M. R. (2007). Long-chain omega-3 fatty acid intake is associated positively with corticolimbic gray matter volume in healthy adults. *Neuroscience Letters*, *421*, 209-212.
- Conklin, S. M., Harris, J. I., Manuck, S. B., Yao, J. K., Hibbeln, J. R., & Muldron, M. F. (2007). Serum w-3 fatty acids are associated with variation in

mood, personality and behaviour in hypercholesterolemic community volunteers. *Psychiatry Research, 152,* 1-10.

- Connor, D. F. (2002) Review: Short acting methylphenidate has short term efficacy in children and adolescents with attention deficit disorder. *Evidence Based Mental Health, 5*(2), 50.
- Corbetta, M., & Shulman, G. L. (2002). Control of goal-directed and stimulusdriven attention in the brain. *Nature Reviews Neuroscience, 3,* 201-215.
- Cornell, D. G., Warren, J., Hawk, G., Stafford, E., Oram, G., & Pine, D. (1996). Psychopathy in instrumental and reactive violent offenders. *Journal of consulting and clinical psychology*, 64(4), 783.
- Correll, J., Urland, G. R., & Ito, T. A. (2006). Event-related potentials and the decision to shoot: The role of threat perception and cognitive control. *Journal of Experimental Social Psychology, 42*, 120-128.
- Costa, L., Bauer, L. O., Kuperman, S., Porjesz, B., O'Connor, S., Hesselbrock,
 V., ... Begleiter, H. (2000). Frontal P300 decrements, alcohol
 dependence, and antisocial personality disorder. *Biological Psychiatry*,
 47, 1064-1071.
- Coxon, J. P., Stinear, C. M., & Byblow, W. D. (2006). Intracortical inhibition during volitional inhibition of prepared action. *Journal of Neurophysiology, 95,* 3371-3383.
- Coyne, S. M., Archer, J., & Eslea, M. (2006). "We're not friends anymore! Unless" The frequency and harmfulness of indirect, relational, and social aggression. *Aggressive Behavior, 32,* 294-307.
- Crawford, M. A. (2006). Docosahexaenoic acid in neural signalling systems. *Nutritional Health, 18*(3), 263-276.
- Crawford, M. A., Golfetto, I., Ghebremeskel, K., Min, Y., Moodley, T., Poston,
 L., & Schmidt, W. (2003). The potential role for arachidonic and
 docosahexaenoic acids in protection against some central nervous
 system injuries in preterm infants. *Lipids, 38*, 303-315.
- Crawford, M. A., Hassam, A. G., Williams, G., & Whitehouse, W. (1977). Fetal accumulation of long-chain polyunsaturated fatty acids. *Advancements in Experimental Medicine and Biology, 83,* 135-143.

- Crick, N. R., & Dodge, K. A. (1994). A review and reformulation of socialinformation processing mechanisms in children's development. *Psychological Bulletin, 115,* 74-101.
- Crick, N. R., & Grotpeter, J. K. (1995). Relational aggression, gender, and social-psychological adjustment. *Child Development, 66*(3), 710-722.
- Crist, R., Wu, C. T., Karp, C., & Woldorff, M. (2008). Face processing is gated by visual spatial attention. *Frontiers in Human Neuroscience, 1.*
- Crottaz-Herbette, S., & Menon, V. (2006). Where and when the anterior cingulate cortex modulates attentional response: combined fMRI and ERP evidence. *Journal of Cognitive Neuroscience, 18,* 766-780.
- Crutcher, M. D., & Alexander, G. E. (1990). Movement-related neuronal activity selectively coding either direction or muscle pattern in three motor areas of the monkey. *Journal of Neurophysiology, 64*(1), 151-163.
- Cusi, A. M., Nazarov, A., Holshausen, K., Macqueen, G. M., & McKinnon, M. C. (2012). Systematic review of the neural basis of social cognition in patients with mood disorders. *Journal of Psychiatry Neuroscience*, *37*(3), 154-169.
- da Silva, D. R., Rijo, D., & Salekin, R. T. (2015). The evolutionary roots of psychopathy. *Aggression and Violent Behavior, 21,* 85-96.
- Dadds, M. R., El Masry, Y., Wimalaweera, S., & Guastella, A. J. (2008).
 Reduced eye gaze explains fear blindness in childhood psychopathic traits. *Journal of the American Academy of Child and Adolescent Psychiatry*, 47, 455-463.
- Dadds, M. R., Fraser, J., Frost, A., & Hawes, D. J. (2005). Disentangling the underlying dimensions of psychopathy and conduct problems in childhood: A community study. *Journal of Consulting and Clinical Psychology*, *73*, 400-410.
- Dadds, M. R., Hawes, D. J., Frost, A. D., Vassallo, S., Bunn, P., Hunter, K., & Merz, S. (2009). Learning to 'talk the talk: the relationship of psychopathic traits to deficits in empathy across childhood. *Journal of Child Psychology and Psychiatry, 50*(5), 599-606.

- Dadds, M. R., Perry, Y., Hawes, D. J., Merz, S., Riddell, A. C., Haines, D. J., ...Abeygunawardane, A. I. (2006). Attention to the eyes and fearrecognition deficits in child psychopathy. *British Journal of Psychiatry*, *189*, 280-281.
- Dai, Q., & Feng, Z. (2012). More excited for negative facial expressions in depression: evidence from an event-related potential study. *Clinical Neurophysiology*, *123*(11), 2172-2179.
- Damasio, A. (1994). *Descartes' error: emotions, reason, and the human brain.* New York, NY: Avon Books.
- Damasio, A. (2003). *Looking for Spinoza. Joy, sorrow, and the feeling brain.* New York, NY: Harcourt.
- Danthiir, V., Burns, N. R., Nettelbeck, T., Wilson, C., & Wittert, G. (2011). The older people, omega-3, and cognitive health (EPOCH) trial design and methodology: A randomised, double-blind, controlled trial investigating the effect of long-chain omega-3 fatty acids on cognitive ageing and wellbeing in cognitively healthy older adults. *Nutrition Journal, 10,* 117-135.
- Dapretto, M., Davies, M. S., Pfeifer, J. H., Scott, A. A., Sigman, M., Bookheimer, S. Y., & Iacoboni, M. (2006). Understanding emotions in others: mirror neuron dysfunction in children with autism spectrum disorders. *Nature Neuroscience*, *9*, 28-30.
- Darwin, C. R. (1872). *The expression of the emotions in man and animals.* London, UK: John Murray.
- Davidson, R. J., Jackson, D. C., & Kalin, N. H. (2000). Emotion, plasticity, context, and regulation: perspectives from affective neuroscience. *Psychological Bulletin*, 126(6), 890-909.
- Davidson, R. J., Putnam, K. M., & Larson, C. L. (2000). Dysfunction in the neural circuitry of emotion regulation a possible prelude to violence. *Science*, *289*, 591-594.
- Davis, M. H. (1983). The effects of dispositional empathy on emotional reactions and helping: A multidimensional approach. *Journal of Personality, 51,* 167-184.

- Dawel, A., O'Kearney, R., McKone, E., & Palermo, R. (2012). Not just fear and sadness: meta-analytic evidence of pervasive emotion recognition deficits for facial and vocal expressions in psychopathy. *Neuroscience* and Biobehavioral Reviews, 36, 2288-2304.
- Dawkins, R. (2006). *The Selfish Gene* (30th anniversary edition). Oxford, UK: Oxford University Press.
- Dawson, G., Webb, S. J., Carver, L., Panagiotides, H., & McPartland, J. (2004).
 Young children with autism show atypical brain responses to fearful versus neutral facial expressions of emotion. *Developmental Science*, *7*(3), 340-359.
- de Brito, S. A., Mechelli, A., Wilke, M., Laurens, K. R., Jones, A. P., Barker, G. J., ... Viding, E. (2009). Size matters: Increased grey matter in boys with conduct problems and callous-unemotional traits. *Brain, 132*(4), 843-852.
- de Groot, R., Costanza, R., Broeck, D. V. D., Aronson, J., Burkhard, B., Gomez-Baggethun, E., ... Zurlini, G. (2011). A global partnership for ecosystem services. *Solutions*, *2*, 42-43.
- de LaRosa, B. L., Spence, J. S., Shakal, S. K. M., Motes, M. A., Calley, C. S., Calley, V. I., ... Kraut, M. A. (2014). Electrophysiological spatiotemporal dynamics during implicit visual threat processing. *Brain and Cognition*, *91*, 54-61.
- de Jong, R., Coles, M. G., Logan, G. D., & Gratton, G. (1990). In search of the point of no return: the control of response processes. *Journal of Experimental Psychology, 16,* 164-182.
- de Pascalis, V., Arwari, B., Matteucci, M., & Mazzocco, A. (2005). Effects of emotional visual stimuli on auditory information processing: a test of J.A. Gray's reinforcement sensitivity theory. *Personality and Individual Differences, 38,* 163-176.
- de Velasco, P. C., Mendonca, H. R., Borba, J. M., Andrade Da Costa, B. L., Guedes, R. C., Navarro, D. M., ... Serfaty, C. A. (2012). Nutritional restriction of omega-3 fatty acids alters topographical fine tuning and

leads to a delay in the critical period in the rodent visual system. *Experimental Neurology, 234*, 220-229.

- Decety, J. (2011). The neuroevolution of empathy. *Annals of the New York Academy of Sciences, 1231,* 35-45.
- Decety, J., & Chaminade, T. (2003). Neural correlates of feeling sympathy. *Neuropsychologia, 41,* 127-138.
- Decety, J., & Jackson, P. L. (2004). The functional architecture of human empathy. *Behavioral and Cognitive Neuroscience Review*, *3*, 71-100.
- Decety, J., & Meyer, M. (2008). From emotion resonance to empathic understanding: A social developmental neuroscience account. *Development and Psychopathology, 20,* 1053-1080.
- Decety, J., Michalska, J. J., Akitsuki, Y., & Lahey, B. B. (2009). Atypical empathic responses in adolescents with aggressive conduct disorder: a functional MRI investigation. *Biological Psychology*, *80*(2), 203.
- Decety, J., & Moriguchi, Y. (2007). The empathic brain and its dysfunction in psychiatric populations: Implications for intervention across different clinical conditions. *BioPsychoSocial Medicine*, *1*, 22.
- Deeley, Q. Daly, E., Surguladze, S., Tunstall, N., Mezey, G., Beer, D., ...
 Murphy, D. (2006). Facial emotion processing in criminal psychopathy:
 Preliminary functional magnetic resonance imaging study. *The British Journal of Psychiatry*, *189*, 533-539.
- Deffke, I., Sander, T., Heidenreich, J., Sommer, W., Curio, G., Trahms, L., & Lueschow, A. (2007). MEG/EEG sources of the 170-ms response to faces are co-localized in the fusiform gyrus. *Neuroimage*, *35*, 1495-1501.
- Degabriele, R., Lagopoulos, J., & Malhi, G. (2011). Neural correlates of emotional face processing in bipolar disorder: An event-related potential study. *Journal of Affective Disorders, 133*(1-2), 212-220.
- Del Gaizo, A. L., & Falkenbach, D. M. (2008). Primary and secondary psychopathic-traits and their relationship to perception and experience of emotion. Personality and Individual Differences, 45, 206-212.

- Deldin, P. J., Keller, J., Gergen, J. A., & Miller, G. A. (2000). Right-posterior face processing anomaly in depression. *Journal of Abnormal Psychology, 109,* 116-121.
- Delle-Vigne, D., Wang, W., Kornreich, C., Verbanck, P., & Campanella, S. (2014). Emotional facial expression processing in depression: Data from behavioral and event-related potential studies. *Clinical Neurophysiology*, 44, 169-187.
- Delplanque, S., Lavoie, M. E., Hot, P., Silvert, L., & Sequeira, H. (2004).
 Modulation of cognitive processing by emotional valence studied through event-related potentials in humans. *Neuroscience Letters, 356,* 1-4.
- Delplanque, S., Silvert, L., Hot, P., & Sequeira, H. (2005). Event-related P3a and P3b in response to unpredictable emotional stimuli. *Biological Psychology, 68,* 107-120.
- Delville, Y., David, J. T., Taravosh-Lahn, K., & Wommack, J. C. (2003). Stress and the development of agonistic behavior in golden hamsters. *Hormones and Behavior, 44,* 263-270.
- Delville, Y., de Vries, G. J., & Ferris, C. F. (2000). Neural Connections of the Anterior Hypothalamus and Agonistic Behavior in Golden Hamsters. *Brain, Behavior, and Evolution, 55*, 53-76.
- DeMar Jr, J. C., Ma, K, Bell, J. M., Igarashi, M., Greenstein, D., & Rapoport, S.
 I. (2006). One generation of n-3 polyunsaturated fatty acid deprivation increases depression and aggression test scores in rats. *Journal of Lipid Research*, 47(1), 172-180.
- Demenescu, L. R., Kortekaas, R., den Boer, J. A., & Aleman, A. (2010). Impaired attribution of emotion to facial expressions in anxiety major depression. *PLoS ONE*, *5*(12), e15058.
- Dennis, T. A., & Chen, C. C. (2007). Neurophysiological mechanisms in the emotional modulation of attention: the interplay between threat sensitivity and attentional control. *Biological Psychology*, *76*, 1-10.
- Denson, T. F., von Hipple, W., Kemp, R. I., & Teo, L. S. (2010). Glucose consumption decreases impulsive aggression in response to

provocation in aggressive individuals. *Journal of Experimental Social Psychology, 46,* 1023-1028.

- Derrfuss, J., Brass, M., & van Cramon, C. Y. (2004). Cognitive control in the posterior frontolateral cortex: evidence from common activations in task coordination, interference control, and working memory. *Neuroimage*, 23, 604-612.
- Desimone, R., & Duncan, J. (1995). Neural mechanisms of selective visual attention. *Annual Review of Neuroscience 18*, 193-222.
- Devinsky, O., Morrell, M. J., & Vogt, B. (1995). Contributions of anterior cingulate cortex to behavior. *Brain*, *118*, 279-306.
- DeWall, N. C., Anderson, C. A., & Bushman, B. J. (2011). The General Aggression Model: Theoretical Extensions to Violence. *Psychology of Violence*, 1(3), 245-258.
- DiGiuseppe, R. (1995). Developing the therapeutic alliance with angry clients. In
 H. Kassinove (Ed.), Anger disorders: definitions, diagnosis, and
 treatment (pp. 131-149). Washington, DC: Taylor & Francis.
- Dimoska, A., Johnstone, S. J., & Barry, R. J. (2006). The auditory-evoked N2 and P3 690 components in the stop-signal task: indices of inhibition, response-conflict or error-detection? *Brain Cognition, 62*, 98-112.
- Dimoska, A., Johnstone, S. J., Barry, R. J., & Clarke, A. R. (2003). Inhibitory motor control in children with attention-deficit/hyperactivity disorder: event-related potentials in the stop-signal paradigm. *Biological Psychiatry, 54*, 1345-1354.
- Dolan, M. C., & Anderson, I. M. (2003): The relationship between serotonergic function and the Psychopathy Checklist: Screening version. *Journal of Psychopharmacology*, 17, 216-222.
- Dolan, M. C., Anderson, I. M., & Deakin, J. F. (2001). Relationship between 5-HT function and impulsivity and aggression in male offenders with personality disorders. *British Journal of Psychiatry*, *178*, 352-359.
- Dolan, M. C., & Fullam, R. S. (2006). Face affect recognition deficits in personality-disordered offenders: association with psychopathy. *Psychological Medicine*, *36*(11), 1563-1569.

- Dolan, M. C., & Fullam, R. S. (2009). Psychopathy and Functional Magnetic Resonance Imaging Blood Oxygenation Level-Dependent Responses to Emotional Faces in Violent Patients with Schizophrenia. *Biological Psychiatry, 66,* 570-577.
- Dollard, J., Doob, L., Miller, N., Mowrer, O., & Sears, R. (1939). *Frustration and aggression*. New Haven, CT: Yale University Press.
- Donahue, J. J., McClure, K. S., & Moon, S. M. (2014). The relationship between emotion regulation difficulties and psychopathic personality characteristics. *Personality Disorders: Theory, Research, and Treatment, 5*(2), 186-194.
- Donchin, E. (1978). Use of scalp distribution as a dependent variable in event-related potential studies. Excerpts of pre conference correspondence.
 In. D. Otto (Ed.), *Multidisciplinary Perspectives in Event-Related Brain Potentials Research* (pp. 501-510). Washington, DC: U.S. Government Printing Office.
- Donchin, E., & Coles, M. G. H. (1988). Is the P300 component a manifestation of cognitive updating? *The Behavioral and Brain Sciences*, *11*, 357-427.
- Donegan, N. H., Sanislow, C. A., Blumberg, H. P., Fulbright, R. K., Lacadie, C., Skudlarski, P., ... Wexler, B. E. (2003). Amygdala hyperreactivity in borderline personality disorder: implications for emotional dysregulation. *Biological Psychiatry*, *54*, 1284-1293.
- Donkers, F. C. L., & van Boxtel, G. J. M. (2004). The N2 in go/no-go tasks reflects conflict monitoring not response inhibition. *Brain Cognition, 56,* 165-176.
- Donohoe, R. T., & Benton, D. (1999). Blood glucose control and aggressiveness in females. *Personality and Individual Differences, 26,* 905-911.
- Dougherty, D. D., Rauch, S. L., Deckersbach, T., Marci, C., Loh, R., Shin, L. M.
 ... Fava, M. (2004). Ventromedial prefrontal cortex and amygdala
 dysfunction during an anger induction Positron Emission Tomography
 Study in patients with major depressive disorder with anger attacks.
 Archive of General Psychiatry, 61, 795-804.

- Dougherty, D. D., Shin, L. M., Alpert, N. M., Pitman, R. K., Orr, S. P., Lasko, M. P., ... Rauch, S. L. (1999). Anger in healthy men: a PET study using script-driven imagery, *Biological Psychiatry*, 46, 466-472.
- Drevets, W. C., & Raichle, M. E. (1998). Reciprocal suppression of regional cerebral blood flow during emotional versus higher cognitive processes: implications for interactions between emotion and cognition. *Cognition* & Emotion, 12, 353-385.
- Du, J., Li, J., Wang, Y., Jiang, Q., Livesley, J. W., Jang, K. L., ... Wang, W. (2006). Event-related potentials in adolescents with combined ADHD and CD disorder: A single stimulus paradigm. *Brain and Cognition*, 60(1), 70-75.
- Dum, R. P., & Strick, P. L. (2005). Frontal lobe inputs to the digit representations in the motor areas on the lateral surface of the hemisphere. *Journal of Neuroscience*, 25, 1375-1386.
- Duncan, C. C., Barry, R. J., Connolly, J. F., Fischer, C., Michie, P. T., Näätänen, R., ... van Petten, C. (2009). Event-related potentials in clinical research: guidelines for eliciting, recording, and quantifying mismatch negativity, P300, and N400. *Clinical Neurophysiology, 120,* 1883-1908.
- Duncan, C., Kosmidis, M. H., & Mirsky, A. F. (2005). Closed head injury-related information processing deficits: an event-related potential analysis. *International Journal of Psychophysiology*, 58, 133-157.
- Dutton, D. G., & Karakanta, C. (2013) Depression as a risk marker for aggression: A critical review. *Aggression & Violent Behavior, 18,* 310-319.
- Dyall, S. C. (2015). Long-chain omega-3 fatty acids and the brain: a review of the independent and shared effects of EPA, DPA and DHA. *Frontiers in Aging Neuroscience*, *7*, 52.
- Eckhardt, I. E., & Cohen, D. J. (1997). Attention to anger-relevant and irrelevant stimuli following naturalistic insult. *Personality and Individual Differences, 23,* 619-629.

- Eckhardt, C. I., & Deffenbacher, J. L. (1995). Diagnosis of anger disorders. In H. Kassinove (Ed.), *Anger disorders: Definition, diagnosis, and treatment* (pp. 27-48). Washington, DC: Taylor & Francis.
- Edwards, R., Peet, M., Shay, J., & Horrobin, D. (1998). Omega-3 polyunsaturated fatty acid levels in the diet and in red blood cell membranes of depressed patients. *Journal of Affective Disorders, 48*, 149-155.
- Eger, E., Sterzer, P., Russ, M. O., Giraud, A. L., & Kleinschmidt, A. (2003). A supramodal number representation in human intraparietal cortex. *Neuron, 37,* 719-725.
- Egner, T., & Hirsch, J. (2005). Cognitive control mechanisms resolve conflict through cortical amplification of task-relevant information. *Nature Neuroscience, 8,* 1784-1790.
- Egner, T., & Hirsch, J. (2005). The neural correlates and functional integration of cognitive control in the Stroop task. *NeuroImage, 24*, 539-547.
- Eibl-Eibesfeldt, I. (1970). *Ethology. The Biology of Behavior.* New York, NY: Holt, Rinehart and Winston, Inc.
- Eichele, T., Debener, S., Calhoun, V. D., Specht, K., Engel, A. K., Hugdahl, K.,
 ... Ullsperger, M. (2008). Prediction of human errors by maladaptive changes in event related brain networks. *Proceedings of the National Academy of Sciences, 105,* 6173-6178.
- Eichele, H., Juvodden, H. T., Ullsperger, M., & Eichele, T. (2010).Maladaptation of event related EEG responses preceding performance errors. *Frontiers in Human Neuroscience*, *4*, 65.
- Eimer, M. (1993). Effects of attention and stimulus probability on ERPs in a go/Nogo task. *Biological Psychology, 35,* 123-138.
- Eimer, M. (2000a). Attentional modulations of event-related brain potentials sensitive to faces. *Cognitive Neuropsychology*, *17*, 103-116.
- Eimer, M. (2000b). Effects of face inversion on the structural encoding and recognition of faces Evidence from event-related brain potentials. *Cognitive Brain Research*, *10*, 145-158.

- Eimer, M. (2000c). Event-related brain potentials distinguish processing stages involved in face perception and recognition. *Clinical Neurophysiology, 111*(4), 694-705.
- Eimer, M., & Holmes, A. (2002). An ERP study on the time course of emotional face processing. *Cognitive Neuroscience and Neuropsychology, 13,* 427-431.
- Eimer, M., & Holmes, A. (2007). Event-related brain potential correlates of emotional face processing. *Neuropsychologia*, *45*(1), 15-31.
- Eimer, M., Holmes, A., & McGlone, F. P. (2003). The role of spatial attention in the processing of facial expression: an ERP study of rapid brain responses to six basic emotions. *Cognitive, Affective & Behavioral Neuroscience, 3,* 97-110.
- Eisenbarth, H., & Alpers, G. W. (2011). Happy mouth and sad eyes: Scanning emotional facial expressions. *Emotion*, *11*(4), 860-865.
- Eisenbarth, H., Alpers, G. W., Segrè, D., Calogero, A., & Angrilli, A. (2008). Categorization and evaluation of emotional faces in psychopathic women. *Psychiatry Research*, *159*(1-2), 189-195.
- Eisenbarth, H., Angrilli, A., Calogero, A., Harper, J., Olson, L. A., & Bernat, E. (2013). Reduced negative affect response in female psychopaths. *Biological Psychology*, *94*, 310-318.
- Eisenberg, N. (2000). Emotion, regulation, and moral development. *Annual Review of Psychology, 51,* 556-697.
- Ekman, P. (1999). Basic emotions. In T. Dalgleish & T. Power (Eds.), *The handbook of cognition and emotion* (pp. 45-60). New York, NY: John Wiley & Sons.
- Ekman, P. (1993). Facial expression of emotion. *American Psychologist, 48,* 384-392.
- Elfenbein, H. A., & Ambady, N. (2002). On the universality and cultural specificity of emotion recognition: A meta-analysis. *Psychological Bulletin, 128*, 203-235.

- Emanuele, E., Brondino, N., Re, S., Bertona, M., & Gerald, D. (2009). Serum omega-3 fatty acids are associated with ultimatum bargaining behaviour. *Physiology & Behavior, 96,* 180-183.
- Endres, S., Ghorbani, R., Kelley, V., Georgilis, K., Lonnemann, G., van der Meer, J. W., ... Weber, P. C. (1989). The effect of dietary supplementation with n-3 polyunsaturated fatty acids on the synthesis of interleukin-1 and tumor necrosis factor by mononuclear cells. *The New England Journal of Medicine: Research and Review, 320,* 265-271,
- Enriquez-Geppert, S., Konrad, C., Pantev, C., & Huster, R. J. (2010). Conflict and inhibition differentially affect the N200/P300 complex in a combined go/nogo and stop-signal task. *NeuroImage, 51,* 877-887.
- Essau, C. A., Sasagawa, S., & Frick, P. J. (2006). Callous-unemotional traits in community sample of adolescents. *Assessment, 13*, 454-469.
- Etkin, A., Egner, T., & Kalisch, R. (2011). Emotional processing in anterior cingulate and medial prefrontal cortex. *Trends in Cognitive Science*, *15*(2), 85-93.
- Etkin, A., Egner, T., Peraza, D. M., Kandel, E. R., & Hirsch, J. (2006). Resolving emotional conflict: a role for the rostral anterior cingulate cortex in modulating activity in the amygdala. *Neuron*, *51*(6), 871-882.
- Ewbank, M. P., Lawrence, A. D., Passamonti, L., Keane, J., Peers, P. V., & Calder, A. (2009). Anxiety predicts a differential neural response to attended and unattended facial signals of anger and fear. *Neuroimage*, 44, 1144-1151.
- Fairchild, G., Hagan, C. C., Walsh, N. D., Passamonti, L., Calder, A. J., & Goodyer, I. M. (2013). Brain structure abnormalities in adolescent girls with conduct disorder. *Journal of Child Psychology and Psychiatry*, 54, 86-95.
- Fairchild, G., van Goozen, S. H. M., Calder, A. J., Stollery, S. J., & Goodyer, I.
 M. (2009). Deficits in facial expression recognition in male adolescents with early-onset or adolescence-onset conduct disorder. *Journal of Child Psychology and Psychiatry, and Allied Disciplines, 50,* 627-636.

- Fairhall, S. L., & Ishai, A. (2007). Effective connectivity within the distributed cortical network for face perception. *Cerebral Cortex, 17*, 2400-2406.
- Falkenstein, M., Hoormann, J., & Hohnsbein, J. (1999). ERP components in go/nogo tasks and their relation to inhibition. *Acta Psychologica*, 101, 267-291.
- Fallgatter, A. J., Ehlis, A. C., Rösler, M., Strik, W. K., Blocher, D., & Herrmann,
 M. J. (2005). Diminished prefrontal brain function in adults with attention
 deficit hyperactivity disorder (ADHD) related psychopathology in
 childhood. *Psychiatry Research and Neuroimaging*, *138*, 157-169.
- Fanti, K. A., Frick, P. J., & Georgiou, S. (2009). Linking callous-unemotional traits to instrumental and non-instrumental forms of aggression. *Journal* of Psychopathology and Behavioral Assessment, 31(4), 285-298.
- Farah, M. J. (1990). Visual Agnosia: Disorders of Object Recognition and What They Tell Us About Normal Vision. Cambridge, MA: MIT Press.
- Farthing, G. W. (2005). Attitudes toward heroic and nonheroic physical risk takers as mates and as friends. *Evolution and Human Behavior, 26*, 171-185.
- Faul, F., Erdfelder, E., Lang, A., G., & Buchner, A. (2007). G*Power 3: A flexible statistical power analysis program for the social, behavioral, and biomedical sciences. *Behavior Research Methods*, 39, 175-191.
- Fawzi, W. W., Rifas-Shiman, S. L., Rich-Edwards, J. W., Willett, W. C., & Gillman, M. W (2004). Calibration of a semi-quantitative food frequency questionnaire in early pregnancy. *Annals of Epidemiology*, *14*, 754-762.
- Fedorova, I., & Salem, N., Jr. (2006). Omega-3 fatty acids and rodent behavior. *Prostaglandins, Leukotrienes and Essential Fatty Acids, 75,* 271-289.
- Fei-Fei, L., Iyer, A., Koch, C., & Perona, P. (2007). What do we see in a glance of a scene. *Journal of Vision 7(1-10), 10,* 1-29.
- Feilhauer, J., & Cima, M. (2013). Youth psychopathy: Differential correlates of callous-unemotional traits, narcissism, and impulsivity. *Forensic Science International, 224,* 1-7.
- Feilhauer, J., Cima, M., & Arntz, A. (2012). Assessing callous-unemotional traits across different groups of youths: Further cross-cultural validation of the
Inventory of Callous-Unemotional Traits. *International Journal of Law and Psychiatry, 35,* 251-262.

- Feilhauer, J., Cima, M., Korebrits, A., & Nicolson, N. A. (2013). Salivary cortisol and psychopathy dimensions in detained antisocial adolescents. *Psychoneuroendocrinology*, 38, 1586-1595.
- Feng, Z., Zou, X., Jia, H., Li, X., Zhu, Z., Liu X., ... Liu, J. (2012). Maternal docosahexaenoic acid feeding protects against impairment of learning and memory and oxidative stress in prenatally stressed rats: possible role of neuronal mitochondria metabolism. *Antioxidants & Redox Signaling, 16*, 275-289.
- Feldmann-Wüstefeld, T., Schmidt-Daffy, M., & Schubö, A. (2011). Neural evidence for the threat detection advantage: Differential attention allocation to angry and happy faces. *Psychophysiology*, 48, 697-707.
- Ferguson, C. J., & Dyck, D. (2012). Paradigm change in aggression research: The time has come to retire the general aggression model. *Aggression and Violent Behavior*, 17, 220-228.
- Fernandes, L. O., Keller, J., Giese-Davis, J. E., Hicks, B. D., Klein, D. N., & Miller, G. A. (1999). Converging evidence for a cognitive anomaly in early psychopathology. *Psychophysiology*, *36*, 511-521.
- Ferreira, C. F., Bernardi, J. R., Krolow, R., Arcego, D. M., Fries, G. R., de Aquiar, B. W., Senter, G., ... Salmaz, C. (2013). Vulnerability to dietary n-3 polyunsaturated fatty acid deficiency after exposure to early stress in rats. *Pharmacology, Biochemistry and Behavior, 107,* 11-19.
- Feshbach, S. (1964). The function of aggression and the regulation of aggressive drive. *Psychological Review, 71,* 257-272.
- Feshbach, N. D., & Feshbach, S. (1982). Empathy training and the regulation of aggression: Potentialities and limitations. *Academic Psychology Bulletin, 4,* 399-413.
- Fiennes, R. N. T. W., Sinclair, A. J., & Crawford, M. A. (1973). Essential fatty acid studies in primates. Linolenic acid requirements of capuchins. *Journal of Medicine in Primates*, 2, 155-169.

- Fincham, F. D., Cui, M., Braithwaite, S., & Pasley, K. (2008). Attitudes Toward Intimate Partner Violence in Dating Relationships. *Psychological Assessment, 20*(30), 260-269.
- Finkel, E. J., DeWall, C. N., Slotter, E., Oaten, M. B., & Foshee, V. A. (2009). Self-regulatory failure and intimate partner violence perpetration. *Journal of Personality and Social Psychology*, 97, 483-499.
- Fisher, L., & Blair, R. J. (1998). Cognitive impairment and its relationship to psychopathic tendencies in children with emotional and behavioral difficulties. *Journal of Abnormal Child Psychology*, 26(6), 511-519.
- Fite, P. J., Stoppelbein, L., & Greening, L. (2009). Proactive and reactive aggression in a child psychiatric inpatient population: relations to psychopathic characteristics. *Criminal Justice & Behavior, 36*(5), 481-493.
- Fitzgerald, P. B., Oxley, T. J., Laird, A. R., Kulkarni, J., Egan, G. F., & Daskalakis, Z. J. (2006). An analysis of functional neuroimaging studies of dorsolateral prefrontal cortical activity in depression. *Psychiatry Research: Neuroimaging, 148,* 33-45.
- Floden, D., & Stuss, D. T. (2006). Inhibitory control is slowed in patients with right superior medial frontal damage. *Journal of Cognitive Neuroscience, 18,* 1843-1849.
- Flor, H., Birbaumer, N., Hermann, C., Ziegler, S., & Patrick, C. J. (2002). Aversive pavlovian conditioning in psychopaths: Peripheral and central correlates. *Psychophysiology*, *39*, 505-518.
- Flores-Mancilla, L. E., Hernández-González, M., Guevara, M. A., Benavides-Haro, D. E., & Martínez-Arteaga, P. (2014). Long-term fish oil supplementation attenuates seizure activity in the amygdala induced by 3-mercaptopropionic acid in adult male rats. *Epilepsy & Behavior, 33*, 126-134.
- Folstein, J. R., & van Petten, C. (2008). Influence of cognitive control and mismatch on the N2 component of the ERP: a review. *Psychophysiology*, 45, 152-170.

- Fontani, G., Corradeschi, F., Felici, A., Alfatti, F., Migliorini, S., & Lodi, L. (2005a). Cognitive and physiological effects of omega-3 polyunsaturated fatty acid supplementation in healthy subjects. *European Journal of Clinical Investigation*, 35(11), 691-699.
- Fontani, G., Corradeschi, F., Felici, A., Alfatti, F., Bugarini, R., Fiaschi, A. I., ... Berra, B. (2005b). Blood profiles, body fat and mood state in healthy subjects on different diets supplemented with omega-3 polyunsaturated fatty acids. *European Journal of Clinical Investigation*, 35(8), 499-507
- Food and Agriculture Organization of the United Nations. (2010). *Fats and fatty acids in human nutrition: Report of an expert consultation.* Rome, IT: FAO.
- Fox, A. S., Oler, J. A., Tromp, D. P. M., Fudge, J. L., & Kalin, N. H. (2015). Extending the amygdala in theories of threat processing. *Trends in Neurosciences*, 38(5), 319-329.
- Fox, C. J., Iaria, G., & Barton, J. J. S. (2008). Defining the Face Processing Network: Optimization of the Functional Localizer in fMRI. *Human Brain Mapping, 30,* 1637-1651.
- Fox, E. (2008). *Emotion Science: Cognitive and Neuroscientific Approaches to Understanding Human Emotions*. Houndsmills, UK: Palgrave Macmillan.
- Fox, E., Griggs, L., & Mouchlianitis, E. (2007). The Detection of Fear-Relevant Stimuli: Are Guns Noticed as Quickly as Snakes? *Emotion*, 7(4), 691-696.
- Fox, E., Lester, V., Russo, R., Bowles, R. J., Pichler, A., & Dutton, K. (2000).
 Facial Expressions of Emotion: Are Angry Faces Detected More Efficiently? *Cognition and Emotion*, *14*(1), 61-92.
- Freeman, M. P., Hibbeln, J. R., Wisner, K. L., Davis, J. M., Mischoulon, D., Peet, M., ... Stoll, A. L. (2006). Omega-3 fatty acids: evidence basis for treatment and future research in psychiatry. *Journal of Clinical Psychiatry, 67*(12), 1954-1967.

- Freese, J. L., & Amaral, D. G. (2009). Neuroanatomy of the primate amygdala. In P. J. Whalen, & E. A. Phelps (Eds.), *The Human Amygdala*. New York, NY: The Guilford Press.
- Freunberger, R., Klimesch, W., Doppelmayr, M., & Höller, Y. (2007). Visual P2 component is related to theta phase-locking. *Neuroscience Letters*, 426(3), 181-186.
- Frick, P. J. (2003). *The Inventory of Callous-Unemotional Traits*. Unpublished rating scale, The University of New Orleans.
- Frick, P. J., Cornell, A. H., Barry, C. T., Bodin, D., & Dane, H. E. (2003).
 Callous-Unemotional Traits and Conduct Problems in the Prediction of Conduct Problem Severity, Aggression, and Self-Report of Delinquency. *Journal of Abnormal Child Psychology*, *31*(4), 457-470.
- Frick, P. J., & Dickens, C. (2006). Current perspectives on conduct disorder. *Current Psychiatry Reports, 8*(1), 59-72.
- Frick, P. J., & Hare, R. D. (2001). *The Antisocial Processes Screening Device.* Toronto, ON: Multi-Health Systems.
- Frick, P. J., Kimonis, E. R., Dandreaux, D. M., & Farell, J. M. (2003). The 4 year stability of psychopathic traits in non-referred youth. *Behavioral Science* and the Law, 21(6), 713-736.
- Frick, P. J., & Morris, A. S. (2004). Temperament and developmental pathways to conduct problems. *Journal of Clinical Child and Adolescent Psychology, 33*, 54-68.
- Frick, P. J., O'Brien, B. S., Wootton, J. M., & McBurnett, K. (1994). Psychopathy and conduct problems in children. *Journal of Abnormal Psychology*, *103*, 700-707.
- Frick, P. J., & Viding E. (2009). Antisocial behavior from a developmental psychopathology perspective. *Developmental Psychopathology*, 21, 1111-1131.
- Frick, P.J., & White, S.F. (2008). Research Review: The importance of callous unemotional traits for developmental models of aggressive and antisocial behaviour. *Journal of Child Psychology & Psychiatry, 49,* 359-375.

- Friston, K. J., Frith, C. D., Liddle, P. F., & Frackowiak, R. S. (1991). Comparing functional (PET) images: the assessment of significant change. *Journal* of Cerebral Blood Flow and Metabolism, 11(4), 690-699.
- Fullam, R. S., & Dolan, M. C. (2006). Emotional information processing in violent patients with schizophrenia: Association with psychopathy and symptomatology. *Psychiatry Research*, 141, 29-37.
- Fulwiler, C. E., King, J. A., & Zhang, N. (2012). Amygdala-orbitofrontal restingstate functional connectivity is associated with trait anger. *Neuroreport*, 23, 606-610.
- Fusar-Poli, P., Placentino, A., Carletti, F., Landi P., Allen, P., Surguladze, S., ... Politi, P. (2009). Functional atlas of emotional faces processing: a voxel-based meta-analysis of 105 functional magnetic resonance imaging studies. *Journal of Psychiatry Neuroscience, 34*(6), 418-432.
- Gadoth, N. (2008). On fish oil and omega-3 supplementation in children: The role of such supplementation on attention and cognitive dysfunction. *Brain and Development, 30,* 309-312.
- Gailliot, M. T., & Baumeister, R. F. (2007). The physiology of willpower: linking blood glucose to self-control. *Personality and Social Psychology Review, 11*(4), 303-327.
- Gallese, V. (2007). Before and below 'theory of mind': embodied simulation and the neural correlates of social cognition. *Philosophical Transactions of the Royal Society B: Biological Sciences, 362,* 659-669.
- Gao, Y., & Raine, A. (2009). P3 event-related potential impairments in antisocial and psychopathic individuals: a meta-analysis. *Biological Psychology*, *82*, 199-210.
- Garandeau, C., & Cillessen, A. (2006). From indirect aggression to invisible aggression: A conceptual view on bullying and peer group manipulation. *Aggression and Violent Behavior, 11*, 641-654.
- Garavan, H., Hester, R., Murphy, K., Fassbender, C., & Kelly, C. (2006). Individual differences in the functional anatomy of inhibitory control. *Brain Research, 1105,* 130-142.

- Garavan, H., Ross, T. J., Murphy, K., Roche, R. A. P., & Stein, E. A. (2002).
 Dissociable executive functions in the dynamic control of behavior: inhibition, error detection and correction. *NeuroImage*, *17*, 1820-1829.
- Garavan, H., Ross, T., & Stein, E. (1999). Right hemispheric dominance of inhibitory control: An event-related functional MRI study. *Proceedings of the National Academy of Sciences, 96,* 8301-8306.
- García-León, A., Reyes, G. A., Vila, J., Pérez, N., Robles, H., & Ramos, M. M.
 (2002), The Aggression Questionnaire: A validation study in student samples. *The Spanish Journal of Psychology*, *5*(1), 45-53.
- Garland, M. R., & Hallahan, B. (2006). Essential fatty acids and their role in conditions characterised by impulsivity. *International Review of Psychiatry, 18*(2), 99-105.
- Garland, M. R., Hallahan, B., McNamara, M., Carney, P. A., Grimes, H.,
 Hibbeln, J. R., ... Conroy, R. M. (2007) Lipids and essential fatty acids in patients presenting with self-harm. *British Journal of Psychiatry*, *190*, 112-117.
- Garland, M., Sacks, F. M., Colditz, G. A., Rimm, E. B., Sampson, L. A., Willett,
 W., & Hunter, D. J. (1998). The relation between dietary intake and
 adipose tissue composition of selected fatty acids in US women. *American Journal of Nutrition*, *67*, 25-30.
- Gauthier, I., Skudlarski, P., Gore, J. C., & Anderson, A. W. (2000). Expertise for cars and birds recruits brain areas involved in face recognition. *Nature Neuroscience, 3,* 191-197.
- Georgieff, M. K. (2011). Long-term brain and behavioral consequences of early iron deficiency. *Nutrition Reviews, 69,* 43-48.
- Gerdes, A. B. M., Wieser, M. J., Bublatzky, F., Kusay, A., Plichta, M. M., & Alpers, G. W. (2013). Emotional sounds modulate early neural processing of emotional pictures. *Frontiers in Psychology*, *4*, 741.
- Gerstle, J. E., Mathias, C. W., & Stanford, M. S. (1998). Auditory P300 and selfreported impulsive aggression. *Progress in Neuro-Psychopharmacology and Biological Psychiatry, 22,* 575-583.

- Gesch, C. B., Hammond, S. M., Hampson, S. E., Eves, A., & Crowder, M. J.
 (2002) Influence of supplementary vitamins, minerals and essential fatty acids on the antisocial behaviour of young adult prisoners.
 Randomised, placebo-controlled trial. *British Journal of Psychiatry*, 181, 22-28.
- Giancola, P. R., Mezzich, A. C., & Tarter, R. E. (1998). Disruptive, delinquent and aggressive behavior in female adolescents with a psychoactive substance use disorder: relation to executive cognitive functioning. *Journal of Studies on Alcohol and Drugs, 59*(5), 560-567.
- Gilbert, P. (2009). *The compassionate mind: A new approach to life's challenges*. London, UK: Constable & Robinson.
- Gilbert, P., Price, J., & Allan, S. (1995). Social comparison, social attractiveness and evolution: How might they be related? *New Ideas in Psychology, 13,* 149-165.
- Gillies, D., Sinn, J. K., Lad, S. S., Leach, M. J., & Ross, M. J. (2012).
 Polyunsaturated fatty acids (PUFA) for attention deficit hyperactivity disorder (ADHD) in children and adolescents. *Cochrane Database Systematic Review*, *11*, 7.
- Glass, S. J., & Newman, J. P. (2006). Recognition of Facial Affect in Psychopathic Offenders. *Journal of Abnormal Psychology*, *115*(4), 815-820.
- Golden, C. J., Jackson, M. L., Peterson-Rohnem, A., & Gontkovsky, S. T.
 (1996). Neuropsychological correlates of violence and aggression: A review of the clinical literature. *Aggression and Violent Behaviour, 1,* 3-25.
- Gomez-Pinilla, F. (2008). Brain foods: the effects of nutrients on brain function. *Nature Reviews Neuroscience, 9*, 568-578.
- Gonsalves, V. M., McLawsen, J. E., Huss, M. T., & Scalora, M. J. (2013). Factor structure and construct validity of the psychopathic personality inventory in a forensic sample. *International Journal of Law and Psychiatry, 36*, 176-184.

- González-Roldán, A. M., Martínez-Jauand, M., Muñoz-García, M. A., Sitges, C., Cifre, I., & Montoya, P. (2011). Temporal dissociation in the brain processing of pain and anger faces with different intensities of emotional expression. *Pain*, *152*(4), 853-859.
- González-Roldán, A. M., Muñoz, M. A., Cifre, I., Sitges, C., & Montoya, P.
 (2013). Altered Psychophysiological Responses to the View of Others' Pain and Anger Faces in Fibromyalgia Patients. *The Journal of Pain*, *14*(7), 709-719.
- Gordon, J. A., & Malmsjo Moore, P. (2005). ADHD among incarcerated youth: an investigation on the congruency with ADHD prevalence and correlates among the general population. *American Journal of Criminal Justice, 30,* 87-97.
- Gorjão, R., Azevedo-Martins, A. K., Rodrigues, H. G., Abdulkader, F., Arcisio-Miranda, M., Procopio, J., ... Curi, R. (2009). Comparative effects of DHA and EPA on cell function. *Pharmacological Therapy*, *122*, 56-64.
- Gotlib, I. H., & Joormann, J. (2010). Cognition and Depression: Current Status and Future Directions. *Annual Review of Clinical Psychology*, 6, 285-312.
- Goubert, L., Craig, K. D., Vervoort, T., Morley, S., Sullivan, M. J., de, C. W. A., ... Crombez, G. (2005). Facing others in pain: the effects of empathy. *Pain, 118,* 285-288.
- Gow, R. V., Matsudaira, T., Taylor, E., Rubia, K., Crawford, M., Ghebremeskel,
 K., ... Sumich, A. L. (2009). Total red blood cell concentrations of o-3
 fatty acids are associated with emotion-elicited neural activity in
 adolescent boys with attention-deficit hyperactivity disorder.
 Prostaglandins, Leukotrienes and Essential Fatty Acids, 80, 151-156.

Gow, R. V., Sumich, A., Vallee-Tourangeau, F., Crawford, M. A.,
 Ghebremeskel, K., Bueno, A. A., ... Rubia, K. (2013). Omega-3 fatty acids are related to abnormal emotion processing in adolescent boys with attention deficit hyperactivity disorder. *Prostaglandins, Leukotrienes and Essential Fatty Acids, 88*(6), 419-429.

- Gow, R. V., Vallee-Tourangeau, F., Crawford, M. A., Taylor, E., Ghebremeskel, K., Bueno, A. A., ... Rubia, K. (2013). Omega-3 fatty acids are inversely related to callous and unemotional traits in adolescent boys with attention deficit hyperactivity disorder. *Prostaglandins, Leukotrienes, and Essential Fatty Acids, 88*(6), 411-418.
- Goyer, P. F., Anderson, P. J., Semple, W. E., Clayton, A. H., King, A. C.,
 Compton-Toth, B. A., ... Cohen, R. M. (1994). Positron-emission
 tomography and personality disorders. *Neuropsychopharmacology*, *10*, 21-28.
- Grafman, J., Schwab, K., Warden, D., Pridgen, A., Brown, H. R., & Salazar, A.
 M. (1996). Frontal lobe injuries, violence, and aggression: A report of the Vietnam Head Injury Study. *Neurology*, *46*, 1231-1238.
- Grantham-McGregor, S., Cheung, Y. B., Cueto, S., Glewwe, P., Richter, L.,
 Strupp, B., & International Child Development Steering Group. (2007).
 Developmental potential in the first 5 years for children in developing countries. *Lancet, 369*(9555), 60-70.
- Green, P., Glozman, S., Kamensky, B., & Yavin, E. (1999). Developmental changes in rat brain membrane lipids and fatty acids. The preferential prenatal accumulation of docosahexaenoic acid. *Journal of Lipid Research, 40*, 960-966.
- Green, M. F., Penn. D. L., Bentall, R., Carpenter, W. T., Gaebel, W., Gur, R. C.,
 ... Heinssen, R. (2008). Social cognition in schizophrenia: an NIMH
 workshop on definitions, assessment, and research opportunities. *Schizophrenia Bulletin, 34*(6), 1211-1220.
- Greenberg, N., Scott, M., & Crews, D. (1984). Role of the amygdala in the reproductive and aggressive behavior of the lizard, Anolis carolinensis. *Physiological Behavior, 32,* 147-151.
- Gregg, T. R., & Siegel, A. (2001). Brain structures and neurotransmitters regulating aggression in cats: implications for human aggression. *Progress in Neuro-Psychopharmacology and Biological Psychiatry, 25,* 91-140.

- Gross, J. (2007). *Handbook of emotion regulation.* New York, NY: Guilford Press.
- Guelker, M. D., Barry, C. T., Barry, T. D., & Malkin, M. L. (2014). Perceived positive outcomes as a mediator between adolescent callousunemotional traits and antisocial behavior. *Personality and Individual Differences, 69,* 129-134.
- Guerra, N. G., & Huesmann, L. R. (2004). A cognitive-ecological model of aggression. *Revue Internationale de Psychologie Sociale, 2,* 177-204.
- Gustafsson, P. A., Birberg-Thornberg, U., Duchén, K., Landgren, M., Malmberg,
 K., Pelling, H., ... Karlsson, T. (2010). EPA supplementation improves
 teacher-rated behaviour and oppositional symptoms in children with
 ADHD. Acta Paediatrica, 99(10), 1540-1549.
- Halas, E. S., Reynolds, G. M., & Sandstead, H. H. (1997). Intra-uterine nutrition and its effects on aggression. *Physiology and Behavior, 19,* 653-661.
- Halgren, E., Baudena, P., Heit, G., Clarke, J. M., Marinkovic, K., & Clarke, M. (1994). Spatio-temporal stages in face and word processing. 1. Depth recorded potentials in the human occipital and parietal lobes. *Journal of Physiology*, *88*, 1-50.
- Halgren, E., Marinkovic, K., & Chauvel, P. (1998). Generators of the late cognitive potentials in auditory and visual oddball tasks. *Electroencephalography and Clinical Neurophysiology, 106*(2), 156-164.
- Halgren, E., Raij, T., Marinkovic, K., Jousmäki, V., & Hari, R. (2000). Cognitive response profile of the human fusiform face area as determined by MEG. *Cerebral Cortex*, *10*, 69-81.
- Hallahan, B., & Garland, M. R. (2004). Essential fatty acids and their role in the treatment of impulsive disorders. *Prostaglandins, Leukotrienes and Essential Fatty Acids, 71,* 211-216.
- Hallahan, B., Hibbeln, J. R., Davis, J. M., & Garland, M. R. (2007). Omega-3 fatty acid supplementation in patients with recurrent self-harm: Single-centre double-blind randomised controlled trial. *British Journal of Psychiatry, 190*, 118-122.

- Hamazaki, T., & Hamazaki, K. (2008). Fish oils and aggression or hostility. *Progress in Lipid research, 47,* 221-232.
- Hamazaki, K., Hamazaki, T., & Inadera, H. (2012). Fatty acid composition in the postmortem amygdala of patients with schizophrenia, bipolar disorder, and major depressive disorder. *Journal of Psychiatric Research, 46,* 1024-1028.
- Hamazaki, T., & Hirayama, S. (2004). The effect of docosahexaenoic acidcontaining food administration on symptoms of attentiondeficit/hyperactivity disorder - a placebo-controlled double-blind study. *European Journal of Clinical Nutrition, 58*, 838.
- Hamazaki, T., Sawazaki, S., Itomura, M., Asaoka, E., Nagao, Y., Nishimura, N., Kobayashi, M. (1996). The effect of docosahexaenoic acid on aggression in young adults. A placebo-controlled double-blind study. *Journal of Clinical Investigations*, *97*, 1129-1133.
- Hamilton, J. P., Siemer, M., & Gotlib, I. H. (2008). Amygdala volume in major depressive disorder: a meta-analysis of magnetic resonance imaging studies. *Molecular Psychiatry*, *13*(11), 993-1000.
- Hansen, B. C., Farivar, R., Thompson, B., & Hess, R. F. (2008). A critical band of phase alignment for discrimination but not recognition of human faces. *Vision Research*, 48, 2523-2536.
- Hansen, C. H., & Hansen, R. D. (1988). Finding the face in the crowd: An anger superiority effect. Journal of Personality and Social Psychology, 54, 917-924.
- Hare, R. D. (1991). *The Hare Psychopathy Checklist-Revised.* Toronto, AL: Multi-Health Systems.
- Hare, R. D. (2003). *The Hare Psychopathy Checklist-Revised, 2nd Edition*. Toronto, ON: Multi- Health Systems.
- Hare, R. D., & Neumann, C. S. (2006). The PCL-R assessment of psychopathy: Development, structural properties, and new directions. In C. Patrick (Ed.), *Handbook of Psychopathy* (pp. 58-90). New York, NY: The Guilford Press.

- Hare, R. D., & Neumann, C. S. (2008). Psychopathy as a clinical and empirical construct. *Annual Review of Clinical Psychology*, *4*, 217-246.
- Hare, T. A., Tottenham, N., Davidson, M. C., Glover, G. H., & Casey, B. J. (2005). Contributions of amygdala and striatal activity in emotion regulation. *Biological Psychiatry*, *57*, 624-632.
- Harmon-Jones, E., Barratt, E. S., & Wigg, C. (1997). Impulsiveness, aggression, reading, and the P300 of the event-related potential.
 Personality and Individual Differences, 22, 439-445.
- Harpending, H. C., & Sobus, J. (1987). Sociopathy as an adaptation. *Ethology and Sociobiology*, *8*(1), 63-72.
- Harpur, T. J., Hare, R. D., & Hakstian, A. R. (1989). Two-factor conceptualization of psychopathy: Construct validity and assessment implications. *Psychological Assessment: A Journal of Consulting and Clinical Psychology*, 1(1), 6-17.
- Hart, S. D., & Dempster, R. J. (1997). Impulsivity and psychopathy. In C. D.
 Webster & M. A. Jackson (Eds.), *Impulsivity: Theory, assessment, and treatment* (pp. 212-232). New York, NY: Guilford.
- Hartley, C. A., & Phelps, E. A. (2009). Changing Fear: The Neurocircuitry of Emotion Regulation. *Neuropsychopharmacology*, *35*(1), 136-146.
- Hastings, P. D., Nuselovici, J. N., Utendale, W. T., Coutya, J., McShane, K. E.,
 & Sullivan, C. (2008). Applying the polyvagal theory to children's emotion regulation: Social context, socialization, and adjustment. *Biological Psychology*, *79*, 299-306.
- Hawes, D. J., Brennan, J., & Dadds, M. R. (2009). Cortisol, callous-unemotional traits, and pathways to antisocial behavior. *Current Opinion in Psychiatry, 22*, 357-362.
- Haxby, J. V., Gobbini, M. I., Furey, M. L., Ishai, A., Schouten, J. L., & Pietrini, P. (2001). Distributed and overlapping representations of faces and objects in ventral temporal cortex. *Science*, *293*, 2425-2430.
- Haxby, J. V., Hoffman, E. A., & Gobbini, M. I. (2000). The distributed human neural system for face perception. *Trends in Cognitive Science*, *46*, 223-233.

- Haxby, J. V., Hoffman, E. A., & Gobbini, M. I. (2002). Human neural systems for face recognition and social communication. *Biological Psychiatry*, 51, 59-67.
- He, C., Qu, X., Cui, L., Wang, J., & Kang, J. X. (2009). Improved spatial learning performance of fat-1 mice is associated with enhanced neurogenesis and neuritogenesis by docosahexaenoic acid. *Proceedings of the National Academy of Sciences: USA, 106,* 11370-11375.
- Heinze, H. J., Münte, T. F., Gobiet, W., Niemann, H., & Ruff, R. M. (1992).
 Parallel and serial visual search after closed head injury: electrophysiological evidence for perceptual dysfunctions. *Neuropsychologia, 30*(6), 495-514.
- Heiser, M., Iacoboni, M., Maeda, F., Marcus, J., & Mazziotta, J. C. (2003). The essential role of Broca's area in imitation. *European Journal of Neuroscience*, *17*, 1123-1128.

Helmut, L. (2003). Fear and trembling in the amygdala. Science, 300, 568-569.

- Herrmann, M. J., Aranda, D., Ellgring, H., Mueller, T. J., Strik, W. K., Heidrich, A., ... Fallgatter, A. J. (2002). Face-specific event-related potential in humans is independent from facial expression. *International Journal of Psychophysiology*, 45, 241-244.
- Herrmann, C. S., Busch, N. A., & Grigutsch, M. (2005). EEG oscillations and wavelet analysis. In: Handy, T. (Ed.), *Event-related potentials: a methods handbook* (pp. 229-260). Cambridge, MA: MIT Press.
- Hervé, H. F., Hayes, P. J., & Hare, R. D. (2003). Psychopathy and sensitivity to the emotional polarity of metaphorical statements. *Personality and Individual Differences*, 35(7), 1497-1507.
- Hess, W. R. (1928). Stammganglien-Reizversuche. Berichte über die gesamte. *Physiologie und experimentelle Pharmakologie, 42*, 554–555.
- Hewig, J., Trippe, R., Hecht, H., Coles, M. G., Holroyd, C. B., & Miltner, W. H.,(2007). Decision-making in Blackjack: an electrophysiological analysis. *Cerebral Cortex, 17*, 865-877.

- Hiatt, K. D., Lorenz, A. R., & Newman, J. P. (2002). Assessment of emotion and language processing in psychopathic offenders: Results from a dichotic listening task. *Personality and Individual Differences*, *32*, 1255-1268.
- Hibbeln, J. R. (2001). Seafood consumption and homicide mortality. A crossnational ecological analysis. World Review of Nutrition & Dietetics, 88, 41-46.
- Hibbeln, J. R. (2002). Seafood consumption, the DHA content of mothers' milk and prevalence rates of postpartum depression: A cross-national, ecological analysis. *Journal of Affective Disorders, 69*(1-3), 15-29.
- Hibbeln, J. R., Bissette, G., Umbau, J. C., & George, D. T. (2004). Omega-3 status and cerebrospinal fluid corticotropin releasing hormone in perpetrators of domestic violence. *Biological Psychiatry*, *56*, 895-897.
- Hibbeln, J. R., Ferguson, T. A., & Blasbalg, T. L. (2006). Omega-3 fatty acid deficiencies in neurodevelopment, aggression and autonomic dysregulation: opportunities for intervention. *International Review of Psychiatry, 18*(2), 107-118.
- Hibbeln, J. R., & Gow, R. V. (2014). The potential for military diets to reduce depression, suicide, and impulsive aggression: a review of current evidence for omega-3 and omega-6 fatty acids. *Military Medicine*, *179*(11), 117-128.
- Hibbeln, J. R., & Makino, K. (2002). Omega-3 fats in depressive disorders and violence: the context of evolution and cardiovascular health. In E. R. Skinner (Ed). *Brain Lipids and Disorders in Biological Psychiatry, vol. 35* (pp. 67-111). Amsterdam: Elsevier Science.
- Hibbeln, J. R., Nieminen, L. R., & Lands, W. E. (2004). Increasing homicide rates and linoleic acid consumption among five Western countries, 1961-2000. *Lipids*, 39(12), 1207-1213.
- Hibbeln, J. R., Umhau, J. C., George, D. T., & Salem Jr, N. (1997). Do plasma polyunsaturates predict hostility and depression? World Review of Nutrition & Dietetics, 82, 175–186.

- Hicks, B. M., Bernat, E., Malone, S. M., Iacono, W. G., Patrick, C. J., Krueger,
 R. F., & McGue, M. (2007). Genes mediate the association between P3 amplitude and externalizing disorders. *Psychophysiology*, 44, 98-105.
- Hill, D., & Sargant, W. (1943). A case of matricide. Lancet, 241, 526-527.
- Hillyard, S. A., & Kutas, M. (1983). Electrophysiology of cognitive processing. *Annual Review of Psychology, 34,* 33-61.
- Hillyard, S. A., Mangun, G. R., Woldorff, M. G., & Luck, S. J. (1995). Neural systems mediating selective attention. In M. S. Gazzaniga (Ed.), *The Cognitive Neurosciences* (pp. 665-681). Cambridge, MA: MIT Press.
- Hinojosa, J. A., Mercado, F., & Carretié, L. (2015). N170 sensitivity to facial expression: A meta-analysis. *Neuroscience and Biobehavioral Review*, 55, 498-509.
- Hirayama, S., Hamazaki, T., & Terasawa, K. (2004) Effect of Docosahexaenoic Acid-Containing Food Administration on Symptoms of Attention-Deficit/Hyperactivity Disorder- A Placebo-Controlled Double-Blind Study. *European Journal of Clinical Nutrition, 58,* 467-473.
- Hoaken, P. N., Allaby, D. B., & Earle, J. (2007). Executive cognitive functioning and the recognition of facial expressions of emotion in incarcerated violent offenders, non-violent offenders, and controls. *Aggressive Behavior, 33*(5), 412-421.
- Hoffman, M. L. (1987). The contribution of empathy to justice and moral judgment. In N. Eisenberg & J. Strayer (Eds.), *Empathy and its development* (pp. 47-80). New York, NY: Cambridge University Press.
- Hoffman, M. L. (2000). *Empathy and moral development: Implications for caring and justice.* Cambridge, UK: Cambridge University Press.
- Hoffmann, P., Taube, C., Forster, W., Weinert, D., Grosse, W.R., Balschun, D.,
 & Schuh, J. (1986). Circadian rhythm of eicosanoid formation as affected by dietary linoleate. *Prostaglandins, Leukotrienes and Medicine, 25*, 91-103.
- Holmes, A., Kiss, M., & Eimer, M. (2006). Attention modulates the processing of emotional expression triggered by foveal faces. *Neuroscience Letters*, 394, 48-52.

- Holmes, A., Mogg, K., de Fockert, J., Nielsen, M. K., & Bradley, B. P. (2014).
 Electrophysiological evidence for greater attention to threat when cognitive control resources are depleted. *Cognitive, Affective, & Behavioral Neuroscience, 14*(2), 827-35.
- Holmes, A., Winston, J. S., & Eimer, M. (2005). The role of spatial frequency information for ERP components sensitive to faces and emotional facial expression. *Cognitive Brain Research, 25*, 508-520.
- Holroyd, C. B., & Coles, M. G. (2002). The neural basis of human error processing: reinforcement learning, dopamine, and the error-related negativity. *Psychology Review*, *109*, 679-709.
- Holroyd, C. B., & Coles, M. G. (2008). Dorsal anterior cingulate cortex integrates reinforcement history to guide voluntary behavior. *Cortex, 44,* 548-559.
- Holroyd, C. B., Hajcak, G., & Larsen, J. T. (2006). The good, the bad and the neutral: electrophysiological responses to feedback stimuli. *Brain Research, 1105,* 93-101.
- Holroyd, C. B., Pakzad-Vaezi, K. L., & Krigolson, O. E. (2008). The feedback correct-related positivity: sensitivity of the event-related brain potential to unexpected positive feedback. *Psychophysiology*, 45, 688-697.
- Holtzman, N. S. (2011). Facing a psychopath: Detecting the dark triad from emotionally-neutral faces, using prototypes from the Personality Faceaurus. *Journal of Research in Personality, 45*, 648-654.
- Holub, B. (2002). Clinical nutrition: Omega-3 fatty acids in cardiovascular care. *Canadian Medical Association Journal, 166,* 608-615.
- Hoptman, M. J. (2003). Neuroimaging studies of violence and antisocial behavior. *Journal of Psychiatric Practice, 9,* 265-278.
- Hoptman, M. J., D'Angelo, D., Catalano, D., Mauro, C. J., Shehzad, Z. E., Kelly,
 A. M., ... Milham, M. P. (2010). Amygdalofrontal functional disconnectivity and aggression in schizophrenia. *Schizophrenia Bulletin,* 36(5), 1020-1028.

- Horley, K., Gonsalvez, C., Williams, L., Lazzaro, I., Bahramali, H., & Gordon, E.
 (2001). Event-related potentials to threat related faces in schizophrenia.
 International Journal of Neuroscience, *107*, 113-130.
- Horn, N. R., Dolan, M., Elliott, R., Deakin, J. F. W., & Woodruff, P. W. R. (2003).
 Response inhibition and impulsivity: an fMRI study. *Neuropsychologia*, *41*(14), 1959-1966.
- Horovitz, S. G., Rossion, B., Skudlarski, P., & Gore, J.C. (2004). Parametric design and correlational analyses help integrating fMRI and electrophysiological data during face processing. *NeuroImage, 22,* 1587-1595.
- Horstmann, G., & Bauland, A. (2006). Search asymmetries with real faces: Testing the anger- superiority effect. *Emotion, 6*, 193 – 207.
- Hrboticky, N., MacKinnon, M. J., & Innis, S. M. (1990). Effect of vegetable oil formula rich in linoleic acid on tissue fatty acid accretion in brain, liver, plasma and erythrocytes of infant piglets. *American Journal of Clinical Nutrition, 51,* 173-182.
- Huddy, V., Schweinberger, S., Jentzsch, I., & Burton, A. M. (2003). Matching faces for semantic information and names: an event related brain potentials study. *Cognitive Brain Research*, 17, 314-326.
- Huesmann, L. R. (1988). An information processing model for the development of aggression. *Aggressive Behavior, 14,* 13-24.
- Huesmann, L. R. (1998). The role of social information processing and cognitive schemas in the acquisition and maintenance of habitual aggressive behavior. In R. G. Geen & E. Donnerstein (Eds.), *Human aggression: Theories, research, and implications for policy* (pp. 73-109). New York, NY: Academic Press.
- Huesmann, L. R., & Eron, L. D. (1986). *Television and the aggressive child: A cross-national comparison*. Hillsdale, NJ: Erlbaum.
- Hughes, M. E., Fulham, W. R., Johnston, P. J., & Michie, P. T. (2012). Stopsignal response inhibition in schizophrenia: Behavioural, event-related potential and functional neuroimaging data. *Biological Psychology*, *89*, 220-231.

- Hunter, D. J., Rimm, E. B., Sacks, F. M., Stampfer, M., Colditz, G. A., Litin, L.
 B., & Willett, W. (1992). Comparison of measures of fatty acid intake by subcutaneous fat aspirate, food frequency questionnaire, and diet records in a free-living population of US men. *American Journal of Epidemiology*, *135*, 418-427.
- Huster, R. J., Enriquez-Geppert, S., Lavallee, C. F., Falkenstein, M., & Hermann, C. S. (2013). Electroencephalography of response inhibition tasks: Functional networks and cognitive contributions. *International Journal of Psychophysiology*, *87*, 217-233.
- Huster, R. J., Pils, S. M., Lavallee, C. F., Calhoun, V. D., & Hermann, C. S. (2014). Functional and effective connectivity of stopping. *NeuroImage*, 94, 120-128.
- Hyde, L. W., Shaw, D. S., & Hariri, A. R. (2013). Understanding youth antisocial behavior using neuroscience through a developmental psychopathology lens: Review, integration, and directions for research. *Developmental Review, 33,* 168-223.
- Iacono, W. G., Carlson, S. R., Malone, S. M., & McGue, M. (2002). P3 eventrelated potential amplitude and the risk for disinhibitory disorders in adolescent boys. *Archives of General Psychiatry*, 59, 750-757.
- Iacono, W. G., Malone, S. M., & McGue, M. (2003). Substance use disorders, externalizing psychopathology, and P300 event-related potential amplitude. *International Journal of Psychophysiology*, 48, 147-178.
- Ibáñez, A., Riveros, R., Hurtado, E., Gleichgerrcht, E., Urquina, H., Herrera, E.,
 ... Manes, F. (2012). The face and its emotion: Right N170 deficits in structural processing and early emotional discrimination in schizophrenic patients and relatives. *Psychiatry Research, 195,* 18-26.
- Innis, S. M. (2000). The role of dietary n-6 and n-3 fatty acids in the developing brain. *Developmental Neuroscience, 22,* 474-480.
- Innis, S. M. (2007). Fatty acids and early human development. *Early Human Development*, *83*(12), 761-766.
- Innis, S. M. (2008). Dietary omega 3 fatty acids and the developing brain. *Brain Research, 1237,* 35-43.

- Ioannides, A. A., Liu, L. C., Kwapien, J., Drozdz, S., & Streit, M. (2000). Coupling of regional activations in a human brain during an object and face affect recognition task. *Human Brain Mapping*, *11*, 77-92.
- Iria, C., & Barbosa, F. (2009). Perception of facial expressions of fear: Comparative research with criminal and non-criminal psychopaths. *Journal of Forensic Psychiatry & Psychology, 20*, 67-73.
- Iribarren, C., Markovitz, J. H., Jacobs, D. R. Jr, Schreiner, P. J., Daviglus, M., & Hibbeln, J. R. (2004). Dietary intake of n-3, n-6 fatty acids and fish: relationship with hostility in young adults--the CARDIA study. *European Journal of Clinical Nutrition, 58*(1), 24-31.
- Itier, R. J., & Taylor, M. J. (2002). Inversion and contrast polarity reversal affect both encoding and recognition processes of unfamiliar faces: A repetition study using ERPs. *Neuroimage*, *15*, 353-372.
- Itier, R. J., & Taylor, M. J. (2004). N170 or N1? Spatiotemporal differences between object and face processing using ERPs. *Cerebral Cortex, 14,* 132-142.
- Itier, R. J., Herdman, A. T., George, N., Cheyne, D., & Taylor, M. J. (2006). Inversion and contrast-reversal effects on face processing assessed by MEG. *Brain Research*, *1115*, 108-120.
- Ito, T. A., Larsen, J. T., Smith, N. K., & Cacioppo, J. T. (1998). Negative information weighs more heavily on the brain: The negativity bias in evaluative categorizations. *Journal of Personality and Social Psychology, 75,* 887-900.
- Itomura, M., Hamazaki, K., Sawazaki, S., Kobayashi, M., Terasawa, K., Watanabe, S., & Hamazaki, T. (2005). The effect of fish oil on physical aggression in school children - a randomized, double-blind, placebocontrolled. *Journal of Nutritional Biochemistry, 16*(3), 163-171,
- Izard, C. E. (2007). Basic emotions, natural kinds, emotion schemas, and a new paradigm. *Perspectives on Psychological Science, 2*(3), 260-280.
- Izquierdo, A., & Murray, E. A. (2004). Combined unilateral lesions of the amygdala and orbital prefrontal cortex impair affective processing in rhesus monkeys. *Journal of Neurophysiology*, *91*, 2023-2039.

- Izquierdo, A., Suda, R. K., & Murray, E. A. (2004). Bilateral orbital prefrontal cortex lesions in rhesus monkeys disrupt choices guided by both reward value and reward contingency. *Journal of Neuroscience, 24,* 7540-7548.
- Izquierdo, A., Suda, R. K., & Murray, E. A. (2005). Comparison of the effects of bilateral orbital prefrontal cortex lesions and amygdala lesions on emotional responses in rhesus monkeys. *Journal of Neuroscience, 25,* 8534-8542.
- Jabbi, M., Swart, M., & Keysers, C. (2007). Empathy for positive and negative emotions in the gustatory cortex. *Neuroimage*, *34*, 1744-1753.
- Jacques, C., d'Arripe, O., & Rossion, B. (2007). The time course of the inversion effect during individual face discrimination. *Journal of Vision, 7*, 3.
- James, S., Montgomery, P., & Williams, K. (2011). Omega-3 fatty acids supplementation for autism spectrum disorders (ASD). Cochrane Database of Systematic Reviews 2011, 11.
- Janak, P. H., & Tye, K. M. (2015). From circuits to behaviour in the amygdala. *Nature, 517,* 284-292.
- Jeffreys, D. A. (1989). A face-responsive potential recorded from the human scalp. *Experimental Brain Research*, *78*, 193-202.
- Jeffreys, D. (1996). Evoked potential studies of face and object processing. Visual *Cognition, 3,* 1-38.
- Jezior, K. L., McKenzie, M. E., & Lee, S. S. (2015). Narcissism and Callous-Unemotional Traits Prospectively Predict Child Conduct Problems. *Journal of Clinical Child & Adolescent Psychology.* Advance online publication. Retrieved from

http://www.tandfonline.com/doi/abs/10.1080/15374416.2014.982280

Jodo, E., & Kayama, Y. (1992). Relation of a negative ERP component to response inhibition in a go/Nogo task. *Electroencephalography and Clinical Neurophysiology, 82*, 477-482.

- John, O. P., & Gross, J. J. (2004). Healthy and unhealthy emotion regulation: personality processes, individual differences, and life span development. *Journal of Personality*, *72*(6), 1301-1333.
- Johnson, M., Ostlund, S., Fransson, G., Kadesjo, B., & Gillberg, C. (2009).
 Omega-3/omega-6 fatty acids for attention deficit hyperactivity disorder:
 A randomized placebo-controlled trial in children and adolescents. *Journal of Attention Disorders, 12*(5), 394-401.
- Johnstone, S. J., Dimoska, A., Smith, J. L., Barry, R. J., Pleffer, C. B., Chiswick,
 D., & Clarke, A. R. (2007). The development of stop-signal and go/nogo response inhibition in children aged 7-12 years: Performance and event-related potential indices. *International Journal of Psychophysiology, 63*(1), 25-38.
- Jonas, J., Descoins, M., Koessler, L., Colnat-Coulbois, S., Sauvée, M., Guye, M., ... Maillard, L. (2012). Focal electrical intracerebral stimulation of a face-sensitive area causes transient prosopagnosia. *Neuroscience,* 222, 281-288.
- Jones, A. P., Happé, F. G. E., Gilbert, F., Burnett, S., & Viding, E. (2010). Feeling, caring, knowing: different types of empathy deficit in boys with psychopathic tendencies and autism spectrum disorder. *Journal of Child Psychology and Psychiatry*, *51*(11), 1188-1197,
- Jones, A. P., Laurens, K., Herba, C., Barker, G., & Viding, E. (2009). Amygdala hypoactivity to fearful faces in boys with conduct problems and callousunemotional traits. *American Journal of Psychiatry, 166*, 95-102.
- Joyal, C. C., Dubreucq, J., Gendron, C., & Millaud, F. (2007). Major Mental Disorders and Violence: a Critical Update. *Current Psychiatry Reviews, 3*, 33-50.
- Joyce, C., & Rossion, B. (2005). The face-sensitive N170 and VPP components manifest the same brain processes: The effect of reference electrode site. *Clinical Neurophysiology*, *116*, 2613–2631.
- Jump, D. B. (2002). Dietary polyunsaturated fatty acids and regulation of gene transcription. *Current Opinion in Lipidology, 13,* 155-164.

- Jump, D. B. (2004). Fatty acid regulation of gene transcription. *Critical Reviews in Clinical Laboratory Sciences, 41*, 41-78.
- Jumpsen, J., & Clandinin, M.T. (1997). *Brain Development: Relationship to Dietary Lipid and Lipid Metabolism.* Champaign, IL: AOCS Press.
- Jung, T. P., Makeig, S., Westerfield, W., Townsend, J., Courchesne, E., & Sejnowski, T. J. (2000). Removal of eye activity artifacts from visual event-related potentials in normal and clinical subjects. *Clinical Neurophysiology*, *111*(10), 1745-1758.
- Kanske, P., & Kotz, S. A. (2007). Concreteness in emotional words: ERP evidence from a hemifield study. *Brain Research, 1148*, 138-148.
- Kanske, P., & Kotz, S. A. (2010a). Emotion triggers executive attention: ACC and amygdala responses to emotional words in a conflict task. *Human Brain Mapping, 32*(2), 198-208.
- Kanske, P., & Kotz, S. A. (2010b). Modulation of early conflict processing: N200 responses to emotional words in a flanker task. *Neuropsychologia*, 48, 3661-3664.
- Kappenman, E. S., & Luck, S. J. (2012). ERP Components: The ups and downs of brainwave recording. In S. J. Luck, & E. S. Kappenman (Eds.), *The Oxford handbook of event-related potential components* (pp. 3-30).
 Oxford, UK: Oxford University Press.
- Karr, J. E., Alexander, J. E., & Winningham, R. G. (2011). Omega-3 polyunsaturated fatty acids and cognition throughout the lifespan: a review. *Nutrition Neuroscience*, 14, 216-225.
- Karr, J. E., Grindstaff, T. R., & Alexander, J. E. (2012). Omega-3 polyunsaturated fatty acids and cognition in a college-aged population. *Experimental and Clinical Psychopharmacology, 20*(3), 236-242.
- Kassinove, H., & Sukhodolsky, D. G. (1995). Anger disorders: Basic science and practical issues. In H. Kassinove (Ed.), *Anger disorders: Definitions, diagnosis and treatment* (pp. 1-26). Washington, DC: Taylor & Francis.

- Keil, A., Bradley, M. M., Hauk, O., Rockstroh, B., Elbert, T., & Lang, P. J. (2002). Large-scale neural correlates of affective picture processing. *Psychophysiology*, 39, 641-649.
- Kelloway, E. K., & Barling, J. (1990). Item content versus item wording: Disentangling role conflict and role ambiguity. *Journal of Applied Psychology*, *75*(6), 738–742.
- Kelly, S., & Dunbar, R. I. M. (2001). Who dares wins: Heroism versus altruism in women's mate choice. *Human Nature*, *12*, 89-105.
- Kennedy, D. O., Jackson, P. A., Elliott, J. M., Scholey, A. B., Robertson, B. C., Greer, J., ... Haskell, C. F. (2009), Cognitive and mood effects of 8 weeks' supplementation with 400 mg or 1000 mg of the omega-3 essential fatty acid docosahexaenoic acid (DHA) in healthy children aged 10-12 years. *Nutritional Neuroscience, 12,* 48-56.
- Kerns, J. G., Cohen, J. D., MacDonald 3rd, A. W., Cho, R. Y., Stenger, V. A., & Carter, C.S. (2004). Anterior cingulate conflict monitoring and adjustments in control. *Science*, *303*, 1023-1026.
- Kessler, R. C., Adler, L., Ames, M., Delmer, O., Faraone, S., Hiripi, E., ...
 Walters, E. E. (2005). The World Health Organization Adult ADHD Self-Report Scale (ASRS): a short screening scale for use in the general population. *Psychological Medicine*, *35*, 245-256.
- Kessler, R. C., Adler, L., Barkley, R., Biederman, J., Conners, C. K., Demler, O.,
 ... Zaslavsky, A. M. (2006) The prevalence and correlates of adult
 ADHD in the United States: results from the National Comorbidity
 Survey Replication. *American Journal of Psychiatry*, *163*, 716-723.
- Kiehl, K. A. (2006). A cognitive neuroscience perspective on psychopathy: Evidence for paralimbic system dysfunction. *Psychiatry Research*, 142, 107-128.
- Kiehl, K. A., Bates, A. T., Laurens, K. R., Hare, R. D., & Liddle, P. F. (2006).
 Brain potentials implicate temporal lobe abnormalities in criminal psychopaths. *Journal of Abnormal Psychology*, *115*(3), 443-453.

- Kiehl, K. A., Hare, R. D., Liddle, P. F., & McDonald, J. J. (1999). Reduced P300 responses in criminal psychopaths during a visual oddball task. *Biological Psychiatry*, 45, 1498-1507.
- Kiehl, K. A., Smith, A. M., Hare, R. D., & Liddle, P. F. (2000). An event-related potential investigation of response inhibition in schizophrenia and psychopathy. *Biological Psychiatry*, 48, 210-221.
- Kim, Y. Y., & Jung, Y. S. (2014). Reduced frontal activity during response inhibition in individuals with psychopathic traits: An sLORETA study. *Biological Psychology*, *97*, 49-59.
- Kimonis, E. R., Branch, J., Hagman, B., Graham, N., & Miller, C. (2013). The psychometric properties of the Inventory of Callous-Unemotional Traits in an undergraduate sample. *Psychological Assessment*, 25(1), 84-93.
- Kimonis, E. R., Frick, P. J., Skeem, J. L., Marsee, M. A., Cruise, K., Munoz, L. C., ... & Morris, A. S. (2008). Assessing callous-unemotional traits in adolescent offenders: Validation of the Inventory of Callous-Unemotional Traits. *International Journal of Law and Psychiatry, 31,* 241-252.
- King, S., Waschbusch, D. A., Pelham, W. E., Frankland, B. W., Corkum, P. V., & Jacques, S. (2009). Subtypes of Aggression in Children with Attention Deficit Hyperactivity Disorder: Medication Effects and Comparison with Typical Children. *Journal of Clinical Child & Adolescent Psychology, 38*(5), 619-629.
- Kitajka, K., Sinclair, A. J., Weisinger, R. S., Weisinger, H. S., Mathai, M., Jayasooriya, A. P., ... Puskas, L. G. (2004). Effects of dietary omega-3 polyunsaturated fatty acids on brain gene expression. *Proceedings of the National Academy of Sciences: USA, 101,* 1093-1096.
- Klimesch, W., Hanslmayr, S., Sauseng, P., Gruber, W., Brozinsky, C. J., Kroll,
 N. E. A., ... Doppelmayr, M. (2006). Oscillatory EEG correlates of
 episodic trace decay. *Cerebral Cortex*, *16*, 280-290.
- Knutsen, S. F., Fraser, G. E., Beeson, W. L., Lindsted, K. D., & Shavlik, D. J.(2003). Comparison of adipose tissue fatty acids with dietary fatty acids as measured by 24-hour recall and food frequency questionnaire in

black and white Adventists: the Adventist Health Study. *Annals of Epidemiology, 13,* 119-127.

- Kok, A. (1986). Effects of degradation of visual stimulation on components of the event-related potential (ERP) in go/nogo reaction tasks. *Biological Psychology, 23,* 21-38.
- Kok, A., Ramautar, J. R., de Ruiter, M. B., Band, G. P., & Ridderinkhof, K. R.
 (2004). ERP components associated with successful and unsuccessful stopping in a stop-signal task. *Psychophysiology*, *41*, 9-20.
- Kolakowska, A., Zienkowicz, L., Domiszewski, Z., & Bienkiewicz, G. (2006).
 Lipid charges and sensory quality of whole and gutted rainbow trout during storage in ice. *Acta Ichthyologica et Piscatoria, 36,* 39-47.
- Konishi, S., Nakajima, K., Uchida, I., Kikyo, H., Kameyama, M., & Miyashita, Y. (1999). Common inhibitory mechanism in human inferior prefrontal cortex revealed by event-related fMRI. *Brain*, *122*, 981-991.
- Kopp, B., Rist, F., & Mattler, U. (1996). N200 in the flanker task as a neurobehavioral tool for investigating executive control *Psychophysiology*, 33(3), 282-294.
- Kosson, D. S., Suchy, Y., Mayer, A. R., & Libby, J. (2002). Facial affect recognition in criminal psychopaths. *Emotion, 2*(4), 398-411.
- Kovacsics, C. E., & Gould, T. D. (2010). Shock-induced aggression in mice is modified by lithium. *Pharmacology, Biochemistry and Behavior, 94,* 380-386.

Krämer, U. M., Jansma, H., Tempelmann, C., & Münte, T. F. (2007). Tit-for-tat: the neural basis of reactive aggression. *Neuroimage, 38*, 203-211.

- Krämer, U. M., Knight, R. T., & Münte, T. F. (2011). Electrophysiological evidence for different inhibitory mechanisms when stopping or changing a planned response. *Journal of Cognitive Neuroscience, 23*, 2481-2493.
- Kris-Etherton, P. M., Harris, W. S., & Appel, L. J. (2002). Fish consumption, fish oil, omega-3 fatty acids, and cardiovascular disease. *Circulation*, 106(21), 2747-2757.

- Kris-Etherton, P. M., Taylor, S. T., Yu-Poth, S., Huth, P., Moriarty, K., Fishell, V., ... Etherton, T. D. (2000). *American Journal of Clinical Nutrition*, 71(1), 179-188.
- Krischer, M. K., & Sevecke, K. (2008). Early traumatization and psychopathy in female and male juvenile offender. *International Journal of Law and Psychiatry*, 31, 253-262.
- Kristjánsson, A., & Jóhannesson, O. I. (2014). How priming in visual search affects response time distributions: analyses with ex-Gaussian fits. *Attention, Perception and Psychophysiology, 76*(8), 2199-2211.
- Krolak-Salmon, P., Fischer, C., Vighetto, A., & Mauguiere, F. (2001).
 Processing of facial emotional expression: spatio-temporal data as assessed by scalp event-related potentials. *European Journal of Neuroscience, 13*, 987-994.
- Krombholz, A., Schaefer, F., & Boucsein, W. (2007). Modification of N170 by different emotional expression of schematic faces. *Biological Psychology, 76,* 156-162.
- Krueger, R. F. (1999). The structure of common mental disorders. *Archives of General Psychiatry, 56,* 921-926.
- Kruh, I. P., Frick, P., J., & Clements, C. B. (2005). Historical and personality correlates to the violence patterns of juveniles tried as adults. *Criminal Justice and Behavior, 32*, 69-96.
- Kruk, M. R. (2014). Hypothalamic attack: a wonderful artifact or a useful perspective on escalation and pathology in aggression? A viewpoint. *Neuroscience of Aggression: Current Topics in Behavioral Neurosciences, 17,* 143-188.
- Kruk, M. R., Halasz, J., Meelis, W., & Haller, J. (2004). Fast positive feedback between the adrenocortical stress response and a brain mechanism involved in aggressive behavior. *Behavioral Neuroscience*, *118*, 1062-1070.
- Kruk, M. R., Westphal, K. G. C., van Erp, A. M. M., Van Asperen, J., Cave, B. J., Slater, E., ... Haller, J. (1998). The hypothalamus: Cross-roads of

endocrine and behavioural regulation in grooming and aggression. *Neuroscience & Biobehavioral Reviews, 23,* 163-177.

- Kubak, F. A., & Salekin, R. T. (2009). Psychopathy and anxiety in children and adolescents: new insights on developmental pathways to offending.
 Journal of Psychopathology and Behavioral Assessment, 31, 271-284.
- Kuo, L., & Kuo, Y. (1976). Chinese Folk Tales. Millbrae, CA: Celestial Arts.
- Lands, W. E. M., Libelt, B., Morris, A., Kramer, N. C., Prewitt, T. E., Bowen, P.,
 ... Burns, J. H. (1992). Maintenance of lower proportions of (n 6)
 eicosanoid precursors in phospholipids of human plasma in response to added dietary (n 3) fatty acids. *Biochimica et Biophysica Acta, 1180,* 147-162.
- Lang, P. J., Bradley, M. M., & Cuthbert, B. N. (1990). Emotion, attention, and the startle reflex. *Psychological Review, 97,* 377-398.
- Lansbergen, M. M., Böcker, K. B. E., Bekker, E. M., & Kenemans, J. L. (2007). Neural correlates of stopping and self-reported impulsivity. *Clinical Neurophysiology, 118,* 2089-2103.
- Lappin, J. S., & Eriksen, C. W. (1966). Use of a delayed signal to stop a visual reaction-time response. *Journal of Experimental Psychology*, *72*, 805-811.
- Latinus, M., & Taylor, M. J. (2006). Face processing stages: Impact of difficulty and the separation of effects. *Brain Research, 1123,* 179-187.
- Laugero, K. D., Smilowitz, J. T., German, J. B., Jarcho, M. R., Mendoza, S. P., & Bales, K. L. (2011). Plasma omega-3 polyunsaturated fatty acid status and monounsaturated fatty acids are altered by chronic social stress and predict endocrine responses to acute stress in titi monkeys. *Prostaglandins, Leukotrienes and Essential Fatty Acids, 84,* 71-78.
- Laviolle, B., Froger-Bompas, C., Guillo, P., Sevestre, A., Letellier, C., Pouchard, M, ... Paillard, F. (2005). Relative validity and reproducibility of a 14-item semi-quantitative food frequency questionnaire for cardiovascular prevention. *European Journal of Cardiovascular Prevention and Rehabilitation, 12*, 587-595.

- Lavric, A., Pizzagalli, D. A., & Forstmeier, S. (2004). When 'go' and 'nogo' are equally frequent: ERP components and cortical tomography. *European Journal of Neuroscience, 20,* 2483-2488.
- LeDoux, J. E. (1996). *The Emotional Brain.* New York, NY: Simon and Schuster.
- LeDoux, J. E. (2000). Emotion circuits in the brain. *Annual Review of Neuroscience*, *23*, 155-184.
- Lee, T. M., Chan, S. C., & Raine, A. (2009). Hyperresponsivity to threat stimuli in domestic violence offenders: A functional magnetic resonance imaging study. *Journal of Clinical Psychiatry*, *70*, 36-45.
- Légaré, N., Brosseau, E., & Joyal, C. C. (2007). Omega-3 and violence in schizophrenia. *Schizophrenia Research, 96*(1-3), 269.
- Leist, T., & Dadds, M. R. (2009). Adolescents' ability to read different emotional faces relates to their history of maltreatment and type of psychopathology. *Clinical Child Psychology and Psychiatry*, 14(2), 237-250.
- Lewis, M. D., Granic, I., Lamm, C., Zelazo, P. D., Stieben, J., Todd, R. M., ... Pepler, D. (2008). Changes in the neural bases of emotion regulation associated with clinical improvement in children with behavior problems. *Development and Psychopathology, 20,* 913-939.
- Lewis, M. D., Lamm, C., Segalowitz, S. J., Stieben, J., & Zelazo, P. D. (2006). Neurophysiological correlates of emotion regulation in children and adolescents. *Journal of Cognitive Neuroscience*, *18*(3), 430-443.
- Lewis, M. D., Todd, R. M., & Honsberger, M. J. M. (2007). Event-related potential measures of emotion regulation in early childhood. *Neuroreport: For Rapid Communication of Neuroscience Research, 18*, 61-65.
- Li, C. S. R., Huang, C., Constable, R. T., & Sinha, R. (2006). Imaging response inhibition in a stop-signal task: neural correlates independent of signal monitoring and post-response processing. *Journal of Neuroscience, 26,* 186-192.

- Li, J. H., Ko, H. C., Weng, L.J., Liau, L. C., & Lu, R. B. (2002). The development of an impulsiveness scale: psychometric properties and relation to antisocial personal disorder. *Chinese Journal of Psychology, 44,* 109-119.
- Li, D., Weisinger, H. S., Weisinger, R. S., Mathai, M., Armitage, J. A., Vingrys,
 A. J., & Sinclair, A. J. (2006). Omega 6 to omega 3 fatty acid imbalance early in life leads to persistent reductions in DHA levels in glycerophospholipids in rat hypothalamus even after long-term omega 3 fatty acid repletion. *Prostaglandins, Leukotrienes, and Essential Fatty Acids, 74,* 391-399.
- Lijffijt, M., Cox, B., Acas, M. D., Lane, S. D., Moeller, F. G. & Swann, A. C., (2012). Differential relationships of impulsivity or antisocial symptoms on P50, N100, or P200 auditory sensory gating in controls and antisocial personality disorder. *Journal of Psychiatric Research, 46,* 743-750.
- Linkenkaer-Hansen, K., Palva, J. M., Sams, M., Hietanen, J. K., Aronen, H. J., & Ilmoniemi, R. J. (1998). Face-selective processing in human extrastriate cortex around 120 ms after stimulus onset revealed by magneto- and electroencephalography. *Neuroscience Letters, 253,* 147-150.
- Liotti, M., Pliszka, S., Perez, R., Kothmann, D., & Woldorff, M. (2005). Abnormal brain activity related to performance monitoring and error. *Cortex, 41*(3), 377-388.
- Lilienfeld, S. O., & Widows, M. R. (2005). *Professional manual for the Psychopathic Personality Inventory-Revised (PPI-R).* Lutz, FL: Psychological Assessment Resources.
- Lish, J. D., Zimmerman, M., Farber, N. J., Lush, D. T., Kuzma, M. A., & Plescia, G. (1996). Suicide screening in a primary care setting at a VeteransAffairs medical center. *Psychosomatics*, *37*, 413-424.
- Litman, B. J., Niu, S. L., Polozova, A., & Mitchell, D. C. (2001). The role of docosahexaenoic acid containing phospholipids in modulating G

protein-coupled signaling pathways: visual transduction. *Journal of Molecular Neuroscience, 16*, 237-242.

- Liu, Y., Zhan, X., Li, W., Han, H., Wang, H., Hou, J., ... Wang, Y. (2015). The trait anger affects conflict inhibition: a go/Nogo ERP study. *Frontiers in Human Neuroscience*, 8, 1076.
- Lochman, J. E., & Dodge, K. A. (1994). Social-cognitive processes of severely violent, moderately aggressive, and nonaggressive boys. *Journal of Consulting and Clinical Psychology, 62,* 366-374.
- Loeber, R., & Schmaling, K. (1985). Empirical evidence for overt and covert patterns of antisocial conduct problems. *Journal of Abnormal Child Psychology, 13,* 337-352.
- Logan, A. C. (2005). Omega-3 and depression research: Hold the olive oil. *Prostaglandins, Leukotrienes and Essential Fatty Acids, 72*(6), 441.
- Logan, G. D. (1994). On the ability to inhibit thought and action: a users' guide to the stop signal paradigm. In D. Dagenbach, & T. H. Carr (Eds.), *Inhibitory processes in attention, memory, and language.* San Diego, CA: Academic Press.
- Logan, G. D., & Cowan, W. B. (1984). On the ability to inhibit thought and action: a theory of an act of control. *Psychological Review, 91*(3), 295-327.
- Long, S. J., & Benton, D. (2013). A double-blind trial of the effect of docosahexaenoic acid and vitamin and mineral supplementation on aggression, impulsivity, and stress. *Human Psychopharmacology*, *28*(3), 238-47.
- Loney, B. R., Butler, M. A., Lima, E. N., Counts, C. A., & Eckel, L. A. (2006). The relation between salivary cortisol, callous-unemotional traits, and conduct problems in an adolescent non-referred sample. *Journal of Child Psychology and Psychiatry, 47*(1), 30-36.
- Loney, B. R., Frick, P. J., Clements, C. B., Ellis, M. L., & Kerlin, K. (2003). Callous- unemotional traits, impulsivity, and emotional processing in antisocial adolescents. *Journal of Clinical Child and Adolescent Psychology, 32,* 139-152.

Lopez-Duran, N. L., Kovacs, M., & George, C. J. (2009). Hypothalamic-pituitaryadrenal axis dysregulation in depressed children and adolescents: A meta-analysis. *Psychoneuroendocrinology*, *34*(9), 1272-1283.

Lorenz, K. (1966). On Aggression. London, UK: Methuen.

Lorenz, K. (1981). The Foundations of Ethology. New York, NY: Springer.

- Lotrich, F. E., Sears, B., & McNamara, R. K. (2013). Anger induced by interferon-alpha is moderated by ratio of arachidonic acid to omega-3 fatty acids. *Journal of Psychosomatic Research, 75,* 475-483.
- Lozier, L. M., Cardinale, E. M., Vanmeter, J. W., & Marsh, A. A. (2014). Mediation of the relationship between callous-unemotional traits and proactive aggression by amygdala response to fear among children with conduct problems. *JAMA Psychiatry*, *71*(6), 627-636.
- Luchtman, D. W., & Song, C. (2012). Cognitive enhancement by omega-3 fatty acids from childhood to old age: findings from animal and clinical studies. *Neuropharmacology, 64*, 550-565.
- Luck, S. J. (2004). Ten Simple Rules for Designing and Interpreting ERP Experiments. In T. C. Handy (Ed.), *Event-Related Potentials: A Methods Handbook* (pp. 17-32). Cambridge, MA: MIT Press.
- Luck, S. J. (2005). *An introduction to the event-related potential technique.* Massachusetts Institute of Technology.
- Luck, S. J. (2006). The operation of attention—millisecond by millisecond—over the first half second. In H. Ogmen & B. G. Breitmeyer (Eds.), *The first half second: The microgenesis and temporal dynamics of unconscious and conscious visual processes* (pp. 187-206). Cambridge, MA: MIT Press.
- Luck, S. J. (2012). Event-related potentials. *APA Handbook of Research Methods in Psychology, 1,* 523-546.
- Luck, S. J., & Hillyard, S. A. (1990). Electrophysiological evidence for parallel and serial processing during visual search. *Perception and Psychophysics, 48,* 603-617.
- Luck, S. J., & Kappenman, E. K. (2012). *The Oxford Handbook of (Event)-Related Potential Components.* New York, NY: Oxford University Press.

- Luijten, M., Veltman, D. J., van den Brink, W., Hester, R., Field, M., Smits, M., & Franken, I. H. A. (2011). Neurobiological substrate of smoking-related attentional bias. *NeuroImage*, *54*, 2374-2381.
- Lykken, D. T. (1957). A study of anxiety in the sociopathic personality. *The Journal of Abnormal and Social Psychology, 55*(1), 6-10.
- Lynam, D. R. (1997). Pursuing the psychopath: Capturing the fledgling psychopath in a nomological net. *Journal of Abnormal Psychology*, *106*, 425-438.
- MacLeod, C. M. (2007). The concept of inhibition in cognition. In D. S. Gorfein,
 & C. M. MacLeod (Eds.), *Inhibition in Cognition* (pp. 3-23). Washington,
 DC: American Psychological Society.
- Magnuski, M., & Gola, M. (2013). It's not only in the eyes: nonlinear relationship between face orientation and N170 amplitude irrespective of eye presence. *International Journal of Psychophysiology*, *89*(3), 358-365.
- Malone, R. P., Luebbert, J. F., Pena-Ariet, M., Biesecker, K., & Delaney, M. A. (1994). The Overt Aggression Scale in a study of lithium in aggressive conduct disorder. *Psychopharmacology Bulletin*, 30, 215-218.
- Malone, R. P., Delaney, M. A., Luebbert, J. F., Cater, J., & Campbell, M. (2000).
 A double-blind placebo-controlled study of lithium in hospitalized aggressive children and adolescents with conduct disorder. *Archives of General Psychiatry, 57*, 649-654.
- Malone, A., Carroll, A., & Murphy, B. P. (2012). Facial affect recognition deficits: A potential contributor to aggression in psychotic illness. *Aggression* and Violent Behavior, 17, 27-35.
- Marangell, L. B., Martinez, J. M., Zboyan, H. A., Kertz, B., Kim, H. F., & Puryear, L. J. (2003). A double-blind, placebo-controlled study of the omega-3 fatty acid docosahexaenoic acid in the treatment of major depression. *Journal of Psychiatry*, *160*, 996-998.
- Mardaga, S., & Hansenne, M. (2009). Do personality traits modulate the effect of emotional visual stimuli on auditory information processing? *Journal of Individual Differences, 30*(1), 28-34.

- Mars, R. B., Klein, M. C., Neubert, F. X., Olivier, E., Buch, E. R., Boorman, E.
 D., & Rushworth, M. F. (2009). Short-latency influence of medial frontal cortex on primary motor cortex during action selection under conflict. *The Journal of Neuroscience, 29*(21), 6926-6931.
- Marsh, A. A., & Ambady, N. (2007). The influence of the fear facial expression on prosocial responding. *Journal of Cognitive Emotion*, 21, 225-247.
- Marsh, A. A., & Blair, R. J. R. (2008). Deficits in facial affect recognition among antisocial populations: a meta-analysis. *Neuroscience and Biobehavioral Reviews*, 32(3), 454-465.
- Marsh, A. A., Finger, E. C., Mitchell, D. G., Reid, M. E., Sims, C., Kosson, D. S., ... Blair, R. J. (2008). Reduced amygdala response to fearful expressions in children and adolescents with callous-unemotional traits and disruptive behavior disorders. *American Journal of Psychiatry*, 165, 712-720.
- Marsh, A. A., Finger, E. C., Fowler, K. A., Jurkowitz, I. T. N., Schechter, J. T., Yu, H. H., ... Blair, R. J. R. (2011). Reduced amygdala-orbitofrontal connectivity during moral judgments in youths with disruptive behavior disorders and psychopathic traits. *Psychiatry Research: Neuroimaging*, 194, 279-286.
- Marsh, D. M., Dougherty, D. M., Mathias, C. W., Moeller, F. G., & Hicks, L. R. (2002). Comparison of women with high and low trait impulsivity using laboratory impulsivity models of response-inhibition and reward-choice. *Personality and Individual Differences, 33,* 1291-1310.
- Marshall, W. L., Hudson, S. M., Jones, R., & Fernandez, Y. M. (1995). Empathy in sex offenders. *Clinical Psychology Review*, *15*, 99-113.
- Marszalek, J. R., & Lodish, H. F. (2005). Docosahexaenoic acid, fatty acid interacting proteins, and neuronal function: breastmilk and fish are good for you. *Annual Review of Cell and Developmental Biology, 21*, 633-657.
- Martinez, M. (1992). Tissue levels of polyunsaturated fatty acids during early human development. *Journal of Pediatrics, 120,* 129-138.

- Mathias, C. W., & Stanford, M. S. (1999). P300 under standard and surprise conditions in self-reported impulsive aggression. *Progress in Neuro-Psychopharmacological and Biological Psychiatry*, 23, 1037-1051.
- Mathieu, G., Denis, S., Lavialle, M., & Vancassel, S. (2008). Synergistic effects of stress and omega-3 fatty acid deprivation on emotional response and brain lipid composition in adult rats. *Prostaglandins, Leukotrienes, and Essential Fatty Acids, 78*, 391-401.
- Matsumoto, M., Matsumoto, K., Abe, H., & Tanaka, K. (2007). Medial prefrontal cell activity signaling prediction errors of action values. *Nature Neuroscience*, *10*, 647-656.
- Mattia, M., Spadacenta, S., Pavone, L., Quarato, P., Esposito, V., Sparano, A.,
 ... Mirabella, G. (2012). Stop-event-related potentials from intracranial electrodes reveal a key role of premotor and motor cortices in stopping ongoing movements. *Frontiers in Neuroengineering*, *5*, 12.
- Matthies, S., Rüsch, N., Weber, M., Lieb, K., Philipsen, A., ... van Elst, L. T. (2012). Small amygdala-high aggression? The role of the amygdala in modulating aggression in healthy subjects. *World Journal of Biological Psychiatry*, 13(1), 75-81.
- Maughan, B., & Rutter, M. (2001). Antisocial children grown up. In J. Hill, & B.
 Maughan (Eds.), *Conduct Disorders in Childhood and Adolescence* (pp. 507-552). Cambridge, UK: Cambridge University Press.
- Mazereeuw, G., Lanctôt, K. L., Chau, S. A., Swardfager, W., & Herrmann N. (2012). Effects of omega-3 fatty acids on cognitive performance: a meta-analysis. *Neurobiology of Aging*, 1482, 17-29.
- McBurnett, K., Raine, A., Stouthamer-Loeber, M., Loeber, R., Kumar, A. M., Kumar, M., & Lahey, B. B. (2005). Mood and Hormone Responses to Psychological Challenge in Adolescent Males with Conduct Problems. *Biological Psychiatry*, *57*(10), 1109-1116.
- McCarthy, G., Puce, A., Gore, J. C., & Allison, T. (1997). Face-specific processing in the human fusiform gyrus. *Journal of Cognitive Neuroscience*, *9*, 605-610.

- McCloskey. M. S., Berman, M. E., Noblett, K. L., & Coccaro, E. F. (2006).
 Intermittent explosive disorder-integrated research diagnostic criteria: convergent and discriminant validity. *Journal of Psychiatric Research*, 40, 231-242.
- McClure, E. B. (2000). A meta-analytic review of sex differences in facial expression processing and their development in infants, children, and adolescents. *Psychological Bulletin, 126,* 424-453.
- McCown, W., Johnson, J., & Austin, S. (1986). Inability of delinquents to recognize facial affects. *Journal of Social Behavior & Personality, 1,* 489-496.
- McEllistrem, J. E. (2004). Affective and predatory violence: a bimodal classification system of human aggression and violence. *Aggression and Violent Behavior, 10,* 1-30.
- McNamara, R. K., & Carlson, S. E. (2006). Role of omega-3 fatty acids in brain development and function: potential implications for the pathogenesis and prevention of psychopathology. *Prostaglandins, Leukotrienes, and Essential Fatty Acids, 75,* 329-349.
- McNamara, R. K., Hahn, C., Jandacek, R., Rider, T., Tso, P., Stanford, K. E., & Richardson, N. M. (2007). Selective Deficits in the Omega-3 Fatty Acid Docosahexaenoic Acid in the Postmortem Orbitofrontal Cortex of Patients with Major Depressive Disorder. *Biological Psychiatry, 62,* 17-24.
- McNamara, R. K., Jandacek, R., Rider, T., Tso, P., Cole-Strauss, A., & Lipton, J. W. (2010). Omega-3 fatty acid deficiency increases constitutive pro-inflammatory cytokine production in rats: relationship with central serotonin turnover. *Prostaglandins, Leukotrienes and Essential Fatty Acids, 83*(4-6), 185-191.
- McNamara, R. K., Sullivan, J., & Richtand, N. M. (2008). Omega-3 fatty acid deficiency augments amphetamine-induced behavioral sensitization in adult mice: prevention by chronic lithium treatment. *Journal of Psychiatric Research, 42*(6), 458-468.

- Meaux, E., Roux, S., & Batty, M. (2014). Early visual ERPs are influenced by individual emotional skills. *Social Cognitive and Affective Neuroscience*, 9(8), 1089-1098.
- Meier, N. M., Perrig, W., & Koenig, T. (2012). Neurophysiological correlates of delinquent behaviour in adult subjects with ADHD. *International Journal* of Psychophysiology, 84, 1-16.
- Melcher, T., Born, C., & Gruber, O. (2011). How negative affect influences neural control processes underlying the resolution of cognitive interference: An event-related fMRI study. *Neuroscience Research, 70,* 415-427.
- Melcher, T., & Gruber, O. (2006). Oddball and incongruity effects during Stroop task performance: a comparative fMRI study on selective attention. *Brain Research, 1121*, 136-149.
- Melcher, T., & Gruber, O. (2009). Decomposing interference during Stroop performance into different conflict factors: an event-related fMRI study. *Cortex, 45*, 189-200.
- Menon, V., Adleman, N. E., White, C. D., Glover, G. H., & Reiss, F. L. (2001). Error-related brain activation during a go/no-go response inhibition task. *Human Brain Mapping, 12,* 131-143.
- Mériau, K., Wartenburger, I., Kazzer, P., Prehn, K., Lammers, C. H., van der Meer, E., ... Heekeren, H. R. (2006). A neural network reflecting individual differences in cognitive processing of emotions during perceptual decision making. *Neuroimage*, *33*(3), 1016-1027.
- Meyer, B. J., Byrne, M. K., Collier, C., Parletta, N., Crawford, D., Winberg, P. C.,
 ... Grant, L. (2015). Baseline omega-3 index correlates with aggressive and attention deficit disorder behaviours in adult prisoners. *PLoS One, 10*(3), e0120220.
- Meyer-Lindenberg, A., Buckholtz, J. W., Kolachana, B. A. R. H., Pezawas, L., Blasi, G., Wabnitz, A., ... Weinberger, D. R. (2006). Neural mechanisms of genetic risk for impulsivity and violence in humans. *Proceedings of the National Academy of Sciences of the United States of America*, *103*(16), 6269-6274
- Michael-Titus, A. T., & Priestley, J. V. (2014). Omega-3 fatty acids and traumatic neurological injury: from neuroprotection to neuroplasticity. *Trends in Neurosciences, 37,* 30-38.
- Mickleborough, T. D., Tecklenburg, S. L., Gregory, M. S., Montgomery, S., & Lindley, M. R. (2009). Eicosapentaenoic acid is more effective than docosahexaenoic acid in inhibiting proin □ ammatory mediator production and transcription from LPS-induced human asthmatic alveolar macrophage cells. *Clinical Nutrition, 28*(1), 71-77.
- Miller, P. A., & Eisenberg, N. (1988). The relation of empathy to aggressive and externalizing/antisocial behavior. *Psychological Bulletin*, 103(3), 324-344.
- Milte, C. M., Parletta, N., Buckley, J. D., Coates, A. M., Young, R. M., & Howe,
 P. R. C. (2012). Eicosapentaenoic and docosahexaenoic acids,
 cognition, and behavior in children with attention-deficit/hyperactivity
 disorder: A randomized controlled trial. *Nutrition, 28*, 670-677.
- Miltner, W. H., Lemke, U., Weiss, T., Holroyd, C., Scheffers, M. K., & Coles, M. G. (2003). Implementation of error-processing in the human anterior cingulate cortex: a source analysis of the magnetic equivalent of the error-related negativity. *Biological Psychology*, 64, 157-166.
- Mineka, S., & Cook, M. (1993). Mechanisms involved in the observational conditioning of fear. *Journal of Experimental Psychology: General*, 122, 23-38.
- Mineka, S., Davidson, M., Cook, M., & Keir, R. (1984). Observational conditioning of snake fear in rhesus monkeys. *Journal of Abnormal Psychology*, *93*, 355-372.
- Minzenberg, M. J., Fan, J., New, A. S., Tang, C. Y., & Siever, L. J. (2007). Frontolimbic dysfunction in response to facial emotion in borderline personality disorder: an event-related fMRI study. *Psychiatry Research: Neuroimaging*, 15(3), 231-243.
- Mirabella, G., Pani, P., & Ferraina, S. (2011). Neural correlates of cognitive control of reaching movements in the dorsal premotor cortex of rhesus monkeys. *Journal of Neurophysiology, 106,* 1454-1466.

- Miyake, A., Friedman, N. P., Emerson, M. J., Witzki, A. H., Howerter, A., & Wager, T. D. (2000). The unity and diversity of executive functions and their contributions to complex "frontal lobe" tasks: A latent variable analysis. *Cognitive Psychology*, *41*, 49-100.
- Mnatsakanian, E. V., & Tarkka, I. M. (2004). Familiar-face recognition and comparison: source analysis of scalp-recorded event-related potentials. *Clinical Neurophysiology*, *115*, 880-886.
- Moffitt, T. E. (1993). The neuropsychology of conduct disorder. *Developmental Psychopathology, 5,* 135-151.
- Mogg, K., & Bradley, B. P. (1998). A cognitive-motivational analysis of anxiety, Behaviour, Research and Therapy, 36, 809-848.
- Montagne, B., van Honk, J., Kessels, R. P. C., Frigerio, E., Burt, M., van Zandvoort, M. J. E., ... de Haan, E. (2005). Reduced efficiency in recognizing fear in subjects scoring high on psychopathic personality characteristics. *Personality & Individual Differences, 38*, 5-11.
- Montgomery, P., & Richardson, A. J. (2008). Omega-3 fatty acids for bipolar disorder. *Cochrane Database of Systematic Reviews* 2008. The Cochrane Collaboration: John Wiley & Sons, Ltd.
- Morales-Vives, F., & Vigil-Colet, A. (2010). Are there sex differences in physical aggression in the elderly? *Personality and Individual Differences, 49,* 659-662.
- Morita, Y., Morita, K., Yamamoto, M., Waseda, Y., & Maeda, H. (2001). Effects of facial affect recognition on the auditory P300 in healthy subjects. *Neuroscience Research, 41*(1), 89-95.
- Morgan, A. B., & Lilienfeld, S. O. (2000). A meta-analytic review of the relation between antisocial behavior and neuropsychological measures of executive function. *Clinical Psychology Review, 20*(1), 113-136.
- Morris, J. S., deBonis, M., & Dolan, R. J. (2002). Human amygdala responses to fearful eyes. *Neuroimage, 17*, 214-222.
- Morris, J. S., Frith, C. D., Perrett, D. I., Rowland, D., Young, A. W., Calder, A. J., & Dolan, R. J. (1996). A differential neural response in the human amygdala to fearful and happy facial expressions. *Nature, 31,* 812-815.

- Morris, J. S., Friston, K. J., Buchel, C., Frith, C. D., Young, A. W., Calder, A. J.,
 & Dolan, R. J. (1998). A neuromodulatory role for the human amygdala in processing emotional facial expressions. *Brain*, *121*, 47-57.
- Morris, J. S., Ohman, A., & Dolan, R. J. (1999). A subcortical pathway to the right amygdala mediating "unseen" fear. *Proceedings of the National Academy of Sciences of the United States of America, 96*(4), 1680-1685.
- Morton, J., & Frith, U. (1993). What lesson for dyslexia from Down's syndrome? Comments on Cossu, Rossini, and Marshall (1993). *Cognition, 48,* 289-296.
- Moussas, G., Dadout, G., Douzenis, A., Poulis, E., Tselebi, A., Bratis, D., ...
 Lykouras, L. (2009). The Alcohol Use Disorders Identification Test
 (AUDIT): reliability and validity of the Greek version. *Annals of General Psychiatry*, *8*, 11.
- Mozaffarian, D., & Wu, J. H. (2011). Omega-3 fatty acids and cardiovascular disease: effects on risk factors, molecular pathways, and clinical events. *Journal of the American College of Cardiology*, *58*, 2047-2067.
- Müller, J. L., Gänssbauer, S., Sommer, M., Döhnel, K., Weber, T., Schmidt-Wilcke, T., & Hajak, G. (2008). Gray matter changes in right superior temporal gyrus in criminal psychopaths. Evidence from voxel-based morphometry. *Psychiatry Research*, *163*(3), 213-222.
- Müller, J. L., Sommer, M., Wagner, V., Lange, K., Taschler, H., Röder, C. H., ... Hajak, G. (2003). Abnormalities in Emotion Processing within Cortical and Subcortical Regions in Criminal Psychopaths: Evidence from a Functional Magnetic Resonance Imaging Study Using Pictures with Emotional Content. *Biological Psychiatry*, *54*, 152-162.
- Muñoz, L. C. (2009). Callous-Unemotional traits are related to combined deficits in recognizing afraid faces and body poses. *Journal of the American Academy of Child and Adolescent Psychiatry, 48*(5), 554-562.
- Muñoz, L. C., Qualter, P., & Padgett, G. (2011). Empathy and bullying: Exploring the influence of callous-unemotional traits. *Child Psychiatry* and Human Development, 42, 183-196.

- Munro, G. E. S., Dywan, J., Harris, G. T., McKee, S., Unsal, A., & Segalowitz,
 S. J. (2007). Response inhibition in psychopathy: The frontal N2 and
 P3. *Neuroscience Letters, 418*, 149-153.
- Murphy, F. C., Nimmo-Smith, I., & Lawrence, A. D. (2003). Functional neuroanatomy of emotion: a meta-analysis. *Cognitive, Affective and Behavioral Neuroscience, 3*, 207-233.
- Nachev, P., Wydell, H., O'Neill, K., Husain, M., & Kennard, C. (2007). The role of the pre-supplementary motor area in the control of action. *NeuroImage, 36,* 155-163.
- Nauta, W. J. H. (1971). The problem of the frontal lobes: A reinterpretation. *Journal of Psychiatric Research, 8,* 167-187.
- Nelson, W. M., & Finch, A. J. (2000). *Children's Inventory of Anger.* Los Angeles, CA: Western Psychological Services.
- Nelson, C. A., & Nugent. K. M. (1990). Recognition memory and resource allocation as revealed by children's event-related potential responses to happy and angry faces. *Developmental Psychology*, 26, 171-179.
- Nelson, R. J., & Trainor, B. C. (2007). Neural mechanisms of aggression. *Nature Reviews Neuroscience, 8,* 536-546.
- Nemets, H., Nemets, B., Apter, A., Bracha, Z., & Belmaker, R. H. (2006).
 Omega-3 treatment of childhood depression: a controlled, double-blind pilot study. *American Journal of Psychiatry*, *163*, 1098-1100.
- Neumann, C. S., Hare, R. D., & Newman, J. P. (2007). The superordinate nature of the psychopathy checklist-revised. *Journal of Personality Disorders, 21,* 102-117.
- Neuringer, M., Anderson, G. J., & Connor, W. E. (1988) The essentiality of n-3 fatty acids for the development and function of the retina and brain. *Annual Review of Nutrition, 8,* 517-541.
- New, A. S., Hazlett, E. A., Newmark, R. E., Zhang, J., Triebwasser, J., Meyerson, D., ... Buchsbaum, M. S. (2009). Laboratory induced aggression: a positron emission tomography study of aggressive individuals with borderline personality disorder. *Biological Psychiatry*, 66, 1107-1114.

- Nguyen, V. T., & Cunnington, R. (2014). The superior temporal sulcus and the N170 during face processing: Single trial analysis of concurrent EEG-fMRI. *Neuroimage, 86,* 492-502.
- Nichols, D. F., Betts, L. R., & Wilson, H. R. (2010). Decoding of faces and face components in face-sensitive human visual cortex. *Frontiers in Psychology*, 1, 28.
- Nichols, S. (2001). Mindreading and the cognitive architecture underlying altruistic motivation. *Mind and Language, 16*(4), 425-455.
- Nieuwenhuis, S., Aston-Jones, G., & Cohen, J. D. (2005). Decision making, the P3, and the locus coeruleus-norepinephrine system. *Psychological Bulletin, 131*, 510-532.
- Nieuwenhuis, S., Holroyd, C. B., Mol, N., & Coles, M. G. (2004). Reinforcementrelated brain potentials from medial frontal cortex: origins and functional significance. *Neuroscience and Biobehavior Review, 28,* 441-448.
- Nieuwenhuis, S., Yeung, N., & Cohen, J. D. (2004). Stimuli modality, perceptual overlap, and the go/no-go N2. *Psychophysiology, 41,* 157-160.
- Nieuwenhuis, S., Yeung, N., van den Wildenberg, W., & Ridderinkhof, K. R.
 (2003). Electrophysiological correlates of anterior cingulate function in a go/nogo task: effects of response conflict and trial-type frequency.
 Cognitive Affective, and Behavioral Neuroscience, 3, 17-26.
- Nigbur, R., Ivanova, G., & Stürmer, B. (2011). Theta power as a marker for cognitive interference. *Clinical Neurophysiology*, *122*, 2185-2194.
- Nigg, J. T., & Huang-Pollock, C. L. (2003). An early-onset model of the role of executive functions and intelligence in conduct disorder/delinquency. In
 B. B. Lahey, T. E. Moffitt, & A. Caspi (Eds.), *Causes of Conduct Disorder and Juvenile Delinquency* (pp. 227-253). New York, NY: The Guilford Press.
- Novaco, R. W. (2003). *The Novaco Anger Scale and Provocation Inventory* (*NAS-PI*). Los Angeles, CA: Western Psychological Services.
- Nummenmaa, L., Hyönä, J., & Calvo, M. G. (2010). Semantic categorization precedes affective evaluation of visual scenes. *Journal of Experimental Psychology: General, 139*(2), 222-246.

- Nummenmaa, L., Passamonti, L., Rowe, J., Engell, A. D., & Calder, A. J. (2010). Connectivity analysis reveals a cortical network for eye gaze perception. *Cerebral Cortex*, 20(8), 1780-1787.
- O'Brien, B. S., & Frick, P. J. (1996). Reward dominance: Associations with anxiety, conduct problems, and psychopathy in children. *Journal of Abnormal Child Psychology, 24*, 223-240.
- O'Connell, R. G., Balsters, J. H., Kilcullen, S. M., Campbell, W., Bokde, A. W., Lai, R., ... Robertson, I. H. (2012). A simultaneous ERP/fMRI investigation of the P300 aging effect. *Neurobiology of Aging, 33*, 2448-2461.
- O'Keane, V., Moloney, E., O'Neill, H., O'Connor, A., Smith, C., & Dinan, T. G. (1992). Blunted prolactin responses to d-fenfluramine in sociopathy. Evidence for subsensitivity of central serotonergic function. *British Journal of Psychiatry, 160,* 643-646.
- Oakes, M. E., & Coover, G. D. (1997). Effects of small amygdala lesions on fear, but not aggression, in the rat. *Physiological Behavior, 61,* 45-55.
- Ochsner, K. N., Bunge, S. A., Gross, J. J., & Gabrieli, J. D. (2002). Rethinking feelings: An FMRI study of the cognitive regulation of emotion. *Journal of Cognitive Neuroscience, 14,* 1215-1229.
- Office for National Statistics. (2015). *Crime Statistics, Focus on Violent Crime and Sexual Offences, 2013/14.* London, UK: UK Statistics Authority.
- Ögmen, H., & Breitmeyer, B. (Eds.). (2006). *The first half second: The microgenesis and temporal dynamics of unconscious and conscious visual processes*. Cambridge, MA: MIT Press.
- Öhman, A., Flykt, A., & Esteves, F. (2001). Emotion drives attention: detecting the snake in the grass. *Journal of Experimental Psychology: General*, *130*(3), 466-478.
- Öhman, A., Flykt, A., & Lundqvist, D. (2000). Unconscious emotion:
 Evolutionary perspectives, psychophysiological data and
 neuropsychological mechanisms. In R. D. R. Lane, L. Nadel, & G. L.
 Ahern (Eds.), *Cognitive Neuroscience of Emotion: Series in Affective Science* (pp. 296-327). Oxford, UK: Oxford University Press.

- Öhman, A., & Mineka, S. (2001). Fears, phobias, and preparedness: Toward an evolved module of fear and fear learning. *Psychological Review*, 108, 438-522.
- Oldenburg, J. F., Roger, C., Assecondi, S., Verbruggen, F., & Fias, W. (2012).
 Repetition priming in the stop signal task: The electrophysiology of sequential effects of stopping. *Neuropsychologia*, *50*(12), 2860-2868.
- Olofsson, J. K., & Polich, J. (2007). Affective visual event-related potentials: Arousal, repetition, and time-on-task. *Biological Psychology, 75,* 101-108.
- Öngen, D. E. (2010). Relationships between narcissism and aggression among non-referred Turkish university students. *Procedia Social and Behavioral Sciences, 5,* 410-415.
- Oosterlaan, J., Scheres, A., & Sergeant, J. A. (2005). Which executive functioning deficits are associated with AD/HD, ODD/CD and comorbid AD/HD + ODD/CD. *Journal of Abnormal Child Psychology, 33*(1), 69-85.
- Oquendo, M. A., & Mann, J. J. (2001). Neuroimaging findings in major depression, suicidal behaviour, and aggression. *Clinical Neuroscience Research, 1,* 377-380.
- Overtoom, C., Kenemans, J., Verbaten, M., Kemner, C., van der Molen, M., van Engeland, H., ... Koelega, H. S. (2002). Inhibition in children with attention-deficit/hyperactivity disorder: A psychophysiological study of the stop. *Biological Psychiatry*, *51*, 668-676.
- Owen, C., Rees, A., & Parker, C. (2008). The role of fatty acids in the development and treatment of mood disorders. *Current Opinion in Psychiatry, 21,* 19-24.
- Pajer, K., Gardner, W., Rubin, R. T., Perel, J., & Neal, S. (2001). Decreased cortisol levels in adolescent girls with conduct disorder. *Archives of General Psychiatry*, 58, 297-302.
- Palermo, R., & Rhodes, G. (2007). Are you always on my mind? A review of how face perception and attention interact. *Neuropsychologia*, 45, 75-92.

- Pandey, A. K., Kamarajan, C., Tang, Y., Chorlian, D. B., Roopesh, B. N., Manz, N., ... Porjesz, B. (2012). Neurocognitive deficits in male alcoholics: An ERP/sLORETA analysis of the N2 component in an equal probability go/nogo task. *Biological Psychology*, *89*, 170-182.
- Pardini, D. A., & Byrd, A. L. (2012). Perceptions of aggressive conflicts and others' distress in children with callous-unemotional traits: 'I'll show you who's boss, even if you suffer and I get in trouble'. *Journal of Child Psychiatry and Psychology, 53*, 283-291.
- Pardini, D. A., Lochman, J. E., & Frick, P. J. (2003). Callous/unemotional traits and social-cognitive processes in adjudicated youths. *Journal of the American Academy of Child and Adolescent Psychiatry*, 42(3), 364–371.
- Pardini, D. A., Raine, A., Erickson, K., & Loeber, R. (2014). Lower amygdala volume in men is associated with childhood aggression, early psychopathic traits, and future violence. *Biological Psychiatry*, 75(1), 73-80.
- Parletta, N., Milte, C. M., & Meyer, B. J. (2013). Nutritional modulation of cognitive function and mental health. *Journal of Nutritional Biochemistry, 24*, 725-743.
- Parra, M., Schnaas, L., Meydani, M., Perroni, E., Martíanez, S., & Romieu, I. (2002). Erythrocyte cell membrane phospholipid levels compared against reported dietary intakes of polyunsaturated fatty acids in pregnant Mexican women. *Public Health and Nutrition, 5*, 931-937.
- Parvizi, J., Jacques, C., Foster, B., Withoft, N., Rangarajan, V., Weiner, K., & Grill-Spector, K. (2012). Electrical stimulation of human fusiform faceselective regions distorts face perception. *The Journal of Neuroscience*, *32*(43), 14915-14920.
- Pasalich, D. S., Dadds, M. R., & Hawes, D. J. (2014). Cognitive and affective empathy in children with conduct problems: Additive and interactive effects of callous-unemotional traits and autism spectrum disorders symptoms. *Psychiatry Research*, 219, 625-630.

- Paschall, M., & Fishbein, D. (2002). Executive cognitive functioning and aggression: A public health perspective. *Aggression & Violent Behavior*, 7(3), 215-235.
- Pascoe, M. C., Crewther, S. G., Carey, L. M., & Crewther, D. P. (2011).
 Inflammation and depression: why poststroke depression may be the norm and not the exception. *International Journal of Stroke, 6,* 128-135.
- Passamonti, L., Crockett, M. J., Apergis-Schoute, A. M., Clark, L., Rowe, J. B., Calder, A. J., & Robbins, T. W. (2012). Effects of Acute Tryptophan
 Depletion on Prefrontal-Amygdala Connectivity While Viewing Facial
 Signals of Aggression. *Biological Psychiatry*, *71*(1), 36-43.
- Patel, S. H., & Azzam, P. N. (2005). Characterization of N200 and P300: selected studies of the Event-Related Potential. *International Journal of Medical Sciences*, 2(4), 147-154.
- Patrick, C. J. (2008). Psychophysiological correlates of aggression and violence: an integrative review. *Philosophical Transactions of the Royal Society of London, Series B, 363,* 2543-2555.
- Patrick, C. J. (2010). Operationalizing the triarchic conceptualization of psychopathy: Preliminary descriptions of brief scales for assessment of boldness, meanness, and disinhibition. Tallahassee, FL: Florida State University.
- Patrick, C. J., & Bernat, E. M. (2009). Neurobiology of psychopathy: A twoprocess theory. In G. G. Berntson, & J. T. Cacioppo (Eds.), *Handbook* of neuroscience for the behavioral sciences (pp. 1110-1131). New York, NY: John Wiley & Sons.
- Patrick, C. J., Bernat, E. M., Malone, S. M., Iacono, W. G., Krueger, R. F., & McGue, M. (2006). P300 amplitude as an indicator of externalizing in adolescent males. *Psychophysiology*, 43, 84-92.
- Patten, A. R., Sickmann, H. M., Dyer, R. A., Innis, S. M., & Christie, B. R. (2013). Omega-3 fatty acids can reverse the long-term deficits in hippocampal synaptic plasticity caused by prenatal ethanol exposure. *Neuroscience Letters*, *551*, 7-11.

- Paulhus, D. L., & Williams, K. M. (2002). The Dark Triad of personality: Narcissism, Machiavellianism, and psychopathy. *Journal of Research in Personality*, *36*, 556-563.
- Pawliczek, C. M., Derntl, B., Kellermann, T., Gur, R. C., & Schneider, F. (2013). Anger under control: neural correlates of frustration as a function of trait aggression. *PLOS One*, *8*(10), e78503.
- Pawlosky, R. J., Bacher, J., & Salem, N., Jr. (2001). Ethanol consumption alters electroretinograms and depletes neural tissues of docosahexaenoic acid in rhesus monkeys: Nutritional consequences of a low n-3 fatty acid diet. *Alcoholism, Clinical and Experimental Research, 25*(12), 1758-1765.
- Peet, M., & Horrobin, D. F. (2002). A dose-ranging study of the effects of ethyleicosapentaenoate in patients with ongoing depression despite apparently adequate treatment with standard drugs. *Archive of General Psychiatry, 59*(10), 913-919.
- Perdeci, Z., Gülsün, M., Çelik, C., Erdem, M., Özdemir, B., Özdağ, F., & Kılıç,
 S. (2010). Aggression and The Event-Related Potentials in Antisocial
 Personality Disorder. *Bulletin of Clinical Psychopharmacology, 20*, 300-306.
- Perry, R. J., Rosen, H. R., Kramer, J. H., Beer, J. S., Levenson, R. L., & Miller,
 B. L. (2001). Hemispheric dominance for emotions, empathy and social behaviour: evidence from right and left handers with frontotemporal dementia. *Neurocase*, *7*, 145-160.
- Pessoa, L., & Adolphs, R. (2010). Emotion processing and the amygdala: from a 'low road' to 'many roads' of evaluating biological significance. *Nature Reviews Neuroscience*, *11*, 773-783.
- Pessoa, L., McKenna, M., Gutierrez, E., & Ungerleider, L. G. (2002). Neural processing of emotional faces requires attention. *Proceedings of the National Academy of Sciences, 99*, 11458-11463.
- Pessoa, L., Padmala, S., Kenzer, A., & Bauer, A. (2012). Interactions between cognition and emotion during response inhibition. *Emotion, 12*, 192-197.

- Petrides, M., & Pandya, D. N. (2007). Efferent Association Pathways from the Rostral Prefrontal Cortex in the Macaque Monkey. *The Journal of Neuroscience, 27*(43), 11573-11586.
- Pfefferbaum, A., Ford, J. M., Weller, B. J., & Kopell, B. (1985). ERPs to response production and inhibition. *Electroencephalography and Clinical Neurophysiology, 60,* 423-434.
- Phillips, M. L., Williams, L. W., Heining, M., Herba, C. M., Russell, T., Andrew, C., ... Gray, J. A. (2004). Differential neural responses to overt and covert presentations of facial expressions of fear and disgust. *NeuroImage, 21,* 1484-1496.
- Phillips, M. L., Young, A. L., Scott, S. K., Calder, A. J., Andrew, C., Giampietro, V., ... Gray, J. A. (1998). Neural responses to facial and vocal expressions of fear and disgust. *Proceedings of the Royal Society B*, 265, 1809-1817.
- Phillips, M. L., Young, A. W., Senior, C., Calder, A. J., Rowland, D., Brammer, M., ... David, A. (1997). A specific neural substrate for perceiving facial expressions of disgust. *Nature*, *389*, 495-498.
- Pierce, K., Müller, R. A., Ambrose, J., Allen, G., & Courchesne, E. (2001). People with autism process faces outside the 'fusiform face area.': evidence from FMRI. *Brain*, *124*, 2059-2073.
- Pietrini, P., Guazzelli, M., Basso, G., Jaffe, K., & Grafman, J. (2000). Neural correlates of imaginal aggressive behavior assessed by positron emission tomography in healthy subjects. *American Journal of Psychiatry, 157*(11), 1772-1781.
- Pillutla, M., & Murnighan, J. K. (1996). Unfairness, anger, and spite: Emotional rejections of ultimatum offers. Organizational Behavior and Human Decision Processes, 68, 208-224.
- Pinkham, A. E., Griffin, M., Baron, R., Sasson, N. J., & Gur, R. C. (2010). The face in the crowd effect: Anger superiority when using real faces and multiple identities. *Emotion*, 10(1), 141-146.

- Pitcher, D., Walsh, V., Yovel, G., & Duchaine, B. (2007). TMS evidence for the involvement of the right occipital face area in early face processing. *Current Biology*, 17(18), 1568-1573.
- Pitchford, I. (2001). The Origins of Violence: Is Psychopathy an Adaptation? *The Human Nature Review, 1,* 28-36.
- Pizzagalli, A., Oakes, R., & Davidson, J. (2003) Coupling of theta activity and glucose metabolism in the human rostral anterior cingulate cortex: An EEG/PET study of normal and depressed subjects. *Journal of Psychophysiology, 40,* 939-949.
- Plichta, M. M., Wolf, I., Hohmann, S., Baumeister, S., Boecker, R., Schwarz, A. J., ... Brandeis, D. (2013). Simultaneous EEG and fMRI reveals a causally connected subcortical-cortical network during reward anticipation. *Journal of Neuroscience*, *33*, 14526-14533.
- Pliszka, S., Liotti, M., & Woldorff, M. (2000). Inhibitory control in children with attention-deficit/hyperactivity disorder: Event-related potentials identify the processing component and timing of an impaired right-frontal response-inhibition mechanism. *Biological Psychiatry, 48,* 238-246.
- Ploghaus, A., Tracey, I., Clare, S., Gati, J. S., Rawlins, J. N., & Matthews, P. M. (2000). Learning about pain: The neural substrate of the prediction error for aversive events. *Proceedings of the National Academy of Sciences USA*, *97*(16), 9281-9286.
- Polich, J. (2003). Overview of P3a and P3b. In J. Polich (Ed.), *Detection of change: event-related potential and fMRI findings* (pp. 83-98). Boston, MA: Kluwer.
- Polich, J. (2007). Updating P300: an integrative theory of P3a and P3b. *Clinical Neurophysiology, 118,* 2128-2148.
- Polich, J., & Herbst, K. L. (2000). P300 as a clinical assay: rationale, evaluation, and findings. *International Journal of Psychophysiology, 38,* 3-19.
- Polich, J., & Kok, A. (1995). Cognitive and biological determinants of P300: an integrative review. *Biological Psychology, 41,* 103-146.

- Polich, J., Pollock, V. E., & Bloom, F. E. (1994). Meta-analysis of P300 amplitude from individuals at risk of alcoholism. *Psychological Bulletin*, *115*, 55-73.
- Politi, P., Cena, H., Comelli, M., Marrone, G., Allegri C., Emanuele, E., ... di Nemi, U. (2008). Behavioral Effects of Omega-3 Fatty Acid Supplementation in Young Adults with Severe Autism: An Open Label Study. Archives of Medical research, 39, 682-685.
- Pollitt, E. (2000). Developmental sequel from early nutritional deficiencies: Conclusive and probability judgements. *The Journal of Nutrition, 130*(2), 350-353.
- Pontius, A. A. (2002). Impact of fear-inducing violence on neuropsychological visual-spatial tests in warring hunter-gatherers. Analogies to violent Western environments. *Aggression and Violent Behavior: A Review Journal, 7,* 69-84.
- Pontius, A. A. (2005). Fastest fight/flight reaction via amygdalar visual pathway implicates simple face drawing as its marker: Neuroscientific data consistent with neuropsychological findings. *Aggression and Violent Behavior, 10,* 363-373.
- Porjesz, B., Begleiter, H., & Garozzo, R. (1980). Visual evoked potential correlates of information processing deficits in chronic alcoholics, In H. Begleiter (Ed.), *Biological Effects of Alcohol* (pp. 603-622). New York, NY: Plenum Press.
- Portillo-Reyes, V., Pérez-García, M., Loya-Méndez, Y., & Puente, A. E. (2014).
 Clinical significance of neuropsychological improvement after supplementation with omega-3 in 8-12 years old malnourished Mexican children: A randomized, double-blind, placebo and treatment clinical trial. *Research in Developmental Disabilities, 35,* 861-870.
- Posner, M. I., & DiGirolamo, G. J. (1998). Executive attention: conflict, target detection, and cognitive control. In R. Parasuraman (Ed.), *The Attentive Brain*, (pp. 401-423). Cambridge, MA: MIT Press.
- Posner, M. I., Rueda, M. R., & Kanske, P. (2007). Probing the Mechanisms of Attention. In J. T. Cacioppo, J. G. Tassinary, & G. G. Berntson (Eds.),

Handbook of Psychophysiology (pp. 410-432). Cambridge, UK: Cambridge University Press.

- Potts, G. F. (2004). An ERP index of task relevance evaluation of visual stimuli. *Brain and Cognition, 56,* 5-13.
- Potts, G. F., Dien, J., Hartry-Speiser, A. L., McDougall, L. M., & Tucker, D. M. (1998). Dense sensor array topography of the event-related potential to task-relevant auditory stimuli. *Electroencephalography and Clinical Neurophysiology*, *106*, 444-456.
- Potts, G. F., Liotti, M., Tucker, D. M., & Posner, M. I. (1996). Frontal and inferior temporal cortical activity in visual target detection: Evidence from high spatially sampled event-related potentials. *Brain Topography*, *9*, 3-14.
- Potts, G. F., & Tucker, D. M. (2001). Frontal evaluation and posterior representation in target detection. *Cognitive Brain Research*, 11, 147-156.
- Pourtois, G., Sander, D., Andres, M., Grandjean, D., Reveret, L., Olivier, E., & Vuilleumier, P. (2004). Dissociable roles of the human somatosensory and superior temporal cortices for processing social face signals. *European Journal of Neuroscience, 20*, 3507-3515.
- Poustka, L., Maras, A., Hohm, E., Fellinger, J., Holtmann, M., Banaschewski,
 T., ... Laucht, M. (2010). Negative association between plasma cortisol
 levels and aggression in a high-risk community sample of adolescents.
 Journal of Neural Transmission, *117*, 621-627.
- Preston, S. D., & de Waal, F. B. (2002). Empathy: Its ultimate and proximate bases. *Behavioral and Brain Sciences*, *25*(1-20), 20-71.
- Pridmore, S., Chamber, A., & McArthur, M. (2005). Neuroimaging in psychopathy. Australian and New Zealand Journal of Psychiatry, 39, 856-865.
- Prox, V., Dietrich, D. E., Zhang, Y., Emrich, H. M., & Ohlmeier, M. D. (2007.) Attentional processing in adults with ADHD as reflected by eventrelated potentials. *Neuroscience Letters*, *419*, 236-241.

- Puce, A., & Perrett, D. (2003). Electrophysiology and brain imaging of biological motion. *Philosophical Transactions of the Royal Society B: Biological Sciences, 358,* 435-445.
- Pujol, J., Harrison, B. J., Ortiz, H., Deus, J., Soriano-Mas, C., López-Solà, M.,
 ... Cardoner, N. (2009). Influence of the fusiform gyrus on amygdala response to emotional faces in the non-clinical range of social anxiety. *Psychological Medicine, 39,* 1177-1187.
- Putman, P., Hermans, E., & van Honk, J. (2004). Emotional Stroop performance for masked angry faces: it's BAS, not BIS. *Emotion, 4*, 305-311.
- R Core Team (2012). *R: A language and environment for statistical computing.* R Foundation for Statistical Computing, Vienna, Austria.
- Raaijmakers, M. A. J., Smidts, D. P., Sergeant, J. A., Maassen, G. H.,
 Posthumus, J. A., van Engeland, H., & Matthys, W. (2008). Executive
 Functions in Preschool Children with Aggressive Behavior: Impairments
 in Inhibitory Control. *Journal of Abnormal Child Psychology*, *36*, 9235.
- Raeder, M. B., Steen, V. M., Vollset, S. E., & Bjelland, I. (2007) Associations between cod liver oil use and symptoms of depression: The Hordaland health Study. *Journal of Affective Disorders*, 101, 245-249.
- Raine, A. (1989). Evoked potentials and psychopathy. *International Journal of Psychophysiology. 8,* 1-16.
- Raine, A. (1993). *The psychopathology o f crime: Criminal behavior as a clinical disorder.* San Diego, CA: Academic Press.
- Raine, A. (2013). *The anatomy of violence: The biological roots of crime.* London, UK: Penguin Books Ltd.
- Raine, A., Buchsbaum, M. S., Stanley, J., Lottenberg, S., Abel, L., & Stoddard,
 S. (1994). Selective reductions in prefrontal glucose metabolism in murderers. *Biological Psychiatry*, *36*, 365-373.
- Raine, A., Fung, A., & Lam, B. (2011). Peer victimization partially mediates the schizotypy-aggression relationship in children and adolescents. *Schizophrenia Bulletin, 37*, 937-945.
- Raine, A., Meloy, J. R., Bihrle, S., Stoddard, J., LaCasse, L., & Buchsbaum, M.S. (1998). Reduced prefrontal and increased subcortical brain

functioning assessed using positron emission tomography in predatory and affective murderers. *Behavioral Sciences & the Law, 16,* 319-332.

- Raine, A., Portnoy, J., Liu, J., Mahoomed, T., & Hibbeln, J. R. (2015). Reduction in behavior problems with omega-3 supplementation in children aged 8-16 years: a randomized, double-blind, placebo-controlled, stratified, parallel-group trial. *Journal of Child Psychology and Psychiatry*, *56*(5), 509-520.
- Raine, A., Venables, P. H., & Mednick, S. A. (1997). Low resting heart rate at age 3 years predisposes to aggression at age 11 years: Findings from the Mauritius Joint Child Health Project. *Journal of the American Academy of Child and Adolescent Psychiatry*, *36*, 1457-1465.
- Raison, C. L., & Miller, A. H. (2013). The evolutionary significance of depression in Pathogen Host Defense (PATHOS-D). *Molecular Psychiatry*, 18, 15-37.
- Ramautar, J. R., Kok, A., & Ridderinkhof, K. R. (2006). Effects of stop-signal modality on the N2/P3 complex elicited in the stop-signal paradigm. *Biological Psychology, 76,* 96-109.
- Ramirez, J. M., & Andreu, J. M. (2006). Aggression, and some related psychological constructs (Anger, hostility, and impulsivity. Some comments from a research project). *Neuroscience and Biobehavioral Reviews, 30,* 276-291.
- Ramos-Loyo, J., Gonzalez-Garrido, A. A., Sanchez-Loyo, L. M., Medina, V., & Basar-Eroglu, C. (2009) Event-related potentials and event-related oscillations during identity and facial emotional processing in schizophrenia. *International Journal of Psychophysiology*, *71*(1), 84-90.
- Randall, W. M., & Smith, J. L. (2011). Conflict and inhibition in the cued-go/nogo task. *Clinical Neurophysiology*, *122*, 2400-2407.
- Rapcsak, S. Z., Galper, S. R., Comer, J. F., Reminger, S. L., Nielsen, L., Kaszniak, ... Cohen, R. A. (2000). Fear recognition deficits after focal brain damage: A cautionary note. *Neurology*, *54*, 575-581.
- Raygada, M., Cho, E., & Hilakivi-Clarke, L. (1998). High maternal intake of polyunsaturated fatty acids during pregnancy in mice alters offsprings'

aggressive behavior, immobility in the swim test, locomotor activity and brain protein kinase C activity. *Journal of Nutrition, 128,* 2505-2511.

- Re, S., Zanoletti, M., & Emanuele, E. (2008). Aggressive dogs are characterized by low omega-3 polyunsaturated fatty acid status. *Veterinary Research Communications, 32*, 225–230.
- Realmuto, G., Begleiter, H., Odencrantz, J., & Porjesz, B. (1993). Event-related potential evidence of dysfunction in automatic processing in abstinent alcoholics. *Biological Psychiatry*, *33*, 594-601.
- Rellecke, J., Sommer, W., & Schacht, A. (2013). Emotion effects on the N170: a question of reference. *Brain Topography, 26*, 62-71.
- Renier, L. A., Anurova, I., De Volder, A. G., Carlson, S., VanMeter, J., & Rauschecker, J. P. (2010). Preserved Functional Specialization for Spatial Processing in the Middle Occipital Gyrus of the Early Blind. *Neuron, 68*(1), 138-148.
- Reske, M., Kellermann, T., Habel, U., Shah, N. J., Backes, V., von Wilmsdorff,
 M., ... Schneider, F. (2007). Stability of emotional dysfunctions? A long-term fMRI study in first-episode schizophrenia. *Journal of Psychiatric Research*, *41*(11), 918-927.
- Ressler, K. J., & Nemeroff, C. B. (2000). Role of serotonergic and noradrenergic systems in the pathophysiology of depression and anxiety disorders. *Depression and Anxiety*, *12*(1), 2-19.
- Reyna, C., Ivacevich, M. G. L., Sanchez, A., & Brussino, S. (2011). The Buss-Perry Aggression Questionnaire: Construct validity and gender invariance among Argentinean adolescents. *International Journal of Psychological Research*, 4(2), 30-37.
- Richardson, D. S., & Green, L. R. (2003). Defining direct and indirect aggression: The Richardson Conflict Response Questionnaire. *International Review of Social Psychology*, *16*(3), 11-30.
- Richardson, D. R., Hammock, G. S., Smith, S. M., Gardner, W., & Signo, M. (1994). *Aggressive Behavior, 20*(4), 275-289.
- Richell, R. A., Mitchell, D. G. V., Peschardt, K. S., Winston, J. S., Leonard, A., Dolan, R. J., & Blair, R. J. R. (2005). Trust and distrust: the perception

of trustworthiness of faces in psychopathic and non-psychopathic offenders. *Personality and Individual Differences, 38*, 1735-1744.

- Riemer, S., Maes, M., Christophe, A., & Rief, W. (2010). Lowered w-3 PUFAs are related to major depression, but not somatization syndrome. *Journal of Affective Disorders*, *123*, 173-180.
- Robinson, L., Spencer, M. D., Thomson, L. D. G., Sprengelmeyer, R., Owens,
 D. G. C., Stanfield, A. C., ... Johnstone, E. C. (2012). Facial emotion recognition in Scottish prisoners. *International Journal of Law and Psychiatry*, *35*, 57-61.
- Rolls, E. (1999). The Brain and Emotion. Oxford, UK: Oxford University Press.
- Roose, A., Bijttebier, P., Claes, L., Decoene, S., & Frick, P. J. (2010). Assessing the affective features of psychopathy in adolescence: A further validation of the inventory of callous and unemotional traits. *Assessment, 17*(1), 44-57.
- Roose, A., Bijttebier, P., Van der Oord, S., Claes, L., & Lilienfeld, S. O. (2013).
 Psychopathic traits in youth and associations with temperamental features: Results from a performance-based measure. *Journal of Individual Differences*, *34*, 1-7.
- Rosel, D. R., & Siever, L. J. (2015). The neurobiology of aggression and violence. *CNS Spectrums, 20*(3), 254-279.
- Rossion, B. (2014). Understanding face perception by means of human electrophysiology. *Trends in Cognitive Sciences, 18,* 310-318.
- Rossion, B., Gauthier, I., Tarr, M. J., Despland, P. A., Bruyer, R., Linotte, S., & Crommelinck, M. (2000). The N170 occipito-temporal component is enhanced and delayed to inverted faces but not to inverted objects: an electrophysiological account of face-specific processes in the human brain. *NeuroReport, 11*, 69-74.
- Rossion, B., & Jacques, C. (2011). The N170: Understanding the time course of face perception in the human brain. In S. J. Luck, & E. Kappenman (Eds.), *The Oxford handbook of ERP components* (pp. 115-142). Oxford, UK: Oxford University Press.

- Rossion, B., Joyce, C. A., Cottrell, G. W., & Tarr, M. J. (2003). Early lateralization and orientation tuning for face, word, and object processing in the visual cortex. *NeuroImage*, *20*(3), 1609-1624.
- Rubia, K., Russell, T., Overmeyer, S., Brammer, M. J., Bullmore, E. T., Sharma,
 T., ... Taylor, E. (2001). Mapping motor inhibition: conjunctive brain
 activations across different versions of go/no-go and stop tasks. *Neuroimage, 13*, 250-261.
- Rubia, K., Smith, A. B., Brammer, M. J., & Taylor, E. (2003). Right inferior prefrontal cortex mediates response inhibition while mesial prefrontal cortex is responsible for error detection. *NeuroImage, 20,* 351-358.
- Ruchsow, M., Groen, G., Kiefer, M., Buchheim, A., Walter, H., Martius, P., ... Falkenstein, M. (2008). Response inhibition in borderline personality disorder: event-related potentials in a go/nogo task. *Journal of Neural Transmission, 115,* 127-133.
- Rugg, M. D., & Coles, M. G. H. (1995). Electrophysiology of the mind: Eventrelated brain potentials and cognition. Oxford, UK: Oxford University Press.
- Rüsch, N., van Elst, L. T., Ludaescher, P., Wilke, M., Huppertz, H. J., Thiel, T.,
 ... Ebert, D. (2003). A voxel-based morphometric MRI study in female patients with borderline personality disorder. *NeuroImage, 20,* 385-392.
- Russell, J. A. (1994). Is there universal recognition of emotion from facial expression? A review of the cross-cultural studies. *Psychological Bulletin, 115,* 102-141.
- Russell, F. D., & Bürgin-Maunder, C. S. (2012). Distinguishing health benefits of eicosapentaenoic and docosahexaenoic acids. *Marine Drugs*, 10, 2535-2559.
- Ryan, A., Astwood, J., Gautier, S., Kuratko, C., Nelson, E., & Salem, N. (2010).
 Effects of long-chain polyunsaturated fatty acid supplementation on neurodevelopment in childhood: a review of human studies.
 Prostaglandins Leukot Essent Fatty Acids, 82, 305-314.
- Rylands, A. J., Hinz, R., Jones, M., Holmes, S. E., Feldmann, M., Brown, G., ... Talbot, P. S. (2012). Pre- and Postsynaptic Serotonergic Differences in

Males with Extreme Levels of Impulsive Aggression Without Callous Unemotional Traits: A Positron Emission Tomography Study Using 11C-DASB and 11C-MDL 100907. *Biological Psychiatry, 72,* 1004-1011.

- Sabatinelli, D., Bradley, M. M., Fitzsimmons, J. R., & Lang, P. J. (2005). Parallel amygdala and inferotemporal activation reflect emotional intensity and fear relevance. *NeuroImage*, *24*, *1*265-1270.
- Sable, P. S., Kale, A. A., & Joshi, S. R. (2013). Prenatal omega 3 fatty acid supplementation to a micronutrient imbalanced diet protects brain neurotrophins in both the cortex and hippocampus in the adult rat offspring. *Metabolism Clinical and Experimental*, *62*, 1607-1622.
- Saddoris, M. P., Gallagher, M., & Schoenbaum, G. (2005). Rapid associative encoding in basolateral amygdala depends on connections with orbitofrontal cortex. *Neuron, 46,* 321-331.
- Sadeh, B., Podlipsky, I., Zhdanov, A., & Yovel, G. (2010). Event-related potential and functional MRI measures of face-selectivity are highly correlated: a simultaneous ERP-fMRI investigation. *Human Brain Mapping, 31*(10), 1490-1501.
- Sah, P., Faber, E. S. L., Lopez de Armentia, M., & Power, J. (2003). The amygdaloid complex: anatomy and physiology. *Physiological Reviews*, *83*(3), 803-834.
- Salem Jr., N., Litman, B., Kim, H., & Gawrisch, K. (2001). Mechanisms of action of docosahexaenoic acid in the nervous system. *Lipids, 36*(9), 945-959.
- Salmivalli, C. (2010). Bullying and the peer group: A review. *Aggression and Violent Behavior, 15,* 112-120.
- SanGiovanni, J. P., & Chew, E. Y. (2005). The role of omega-3 long-chain polyunsaturated fatty acids in health and disease of the retina. *Progress in Retinal and Eye Research, 24,* 87-138.
- Sato, W., Kochiyama, T., Uono, S., & Yoshikawa, S. (2008). Time course of superior temporal sulcus activity in response to eye gaze: a combined fMRI and MEG study. *Social Cognitive and Affective Neuroscience, 3*(3), 224-232.

- Sato, W., Toichi, M., Uono, S., & Kochiyama, T. (2012). Impaired social brain network for processing dynamic facial expressions in autism spectrum disorders. *BMC Neuroscience*, *13*, 99.
- Sato, W., Uono, S., Matsuura, N., & Toichi, M. (2009). Misrecognition of facial expressions in delinquents. *Child and Adolescent Psychiatry and Mental Health, 3*, 27.
- Saunders, J. B., Aasland, O. G., Babor, T. F., de la Fuente, J. R., & Grant, M. (1993). Development of the Alcohol Use Disorders Identification Test (AUDIT): WHO Collaborative Project on Early Detection of Persons with Harmful Alcohol Consumption--II. *Addiction, 88*(6), 791-804.
- Sauseng, P., Griesmayr, B., Freunberger, R., & Klimesch, W. (2010). Control mechanisms in working memory: a possible function of EEG theta oscillations. *Neuroscience and Biobehavioral Reviews, 34*, 1015-1022.
- Sawada, R., Sato, W., Uono, S., Kochiyama, T., & Toichi, M. (2014). Electrophysiological correlates of the efficient detection of emotional facial expressions. *Brain Research*, 1560, 60-72.
- Schachar, R., & Logan, G. (1990). Impulsivity and inhibitory control in normal development and childhood psychopathology. *Developmental Psychology, 26,* 710-720.
- Schall, J. D., Stuphorn, V., & Brown, J. W. (2002). Monitoring and control of action by the frontal lobes. *Neuron, 36,* 309-322.
- Schapkin, S. A., Gusev, S. A., & Kuhl, J. (2000). Categorization of unilaterally presented emotional words: an ERP analysis. *Acta Neurobiologiae Experimentalis, 60*(1), 17-28.
- Schauer, M., & Elbert, T. (2010). Dissociation Following Traumatic Stress Etiology and Treatment. *Journal of Psychology, 218*(2), 109-127.
- Schiller, D., & Delgado, M. R. (2010). Overlapping neural systems mediating extinction, reversal and regulation of fear. *Trends in Cognitive Science*, 14, 268-276.
- Schmajuk, M., Liotti, M., Busse, L., & Woldorff, M. G. (2006). Electrophysiological activity underlying inhibitory control processes in normal adults. *Neuropsychologia*, 44(3), 384-395.

- Schmiedt-Fehr, C., & Basar-Eroglu, C. (2011). Event-related delta and theta brain oscillations reflect age-related changes in both a general and a specific neuronal inhibitory mechanism. *Clinical Neurophysiology*, *122*, 1156-1167.
- Schneider, S., Peters, J., Bromberg, U., Brassen, S., Menz, M. M., Miedl, S. F., ... IMAGEN consortium. (2011). Boys do it the right way: Sexdependent amygdala lateralization during face processing in adolescents. *NeuroImage, 56,* 1847-1853.
- Schoenthaler, S. J., Amos, S., Doraz, W., Kelly, M. A., Muedeking, G., & Wakefield, J. (1997). The effect of randomized vitamin-mineral supplementation on violent and non-violent anti-social behavior among incarcerated juveniles. *Journal of Nutritional and Environmental Medicine*, *7*, 343-352.
- Schram, L. B., Nielsen, C. J., Porsgaard, T., Nielsen, N. S., Holm, R., & Mu, H. (2007). Food matrices affect the bioavailability of (n-3) polyunsaturated fatty acids in a single meal study in humans. *Food Research International, 40,* 1062-1068.
- Schulreich, S., Pfabigan, D. M., Derntl, B., & Sailer, U. (2013). Fearless
 Dominance and reduced feedback-related negativity amplitudes in a time-estimation task Further neuroscientific evidence for dual-process
 models of psychopathy. *Biological Psychology*, *93*, 352-363.
- Schulte-Ruther, M., Markowitsch, H. J., Fink, G. R., & Piefke, M. (2007). Mirror neuron and theory of mind mechanisms involved in face-to-face interactions: a functional magnetic resonance imaging approach to empathy. *Journal of Cognitive Neuroscience*, 19, 1354-1372.
- Schultz, R. T., Grelotti, D. J., Klin, A., Kleinman, J., van der Gaag, C., Marois,
 R., & Skudlarski, P. (2003). The role of the fusiform face area in social cognition: implications for the pathobiology of autism. *Philosophical Transactions of the Royal Society B: Biological Sciences, 358*(1430), 415-427.

- Schultz, D., Izard, C. E., & Bear, G. G. (2004). Emotionality, emotion information processing, and aggression. *Development and Psychopathology*, *16*, 371-387.
- Schulz, C., Kaufmann, J. M., Kurt, A., & Schweinberger, S. R. (2012). Faces forming traces: Neurophysiological correlates of learning naturally distinctive and caricatured faces. *NeuroImage*, *63*(1), 491-500.
- Schutter, D. J. L. G., & Harmon-Jones, E. (2013). The corpus callosum: A commissural road to anger and aggression. *Neuroscience and Biobehavioral Reviews*, 37, 2481-2488.
- Schwenck, C., Mergenthaler, J., Keller, K., Zech, J., Salehi, S., Taurines, R., ... Freitag, C. M. (2012). Empathy in children with autism and conduct disorder: group-specific profiles and developmental aspects. *Journal of Child Psychology and Psychiatry, 53*(6), 651-659.
- Scott, S., Knapp, M., Henderson, J., & Maughan, B. (2001). Financial cost of social exclusion: follow up study of antisocial children into adulthood. *British Medical Journal, 323,* 1-5.
- Scott, S. K., Young, A. W., Calder, A. J., Hellawell, D. J., Aggleton, J. P., & Johnson, M. (1997). Impaired auditory recognition of fear and anger following bilateral amygdala lesions. *Nature*, 385, 254-257.
- Seara-Cardoso, A., Dolberg, H., Neumann, C., Roiser, J. P., & Viding, E.(2013). Empathy, morality and psychopathic traits in women.*Personality and Individual Differences, 55,* 328-333.
- Seguin, J. R., Nagin, D., Assaad, J. M., & Tremblay, R. E. (2004). Cognitiveneuropsychological function in chronic physical aggression and hyperactivity. *Journal of Abnormal Psychology*, *113*, 603-613.
- Seidel, E., Pfabigan, D. M., Keckeis, K., Wucherer, A. M., Jahn, T., Lamm, C., & Derntl, B. (2013). Empathic competencies in violent offenders. *Psychiatry Research, 210,* 1168-1175.
- Seitz, R. J., Schäfer, R., Scherfeld, D., Friederichs, S., Popp, K., Wittsack, H. J.,
 ... Franz, M. (2008). Valuing other people's emotional face expression:
 a combined functional magnetic resonance imaging and
 electroencephalography study. *Neuroscience*, *152*, 713-722.

- Seo, H., & Lee, D. (2007). Temporal filtering of reward signals in the dorsal anterior cingulate cortex during a mixed-strategy game. *Journal of Neuroscience*, *27*, 8366-8377.
- Serfaty, C. A., & de Velasco, P. C. (2014). Omega-3 fatty acids and the impact on the development of visual connections and critical periods of plasticity. In R. R. Watson, & F. de Meester (Eds.) *Omega-3 fatty acids in brain and neurological health,* (pp. 109-120). London, UK: Academic Press Inc.
- Sergerie, K., Chochol, C., & Armony, J. L. (2008). The role of the amygdala in emotional processing: a quantitative meta-analysis of functional neuroimaging studies. *Neuroscience and Behavioural Reviews, 32,* 811-830.
- Sewards, T. V., & Sewards, M. A. (2003). Input and output stations of the entorhinal cortex: superficial vs. deep layers or lateral vs. medial divisions? *Brain Research: Brain Research Reviews, 42,* 243-251.
- Shamay-Tsoory, S. G. (2011). The Neural Bases for Empathy. *The Neuroscientist, 17*(1), 18-24.
- Shamay-Tsoory, S. G., Aharon-Peret, J., & Perry, D. (2009). Two systems for empathy: a double dissociation between emotional and cognitive empathy in inferior frontal gyrus versus ventromedial prefrontal lesions. *Brain, 132,* 617-627.
- Shamay-Tsoory, S. G., Harari, H., Aharon-Peretz, J., & Levkovitz, Y. (2010).
 The role of the orbitofrontal cortex in affective theory of mind deficits in criminal offenders with psychopathic tendencies. *Cortex, 46,* 668-677.
- Shamay-Tsoory, S. G., Tomer, R., Berger, B. D., & Aharon-Peretz, J. (2003).
 Characterization of empathy deficits following prefrontal brain damage: the role of the right ventromedial prefrontal cortex. *Journal of Cognitive Neuroscience, 15,* 324-337.
- Shamay-Tsoory, S. G., Tomer, R., Goldsher, D., Berger, B. D., & Aharon-Peretz, J. (2004). Impairment in cognitive and affective empathy in participants with brain lesions: Anatomical and cognitive correlates. *Journal of Clinical and Experimental Neuropsychology, 26,* 1113-1127.

- Shannon, R. W., Patrick, C. J., Venables, N. C., & He, S. (2013). 'Faceness' and Affectivity: Evidence for Genetic Contributions to Distinct Components of Electrocortical Response to Human Faces. *Neuroimage, 83*, 609-615.
- Shapiro, J. P. (2000). *Attitudes toward guns and violence questionnaire* (*AGVQ*): *Manual.* Los Angeles, CA: Western Psychological Services.
- Sheard, M. H. (1971). Effect of lithium on human aggression. *Nature, 230,* 113-114.
- Sheard, M. H., Marini, J. L., Bridges, C. I., & Wagner, E. (1976). The effect of lithium on impulsive aggressive behavior in man. *American Journal of Psychiatry*, 133(12), 1409-1413.
- Shen, I., Lee, D., & Chen, C. (2014). The role of trait impulsivity in response inhibition: Event-related potentials in a stop-signal task. *International Journal of Psychophysiology*, *91*, 80-87.
- Shibata, T., Nishijo, H., Tamura, R., Miyamoto, K., Eifuku, S., Endo, S., & Ono,
 T. (2002) Generators of visual evoked potentials for faces and eyes in
 the human brain as determined by dipole localization. *Brain Topography*, *1*5, 51-63.
- Shorey, R. C., Sherman, A. E., Kivisto, A. J., Elkins, S. R., Rhatigan, D. L., & Moore, T. M. (2011). Gender differences in depression and anxiety among victims of intimate partner violence: The moderating effect of shame-proneness. Journal of Interpersonal Violence, 26, 1834-1850.
- Siegel, A., & Victoroff, J. (2007). Understanding human aggression: New insights from neuroscience. *International Journal of Law and Psychiatry, 32*, 209-215.
- Siever, L. J. (2008). Neurobiology of aggression and violence. *American Journal of Psychiatry, 165,* 429-442.
- Silvers, K. M., Woolley, C. C., Hamilton, F. C., Watts, P. M., & Watson, R. A. (2005). Randomised double-blind placebo-controlled trial of fish oil in the treatment of depression. *Prostaglandins, Leukotrienes and Essential Fatty Acids, 72,* 211-218.

- Silvers, J. A., Wager, T. D., Weber, J., & Ochsner, K. N. (2014). The neural bases of uninstructed negative emotion modulation. *Social Cognitive* and Affective Neuroscience. Advance online publication. doi: 10.1093/scan/nsu016
- Simopoulos, A. P. (2002). The importance of the ratio of omega-6/omega-3 essential fatty acids. *Biomedicine & Pharmacotherapy, 56*, 365-379.
- Simson, R., Vaughan, H. G., & Ritter, W. (1977). The scalp topography of potentials in auditory and visual go/nogo tasks. *Electroencephalography* and Clinical Neurophysiology, 43, 864-875.
- Sinclair A. J., Begg, D., Mathai, M., & Weisinger, R. S. (2007). Omega 3 fatty acids and the brain: review of studies in depression. *Asia Pacific Journal of Clinical Nutrition*, *16*(1), 391-397.
- Singh, M. (2005). Essential fatty acids, DHA and human brain. *Indian Journal of Pediatrics, 72,* 239-242.
- Sinn, N., & Bryan, J. (2007). Effect of supplementation with polyunsaturated fatty acids and micronutrients on learning and behavior problems associated with child ADHD. *Journal of Developmental and Behavioral Pediatrics, 28*(2), 82-91.
- Sinn, N., & Howe, P. R. C. (2008). Mental health benefits of omega-3 fatty acids may be mediated by improvements in cerebral vascular function. *Bioscience Hypotheses, 1,* 103-108.
- Sinn, N., Milte, C., & Howe, P. R. C. (2010). Oiling the Brain: A Review of Randomized Controlled Trials of Omega-3 Fatty Acids in Psychopathology across the Lifespan. *Nutrients, 2*(2), 128-170.
- Smith, N.K., Cacioppo, J. T., Larsen, J. T., & Chartrand, T. L. (2003). May I have your attention, please: electrocortical responses to positive and negative stimuli. *Neuropsychologia*, 41, 171-183.
- Smith, W. L. (2008). Nutritionally essential fatty acids and biologically indispensable cyclooxygenases. *Trends in Biochemical Sciences*, 33, 27-37.

- Soderberg, M., Edlund, C., Kristensson, K., & Dallner, G. (1991). Fatty-Acid Composition of Brain Phospholipids in Aging and in Alzheimer's-Disease. *Lipids*, *26*(6), 421-425.
- Soloff, P. H., Meltzer, C. C., Becker, C., Greer, P. J., Kelly, T. M., & Constantine, D. (2003). Impulsivity and prefrontal hypometabolism in borderline personality disorder. *Psychiatric Research: Neuroimaging*, *123*, 153-163.
- Soloff, P. H., Nutche, J., Goradia, D., & Diwadkar, V. A. (2008). Structural brain abnormalities in borderline personality disorder: A voxel-based morphometry study. *Psychiatry Research: Neuroimaging*, *164*, 223-236.
- Somech, L., & Elizur, Y. (2012). Promoting self-regulation and cooperation in pre- kindergarten children with conduct problems: A randomized controlled trial. *Journal of the American Academy of Child and Adolescent Psychiatry*, *51*, 412-422.
- Somerville, L. H., Whalen, P. J., & Kelley, W. M. (2010). The human bed nucleus of the stria terminalis indexes hypervigilant threat monitoring. *Biological Psychiatry, 68,* 416-424.
- Songur, A., Sarsilmaz, M., Sogut, S., Ozyurt, B., Ozyurt, H., Zararsiz, I., & Turkoglu, A. O. (2004). Hypothalamic superoxide dismutase, xanthine oxidase, nitric oxide, and malondialdehyde in rats fed with fish w-3 fatty acids. *Progress in Neuro-Psychopharmacology & Biological Psychiatry*, 28, 693-698.
- Sotres-Bayon, F., Cain, C. K., & LeDoux, J. E. (2006). Brain mechanisms of fear extinction: Historical perspectives on the contribution of prefrontal cortex. *Biological Psychiatry, 60*, 329-336.
- Sprengelmeyer, R., & Jentzsch, I. (2006). Event Related Potentials and the perception of intensity in facial expressions. *Neuropsychologia, 44,* 2899-2906.
- Stadler, C., Sterzer, P., Schmeck, K., Krebs, A., Kleinschmidt, A., & Poustka, F. (2007). Reduced anterior cingulate activation in aggressive children and adolescents during affective stimulation: association with temperament traits. *Journal of Psychiatric Research*, *41*, 410-417.

- Stanislav, S. W., Crismon, M. L., & Childs, N. L. (1998) Cerebrospinal fluid monoamine metabolites and glucose metabolism in posttraumatic aggression. *Society of Biological Psychiatry*, 43, 619-621.
- Steele, V. R., Claus, E. D., Aharoni, E., Harenski, C., Calhoun, V. D., Pearlson, G., & Kiehl, K. A. (2014). A large scale (N = 102) functional neuroimaging study of error processing in a go/nogo task. *Behavioural Brain Research, 268*, 127-138.
- Stekelenburg, J. J., & de Gelder, B. (2004). The neuronal correlates of perceiving human bodies: an ERP study on the body-inversion effect. *Cognitive Neuroscience and Neuropsychology*, 15, 777-780.
- Stenberg, G., Wiking, S., & Dahl, M., (1998). Judging words at face value: Interference in a word processing task reveals automatic processing of affective facial stimuli. *Cognition and Emotion*, *12*, 755-782.
- Sterzer, P., Stadler, C., Krebs, A., Kleinschmidt, A., & Poustka, F. (2005).
 Abnormal neural responses to emotional visual stimuli in adolescents with conduct disorder. *Biological Psychiatry*, *57*, 7-15.
- Stevens, D., Charman, T., & Blair, R. J. R. (2001). Recognition of emotion in facial expressions and vocal tones in children with psychopathic tendencies. *Journal of Genetic Psychology*, *162*(2), 201-211.
- Stevens, L. J., Zentall, S. S., Abate, M. L., Kuczek, T., & Burgess, J. R. (1996). Omega-3 fatty acids in boys with behavior, learning, and health problems. *Physiological Behaviour*, *59*(4-5), 915-920.
- Stevens, L. J., Zhang, W., Peck, L., Kuczek, T., Grevstad, N., Mahon, A., ... Burgess, J. R. (2003). EFA supplementation in children with inattention, hyperactivity, and other disruptive behaviors. *Lipids, 38*, 1007-1021.
- Stewart, J. L., Silton, R. L., Sass, S. M., Fisher, J. E., Edgar, J. C., Heller, W., & Miller, G. A. (2010). Attentional bias to negative emotion as a function of approach and withdrawal anger styles: An ERP investigation.
 International Journal of Psychophysiology, 76, 9-18.
- Stieben, J., Lewis, M. D., Granic, I., Zelazo, P. D., Segalowitz, S., & Pepler, D. (2007). Neurophysiological mechanisms of emotion regulation for

subtypes of externalizing children. *Development and Psychopathology, 19*(2), 455-480.

- Stockdale, L. A., Morrison, R. G., Kmiecik, M. J., Garbarino, J., & Silton, R. L. (2015). Emotionally anesthetized: media violence induces neural changes during emotional face processing. *Social Cognitive and Affective Neuroscience.* Advance online publication. Retrieved from http://scan.oxfordjournals.org/
- Stoff, D. M., & Cairns, R. B. (1996). Aggression and Violence: Genetic, Neurobiological, and Biosocial Perspectives. Mahwah, NJ: Lawrence Erlbaum Associates.
- Strauss, J., Barr, C. L., George, C. J., Devlin, B., Vetró, A., Kiss, E., ... Kennedy, J. L. (2005). Brain-derived neurotrophic factor variants are associated with childhood-onset mood disorder: confirmation in a Hungarian sample. *Molecular Psychiatry*, 10, 861-867.
- Strik, W. K., Fallgatter, A. J., Brandeis, D., & Pascual-Marqui, R. D. (1998).
 Three-dimensional tomography of event-related potentials during response inhibition: evidence for phasic frontal lobe activation.
 Electroencephalography and Clinical Neurophysiology, 108, 406-413.
- Stroop, J. R. (1992). Studies of interference in serial verbal reactions. *Journal of Experimental Psychology 121*, 15-23.
- Su, G. G., Wen, X., Bailey, S. T., Jiang, W., Rangwala, S. M., Keilbaugh, S. A.,
 ... Wu, G. D. (1999). A novelty therapy for colitis utilizing PPAR gamma ligands to inhibit the epithelial inflammatory response. *Journal of Clinical Investigations, 104,* 383-389.
- Subar, A F., Thompson, F. E., Kipnis, V., Midthune, D., Hurwitz, P., McNutt, S., ... Rosenfeld, S. (2001). Comparative validation of the Block, Willett, and National Cancer Institute food frequency questionnaires: The Eating at America's Table Study. *American Journal of Epidemiology*, *154*, 1089-1099.
- Sublette, M. E., Segal-Isaacson, C. J., Cooper, T. B., Fekri, S., Vanegas, N., Galfalvy, H. C., ... Mann, J. (2011). Validation of a Food Frequency Questionnaire to Assess Intake of n-3 Polyunsaturated Fatty Acids in

Subjects with and without Major Depressive Disorder. *Journal of American Dietetic Association, 111,* 117-123.

- Suga, M., Yamasue, H., Abe, O., Yamasaki, S., Yamada, H., Inoue, H., Kasai, K. (2010). Reduced gray matter volume of Brodmann's Area 45 is associated with severe psychotic symptoms in patients with schizophrenia. *European Archives of Psychiatry and Clinical Neuroscience, 260*(6), 465-473.
- Sumich, A., Castro, A., Anilkumar, A. P. P., Zachariah, E., & Kumari, V. (2013). Neurophysiological correlates of excitement in schizophrenia. *Progress in Neuro-Psychopharmacology & Biological Psychiatry*, *46*, 132-138.
- Sumich, A. L., Kumari, V., Heasman, B.C., Gordon, E., & Brammer, M. (2006).
 Abnormal asymmetry of N200 and P300 event-related potentials in subclinical depression. *Journal of Affective Disorders, 92*, 171-183.
- Sumich, A., Matsudaira, T., Gow, R. V., Ibrahimovic, A., Ghebremeskel, K., Crawford, M., & Taylor, E. (2009). Resting state electroencephalographic correlates with red cell long-chain fatty acids, memory performance and age in adolescent boys with attention deficit hyperactivity disorder. *Neuropharmacology*, *57*, 708-714.
- Sumich, A., Sarkar, S., Hermens, D. F., Kelesidi, K., Taylor, E., & Rubia, K. (2012). Electrophysiological correlates of CU traits show abnormal regressive maturation in adolescents with conduct problems. *Personality and Individual Differences, 53,* 862-867.
- Sundram, F., Deeley, Q., Sarkar, S., Daly, E., Latham, R., Craig, M., ... Murphy,
 D. G. M. (2012). White matter microstructural abnormalities in the
 frontal lobe of adults with antisocial personality disorder. *Cortex, 48,*216-229.
- Summers, C. H., & Greenberg, N. (1994). Somatic correlates of adrenergic activity during aggression in the lizard, Anolis carolinensis. *Hormones and Behavior, 28*, 29-40.
- Sutton, R. S., & Barto, A. G. (1998). Reinforcement learning: an introduction. *IEEE Transactions on Neural Networks, 9*(5), 1054.

- Suzuki, H., Manabe, S., Wada, O., & Crawford, M. A. (1997). Rapid incorporation of docosahexaenoic acid from dietary sources into brain microsomal, synaptosomal and mitochondrial membranes in adult mice. *International Journal of Vitamins and Nutrition Research, 48,* 524-530.
- Swann, N. C., Cai, W., Conner, C. R., Pieters, T. A., Claffey, M. P., George, J. S., ... Tandon, N. (2012). Roles for the pre-supplementary motor area and the right inferior frontal gyrus in stopping action: electrophysiological responses and functional and structural connectivity. *NeuroImage*, *59*, 2860-2870.
- Swann, N., Tandon, N., Canolty, R., Ellmore, T. M., McEvoy, L.K., Dreyer, S., ... Aron, A. R. (2009). Intracranial EEG reveals a time- and frequencyspecific role for the right inferior frontal gyrus and primary motor cortex in stopping initiated responses. *Journal of Neuroscience, 29,* 12675-12685.
- Swick, D., Ashley, V., & Turken, A. U. (2008). Left inferior frontal gyrus is critical for response inhibition. *BMC Neuroscience, 9*, 102.
- Svilaas, A., Ström, E. C., Svilaas, T., Borgejordet, A., Thoresen, M., & Ose, L. (2002). Reproducibility and validity of a short food questionnaire for the assessment of dietary habits. *Nutrition, Metabolism and Cardiovascular Diseases, 12*, 60-70.
- Sylvers, P. D., Brennan, P. A., & Lilienfeld, S. O. (2011). Psychopathic traits and preattentive threat processing in children: a novel test of the fearlessness hypothesis. *Psychological Science, 22*(10), 1280-1287.
- Szuromi, B., Czobor, P., Komlósi, S., & Bitter, I. (2010). P300 deficits in adults with attention deficit hyperactivity disorder: a meta-analysis. *Psychological Medicine*, *41*(7), 1529-1538.
- Taha, A. Y., Cheon, Y., Ma, K., Rapoport, S. I., & Rao, J. S. (2013). Altered fatty acid concentrations in prefrontal cortex of schizophrenic patients. *Journal of Psychiatric Research*, 47, 636-643.
- Takeda, A., Tamano, H., Kan, F., Hanajima, T., Yamada, K., & Oku, N. (2008). Enhancement of social isolation-induced aggressive behavior of young mice by zinc deficiency. *Life Sciences*, *82*, 909-914.

- Takeuchi, T., Fukumoto, Y., & Harada, E. (2002). Influence of a dietary n-3 fatty acid deficiency on the cerebral catecholamine contents, EEG and learning ability in rat. *Behavior and Brain Research, 131*, 193-203.
- Tanabe, Y., Hashimoto, M., Sugioka, K., Maruyama, M., Fujii, Y., Hagiwara, R., ... Shido, O. (2004). Improvement of spatial cognition with dietary docosahexaenoic acid is associated with an increase in Fos expression in rat CA1 hippocampus. *Clinical and Experimental Pharmacology and Physiology, 31,* 700-703.
- Tebartz van Elst, L., Hesslinger, B., Thiel, T., Geiger, E., Haegele, K., Lemieux, L., ... Ebert, D. (2003). Frontolimbic brain abnormalities in patients with borderline personality disorder: A volumetric magnetic resonance imaging study. *Biological Psychiatry*, *54*, 163-171.
- Tekok-Kilic, A., Shucard, J., & Shucard, D. (2001). Stimulus modality and go/nogo effects on P3 parallel visual and auditory continuous performance tasks. *Psychophysiology*, 28, 578-589.
- Tiemeier, H., Breteler, M. M., van Popele, N. M., Hofman, A., & Witteman, J. C. (2003). Late-life depression is associated with arterial stiffness: a population-based study. *Journal of the American Geriatrics Society*, 51(8), 1105-1110.
- Thomas, L. A., Pilyoung, K., Bones, B. L., Hinton, K. E., Milch, H. S., Reynolds, R. C., ... Leibenluft, E. (2013). Elevated amygdala responses to emotional faces in youths with chronic irritability or bipolar disorder. *Neuroimage: Clinical, 2*, 637-645.
- Thomas, S. J., Johnstone, S. J., & Gonsalvez, C. J. (2007). Event-related potentials during an emotional Stroop task. *International Journal of Psychophysiology 63*, 221-231.
- Todd, R. M., Lewis, M. D., Meusel, L. A., & Zelazo, P. D. (2008). The time course of social-emotional processing in early childhood: ERP responses to facial affect and personal familiarity in a go-no-go task. *Neuropsychologia*, 46, 595-613.
- Tofail, F., Kabir, I., Hamadani, J. D., Chowdhury, F., Yesmin, S., Mehreen, F., ... McGregor, S. M. (2008). Effects of prenatal food and micronutrient

supplementation on infant development: a randomized trial from the Maternal and Infant Nutrition Interventions, Matlab (MINIMat) study. *American Journal of Clinical Nutrition, 87*, 704-711.

- Tokdemir, M., Plota, S. A., Acik, Y., Gursu, F., & Cikim, G. (2003). Blood zinc and copper concentrations in criminal and noncriminal schizophrenic men. *Archives of Andrology, 49,* 365-368.
- Tortosa, M. I., Lupiáñez, J., & Ruz, M. (2013). Race, emotion and trust: An ERP study. *Brain Research, 1494,* 44-55.
- Toth, A. L., Varala, K., Henshaw, M. T., Rodriguez-Zas, S. L., Hudson, M. E., & Robinson, G. E. (2010). Brain transcriptomic analysis in paper wasps identifies genes associated with behavior across social insect lineages. *Proceedings of the Royal Society of London B, 277*, 2139-2148.
- Tranel, D., Bechara, A., & Denburg, N. L. (2002). Asymmetric functional roles of right and left ventromedial prefrontal cortices in social conduct, decision- making, and emotional processing. *Cortex, 38,* 589-612.
- Trestman, R. L, Horvath, T., Kalus, O., Peterson, A. E., Coccaro, E., Mitropoulou, V., Siever, L. J. (1996). Event-related potentials in schizotypal personality disorder. *Journal of Neuropsychiatry and Clinical Neuroscience, 8*, 33-40.
- Tsytsarev, S. V., & Grodnitzky, G. R. (1995). Anger and criminality. In H. Kassinove (Ed.), *Anger disorders: Definition, diagnosis and treatment* (pp. 91-108). Philadelphia, PA: Taylor & Francis.
- Tuzun, F., Kumral, A., Ozbal, S., Dilek, M., Tugyan, K., Duman, N., & Ozkan, H.
 (2012). Maternal prenatal omega-3 fatty acid supplementation attenuates hyperoxia-induced apoptosis in the developing rat brain. *International Journal of Developmental Neuroscience, 30,* 315-323.
- Uauy, R., & Castillo, C. (2003). Lipid requirements of infants: implications for nutrient composition of fortified complementary foods. *Journal of Nutrition, 133,* 2962-2972.
- Uauy R, & Dangour, A. D. (2006). Nutrition in brain development and aging: Role of essential fatty acids. *Nutrition Reviews, 64*(5), 24-33.

- Uauy, R., Hoffman, D.R., Peirano, P., Birch, D. G., & Birch, E. E. (2001). Essential fatty acids on visual and brain development. *Lipids, 36,* 885-895.
- Uauy, R., Mena, P., & Rojas, C. (2000). Essential fatty acids in early life: structural and functional role. *Proceedings of the Nutrition Society, 59,* 315.
- Umhau, J. C., Dauphinais, K. M., Patel, S. H., Nahrwold, D. A., Hibbeln, J. R., Rawlings, R. R., & George, D. T. (2006). The relationship between folate and docosahexaenoic acid in men. *European Journal of Clinical Nutrition, 60*(3), 352-357.
- van Bokhoven, I., van Goozen S. H., van Engeland, H., Schaal, B., Arseneault,
 L., Séguin, J. R., ...Tremblay, R. E. (2005). Salivary cortisol and
 aggression in a population-based longitudinal study of adolescent
 males. *Journal of Neural Transmission*, *112*, 1083-1096.
- van Boxtel, G. J., van der Molen, M. W., Jennings, J. R., & Brunia, C. H. (2001). Psychophysiological analysis of inhibitory motor control in the stop-Signal paradigm. *Biological Psychology*, *58*(3), 229-262.
- van der Meer, M. A., Johnson, A., Schmitzer-Torbert, N. C., & Redish, A. D. (2010). Triple dissociation of information processing in dorsal striatum, ventral striatum, and hippocampus on a learned spatial decision task. *Neuron, 67*(1), 25-32.
- van Goozen, S. H., & Fairchild, G. (2008). How can the study of biological processes help design new interventions for children with severe antisocial behavior? *Development and Psychopathology, 20*(3), 941-973.
- van Goozen, S. H. M., Matthys, W., Cohen-Kettenis, P. T., Gispen-de Wied, C., Wiegant, V. M., & van Engeland, H. (1998). Salivary cortisol and cardiovascular activity during stress in oppositional-defiant disorder boys and normal controls. *Biological Psychiatry*, *43*, 531-539.
- van Veen, V., & Carter, C. S. (2002). The timing of action-monitoring processes in the anterior cingulate cortex. *Journal of Cognitive Neuroscience, 14,* 593-602.

- Venables, N. C., Patrick, C. J., Hall, J. R., & Bernat, E. M. (2011). Clarifying relations between dispositional aggression and brain potential response: Overlapping and distinct contributions of impulsivity and stress reactivity. *Biological Psychology*, *86*, 279-288.
- Venna, V. R., Delplanque, D., Allet, C., Belarbi, K., Hamdane, M., & Bordet, R. (2009). PUFA induce antidepressant-like effects in parallel to structural and molecular changes in the hippocampus. *Psychoneuroendocrinology*, 34, 199-211.
- Verbruggen, F., & Logan, G.D. (2008). Automatic and controlled response inhibition: Associative learning in the go/no-go and stop-signal paradigms. *Journal of Experimental Psychology: General*, 137, 649-672.
- Verona, E., Patrick, C. J., Curtin, J. J., Bradley, M. M., & Lang, P. J. (2004). Psychopathy and physiological reaction to emotionally evocative sounds. *Journal of Abnormal Psychology*, *113*, 99-108.
- Viau, V. (2002). Functional cross-talk between the hypothalamic-pituitarygonadal and -adrenal axes. *Journal of Neuroendocrinology, 14,* 506-513.
- Viding, E., & McCrory, E. J. (2012). Why should we care about measuring callous-unemotional traits in children? *British Journal of Psychiatry*, 200, 177-178.
- Vilà-Balló, A., Hdez-Lafuente, P., Rostan, C., Cunillera, T., & Rodriguez-Fornells, A. (2014). Neurophysiological correlates of error monitoring and inhibitory processing in juvenile violent offenders. *Biological Psychology*, *102*, 141-152.
- Virkkunen, M. E. (1985). Urinary free cortisol secretion in habitually violent offenders. *Acta Psychiatrica Scandinavica, 72*, 40-44.
- Virkkunen, M. E., Horrobin, D. F., Jenkins, D. K., & Manku, M. (1987). Plasma phospholipid essential fatty acids and prostaglandins in alcoholic, habitually violent, and impulsive offenders. *Biological Psychiatry*, 22(11), 1087-1096.

- Vitacco, M., Rogers, R., & Neumann, C. (2003). The Antisocial Process Screening Device: An examination of its construct and criterion-related validity. *Assessment, 10,* 143-150.
- Vitale, J. E., & Newman, J. P. (2001). Using the Psychopathy Checklist– Revised with female samples: Reliability, validity, and implications for clinical utility. *Clinical Psychology: Science and Practice*, *8*, 117-132.
- Voigt, R. G., Llorente, A. M., Jensen, C. L., Fraley, J. K., Berretta, M. C., & Heird, W. C. (2001). A randomized, double-blind, placebo-controlled trial of docosahexaenoic acid supplementation in children with attentiondeficit/hyperactivity disorder. *Journal of Pediatrics*, *139*, 189-196.
- Volpe, U., Mucci, A., Bucci, P., Merlotti, E., Galderisi, S., & Maj, M. (2007). The cortical generators of P3a and P3b: a LORETA study. *Brain Research Bulletin, 73*, 220-230.
- Vuilleumier, P. (2000). Faces call for attention: evidence from patients with visual extinction. *Neuropsychologia, 38,* 693-700.
- Vuilleumier, P., Armony, J. L., Driver, J., & Dolan, R. J. (2001). Effects of attention and emotion on face processing in the human brain: an eventrelated fMRI study. *Neuron*, *30*, 829-841.
- Vuilleumier, P., Armony, J. L., Driver, J., & Dolan, R. J. (2003). Distinct spatial frequency sensitivities for processing faces and emotional expressions. *Nature Neuroscience*, 6(6), 624-631.
- Vuilleumier, P., & Driver, J. (2007). Modulation of visual processing by attention and emotion: windows on causal interactions between human brain regions. *Philosophical Transactions of the Royal Society B: Biological Sciences, 362,* 837-855.
- Vuilleumier, P., & Pourtois, G. (2007). Distributed and interactive brain mechanisms during emotion face perception: evidence from functional neuroimaging. *Neuropsychologia*, 45, 174-194.
- Vuilleumier, P., Richardson, M. P., Armony, J. L., Driver, J., & Dolan, R. J. (2004). Distant influences of amygdala lesion on visual cortical activation during emotional face processing. *Nature Neuroscience*, 7, 1271-1278.
- Vuilleumier, P., & Schwartz, S. (2001). Emotional facial expressions capture attention. *Neurology, 56*, 153-158.
- Vyas, A., Mitra, R., Shankaranarayana, R. B. S., & Chattarji, S. (2002). Chronic stress induces contrasting patterns of dendritic remodeling in hippocampal and amygdaloid neurons. *Journal of Neuroscience, 22,* 6810-6818.
- Wager, T. D., Davidson, M. L., Hughes, B. L., Lindquist, M. A., & Ochsner, K. N. (2008). Prefrontal-subcortical pathways mediating successful emotion regulation. *Neuron*, *59*, 1037-1050.
- Wahlund, K., & Kristiansson, M. (2009). Aggression, psychopathy and brain imaging - Review and future recommendations. *International Journal of Law and Psychiatry, 32*(4), 266-271.
- Waiter, G. D., Williams, J. H., Murray, A. D., Gilchrist, A., Perrett, D. I., &
 Whiten, A. (2004). A voxel-based investigation of brain structure in male
 adolescents with autistic spectrum disorder. *NeuroImage, 22*(2), 619-625.
- Waller, R., Gardner, F., & Hyde, L. W. (2013). What are the associations between parenting, callous-unemotional traits, and antisocial behavior in youth? A systematic review of evidence. *Clinical Psychology Review*, 33, 593-608.
- Walsh, W. J., Isaacson, H. R., Rehman, F., & Hall, A. (1997). Elevated blood copper/zinc ratios in assaultive young males. *Physiology and Behavior*, 62(2), 327-329.
- Watanabe, S., Kakigi, R., Koyama, S., & Kirino, E. (1999). It takes longer to recognize the eyes than the whole face in humans. *Neuroreport, 10,* 2193-2198.
- Watanabe, S., Kakigi, R., & Puce, A. (2003). The spatio-temporal dynamics of the face inversion effect: A magneto- and electroencephalographic study. *Neuroscience*, *116*, 879-895.
- Watari, M., Hamazaki, K., Hirata, T., Hamazaki, T., & Okubo, Y. (2010). Hostility of drug-free patients with schizophrenia and n-3 polyunsaturated fatty acid levels in red blood cells. *Psychiatry Research*, *177*, 22-26.

- Weidner, G., Connor, S. L., Hollis, J. F., & Connor, W. E. (1992). Improvements in hostility and depression in relation to dietary change and cholesterol lowering. *Annals of Internal Medicine*, *117*, 820-823.
- Weymar, M., Low, A., & Hamm, A. O. (2011). Emotional memories are resilient to time: evidence from the parietal ERP old/new effect. *Human Brain Mapping*, *32*(4), 632-640.
- Whalen, P. J., Kagan, J., & Cook, R. G., Davis, F. C., Kim, H., Polis, S., ... Johnstone, T. (2004). Human amygdala responsivity to masked fearful eye whites. *Science*, *306*, 2061.
- White, D. J., Congedo, M., Ciorciari, J., & Silberstein, R. B. (2011). Brain oscillatory activity during spatial navigation: theta and gamma activity link medial temporal and parietal regions. *Journal of Cognitive Neuroscience, 24*, 686-697.
- White, S. (2010). Examining the influence of callous-unemotional traits on outcomes in an evidence-based treatment program for Delinquent Adolescents. University of New Orleans Theses and Dissertations. Paper 1161.
- Widenhorn-Müller, K., Schwanda, S., Scholz, E., Spitzer, M., & Bode, H. (2014).
 Effect of supplementation with long-chain ω-3 polyunsaturated fatty acids on behavior and cognition in children with attention deficit/hyperactivity disorder (ADHD): a randomized placebo-controlled intervention trial. *Prostaglandins, Leukotrienes and Essential Fatty Acids, 91*(1-2), 49-60.
- Wilkowski, B. M., & Robinson, M. D. (2010). The anatomy of anger: An integrative cognitive model of trait anger and reactive aggression. *Journal of Personality*, 78, 9-38.
- Williams, T. Y., Boyd, J. C., Cascardi, M. A., & Poythress, N. (1996). Factor structure convergent validity of the aggression questionnaire in an offender population. *Psychological Assessment*, *4*, 398-403.
- Williams, A., Katz, D., Ali, A., Girard, C., Goodman, J., & Bell. I. (2006). Do essential fatty acids have a role in the treatment of depression? *Journal* of Affective Disorders, 93, 117-223.

- Williams, E., Liddell, B. J., Kemp, A. H., Bryant, R. A., Meares, R. A., Peduto, A. S., & Gordon, E. (2006). Amygdala-prefrontal dissociation of subliminal and supraliminal fear. *Human Brain Mapping*, *27*(8), 652-661.
- Williams, L. M., Phillips, M. L., Brammer, M. J., Skerrett, D., Lagopoulos, J., Rennie, C., ... Gordon, E. (2001). Arousal dissociates amygdala and hippocampal fear responses: evidence from simultaneous fMRI and skin conductance recording. *Neuroimage*, *14*, 1070-1079.
- Williams, M. A., McGlone, F., Abbott, D. F., & Mattingley, J. B. (2005).
 Differential amygdala responses to happy and fearful facial expressions depend on selective attention. *NeuroImage, 24,* 417-425.
- Wilson, K., Demetrioff, S., & Porter, S. (2008). A pawn by any other name?
 Social information processing as a function of psychopathic traits.
 Journal of Research in Personality, 42, 1651-1656.
- Wilson, K., Juodis, M., & Porter, S. (2011). Fear and loathing in psychopaths: A meta-analytic investigation of the facial affect recognition deficit. *Criminal Justice and Behavior, 38,* 659-668.
- Winstanley, C. A., Theobald, D. E., Cardinal, R. N., & Robbins, T. W. (2004). Contrasting roles for basolateral amygdala and orbitofrontal cortex in impulsive choice. *Journal of Neuroscience*, *24*, 4718-4722.
- Winston, J. S., Gottfried, J. A., Kilner, J. M., & Dolan, R. J. (2005). Integrated neural representations of odor intensity and affective valence in human amygdala. *The Journal of Neuroscience*, *25*, 8903-8907.
- Winston, J. S., O'Doherty, J., & Dolan, R. J. (2003a). Common and distinct neural responses during direct and incidental processing of multiple facial emotions. *Neuroimage*, 20, 84-97.
- Winston, J. S., Vuilleumier, P., & Dolan, R. J. (2003b). Effects of low-spatial frequency components of fearful faces on fusiform cortex activity. *Current Biology*, 13, 1824-1829.
- Woldorff, M. G. (1993). Distortion of ERP averages due to overlap from temporally adjacent ERPs: Analysis and correction. *Psychophysiology*, *30*, 98-119.

- Woodman, G. F. (2010). A brief introduction to the use of event-related potentials (ERPs) in studies of perception and attention. *Attention and Perception Psychophysiology*, *72*(8), doi:10.3758/APP.72.8.2031.
- Woodworth, M., & Porter, S. (2002). In cold blood: Characteristics of criminal homicides as a function of psychopathy. *Journal of Abnormal Psychology*, 111, 436-445.
- Woodworth, M., & Waschbusch, D. (2008). Emotional processing in children with conduct problems and callous/unemotional traits. *Child: Care, Health and Development, 34*(2), 234-244.
- World Health Organization (1995). *World Health Statistics Annual.* Geneva, Switzerland: WHO.
- World Health Organization. (2014). *Global status report on violence prevention: 2014*. Geneva, Switzerland: WHO.
- Wurtman, R. J. (2008). Synapse formation and cognitive brain development: effect of docosahexaenoic acid and other dietary constituents. *Metabolism*, 57(2), 6-10.
- Yamanaka, K., & Yamamoto, Y. (2010). Single-trial EEG power and phase dynamics associated with voluntary response inhibition. *Journal of Cognitive Neuroscience, 22*, 714-727.
- Yavin, E., Himovichi, E., & Eilam, R. (2009). Delayed cell migration in the developing rat brain following maternal omega 3 alpha Linolenic acid dietary deficiency. *Neuroscience*, *162*, 1011-1022.
- Yee, C. M., & Miller, G. A. (1987). Affective valence and information processing. In R. Johnson, J. W. Rohrbaugh, & R. Parasuraman (Eds.), *Current trends in event-related potential research* (pp. 300-307). Amsterdam: Elsevier.
- Yehuda, S. (2003). Omega-6/omega-3 ratio and brain-related functions. *World Review of Nutrition Diet, 92,* 37-56.
- Yehuda, S., Rabinovitz, S., Carasso, R. L., & Mostofsky, D. I. (2002). The role of polyunsaturated fatty acids in restoring the aging neuronal membrane. *Neurobiology of Aging, 23*, 843-853.

- Yeung, N., Botvinick, M. M., & Cohen, J. D. (2004). The neural basis of error detection: conflict monitoring and the error-related negativity. *Psychology Review*, *111*, 931-959.
- Youdim, K. A., Martin, A., & Joseph, J. A. (2000). Essential fatty acids and the brain: possible health implications. International Journal of Developmental Neuroscience, 18, 383-399.
- Young, G. S., Conquer, J. A., & Thomas, R. (2005). Effect of randomized supplementation with high dose olive, flax or fish oil on serum phospholipid fatty acid levels in adults with attention deficit hyperactivity disorder. *Reproduction Nutrition Development*, 45(5), 549-558.
- Zaalberg, A., Nijman, H., Bulten, E., Stroosma, L., & van der Staak, C. (2010).
 Effects of nutritional supplements on aggression, rule-breaking, and
 psychopathology among young adult prisoners. *Aggressive Behavior*, *36*, 117-126.
- Zaalberg, A., Wielders, J., Bulten, E., van der Staak, C., Wouters, A., & Nijman, H. (2015). Relationships of diet-related blood parameters and blood lead levels with psychopathology and aggression in forensic psychiatric inpatients. *Criminal Behaviour and Mental Health.* Advance online publication. doi: 10.1002/cbm.1954
- Zahn-Waxler, C., Cole, P. M., Welsh, J. D., & Fox, N. A. (1995).
 Psychophysiological correlates of empathy and prosocial behaviors in preschool children with behavior problems. *Development and Psychopathology, 1*, 27-48.
- Zalcman, S. S., & Siegel, A. (2006). The neurobiology of aggression and rage: Role of cytokines. *Brain, Behavior, and Immunity, 20,* 507-514.
- Zald, D. H. (2003). The human amygdala and the emotional evaluation of sensory stimuli. *Brain Research Reviews, 41,* 88-123.
- Zanarini, M. C., & Frankenburg, F. R. (2003). Omega-3 Fatty acid treatment of women with borderline personality disorder: A double-blind, placebocontrolled pilot study. *American Journal of Psychiatry*, 160, 167-169.

- Zhang, X., Li, T., & Zhou, X. (2008). Brain Responses to Facial Expressions by Adults with Different Attachment-Orientations. *NeuroReport, 19*(4), 437-441.
- Zillmann, D. (1983). Arousal and aggression. In R. G. Geen, & E. Donnerstein, (Eds.), *Aggression: Theoretical and empirical reviews (Vol. 1)*. New York, NY: Academic.
- Zillmann, D. (1988). Cognition-excitation interdependencies in aggressive behavior. *Aggressive Behavior, 14*(1), 51-64.
- Zink, C., Tong, Y., Chen, Q., Bassett, D., Stein, J., & Meyer-Lindenberg, A. (2008). Know your place: neural processing of social hierarchy in humans. *Neuron*, 58, 273-283.
- Zion-Golumbic, E., & Bentin, S. (2007). Dissociated neural mechanisms for face detection and configural encoding: Evidence from N170 and Gammaband oscillation effects. *Cerebral Cortex, 17,* 1741-1749.